#### **REGULATIONS FOR ADMISSIONS**

to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

Section 1. General Regulations for Admissions to Degree Programmes of Higher Education (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes)

#### 1. General Provisions

- 1.1. The Regulations for Admissions to Higher Education Degree Programmes (Bachelor's Degree Programmes, Specialist's Degree Programmes, and Master's Degree Programmes) at St Petersburg University in 2024 (hereinafter referred to as the Regulations for Admissions, Regulations) set out the policy for admitting citizens of the Russian Federation (hereinafter referred to as citizens, individuals, enrollees, applicants), as well as foreign nationals, stateless persons, including compatriots residing abroad (hereinafter referred to as foreign nationals; persons; applicants) at St Petersburg State University (hereinafter referred to as St Petersburg University), including the Branch of St Petersburg University in Tashkent, to Programmes (Bachelor's Programmes, Specialist's Programmes, Programmes) (hereinafter referred to as degree programmes, main educational programmes; programmes of higher education) on the government-funded places, as well as under contracts for chargeable educational services paid for by individuals and/or legal entities (hereinafter referred to as contracts for chargeable educational services). These Regulations also specify the particulars of admissions tests for physically challenged and disabled applicants.
- 1.2. These Regulations have been developed in compliance with Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 2012, federal Law No 19-FZ "About the particulars of legal regulation of relations in education and research due to the admission to the Russian Federation of Donetsk People's Republic, Lugansk People's Republic, Zaporozhye Region, Kherson Region and the formation of new constituting entities within the Russian Federation - Donetsk People's Republic, Lugansk People's Republic, Zaporozhye Region, and Kherson Region' and about amendments introduced to certain legislative acts of the Russian Federation' (hereinafter referred to as Act No 19-FZ) dated 17 February 2023, Order No 231 'On approval of the particulars of enrolment to organisations implementing educational activities in bachelor's degree programmes, specialist's degree programmes, master's degree programmes and programmes for training research and teaching staff in aspirantura (postgraduate military courses) as provided by Part 7 and Part 8 of Article 5 of Federal Law No 19-FZ 'About the particulars of legal regulation of relations in education and research due to the admission to the Russian Federation of Donetsk People's Republic, Lugansk People's Republic, Zaporozhye Region, Kherson Region and the formation of new constituting entities within the Russian Federation - Donetsk People's Republic, Lugansk People's Republic, Zaporozhye Region, and Kherson Region' and about amendments introduced to certain legislative acts of the Russian Federation' dated 17 February 2023' of the Ministry of Science and Higher Education of the Russian Federation dated 1 March 2023; Decree No 2150 'About establishing quota for educating foreign citizens and stateless persons in the Russian Federation' dated 18 December 2020; Procedures for Admission to Degree Programmes -Bachelor's Degree Programmes, Specialist's Degree Programmes, Master's Degree Programmes approved by Order No 1076 of the Ministry of Science and Higher Education of the Russian Federation dated 21 August 2020 (hereinafter referred to as the Procedures for Admission, the Procedures), the Charter of the Federal State Budgetary Educational Institution of Higher Education 'St Petersburg State University'.
- 1.3. Particulars of the admissions of foreign applicants including those having equal rights with Russian citizens when applying for degree programmes of higher education shall be

governed by provisions of Section 3 herein. Any Russian citizen with an additional non-Russian citizenship shall only be considered as a Russian citizen when defining conditions for the admission to St Petersburg University, except cases envisaged by international agreements of the Russian Federation or federal laws.

- 1.4. An order on enrolment concerning a Russian Federation national being a foreign national at the same time who has failed to inform St Petersburg University thereof when applying for the admission can be duly cancelled due to such applicant's failure to provide true and/or full information of themselves to St Petersburg University Admissions Committee (in accordance with the sub-clause 1, Part 2, Article 61 of Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 2012; such an enrolled person shall be subject to expulsion for the violation of the St Petersburg University Regulations, which led to their illegal enrolment to St Petersburg University).
- 1.5. The responsibility for organising admissions to degree programmes of higher education rests with the St Petersburg University Admissions Committee (hereinafter referred to as the Admissions Committee). The Rector of St Petersburg University is a Chair of the Admissions Committee. The Admissions Committee's composition, competences, and operating procedures shall be defined by the Provisions for the Admissions Committee.
- 1.6. The Admissions Committee shall be responsible for publishing regulations for admissions to degree programmes of higher education at St Petersburg University, as well as any other required information, on the official St Petersburg University web site http://www.spbu.ru(hereinafter referred to as 'the St Petersburg University official web site') in the Admission Committee's web page http://www.abiturient.spbu.ru. The specified resource herein shall serve as the only official resource of information on admissions policy for degree programmes of higher education at St Petersburg University, as well as of the Admissions Committee's activities.
- 1.7. Enrolment to the first year of study in bachelor's degree programmes, specialist's degree programmes and master's degree programmes at St Petersburg University is organised to full-time and part-time evening types of study.

## 2. Applying for Admission

- 2 .1. Admissions to St Petersburg University degree programmes of higher education at all levels, types and conditions of enrolment is conducted on a competitive basis upon personal applications and admissions tests results (unless otherwise specified by the laws of the Russian Federation and these Regulations).
- 2.2 Admission to degree programmes of higher education at St Petersburg University is conducted on grounds of the submitted personal application in the name of the Rector of the University. Personal applications shall be submitted by applicants to the Admissions Committee. The application must include: the Insurance Number of the Individual Ledger Account (Insurance Number of Compulsory Pension Insurance) (hereinafter referred to as Insurance Number of the individual Ledger Account) (if applicable).
- 2.3. Applicants shall submit their applications either electronically through the applicant's Personal Account, or through the Super Service 'Online Admission' of the federal state information system 'The Unified Portal of State and Municipal Services (functions)' (hereinafter referred to as UPSMS). To submit their applications through the Personal Account, the applicants must register on the official web site of St Petersburg University within the deadlines set out in the relevant sections of the Regulations and obtain a login and a password to their Personal Account. Logins and passwords must not be disclosed to third parties. The responsibility for Personal Account login and password security after they have obtained them shall rest with the applicants. To submit their application form, the applicant must type information in all boxes available, print the application out, and confirm the completeness and reliability of the provided information by putting their handwritten signature.

- 2 .4. The applicant can submit their personal application and other required documents in the following manner:
  - 2.4.1. in person
  - 2.4.2. through general post service operators;
  - 2.4.3. electronically through their Personal Account;
- 2.4.4. electronically, through the Unified Portal of State and Municipal Services (if applicable).

Requirements to the documents submitted by the applicants electronically, as well as the details of the procedure of submitting the documents for the competitive selection are specified in the relevant sections of the Regulations.

In case Russian citizens with the valid Insurance Number of the Individual Ledger Account use the UPSMS, they can submit their admission application and documents electronically through the UPSMS only.

- 2 .5. Admission applications and other documents submitted in the manner not specified by clause 2.4. of the Regulations, including those submitted via email or fax, shall not be accepted and considered by the Admissions Committee.
- 2 .6. An application receipt shall be issued to the applicant (their legal representatives) submitting the documents in person.
- 2.7. Applicants submitting their documents via general post service shall send their documents to: St Petersburg University Admissions Committee, 7/9 Universitetskaya Embankment, St Petersburg, 199034, Russian Federation
- 2.8. The applications either sent by general post service operators or submitted electronically via the St Petersburg University official web site or submitted through the UPSMS shall be accepted within the same time frame as those submitted in person.
- 2 .9. The following documents shall confirm that the application documents sent by the applicant to the Admissions Committee by general post have been accepted:
- 2.9.1 a return receipt signed by an authorised St Petersburg University official and carrying the date stamp of the destination post office with a signed list of enclosures carrying the date stamp of the post office from which the letter was sent;
- 2.9.2 an invoice (or a copy thereof) signed by an authorised St Petersburg University official (when sent by express mail with the postal item delivery directly at the Admissions Committee's address).
- 2.10. To ensure informing the applicants from the first day of accepting admission applications before enrolment, the Admissions Committee shall update the information about the number of admission applications submitted and the lists of individuals who have submitted documents required for enrolment within each competitive selection on a daily basis at least before 6pm Moscow time on the official web site of St Petersburg University in the manner set forth in clause 1.6 herein.
- 2.11. The admission applications must be submitted in Russian. Documents in foreign languages must be submitted translated into Russian and certified in the prescribed manner. Documents obtained in a foreign country must be submitted either legalised in the manner prescribed by the legislation of the Russian Federation or with Apostille affixing (except cases when the legalisation and Apostille are not required pursuant to the provisions of the legislation of the Russian Federation and/or international agreement). By signing their application, the applicant shall confirm that:
- 2.11.1. They have read the prescription to provide accurate data and originals of documents in the admission application;
- 2.11.2. They have read these Regulations, as well as other regulations and information specified in Part 2, Article 55 of Federal Law No 273-FZ 'On Education in the Russian Federation' dated 29 December 20212;
- 2.11.3. They apply for a degree for the first time when applying for enrolment to government-funded places unless otherwise has been stipulated by the legislation of the Russian

Federation in terms of applying for a degree programme on government-funded places with a respective degree:

- 2.11.3.1. to master bachelor's degree programmes or specialist's degree programmes, they have no bachelor's degree or specialist's degree, or master's degree;
- 2.11.3.2. to master master's degree programmes, they have no specialist's degree or master's degree except those applicants who possess a professional specialist's degree that is confirmed by conferment on them the qualification of a specialist;
  - 2.21.4. When admitting to bachelor's degree or specialist's degree programmes:
- 2.11.4.1. They apply for parallel admission to not more than five (5) educational organisations of higher education, including St Petersburg University;
- 2.11.4.2. They apply for parallel admission at St Petersburg University to not more than three (3) specialities and/or fields of study;
- 2.21.5. When being entitled for the admission to bachelor's degree programmes or specialist's degree programmes to the government-funded places without taking admissions tests pursuant to Part 4 and/or 12, Article 71 of Federal Law No 273-FZ 'On Education in the Russian Federation' or being entitled for the admission without taking admissions tests pursuant to the results of school Olympiads, included in the list of the Ministry of Education of the Russian Federation:
- 2.11.5.1. They submit their admission application based on the relevant particular right only to St Petersburg University;
- 2.11.5.2. They exercise their relevant particular right for the admission to only one degree programme when submitting their admission application to several degree programmes at St Petersburg University.
- 2.12. In their admission application, the applicant shall also provide their handwritten signature confirming their awareness of audio and video recording taking place during admissions tests.
- 2 .13. Applicants' admission application and other documents required for the admission shall be accepted only if applicants provide their consent for processing their personal data, which, above all, contains the consent for processing their personal data which they agree to distribute (disclose to an unidentified number of persons), in compliance with the provision of Article 10.1 of Federal Law No 152-FZ 'On Personal Data' dated 27 July 2006.
- 2 .14. Applicants to St Petersburg University degree programmes of higher education are eligible to apply for admission (to enter the competitive selection) to different types (full-time and, if applicable, part-time evening) and conditions (government-funded places or places under contracts for chargeable educational services). The application submitted to different types and conditions of enrolment within the same educational programme shall be considered as an application to enter the competitive selection for one degree programme.
- 2 .15. In their admission application, applicants should indicate the priority of the enrolment in terms of the conditions available for enrolment (to different degree programmes (groups of degree programmes, depending on the format of the competitive selection), to different types of study; priority of the enrolment in terms of each degree programme within the group of degree programmes to which a single competitive selection has been organised) to government-funded places and places under contracts for chargeable educational services separately. The order of enrolment priorities should be indicated by numbers. The higher is the number, the lower is the enrolment priority.
- 2 .16. Applicants can introduce amendments to their admission application(s), including those concerning enrolment priorities, before the deadline for submitting documents (deadline for submitting admission applications) inclusively. To do so, they should update their Personal Account and the UPSMS (if applicable).
- 2.17. To enter the competition for degree programmes of higher education at St Petersburg University, an applicant must submit the following documents:

- 2.17.1. A copy of an identity document with identification of their citizenship (including a copy of their passport of a Russian citizen identifying their Russian identity abroad or a copy of the document identifying the foreign national in the Russian Federation or the identity of the stateless person in the Russian Federation according to the provisions of Federal Law No 115-FZ 'On Legal Status of Foreign Citizens in the Russian Federation' dated 25 July 2002.
- 2.17.2. An admission application (applications) to a degree programme of the appropriate level in the format established by the Admissions Committee;
- 2.17.3. An applicant's photograph (must be taken in the current calendar year) for persons applying for admissions upon the results of the competitive selection held by St Petersburg University independently;
- 2.17.4. An origin or a copy of the standard education credentials, including education credentials obtained in a foreign country accompanied by a certificate confirming the recognition of the foreign education, except the cases when such a recognition is not required according to the legislation of the Russian Federation and/or international treaties (hereinafter referred to as 'standard credentials'); applicants can provide one or several standard credentials. If they provide several standard credentials, applicants use the original of only one standard credential from the mentioned documents for the enrolment to government-funded place;
- 2.17.5. Documents confirming their personal achievements whose results shall be considered when applying for the admission (at the discretion of the applicant);
- 2.17.6. A document confirming their registration in the system of the individual (personified) accounting (if applicable);
- 2.17.7 A document confirming the completion of the centralised testing (exam) held in the Republic of Belarus if the results of the centralised testing are being demanded.
- 2.18. In case the admission application is being submitted through the UPSMS, the standard credential shall be deemed submitted in a copy if its information has been confirmed through the federal information system 'Federal Register of Information of Education or/and Qualification Credentials, Credentials of Training' (hereinafter referred to as FREC). Other documents shall be deemed submitted in copies if their information has been confirmed by UPSMS or by other public information systems. If the standard credential information has no confirmation in the FREC, the other document information has no confirmation in UPSMS database or other public information resources, applicants shall provide the original or a copy (digital image) of the respective document.
- 2 .19. To be provided with special support when taking admissions tests, persons with a disability or additional support needs or heath restrictions must provide documents confirming their disability or additional support needs, or health restrictions.
- 2 .20. The applicant shall be excluded from the competitive selection if they have failed to:
  - 2.20.1. Submit a complete set f application documents;
  - 2.20.2. Put their handwritten signature on the admission application;
- 2.20.3. Provide legible copies of the documents (including legible scanned copies of the documents when submitting them electronically through the official portal of St Petersburg University);
  - 2.20.4. Provide accurate information;
- 2.20.5. Comply with the competition requirements for degree programmes established by the legislation of the Russian Federation, Procedures for Admission and these Regulations;
- 2.20.6. Provide documents for competition within the application submission deadline established by these Regulations;
- 2.20.7. Score points above the reference value specified by the Admissions Committee to confirm their successful completion of the admissions tests in a discipline when applying for a programme with the respective type (full-time or part-time) and conditions (government-funded places or places under contracts for chargeable educational services) of enrolment.

- 2 .21.St Petersburg University shall undertake to validate the accuracy of the provided in the admission application information and the authenticity of the submitted documents, among other things, by contacting the relevant information systems, public and municipal administration bodies and organisations.
- 2 .22. If applicants have submitted illegible document copies or document copies with significant parts of text omitted (in particular, the seal, the authorised representative's signature, document details, etc. are missing or illegible), such documents shall neither be accepted nor considered.
- 2 .23. Pursuant to Decree No 697 of the Government of the Russian Federation dated 14 August 2013, the applicants applying for the admission to the specialities and fields of study included in the list of such specialities and fields of study that require mandatory preliminary medical examinations in the manner prescribed for entering an employment or labour contract within the relevant position or speciality must undergo mandatory preliminary medical examinations in the manner prescribed for entering an employment or labour contract within the relevant position, profession or speciality. The applicants when applying for admission to respective specialities or subject areas are entitled to provide a medical statement (medical professional advisory opinion).
- 2 .24. Applicants who have consciously submitted forged documents to the Admissions Committee bear responsibility under the laws of the Russian Federation.
- 2.2.5. Applicants, at any point of admission procedures, shall have the right to file a written application to revoke their documents, including the originals of the standard credentials. Applicants enrolled to study shall have the right to file an application to terminate their enrolment.

Applicants enrolled to government-funded places, who wish to revoke the original of standard credentials, should file their revocation application and application to terminate their enrolment simultaneously.

Persons specified in clauses 7.2.3.6, 7.2.3.7 of the Regulations, who have provided their consents for enrolment, have the right to revoke their consent for enrolment. In case the mentioned persons wish to submit their consents for enrolment to the places within the reference quota to another organisation, they should revoke their current consent for enrolment to the places within the reference quota at St Petersburg University before submitting a new one. Persons enlisted for enrolment, who have revoked their consent for enrolment, shall be withdrawn from the applicants enrolled to study. Persons, who have revoked their consent for enrolment, shall not be withdrawn from the list of those who have submitted their documents and from the competition list.

- 2 .26. Before the deadline for admissions to the government-funds places, the documents or originals of the standard credentials submitted shall be returned to applicants upon their application revoking either documents or the originals submitted only in person to St Petersburg University, correspondingly:
- 2.26.1 in the course of two hours after the application has been submitted in case the revocation application is submitted earlier than two hours before the end of the working day.
- 2.26.2. in the course of the first two hours of the next working day in case the revocation application is submitted within two hours before the end of the working day.
- 2 .27. After the deadline for admissions to the government-funds places, the originals of documents or originals of the standard credentials submitted shall be returned to applicants upon their application revoking either documents or the originals, submitted to St Petersburg University, in the course of one working day. If it is impossible to return the mentioned originals, they shall remain in storage at St Petersburg University.
- 2 .28. The following persons shall be excluded from competition: those who have submitted a statement of their refusal to participate in the competition, or have withdrawn their documents after the document submission deadline set by herein, as well as those having received scores below the minimum score points that could confirm their successful completion

of the relevant admissions tests during the admissions tests or the additional admissions tests sessions.

- 2 .29. It shall not be allowed to change the type of competition based on documents submitted to the Admissions Committee after the admission applications submission deadline.
- 2 .30. Unless otherwise specified by the legislation of the Russian Federation and these Regulations, **6 pm**, **Moscow Time** shall be the deadline for accepting documents required for entering the competitive selection and for the enrolment on the day specified by these Regulations.

# 3. Procedure of Defining the Number of Places for Admission to Degree Programmes of Higher Education

- 3.1. The admission of applicants to degree programmes of higher professional education to government-funded places at St Petersburg University is held in compliance with the admission quotas set for St Petersburg University by the relevant federal executive authority.
- 3.2. Within the limits for government-funded places, St Petersburg University enrols to dedicated education those applicants, who have pass the competition as prescribed by the Procedures for Admission and provided their consents for signing agreements for dedicated education with:
- 3.2.1. federal public authorities, public authorities of constituent entities of the Russian Federation, local self-governance authorities;
  - 3.2.2. public and municipal institutions, unitary enterprises;
  - 3.2.3. state-owned corporations;
  - 3.2.4. state-owned companies;
- 3.2.5. organisations included in the consolidated register of organisations of the military-industrial complex, formed in accordance with Part 2, Article 21 of Federal Law No 488-FZ 'On Industrial Policy in the Russian Federation' dated 31 December 2014;
- 3.2.6.economic societies whose equity includes a share of the Russian Federation, a constituent entity of the Russian Federation or a municipal entity;
- 3.2.7. joint stock companies whose stock has been owned or entrusted to a state-owned corporation;
- 3.2.8. Subsidiaries of economic societies, specified in clauses 3.2.4, 3.2.6, 3.2.7 of these Regulations;
- 3.2.9. organisations established by state-owned corporations or transferred to state-owned corporations in compliance with the provisions of federal laws about such specified corporations;
- 3.2.10. organisations recognised as producers of agricultural goods in compliance with the provisions of Part 1 Article 3 Federal Law No 264-FZ 'On Development of Agriculture' dated 29 December 2006 in fields f study and specialities correspondent to agriculture, fishery and engineering (subject to being in this status for at least three (3) years).
- 3.2.11. organisations-residents of the territories of priority development, special economic zone, territorial development zone, Arctic zone, free port of Vladivostok, or organisations that have received the status of a project participant in the territories of the Skolkovo innovation centre, the international medical cluster, innovative scientific and technological centres, as well as the status of a participant in the Military Innovation Technopolis "Era" of the Ministry of Defense of the Russian Federation (subject to being in this status for at least three (3) years).
- 3.3. Apart from admissions within the reference admissions quotas, St Petersburg University accepts applicants to the places under contracts for chargeable educational services within the limits set forth by the Admissions Committee. The limits set forth for admissions to the places under contracts for chargeable educational services can be exceeded by the decision of the Admissions Committee. Upon such a decision, enrolled are all applicants who have

scored at least the minimum number of points and the minimum number competition points (points scored for each admissions test and for individual achievements) sufficient for being enrolled.

- 3.4. The quantity of the places within each competitive selection for the admission to a separate degree programme of higher education, as well as within unified competitive selection for the admission to a totality of degree programmes regarding all types and conditions of study open for the admissions at St Petersburg University is determined by the Admissions Committee in the manner prescribed by the federal legislation and by-laws.
- 3.5. Tuition fee for the places under contracts for chargeable educational services, duly approved in the manner prescribed at St Petersburg University, shall be announced by the Admissions Committee not later than 1 June 2024.
- 3.6. If there are vacancies after the deadline for enrolment, St Petersburg University can additionally enrol to study on these vacant places applicants based on the competition lists. Within the additional enrolment to government-funded places to bachelor's degree programmes or specialist's degree programmes, the originals of standard credentials shall be accepted (marks confirming the provision of originals of standard credentials shall be entered into the UPSMS if there is a correspondent confirmation in the FREC) starting from 10 August 2024 (hereinafter refereed to as the 'confirming UPSMS marks').
- 3.7. St Petersburg University may announce additional admissions to vacant places. Additional admissions and enrolments shall be carried out in line with the provisions of the Regulations. In this case, the information about additional admissions shall be published not later than 15 August 2024 in the manner prescribed by clause π. 1.6 herein. Additional enrolment shall be completed before 29 August 2024.
- 3.8. St Petersburg University conducts admissions to the places within the sponsored reference quota in compliance with the provisions of the federal legislation and by-laws, including Section 12 of the Regulations.
- 3.9. Vacancies within the particular quota, special quota or sponsored quota can be filled in by enrolling applicants applying for enrolment to the places within the admissions reference quota without admissions tests or upon the results of admissions tests.

#### 4. Procedures of Admissions Testing

- 4.1. The list and forms of St Petersburg University admissions tests, as well as admissions syllabi for degree programmes of higher education, are approved by the decision of the Admissions Committee. If it is allowed to select disciplines when taking admissions tests for admission to bachelor or specialist programmes, they can select one or several disciplines.
- 4.2. One and the same nomenclature of admissions tests shall be developed for admission to all types (full-time or part-time) and conditions (government-funded places or places on the fee-paying (contractual) basis paid for by individuals and/or legal entities) of enrolment to each degree programme.
  - 4.3. All admissions tests for degree programmes at St Petersburg University shall:
  - 4.3.1. be evaluated using 100-point scale;
- 4.3.2. be held either in Russian or in Russian and a foreign language, or in a foreign language (depending on the discipline of the admissions test and language of instruction of the implemented degree programme) in accordance with the information publicised in the manner prescribed by clause 1.6. herein;
- 4.3.3. contain elements of testing the creative (professional) skills and/or knowledge of applicants in the written form and/or provide audio- or video recording of the admissions test;
  - 4.3.4. be held using distance teaching and learning technologies exclusively.
- 4.4. Within admissions to degree programmes of higher education, not later than 1 November 2023 the Admissions Committee shall announce the minimum score of points

required for successful completion of each admissions test for every separate degree programme or for a totality of degree programmes with a unified competitive selection.

- 4.5. The schedule of all admissions tests for degree programmes of higher education (with the subject, date, time, place of examination, counselling sessions (if any), results announcement day) shall be approved by the Admissions Committee and made available to the applicant not later than 1 June 2024. The schedule shall not indicate information of admissions tests board members. The results of admissions tests shall be announced within the time appointed by the admissions tests schedule.
- 4.6. Persons who have failed to undergo admissions tests due to favourable excuse (illness or other circumstances confirmed by appropriate documents) shall be allowed to undergo respective admissions tests during next admissions test period in case the admissions tests are held as the groups are formed or before the admissions tests final deadline, or on the admissions tests reserved day based on the filed with the Admission Committee written application accompanied by the confirming evidences. The application with the evidence shall be submitted before the deadline established by these Regulations. The applicant is allowed to undergo only one admissions test on one admissions test reserved day.
- 4.7. If admissions tests for the chosen type and/or condition of enrolment to degree programmes of higher education are organised in several streams, the applicant may take each admission test only once, i.e. in one of the streams.
- 4.8. All admissions tests shall be held in compliance with these Regulations for Admissions Tests for Degree Programmes of Higher Education at St Petersburg University in 2024 as outlined in Annex 1 hereto.
- 4.9. St Petersburg University shall ensure that the applicants' admissions test papers be accepted for the evaluation depersonalised so that their authorship cannot be determined.
- 4.10. Should they disagree with the received score of points for their admissions tests or should they suspect the Regulations have been violated during admission testing, the applicant shall have the right to file a reasonable written appeal (an appeal) with the St Petersburg University Appeal Panel in the manner prescribed by these Regulations for reviewing written works and filing appeals, which constitutes Annex 2 hereto.

#### 5. Applicants' Individual Achievements Evidence Recognition

- 5.1. Applicants have the right to provide within the deadline, established in clauses 7.22, 8.9 8.10 of these Regulations, information about their individual achievements, which is to be recognised for the purpose of enrolment.
- 5.2. Individual achievements shall be recognised and scored by points for individual achievements to be added to the competition score of points or to be accounted to in case other competitive selection criteria are equal.
- 5.3. Applicants' individual achievement evidence list and recognition procedures are specified:
- 5.3.1. in case of applying for bachelor's degree or specialist's degree programmes, in Chapter 7 of the Regulations for Admissions;
- 5.3.2. in case of applying for master's degree programmes, in Chapter 8 herein and Annex 3 hereto.
- 5.4. Within admissions to master's degree programmes regarding a particular educational programme, recognised can be results of mastering any previous educational programmes as an individual achievement (degree diploma grade average point and graduation project grade). In such a case, only one education credential shall be considered for the purpose of scoring points for individual achievements within the competition for a field of study. The calculation of the score average shall be carried out based on the supplement to the standard credential. If the supplement to the standard credential is missing, the score average for a particular applicant for the purpose of ranking shall be zero (0).

5.5. In case applicants submit false evidences of their individual achievements or documents with deliberately false information, they bear responsibility in accordance with the legislation of the Russian Federation. If such facts have been detected, applicants shall be excluded from the competition.

#### 6. Enrolment Procedures

- 6.1. Results of the admissions testing (if applicable) are used to form a ranking list of applicants within each competitive selection (hereinafter referred to as the competition list):
- 6.1.1. competition lists of applicants applying for studying in bachelor's degree or specialist's degree programmes without taking admissions testing to the government-funded places and to the places under contracts for chargeable educational services;
- 6.1.2. competition lists of applicants who have successfully completed the admissions tests for each competitive selection.
  - 6.2. The competition lists shall indicate:
- 6.2.1. the Individual Insurance Number of Personal Ledger Account or a unique code assigned to the applicants (in case the Individual Insurance Number of Personal Ledger Account is not available); for persons applying to study in bachelor's degree or specialist's degree programmes to the places within a special quota just a unique code;
- 6.2.2. regarding each applicant applying for the enrolment to bachelor's or specialist's degree programme without taking admissions tests (except those who apply for the enrolment within a special quota):
  - 6.2.2.1. the grounds for enrolment without taking admissions tests;
  - 6.2.2.2. points scored for individual achievements;
  - 6.2.2.3. grounds for preferential enrolment rights;
- 6.2.3. regarding each applicant applying for the enrolment to bachelor's or specialist's degree programme without taking admissions tests, those applying for the enrolment within a special quota:
- 6.2.3.1. the total number of points (scored for additional admissions tests of creative and/or professional nature and for individual achievements) (if additional tests of creative and/or professional nature are held)
- 6.2.3.2. the total number of points scored for additional admissions tests of creative and/or professional nature (if applicable);
- 6.2.3.3. the number of points scored for each additional admissions test of creative and/or professional nature (if applicable);
  - 6.2.3.4. points scored for individual achievements;
  - 6.2.3.5. grounds for preferential enrolment rights;
  - 6.2.4. per each applicant based on the results of admissions tests:
- 6.2.4.1. the total number of competition points calculated as the total number of points assigned for each admissions test plus the points assigned for personal achievements;
  - 6.2.4.2. the total number of points scored for admissions tests;
  - 6.2.4.3. points scored for each admissions test;
  - 6.2.4.4. points scored for individual achievements;
- 6.2.4.5. grounds for preferential enrolment rights (regarding bachelor's degree and specialist's degree programmes);
  - 6.2.5. within admission to the government-funded places:
- 6.2.5.1. availability of the standard credential original submitted to St Petersburg University (confirming UPSMS marks);
- 6.2.5.2. availability of the consent for enrolment with the explanation of reasons why it is impossible to submit the standard credential original for persons specified in clauses 7.2.3.6, 7.2.3.7 herein;

- 6.2.6. within admissions to places under contracts for chargeable educational services the availability of the signed contract for chargeable educational services at St Petersburg University;
  - 6.2.7. enrolment priority.
- 6.3. The competition lists shall indicate neither first names, nor middle names, nor family names of the applicants.
- 6.4. The competition lists of the applicants for enrolment shall be publicised on the St Petersburg University website in the manner prescribed by clause  $\pi$ . 1.6 of the Regulations, including UPSMS (if applicable) and updated if necessary at least five (5) times before the day the orders on enrolment based on each competitive selection are issued.
- 6.5. Enrolment to degree programmes of higher education to the respective type and/or conditions of study shall be carried out based on competitive selection results within admissions quotas or limits for places under contracts for chargeable educational services (depending on the types and conditions specified in the submitted admission application) established for corresponding degree programme of higher education and respective type and/or condition of study.
- 6.6. Enrolled shall be those who have participated in the competition for a degree programme of higher education in a main field of study in a respective type and/or conditions of enrolment and successfully completed admissions tests with the maximum points obtained for the results of admissions tests and for individual achievements; or those who are eligible, according to the laws of the Russian Federation, to enrolment without taking any admissions tests if such persons, in the manner prescribed by these Regulations:
- 6.6.1 within admissions to the government-funded places in bachelor's or specialist's degree programmes have provided one of the following:
- 6.6.1.1 the original of the standard credential in case they have failed to provide the consent for procession personal data or provided the confirming UPSMS mark (if the document information has been confirmed by FREC or St Petersburg University);
- 6.6.1.2. the consent for enrolment with the explanation of reasons why it is impossible to submit the standard credential original for persons specified in clauses 7.2.3.6, 7.2.3.7 herein;
- 6.6.2 within admissions to the government-funded places in master's degree programmes have provided one of the following:
- 6.6.2.1 the original of their standard credential or provided the confirmation mark in the information system of St Petersburg University about the submission of the original of their standard credential to St Petersburg University (by doing so in the information system of St Petersburg University, they confirm that they have submitted no originals of their standard credentials to other organisations for enrolment to master's degree programmes on government-funded places and provided no corresponding confirming mark in information systems of other organisations);
- 6.6.2.2. the consent for enrolment with reasonable explanation why it is impossible to submit the standard credential original for persons specified in clauses 7.2.3.6 of these Regulations;
- 6.6.3 within admissions to the places under contracts for chargeable educational services in bachelor's degree or specialist's degree programmes, they have provided the signed contract for chargeable educational services; at the same time the standard credential information is confirmed by confirming UPSMS mark or by St Petersburg University, or St Petersburg University possesses a duly certified copy of the standard credential submitted by the applicant (a copy certified by St Petersburg University based on the respective original submitted by the applicant), or, within admissions of the persons specified in clauses 7.2.3.6, 7.2.3.7 of these Regulations, there is a copy of the standard credential;
- 6.6.4 within admissions to places under contracts for chargeable educational services the availability of the signed contract for chargeable educational services;

- 6.7. Applicants applying for the admission to government-funded places shall be enrolled in accordance with the highest priority of enrolment within which they are eligible to be enrolled based on the competitive selection to the specified places. Applicants applying for the admission to the places under contracts for chargeable educational services shall be enrolled within one or several enrolment priorities. Enrolment to the places under contracts for chargeable educational services shall be conducted regradless enrolments to the government-funded places.
- 6 .8. Enrolment to degree programmes of higher education at St Petersburg University shall proceed under the order issued by the authorised official based on the decision of the Admissions Committee upon the completion of admissions for the respective degree programmes.
- 6.9. Applicants applying for admission the degree programme of higher education can be enrolled to St Petersburg university to the government-funded places only in one degree programme and only under one type of study.
- 6.10. Enrolments shall be conducted based on the enrolment priorities specified in admission applications and competition lists until all vacant places are filled in.
- 6.11. Applicants shall be informed about their enrolment by making enrolment lists within each competitive selection available in the manner prescribed by clause 1.6 hereof. The above mentioned lists shall contain either the Individual Insurance Number of Personal Ledger Account (if applicable) or the unique code assigned to the applicant (if the Individual Insurance Number of Personal Ledger Account is not available). Concerning persons admitted within the limits of the special quota their unique code and the total score of competitive selection points, the points for completed admissions tests and the points for their individual achievements. First names, middle names, and family names of the applicants shall not be indicated in the lists.

# Section 2. The Particulars of Admissions to Degree Programmes of Higher Education at Various Levels

### 7. Particulars of Admissions to Bachelor's Degree and Specialist's Degree Programmes

- 7 .1. It is allowed to admit persons to study at bachelor's degree or master's degree levels who possess credentials of secondary (complete) general education, or of secondary vocational education and a qualification, or of higher education and a qualification. Credentials of primary vocational education confirming secondary (complete) general education or credentials of primary vocational education based on the secondary (complete) general education shall be equalled to the credentials of secondary vocational education and a qualification. Person with degree documents and qualification credentials shall be admitted to master bachelor's degree or specialist's degree programmes only on the places under contracts for chargeable educational services unless stipulated by the legislation of the Russian Federation provision for mastering degree programmes on the government-funded places by persons with degrees.
- 7 .2. The admissions to bachelor's degree and specialist's degree programmes at St Petersburg University shall be held in the following manner:
- 7.2.1. on the basis of the 100-point scale USE results obtained in general education subjects correspondent to the degree programme to which the admission is held;
- 7.2.2. on the basis of the results of the centralised testing held in the Republic of Belarus and/or centralised testing, taken by Belarus citizens in 2024 or 2023; at the same time:
- 7.2.2.1. the results of the centralised testing (exam) are being recognised as the results for general education-based admissions tests if the applicant has taken no USE in the corresponding general education subject in the year when the centralised testing certificate was obtained;
- 7.2.2.2. the 100-point scale results for the centralised testing (exam) is being recognised as the result of the admissions test the points for which has been converted into the 100-point scale;
- 7.2.2.3. the congruity between the centralised testing (exam) and general education-based admissions tests has been set out in Annex 5 hereof;
- 7.2.2.4. to enter the competitive selection for bachelor's degree or specialist's degree programmes that implies admissions testing based on general education subjects such as 'Literature', 'History', Informatics and ICT', which has no congruent subjects within the centralised testing (exam) and when there is no availabilities of other optional subjects, Belarus citizens must provide results for the specified subjects of general education or, as provided by sub-clause 7.2.3.2 of these Regulations, must take admissions tests held at the discretion of St Petersburg University in these subjects;
- 7.2.3. on the basis of the results of admissions tests held at the discretion of St Petersburg University in general education subjects, which the following categories of citizens are eligible to take:
- 7.2.3.1. applicants with a disability or additional support needs, including disable children;
  - 7.2.3.2. international applicants;
- 7.2.3.3. applicants with education credentials obtained from an international organisation (with general education disciplines the applicants have not undertaken USE in 2024);
- 7.2.3.4. applicants with degrees (for applicants applying for admission on the basis of secondary vocational education, the USE shall be considered as admissions test, except instances specified in sub-clause 7.2.5.1 of these Regulations);
- 7.2.3.5. persons applying for the admission within the special quota and specified in clause 7.10.2 of these Regulations (the admissions test results obtained by applicants under the provision of the clause herein, which at the same time are not citizens specified in clauses 7.2.3.1 7.2.3.3 of these Regulations, shall not be recognised for admissions to the places

within the special quota, to the places within the sponsored quota, to the main places within admission reference quota, to the places under contracts for chargeable educational services);

- 7.2.3.6. persons who have completed mastering general secondary education programmes and successfully competed the final state assessment in the territory of the Donetsk People's republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region before they have been accepted to the Russian Federation;
- 7.2.3.7. persons who have passed the final state assessment in general secondary educational programmes under special conditions, provided by Federal Law No 19-FZ for persons:
- 7.2.3.7.1. being trained in organisations implementing educational activities located in the territory of the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region;
- 7.2.3.7.2. having been trained in organisations implementing educational activities located in the territory of the Donetsk people's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region, and from 2021/22 academic year admitted to study in organisations implementing educational activities in the state-accredited educational programmes of the basic general and secondary general education located in the territory of the Russian Federation (except territories of the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region before they have been accepted to the Russian Federation);
- 7.2.4. upon the results of additional practical creative admissions tests and/or additional profession-oriented admissions tests in degree programme requiring applicants to have either particular creativity or physical and/or psychological characteristics;
  - 7.2.5. based o the results of the vocational education-based admissions tests:
- 7.2.5.1. according to the syllabus of the secondary vocational education programme in the speciality 'Nursery', related bachelor's degree in the field of study 34.02.01 'Nursing' (within admission to study in bachelor's degree programme in the field of study 34.03.01 'Nursing' to the places for applicants applying for study on the basis of the secondary vocational education);
- 7.2.5.2. for applicants applying for study in bachelor's degree programme implemented in part-time evening format (for persons with degrees) in the field of study 40.03.01 'Law'.
- 7.3. Certain categories of persons eligible to enter St Petersburg University without taking any admissions tests to government-funded places within the special quota, to government-funded places within the particular quota, to government-funded places within the sponsored quota enjoying a priority enrolment right are specified by the laws of the Russian Federation.
  - 7.4. The right to be enrolled witout taking admissions tests is granted to:
- 7.4.1 winners and awardees of the final round of the All-Russian School Olympiad (hereinafter referred to as the winners and awardees of the All-Russian Olympiad), members of Russian national teams, duly formed in the manner defined by the federal executive agency in charge of the development of the governmental policy and legal regulation for general secondary education, who have participated in international Olympiads in general education subjects (hereinafter referred to as the members of Russian national teams) in specialities and/or subject areas corresponding to the profile of the All-Russian School Olympiad within four (4) years following the year of the relevant Olympiad;
- 7.4.2 winners and awardees of Olympic Games, Paralympic Games and Deaflympic Games, world champions, European champions, first-place finishers in the world or European championships in sports admitted to the Olympic Games, Paralympic Games and Deaflympic Games, to specialities and/or fields of study in physical culture and sports;
- 7.4.3 persons recognised as citizens of the Russian Federation, as well as persons constantly residing in the territory of the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region before they have been accepted to the

Russian Federation, who are Stage 4 winners or awardees of the all-Ukrainian School Student Olympiad, a republican Olympiad of school students of the the Donetsk People's Republic, the Lugansk People's Republic, members of national teams of the Ukraine, composed in the manner prescribed by the legislation of the Ukraine, and participated in international Olympiads in general education subjects - within four (4) years following the year of the relevant Olympiad.

7.5. The correspondence of the All-Russian School Olympiad profiles, international Olympiads in general education subjects to the field of study and specialities to which St Petersburg University conducts the admission within degree programmes, is defined in Annex 6 hereto. The correspondence of the fields of study and specialities to which St Petersburg University conducts admission, to the All-Ukrainian School Student Olympiad profiles, international Olympiads in general education subjects is defined in Annex 8 to these Regulations. The correspondence of the fields of study and specialities to which St Petersburg University conducts admission, to the profiles of republican school student Olympiads of the Donetsk People's Republic, the Lugansk People's Republic, is defined in Annex 9 to these Regulations.

Persons specified in clauses 7.4, 7.6.1 of these Regulations, in the course of four (4) years following the year of the respective Olympiad, shall be granted the particular right, determined in sub-clause 7.6.2 of these Regulations if the general education-based admissions test corresponds to the Olympiads profile or physical culture and sports area. The correspondence of the profiles of the All-Russia School Student Olympiad, republican Olympiads of the the Donetsk People's Republic, the Lugansk People's Republic, international Olympiads in general education subjects to general education-based admissions tests is defined in Annex 4 to these Regulation. The correspondence of the profiles of the all-Ukrainian School Student Olympiad, international Olympiads in general education subjects to the general education-based admissions tests is defined in Annex 10 hereto. The correspondence of the school student Olympiad profiles held in accordance with the Procedures for School Olympiads (hereinafter referred to as 'the Procedures for School Olympiads') adopted by the Ministry of Education of the Russian Federation to the general education admissions tests has been set forth by the Procedure of providing particular rights to the winners and awardees of school Olympiads held in accordance with the Procedures for School Olympiads (Annex7 hereto).

- 7.6. By the decision of the Admissions Committee responsible for the admissions to bachelor or specialist programmes, the winners and awardees of school Olympiads held in 2023/2024, 2022/2023, 2021/2022, 2020/2021, 2019/2020 school years and included in the list of the school Olympiads for the relevant school year, approved by the Ministry of Education of the Russian Federation may enjoy one of the following particular rights (only those who were in 10th or 11th year students at the time of the Olympiad):
- 7.6.1 to be enrolled without any admissions tests to a bachelor's (specialist's) degree programme in the field of study corresponding to the Olympiad's profile; the correspondence of degree programmes (fields of study, specialities) implemented by St Petersburg University to the Olympiad's profile shall be defined by the decision of the Admissions Committee;
- 7.6.2 to be equalled to the persons with maximum grades for the Unified State Examination in any general education subject corresponding to the Olympiad's profile or to the persons who have successfully completed practical creative admissions test and/or profession-oriented admissions test in the manner established by the Admissions Committee.
- 7 .7. Decisions of the Admissions Committee mentioned in clauses 7.6.1. and 7.6.2. herein are outlined in Annex 7 to these Regulations.
- 7.8. The following categories of applicants shall be eligible for the admission to degree programmes of higher education to the government-funded places within the limits of the particular quotas: orphaned children or children deprived of parental care as well as persons coming from the category of orphaned children or children deprived of parental care, disabled

children, 1st and 2nd group of disability, those who are disabled from childhood, disabled combatants due to traumas or diseases resulted from their military service, and combat veterans specified in Sub-clauses 1 - 4, Clause 1, Article 3 of Federal Law No 5-FZ 'On Veterans' dated 12 January 1995.

- 7 .9. The right to be enrolled to bachelor's degree and specialist's degree programmes to the government-funded places within the particular quota is granted to:
  - 7.9.1. Heroes of the Russian Federation, persons awarded three Orders of Courage;
- 7.9.2 citizens who are (were) in the military service in the Armed Forces of the Russian Federation, citizens who are (were) in the military service of the Federal service of the National Guard Forces of the Russian Federation, in military troops and bodies, specified in clause 6 Article 1 of federal Law No 61-FZ 'On Defence' dated 31 May 1996 provided they participated in the Special Military Operation in the territory of the Ukraine, the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region and/or accomplished tasks in countering military invasion to the territory of the Russian Federation in the course of the armed provocation at the state border of the Russian Federation and border territories of the constituent entities of the Russian Federation, adjacent to the regions of the Special Military Operation in the territory of the Ukraine, he Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region; situated (were situated) in the specified territories officers (employees) of the law-enforcement bodies of the Russian Federation; citizens accomplishing (having accomplished) official and other similar functions in the specified territories;

7.9.3 citizens who were called for military service by mobilisation in the Armed Forces of the Russian Federation, citizens who have entered contracts for voluntary assistance in carrying out tasks assigned to the Armed Forces provided they participated in the Special Military Operation in the territory of the Ukraine, the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region and/or accomplished tasks in countering military invasion to the territory of the Russian Federation in the course of the armed provocation at the state border of the Russian Federation and border territories of the constituent entities of the Russian Federation, adjacent to the regions of the Special Military Operation in the territory of the Ukraine, he Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region; citizens who have entered contracts (or having other legal relations) with an organisation facilitating the accomplishing of tasks assigned to the Armed Forces of the Russian Federation provided they participated in the Special Military Operation in the specified territories;

7.9.4 persons who, under the decisions of the public authorities of the Donetsk People's Republic, the Lugansk People's Republic, participated in military operations as part of the Armed Forces of the Donetsk People's Republic, the People's Militia of the Lugansk People's Republic, military troops and bodies of the Donetsk People's Republic, the Lugansk People's Republic starting from 11 May 2014;

- 7.9.5 children of the persons specified in clauses 7.9.2 7.9.4 of these Regulations;
- 7.9.6 children of military service persons, employees of the federal executive bodies and federal public agencies, which implies military service; employees of the Ministry of Internal Affairs; employees of the penal system of the Russian Federation, commissioned to other states by public authorities of the Russian Federation and participated in military operations in the course of performing their service in these states;
- 7.9.7 children of health care providers diseased in a result of infection with a new coronavirus infection (COVID-19) while performing their work duties within admissions to degree programmes of medical and pharmaceutical education.
  - 7.10. Enrolled to the places within particular quota:
- 7.10.1 without taking admissions tests (except additional admissions tests of creative and/or professional nature) shall be:
  - 7.10.1.1. Heroes of the Russian Federation, persons awarded three Orders of Courage;

- 7.10.1.2 children of health care providers diseased in a result of infection with a new coronavirus infection (COVID-19) while performing their work duties within admissions to degree programmes of medical and pharmaceutical education;
- 7.9.5 children of the persons specified in clauses 7.9.2 7.9.4 of these Regulations, children of military service persons specified in clauses 7.9.6 of these Regulations if the specified persons, military service persons, employees diseased or became injured (a wound, an injury, a concussion) or ill in the course of performing military service duties (official duties) in the course of the Special Military operation (military operations in the territories of foreign states), or awarded the title of Hero of the Russian Federation or three Orders of Courage;
- 7.10.2 upon results for USE or admissions test held at the discretion of St Petersburg University, based on the applicants' choice, persons applying for admission to study on the places within the particular quota not mentioned in clause 7.10.1 of these Regulations.
- 7 .11. The following persons are eligible for preferential enrolment when applying for admission to bachelor's and specialist's degrees programmes:
- 7.11.1 orphaned children and children deprived of parental care, as well as persons coming from the category of orphaned children or children deprived of parental care;
  - 7.11.2 disabled children, persons with disability group 1 or 2;
- 7.11.3 individuals under 20 (twenty) years of age with only one parent disability group 1, if the family average income is lower than the minimum subsistence level specified for the constituent entity of the Russian Federation where these person reside;
- 7.11.4 individuals who were exposed to radiation as a result of the disaster at the Chernobyl Nuclear Power Plant and are eligible to enjoy the rights under Federal Law No 1244-1 'On Social Protection of Individuals Exposed to Radiation as a Result of the Disaster at the Chernobyl Nuclear Power Plant' dated 15 May 1991;
- 7.11.5 children of military combatants killed on duty or dead as the result of an injury (a wound, a trauma, a concussion) or diseases caused by their military service, including participation in counter-terrorist operations and/or other anti-terrorism efforts;
- 7.11.6 children of deceased (fallen) Heroes of the Soviet Union, Heroes of the Russian Federation and Full Cavaliers of the Order of Glory;
- 7.11.7 children of employees of law enforcement agencies, Federal Service of the National Guard Forces of the Russian Federation, institutions and bodies of the penal enforcement system, Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters, drugs and psychotropics substances control agencies, customs agencies, Investigative Committee of the Russian Federation who have fallen (died) as the result of an injury or other health damage received by them in line of duty or as the result of a disease received during their service in the said institutions and agencies, as well as children having been dependent upon them;
- 7.11.8 children of public prosecution officers, who have fallen (died) as the result of an injury or other health damage received by them during their public prosecution service or after dismissal as a result of personal injury related to their employment activity;
- 7.11.9 individuals on contractual military service whose term of continuous military service is three years or more, as well as conscripts with recommendations for admission from their commanders issued in the manner set forth by the federal executive authority or the federal state authority which implies military service under federal legislation;
- 7.11.10 individuals who have been on contractual military service for three years or longer in the Armed Forces of the Russian Federation, other troops, military units and bodies, appointed at military posts and dismissed from military service at the grounds specified in Subclauses 'b' –'d' of Clause 1, Sub-clause 'a' of Clause 2 and Sub-clauses 'a' 'c' of Clause 3 of Article 51 of Federal Law No. 53-FZ 'On Military Duty and Military Service' dated 28 March 1998;

- 7.11.11 disabled war veterans, combatant as well as combat veterans belonging to persons specified in Sub-clauses 1-4 of Clause 1 of Article 3 of Federal Law No. 5-FZ 'On Veterans' dated 12 January 1995;
- 7.11.12 individuals who personally participated in nuclear weapon, combat radioactive substances in the atmosphere, and underground nuclear weapon testing, in manoeuvres using such armoury and combat radioactive substances before the date of the actual ending of the mentioned tests and manoeuvres, direct participants in the elimination of radiation accidents at nuclear facilities on surface vessels and submarines and other military sites, direct participants in carrying out work in collecting and disposing radioactive substances, as well as direct participants in eliminating consequences of accidents (military personnel and civilians employed by the Armed Forces of the Russian Federation, military personnel of the Ministry of Internal Affairs of the Russian Federation or federal state authorities, military personnel and employees of the National Guard Forces, individuals who served in rail road troops and other military formations, employees of the law enforcement agencies of the Russian Federation, and the Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters);
- 7.11.13 military personnel, employees of the Federal Service of the National Guard Forces of the Russian Federation, employees of law enforcement agencies of the Russian Federation, penal enforcement system, the Ministry of the Russian Federation for Civil Defence, Emergency Situations and Liquidation of Consequences of Natural Disasters who have accomplished missions in the conditions of the armed conflict in the Chechen Republic and adjacent territories belonging to the armed conflict area and the said military personnel accomplishing missions during counter-terrorist operations in the North Caucasus region;
- 7.11.14 individuals who have graduated from educational organisations, vocational educational organisations under the supervision of the federal state bodies and implementing non-degree programmes that aim to train junior students to military or other public service;
  - 7.11.15 persons specified in clauses 7.9.1, 7.9.5, 7.9.6 of these Regulations.
- 7.12. As provided by the legislation of the Russian Federation, individuals eligible for admission to study at St Petersburg University without taking any admissions tests and specified in clauses 7.4 and 7.6.1 of these Regulations, within the admission to bachelor's degree or specialist's degree programmes on the government-funded places, can exercise the provided right of being admitted to only one educational institution of higher education and to only one degree programme upon their choice (regardless the number of evidences confirming the respective particular right).
- 7 .13. Applicants in their admission applications must specify one degree programme of the highest priority to which they wish to be admitted under the provided special right. If such persons apply for admission at other educational organisations that implement degree programmes of higher education or to other bachelor or specialist programmes, they shall take part in the general competitive selection according to the Regulations. These special rights within the admissions to bachelor's degree or specialist's degree programmes provided by Federal Law No 273-FZ 'On Education in the Russian Federation" dated 29 December 2012 shall not apply to compatriots without citizenship of the Russian Federation unless otherwise has been foreseen by international treaties.
- 7 .14. Disabled children, persons of Disability group 1 and 2, lifelong disabled persons; persons disabled as a result of war injuries or diseases received during their military service eligible for admission on a government-funded basis within the established quota, shall submit, at their discretion, either the original or a copy of the statement of physical disability.
- 7 .15. Orphaned children, children deprived of parental care, as well as those from the category of orphaned children and children deprived of parental care, war veterans from the category specified in sub-clauses 1 4 Clause 1 Article 3 of Federal Law No 5-FZ 'On Veterans' dated 12 January 1995 entitled for the enrolment to government-funded places within the established quota must provide a copy of the document confirming their corresponding status.

- 7.16. Persons eligible for enrolment to study at St Petersburg University without taking admissions tests as provided for by the legislation of the Russian Federation must provide a document confirming their eligibility for such a particular right when submitting their admission applications.
- 7 .17. Persons entering the competitive selection within the special admission quota or within the sponsored admission quota must provide a document confirming their eligibility for respective right for the admission within a quota when submitting their admission applications.
- 7.18. Persons entering the competitive selection within the particular quota must indicate the fact that they belong to the number of persons specified in clause 7.9 of these Regulations in their admission applications and within the deadline defined in clause 7.41.2.1 of these Regulations must provide the original of the document issued by an authorised public agency (organisation) that confirms the right to be enrolled to study within the particular quota.
- 7 .19. Persons eligible for the priority right must provide a document confirming their relevant right when submitting their admission applications.
- 7 .20. Following the Procedures for School Olympiads, the winners and awardees of the 2023/2024, 2022/2023, 2021/2022, 2020/2021, 2019/2020 school Olympiads held in accordance with the Procedures for School Olympiads and included into the List of School Olympiads duly approved by the Russian Federation Ministry of Education are eligible to submit, when applying for admission, an electronic image of the school Olympiad winner's (awardee's) diploma, together with its original or a copy, formed by services of the Russian Council of School Olympiads official portal.
- 7.21. To take admissions tests for the enrolment to bachelor's or specialist's degree programmes conducted at the discretion of St Petersburg University, persons with a disability or additional support needs must provide documents confirming their disability or additional support needs.
- 7.22. The acceptance of personal applications for entering the competition and other documents from persons applying for admission to bachelor's and/or specialist's degree programmes starts on **June 20, 2024** and ends on:
- 7.22.1. **7 July 2022,** for persons applying for the admissions to degree programmes requiring completion of practical creative admissions tests and/or profession-oriented additional admissions tests;
- 7.22.2. **10 July 2022,** for persons applying for admissions based on the results of admissions tests held at the discretion of St Petersburg University;
- 7.22.3. **25 July 2024**, for persons applying for admissions to study without taking admissions tests held at the discretion of St Petersburg University, including those applying for the admission without taking admissions tests and those applying for the admission without taking admissions tests to the places within the particular quota (in case there are no additional admissions tests of the creative and/or professional nature);
- 7.22.4. 10 August 2024, for persons applying for admissions to study on the places under contracts for chargeable educational services in the fields of study (specialities) to which admissions are conducted based only on the results for USE, as well as for those who submitted their admission application and documents within the deadline established by clauses 7.22.1 and 7.22.2 of these Regulations only to government-funded places and successfully completed additional admissions tests of creative and/or professional nature or other admissions tests, held at the discretion of St Petersburg University (accepted shall be applications to the degree programmes from those applicants who have successfully completed their admissions tests for the similar programmes).
- 7 .23. Within the admissions to the government-funded places and to the places under contracts for chargeable educational services, applicants should submit two admission applications: in one of the two they should indicate competitive selections for the government-funded places and in another one competitive selections for the places under contracts for chargeable educational services.

- 7 .24. Applicants shall have the right to enter competitions for bachelor's or specialist's degree programmes in as many as three fields of study or specialities.
- 7 .25. Applicants can submit their documents required for entering the competition as follows:
  - 7.25.1 in person;
  - 7. 25. 2 through general post service operators;
- 7.25.3 electronically, through the applicants' Personal Account on the St Petersburg University official site.
- 7 26. Should the applicant for bachelor's degree or specialist's degree programme(s) choose to submit the specified documents electronically, the documents must be scanned and attached to the registration form through the applicant's Personal Account:
- 7.26.1 a digital image of an identity document with the indication of their citizenship (the following pages must be scanned: the applicant's photograph, family name, first name, middle name, gender, date and place of birth, citizenship, document series and number, information about the issuing authority and the applicant's place of registration);
- 7.26.2 a digital image of the admission application for the relevant degree programme containing the applicant's handwriting signature;
- 7.26.3 the applicant's photograph (taken in the current calendar year, with vertical resolution of at least 800 pixels and horizontal resolution of at least 600 pixels);
  - 7.26.4 the electronic image of their standard education credential;
- 7.26.5 a digital image of the document confirming their registration in the system of the individual (personified) accounting (if available);
- 7.26.6 digital images of the documents confirming the applicant's special rights and/or privileges when applying for the admission (if available);
- 7.26.7 digital images of the documents confirming the applicant's individual achievements (if available)
  - 7.26.8 other documents required by the Regulations.
- 7 27. An electronic image of a document in cross-platform open format for electronic documents PDF (PortabelDocumentFiles) must ensure the visual identity with its original on 1:1 scale. The quality of the provided digital images must ensure legibility of the entire document and their details. If the hard copy of any document consists of two or more pages, its digital image must be provided in one file. It is advisable to make scans of documents with the resolution of 300 dpi. It is not allowed to provide illegible scanned images of the documents as well as images that lack significant sections of the document (texts, signatures, seals or stamps, etc.).
- 7 28. The applicant applying for the admission to bachelor or specialist programmes at St Petersburg University must provide the original of either document as their education credential as specified in Clause 7.1. herein taking into account provisions of Clause 4 of the Regulations.
- 7 29. The admissions tests results have to at least be equivalent to the reference score of points specified by the Admissions Committee for passing admissions tests in respective subjects.
- 7 30. If there are several valid evidences of the admissions tests results as provided by clause 7.2 of these Regulations, the applicant's highest score obtained for admissions tests, which constitutes at least the minimum passing score of points, shall be recognised as the result for admissions tests.
- 7 31. The Admissions Committee shall appoint at least three (3) admissions tests from the List of the admissions tests for pursuing higher education degree in bachelor's or specialist's degrees programmes approved by the Ministry of Science and Higher Education of the Russian Federation (hereinafter referred to as the List of admissions tests), including a mandatory admissions test in Russian and an admissions test in the general education discipline defined in

the List of admissions tests for the respective degree programme (speciality) as the mandatory one.

- 7 32. Should there additional practical creative and/or profession-oriented admissions tests be held in respective degree programmes open for admission, the Admissions Committee shall appoint at least two (2) admissions tests from the List of the admissions tests.
- 7 33. Admissions test held at the discretion of St Petersburg University shall start **after 8 July 2024** and end on **25 July 2024**. Additional admissions tests shall be held in several stages as the admissions test groups of applicants, who have submitted required documents, are formed. Applications for taking admissions tests on the reserved test day shall be submitted by the applicants before 12 am (Moscow Standard Time) on 20 July 2024.
- 7 34. An applicant shall take the general education admissions test held at the discretion of St Petersburg University only once regardless the degree programme, the type (full-time, evening part-time) or the condition (government-funded places or places under contracts for chargeable educational services) of study they apply for. The obtained results shall be recognised within admissions to all degree programmes and types and conditions of study for which competition the applicant has submitted their application, except the results for admissions tests, taken in the manner provided by clause 7.2.3.5 of these Regulations, that shall not be recognised within admissions to the places within the special quota, to the places within the sponsored quota, to the main places within the reference quota, to the places under contracts for chargeable educational services.
- 7 35. An applicant shall take the additional practical creative or profession-specified admissions test set for the respective degree programme and held at the discretion of St Petersburg University only once regardless the degree programme, the type (full-time, evening part-time) or the condition (government-funded places or places on the fee-paying basis paid for by individuals or legal entities) of study they apply for. The resulting scores obtained for practical creative or profession-specified admissions tests in different disciplines passed when applying for one degree programme shall not be recognised for resulting scores when applying to other degree programmes.
- 7 36. When admitting persons to bachelor's and specialist's degrees programmes, St Petersburg University shall assign points for the following individual achievements:
- 7.36.1 for the status of a winner and awardee of Olympic Games, Paralympic Games and Deaflympic Games, a world champion, a European champion, a first-place finisher in a world or European championship in sports admitted to the Olympic Games, Paralympic Games and Deaflympic Games 10 points;
- 7.36.2 for being in military service under the conscription, as well as in military service under the contract, in military service under mobilisation to the Armed Forces of the Russian Federation, being in volunteer troops under the contract for voluntary facilitating in accomplishing tasks assigned to the Armed Forces of the Russian Federation in the course of the Special Military Operation in the territory of the Ukraine, the Donetsk People's Republic, the Lugansk People's Republic, Zaporozhye Region and Kherson Region 10 points;
- 7.36.3 for the status of a world champion, a European champion, a first-place finisher in a world or European championship in sports not admitted to the Olympic Games, Paralympic Games, or Deaflympics 5 points;
- 7.36.4 for a gold, silver, or bronze insignia of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO) (hereinafter respectively referred to as GTO badge, GTO complex) received by the applicant in accordance with Order No 16 'On the Procedure for Awarding Persons Who Fulfilled the Test Standards of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO) the Corresponding Insignia of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO)' approved by the Ministry of Sports, Tourism and Youth Policy of the Russian Federation dated 14 January 2016 if the applicant has been awarded a GTO badge for the fulfilment of the GTO complex, according to the Regulations for the All-Russia Physical

Culture and Sports Complex 'Ready for Labour and Defence' (GTO) approved by Decree No 540 of the Government of the Russian Federation dated 11 June 2014 for the age group of the population of the Russian Federation if the applicant belongs(ed) in 2024 and/or 2023 to this age group (the points for the GTO badge shall be scored only once: the presence of a gold and silver, or gold and bronze, or silver and bronze badge at the same time shall not increase the number of the scored points; the GTO badge shall be confirmed by the GTO badge certificate or information posted on the official website of the All-Russia Physical Culture and Sports Complex 'Ready for Labour and Defence' (GTO) on the Internet, or by a copy of the order (extracts from the order) of the Ministry of Sports, Tourism and Youth Policy on awarding a gold GTO badge certified by a respective official, or by a copy of the order (extracts from the order) on awarding a silver or a bronze GTO badge of a public executive body of the entity constituting the Russian Federation) - 2 points;

7.36.5. for other sports achievements:

7.36.5.1. for the membership in a Russian national Olympic team in 2023 and/or in 2024 (the main, under-21, junior and youth teams as well as the reserve squad)— **5 points**;

7.36.5.2. for the membership in a Russian national non-Olympic team in 2023 and 2024 (the main, under-21, junior and youth teams as well as the reserve squad)— **3 point**;

7.36.5.3. for the insignia 'Unarmeyskoye Valour', for a certificate to the badge 'Unarmeyskoye Valour' issued in accordance with the Regulations for the procedure for registering, issuing, and awarding the badges of 'Unarmeyskoye Valour' of three degrees of VVPOD 'UNARMIA' approved by the decision of the General Staff of the VVPOD 'UNARMIA' (Minutes No 29 dated 27 May 2021, for the first, second or third place in the All-Russia and international stages of military sports games, sports Olympiads - 1 point;

7.36.5.4. for the status of a winner of the All-Russia military and sports game 'Pobeda' - **1 point**;

7. 36. 6 for the presence of the obtained by individuals, specified in clauses 7.2.3.6 and 7.2.3.7 of these Regulations, education credentials or education credentials plus qualification credentials with distinction (a certificate of secondary general education with distinction, a certificate of secondary (complete) general education with distinction, a certificate of secondary (complete) general education for the gold (silver) medal awardees; for a diploma of secondary vocational education with distinction, a diploma of primary vocational education with distinction, a diploma of primary vocational education for gold (silver) medal awardees) from educational institutions of the Russian Federation, as well as of the Donetsk People's Republic, the Lugansk People's Republic, and the Ukraine - 3 points;

7.36.7. for the awards from school Olympiads (that grant no particular rights and/or particular privilege when applying for the admission to relevant types and conditions of study) and other intellectual and/or creative contests, or sports events held in compliance with Part 2 Article 77 of Federal Law No 273-FZ 'On Education in the Russian Federation' of 29 December 2012 aimed to identifying and supporting persons presented outstanding abilities:

7.36.7.1 for a diploma of the winner of the regional (3rd) round of the All-Russian School Olympiad in a discipline corresponding to the first or second discipline in the list of admissions tests specified for admission to the relevant bachelor's or specialist's degrees programmes (see Annex 4 hereto to find out disciplines of the All-Russian School Olympiad correspondent to disciplines of the admissions tests) during four (4) subsequent years after the respective Olympiad was held, — 5 points;

7.36.7.2 for a diploma of an awardee of the regional (3rd) round of the All-Russian School Olympiad in a discipline corresponding to the first or second discipline in the list of admissions tests specified for admission to the relevant bachelor's or specialist's degree programmes during four (4) subsequent years after the respective Olympiad was held—4 points;

7.36.7.3 for a diploma of the winner of a school olympiad included in the list of the Ministry of Education during four (4) subsequent years after the respective Olympiad was held - **5 points**;

- 7.36.7.4 for a diploma of an awardee of a school students' Olympiad included in the list of the Ministry of Education during four (4) subsequent years after the respective Olympiad was held **4 points**;
- 7.36.7.5 for a diploma of the winner of the St Petersburg University School Students' Olympiad in the disciplines (sets of disciplines), not included in the current list of the Ministry of Science and Higher Education for the respective academic year, during four (4) subsequent years after the respective Olympiad was held **5 points**;
- 7.36.7.6 for a diploma of an awardee of the St Petersburg University School Students' Olympiad in the disciplines (sets of disciplines), not included in the current list of the Ministry of Science and Higher Education for the respective academic year, during four (4) subsequent years after the respective Olympiad was held **3 points**;
- 7.36.7.7. for a diploma of the winner of the School Olympiad in the History of Engineering held by St Petersburg University and the Central Rail Road Transport Museum of the Russian Federation **5 points**;
- 7.36.7.8. for a diploma of an awardee of the School Olympiad in the History of Engineering held by St Petersburg University and the Central Rail Road Transport Museum of the Russian Federation **3 points**;
- 7.36.7.9 for a diploma of the winner of the School Olympiad in the History of St Petersburg held by St Petersburg University and St Petersburg State Budgetary Institution of Culture 'Museum and Exhibition Centre' (Historical Park 'Russia My History') 5 points;
- 7.36.7.10 for a diploma of an awardee of the School Olympiad in the History of St Petersburg held by St Petersburg University and St Petersburg State Budgetary Institution of Culture 'Museum and Exhibition Centre' (Historical Park 'Russia My History') **3 points**;
- 7.36.11 for a diploma of the winner of the School Olympiad in technopreneurship 'TechoOlympus' held by St Petersburg University and Federal State Budgetary Institution of Additional Education 'Centre for the Development of Creativity and Scientific and Technological Initiatives for Children and Youth' (St Petersburg Quantorium) **5 points**;
- 7.36.7.12. for the status of the winner of the International Contest 'Literary Destinies of Belarus and Russia: From Dialogue to Understanding' held by St Petersburg University with contributions by the representative office of the Federal Agency for Affairs of the Commonwealth of Independent States, Compatriots Residing Abroad and International Humanitarian Co-Operation in the Republic of Belarus **5 points**;
- 7.36.7.13. a diploma of the winner of the All-Russia School Contest 'Bolshaya Peremena' **5 points**;
- 7.36.7.14. a diploma of an awardee of the All-Russia School Contest 'Bolshaya Peremena' **3 points**;
  - 7.36.7.15. a diploma of the winner of the All-Russia School Contest 'SberZ' 5 points;
  - 7.36.7.16. a diploma of an awardee of the All-Russia School Contest 'SberZ' 3 points;
- 7.36.7.17. a diploma of the winner of the Interrigional intellectual contest 'Umnozhaya Talanty' **5 points**;
- 7.36.7.18. a diploma of an awardee of the Interrigional intellectual contest 'Umnozhaya Talanty' **3 points**;
- 7.36.7.19 for a diploma of the winner or an awardee of the All-Russia interdisciplinary Olympiad for school students "Natsionalnaya Tekhnologicheskaya Olimpiada' (not applicable for individual achievements points according to clause 7.36.7.3 or clause 7.36.7.4 of these Regulations) in the course of subsequent four (4) years following the year of the respective Olympiad 2 points;
- 7.36.7.20 for a diploma of the winner or an awardee of the All-Russia Great olympiad 'Iskusstvo Tekhnologii Sport' in the course of four (4) years after the year of the respective Olympiad **2 points**;

- 7.36.7.21 for a diploma of the winner of the federal or regional All-Russia contest 'Istoria Mestnogo Samoupravleniia Moego Kraya' (the History of Local Self-Governence of My Region) 2 points;
- 7.36.7.22 a diploma of the winner of the Special Prokect (contest) 'IT School Casts the Best' 2 points;
- 7.36.7.23 a diploma of the winner of the International Engineering Championship 'CASE-IN' 2 points;
- 7.36.7.24 a diploma of the winner of the All-Russia Contest of research and engineering projects of high school students, lyceums, gymnasiums, and freshmen of secondary specialised institutions of Russia and Commonwealth of Independent States 'The Baltic Research and Engineering Contest' 2 points;
- 7.36.7.25 a diploma of an awardee of the All-Russia Contest of research and engineering projects of high school students, lyceums, gymnasiums, and freshmen of secondary specialised institutions of Russia and Commonwealth of Independent States 'The Baltic Research and Engineering Contest' 1 point;
- 7.36.7.26 a diploma of the winner of the Open Contest in programming, held by Autonomous Non-Commercial Organisation of Additional Professional Education 'Yandex Educational Technologies' **3 points**;
- 7.36.7.27 a diploma of an awardee of the Open Contest in programming, held by Autonomous Non-Commercial Organisation of Additional Professional Education 'Yandex Educational Technologies' 1 point;
- 7.15.7. a diploma of the winner (awardee) of the national and/or international championship in professional skills among persons with a disability or additional support needs 'Abilympics' 10 points;
- 7 37. To the persons with several individual achievements out of those outlined in clauses 7.15.5 and 7.15.7, points shall be assigned for one individual achievement, which ensures assigning more points. The total score of points assigned for individual achievements shall not exceed 10 points. Any information on the applicant's individual achievements (specifying the achievement type and confirmation documents details) shall be given by the applicant in their personal application.
- 7 38. The admissions to bachelor's and specialist's degrees programmes within the admissions reference quotas shall be held by stages:
- 7.38.1 the preferential enrolment stage (enrolment without taking admissions tests to places within the special quota, to places within the sponsored quota, to places within the particular quota);
  - 7.38.2 main stage of enrolment
- 7 39. The lists of applicants eligible for admission without any admissions tests are rated as follows:
- 7.39.1 in accordance with the status of persons entitled to admission without taking any admissions tests in the following order:
  - 7.39.1.1. members of national teams who have participated in international Olympiads;
  - 7.39.1.2. winners of the final round of the All-Russian School Olympiad;
  - 7.39.1.3.awardees of the final round of the All-Russian School Olympiad;
- 738.1.4. winners and awardees of Olympic Games, Paralympic Games and Deaflympic Games, world champions, European champions, first-place finishers in the world or European championships in sports admitted to the Olympic Games, Paralympic Games and Deaflympic Games, who, based on the outlined sport achievements, are entitled for enrolment without taking admissions tests;
- 7.39.1.5. winner of school Olympiads included in the list of the Ministry of Education of the Russian Federation;
- 7.39.1.6. awardees of school Olympiads included in the list of the Ministry of Education of the Russian Federation;

- 7.39.2. For persons specified in each Sun-clauses of clause 7.39.1. of the Regulations in the descending order of the points assigned for individual achievements;
- 7.39.3. If there is the equality of the criteria specified in clauses 7.39.1. and 7.39.2. of the Regulations, the higher place shall be given to the applicants with the preferential enrolment right according to clauses 7.11.1 7.11.13, 7.11.15 of these Regulations;
- 7.39.4. If there is the equality of the criteria specified in clauses 7.39.1. 7.39.3. of the Regulations, the higher place shall be given to the applicants with the preferential enrolment right according to clause 7.11.14 herein;
- 7.39.5. If there is the equality of the criteria specified in clauses 7.39.1. 7.39.4. of the Regulations, the higher place shall be given to those applicants with individual achievements that shall be considered upon the equality of the applicants according to other ranking criteria:
- 7.39.5.1. laureates of the World Contest 'Intel International Science and Engineering Fair' (Intel ISEF);
- 7.39.5.2. persons with certificates confirming their accomplishment of the online courses and online programmes at St Petersburg University;
- 7.39.5.3. persons with certificates confirming their accomplishment of the St Petersburg University non-degree programmes;
- 7.39.5.4. winners of the All-Russia Dictation of Victory at either federal or regional levels;
- 7.39.5.5. winners or awardees of the final stage of 'Russia in a Digital World' School Olympiad;
- 7.39.5.6. winners or awardees of the intellectual talk-show 'Legko Skazat' held by St Petersburg University in the framework of the Federal Dedicated Programme 'The Russian Language';
- 7 40. Ranking of applicants who have successfully completed the admissions tests shall be carried out based on the competition lists in the following order:
  - 7.40.1. in the descending order of the total number of competition points;
- 7.40.2. if the total number of competition points is equal in the descending order of the number of competition points assigned for the results of the admissions tests;
- 7.40.3. if the total number of competition points assigned for the results of the admissions tests is equal in the descending order of the number of points assigned for the results of particular admissions tests, in accordance with the priority ranking established by the Admissions Committee for the admissions tests, which shall be made available when publicising the list of degree programmes open for admissions at St Petersburg University in the manner prescribed by clause 1.6. herein;
- 7.40.4. If there is the equality of the ranking criteria specified in clauses 7.40.1. 7.40.3. of the Regulations, the higher place shall be given to the applicants with the preferential enrolment right according to clauses 7.11.1 7.11.13, 7.11.15 of these Regulations;
- 7.40.5. If there is the equality of the ranking criteria specified in clauses 7.40.1. 7.40.4. of the Regulations, the higher place shall be given to the applicants with the preferential enrolment right according to clause 7.11.14 herein;
- 7.40.6. If there is the equality of the criteria specified in clauses 7.40.1. 7.40.5. of the Regulations, the higher place shall be given to those applicants who have, as provided for by clause 7.38.5. herein, individual achievements that shall be considered if there is the equality of the applicants according to other ranking criteria:
- 7 41. The University enrols applicants to bachelor's or specialist's degree programmes to government-funded places within the following deadlines:
- 7. 41. 1 **27 July 2024** publicising the applicants ranking lists in the manner prescribed by the provisions of clause 1.6 of these Regulations;
  - 7.40.2. The preferential enrolment stage:

- 7.41.2.1. **28 July 2024** (at 12pm Moscow time) the deadline for the activities of persons who are to be enrolled during the preferential enrolment stage as specified in clause 6.6.1 hereof:
- 7.41.2.2. **30 July 2024** issuing and publicising orders on the enrolment of applicants eligible for the enrolment without taking admissions tests to the places within the special admission quotas, to the places within the particular admission quota, to the places within the sponsored admission quota who have abided by the provisions of sub-clause 6.6.1 hereof, and making the information available in the manner prescribed by clause 6.11 of these Regulations;

#### 7.41.3. Main enrolment stage:

- 7.41.3.1. **3 August 2024** (at 12pm Moscow time) the deadline for the activities of persons listed for main competition places as specified by sub-clause 6.6.1 of these Regulations.
- 7.41.3.2. **9 August 2024** issuing and publicising orders on the enrolment of applicants included in the lists of the individuals applying for the admission to main competition places who have abided by the provisions of sub-clause 6.6.1 hereof, and making the information available in the manner prescribed by clause 6.11 of these Regulations;
- 7 42. The University **enrols** applicants applying for the enrolment to bachelor's or specialist's degree programmes to the places **under contracts for chargeable educational services** as follows:
- 7. 42. 1 **12 August 2024** announcing and publicising the lists of applicants, who have passed admissions tests and been ranked in the descending order of the scored points within each bachelor's or specialist's degree programme to the places under contracts for chargeable educational services, in the manner prescribed by clause 1.6 herein;
- 7. 42. 2 **15 August 2024** (at 12pm Moscow time) the deadline for the activities specified in clause 6.6.1 of these Regulations;
- 7. 42. 3 **16 August 2024** issuing the orders on enrolments and making the information available in the manner prescribed by clause 6.11 hereof, on the enrolment of the applicants, who have abided by the provisions of sub-clause 6.6.3 of these Regulations and been included in the list specified in clause 7.41.1 of these Regulations.7.10.1 of these Regulations.
- 7 43. Should the admissions be organised for a totality of bachelor's degree programmes within one field of study or an enlarged group of fields of study, the applicants shall participate in competitive selection for the admission to study in each of bachelor's degree programmes that comprise a single competition group.
- 7 44. In case specified in clause 7.43 of these Regulations, the procedures of both launching the particular degree programme and distributing (enrolling) the applicants to the particular degree programme shall be specified by the *Procedures for Distribution (Enrolment)* to Bachelor's Degree Programmes within the Admissions to a Totality of Degree Programmes within a Field of Study or Enlarged Field of Study (a single competitive selection for a group of degree programmes) (Annex 5 hereto).

## 8. Admissions to Master's Degree Programmes

- 8.1. Individuals with higher education that is confirmed by a diploma of their higher education and qualification shall be admitted to pursue master's degree programmes. Individuals with either a master's or a specialist's diploma that confirms their qualification of a specialist shall only be admitted to pursue master's degree programmes on the places under contracts for chargeable educational services.
- 8.2. Applicants applying for the admission to master's degree programmes shall also have an opportunity to submit their applications and other required documents either in person, or via general post service operator, or electronically. The deadlines for accepting applications and other documents for the competitive selection, deadlines for taking admissions tests and enrolments may differ depending on the type of the enrolment to different master's degree programmes open for admissions.

- 8.3. Individuals applying for enrolment to master's degree programmes shall submit only one admission application. In their admission applications, the applicants have the right to specify any quantity of master's degree programmes, to which the University arranges separate enrolment.
- 8.4. When submitting their admission applications, the applicants for master's degree programmes shall provide the original or a copy (digital image) of their standard credential as their standard credential as provided for by Clause 4 of these Regulations. Should such documents be unavailable by the deadline for accepting documents as specified in Clause 8.9. herein upon the circumstances beyond the applicants control, the applicant shall have the right to provide a standard transcript of academic records confirmed by the seal of the educational institution of higher education stating that they have fulfilled the curriculum and passed the State Final Assessment in a degree programme of higher education and specifying disciplines mastered, assessment grades awarded, and the date of issuance of the standard education credential.
- 8.5. The applicants when applying for the admission to the competitive selection have the right to submit their documents specified in Clause 8.7 herein electronically through their Personal Account on the St Petersburg University official website. To do so, the applicants shall undertake to register on the St Petersburg University official website and receive their personal login and password ensuring their access to the Personal Account. Logins and passwords must not be disclosed to third parties. The responsibility for their Personal Account login and password security after they have obtained them shall rest with the applicants.
- 8.6. Applications for admission to master's degree programmes shall be filled in electronically. To submit their application form, the applicant must type information in all boxes required, print the application out, and confirm the completeness and reliability of the provided information by putting their handwritten signature. Applicants shall submit their signed admission application with the required documents to the Admissions Committee either in person, or through the general post service operator, or electronically in the manner prescribed by Clause 8.7. herein. Admission applications and other documents submitted in the manner not specified herein, including those submitted via email or fax, shall not be accepted and considered by the Admissions Committee.
- 8.7. If the applicant for admission to master's degree programmes chooses to submit the specified documents electronically, they must submit their respective documents as scanned copies and attach them to their registration form:
- 8.7.1. the electronic image of an identity document with the indication of their citizenship (the following pages must be scanned: the applicant's photograph, family name, first name, middle name, gender, date and place of birth, citizenship, document series and number, information about the issuing authority and the applicant's place of registration);
- 8.7.2. the electronic image of the admission application for respective degree programmes;
  - 8.7.3. the electronic image of their standard credential with a supplement;
- 8.7.4. the electronic image of the document confirming their registration with the individual (personified) accounting system carrying their Insurance Number of the Individual Ledger Account (if applicable);
- 8.7.5. the applicant's photograph (taken in the current calendar year, with vertical resolution of at least 800 pixels and horizontal resolution of at least 600 pixels);
- 8.7.6. the electronic image of the documents for the competitive selection of documents (portfolio) in compliance with the admissions test syllabus;
- 8.7.7. the electronic images of the documents confirming the applicant's individual achievements (if applicable);
  - 8.7.8. other required documents.
- 8.8. An electronic image of a document in the cross-platform open format for electronic documents PDF (PortabelDocumentFiles) must ensure the visual identity with its

- original on 1:1 scale. The quality of the provided digital images must ensure legibility of the entire document and their details. If the hard copy of any document consists of two or more pages, its digital image must be provided in one file. It is required to make scans of documents with the resolution of at least 300 dpi. It is not allowed to provide illegible scanned images of the documents as well as images that lack significant sections of the document (texts, signatures, seals or stamps, etc.).
- 8.9. The University begins accepting the admission applications and other documents:
  - 8.9.1. **15 March 2024** when submitted in digital format;
  - 8.9.2. **1 July 2024** if submitted in person or through the general post service operator.
- 8.10. The University finishes accepting documents for the competitive selection and other documents submitted either in digital format, or in person, or by general post operators:
- 8.10.1. **20 July 2024**, **at 6pm (Moscow Time)** submitted for admissions to the full-time master's degree programmes;
- 8.10.2. **6 September 2024**, **at 6pm (Moscow Time)** submitted for admissions to part-time evening master's degree programmes.
  - 8.11. For full-time master's degree programmes:
- 8.11.1. the admissions tests shall be conducted from **July 22, 2024 to August 6, 2024**; applications for taking the admissions tests on a reserve day shall be accepted until 12pm, Moscow time, of 25 July 2024 in case admissions tests dates for different degree programmes of St Petersburg University coincide with each other, and until 12pm, Moscow time, of 29 July 2024 in case there are any other justified reasons (an illness, etc.).
  - 8.11.2. applicants shall be enrolled as students within the following deadlines:
- 8.11.2.1. **12 August 2024:** publishing, in the manner set forth in clause 1.6 of the Regulations, the ranking list of applicants sorted in the descending order by the score of competition points (subject to specifying) calculated as the totality of points for each admissions test and for individual achievements; the list of persons who have successfully completed admissions tests and whose enrolment may be considered by the Admissions Committee within various conditions of the competition (general competition, sponsored quota, etc.) and terms and conditions of study;
  - 8.11.2.2. **16** August **2024**, 6pm Moscow time:
- 8.11.2.2.1. the deadline for the activities specified in sub-clause 6.2.2. hereof within admissions to government-funded places;
- 8.11.2.2.2. the deadline for the activities specified in clause 6.6.4 of these Regulations with admissions under contracts for chargeable educational services;
- 8.11.2.3. **19 August 202**4 issuing the orders on enrolments, and making the information available in the manner prescribed by clause 6.11 hereof, on the enrolment of the applicants, who have abided by the provisions of sub-clauses 6.6.2, 6.6.4 of these Regulations.
- 8.12. Applicants to full-time master programmes who have successfully completed all admissions tests but lost the competition for government-funded places shall have the right to submit, together with the original or a copy of their standard education credential and the enrolment consent application, an application to participate in the competition for places under contracts for chargeable educational services until 6pm Moscow time, **on 16 August 2022**, as well as to conclude a contract for the provision of chargeable educational services.
  - 8.13. For **part-time evening** master's degree programmes:
- 8.13.1. the admissions tests shall be conducted from **2024 to 14 September 2024**; applications for taking the admissions tests on a reserve day shall be accepted until 12pm, September, of 25 July 2024 in case admissions tests dates for different degree programmes of St Petersburg University coincide with each other, and until 12pm, Moscow time, of 29 July 2024 in case there are any other justified reasons (an illness, etc.).
  - 8.13.2. applicants shall be enrolled as students within the following deadlines:

- 8.13.2.1. **16 September 2023:** publishing, in the manner set forth in clause 1.6 of the Regulations, the complete ranking list of applicants sorted in the descending order by the score of competition points they have received (the number of points specified) calculated as the totality of points for each admissions test and for individual achievements; the list of persons who have successfully completed the admissions tests and whose enrolment may be considered by the Admissions Committee within various conditions of the competitive selection (general competition, sponsored quotas, etc.) and terms and conditions of study;
  - 8.13.2.2. 19 September 2024, 6pm Moscow Time -
- 8.13.2.2.1. the deadline for the activities specified in sub-clause 6.2.2. hereof within admissions to government-funded places;
- 8.13.2.2.2. the deadline for the activities specified in sub-clause 6.6.4 of these Regulations with admissions under contracts for chargeable educational services;
- 8.13.2.3. **20 September 2024** issuing the orders, and making the information available in the manner prescribed by clause 6.11 hereof, on the enrolment of the applicants, who have abided by the provisions of sub-clauses 6.6.2, 6.6.4 of these Regulations.
- 8.14. In the competitive list, the applicants who have successfully passed the admissions tests shall be ranked as follows:
- 8.14.1. in the descending order of the competitive points calculated as the sum of the score points for each admissions test and for individual achievements
  - 8.14.2. in the descending order of the scored points assigned for the admissions tests;
- 8.14.3. in the descending order of the scored points assigned for separate admissions tests in accordance with the priority of the admissions tests established by the Admissions Committee;
- 8.14.4. based on individual achievements if other criteria considered and specified in Clauses 8.14.1. 8.14.3. of these Regulations are equal:
- 8.14.4.1. based on the average score of points of the standard credential provided by the applicant when applying for competitive selection. The calculation of the score average shall be carried out based on the supplement to the standard credential. If the supplement to the standard credential is missing, the score average for a particular applicant for the purpose of ranking shall be zero (0).
- 8.14.4.2. based o the number of publications on the topics correspondent to the enlarged group of subject areas (specialities) within which the admission to a degree programme is held in the journals indexed in the Web of Science Core Collection and Scopus scientometric databases, information of which has been provided by the applicant;
- 8.14.4.3. based on the number of publications on the topics correspondent to the enlarged group of subject areas (specialities) within which the admission to a degree programme is held in the journals indexed in the Russian Science Citation Index scientometric databases, information of which has been provided by the applicant;
- 8.14.4.4. based o the number of scientific events on the topic correspondent to the enlarged group of subject areas and specialities within which the admission to a degree programme is held, information of which has been provided by the applicant;
- 8.14.4.5. based on the status of a diploma holder of the St Petersburg University 'Petropolitan Science (Re)Search Open International Olympiad for students and young scientists or the 'I am a Professional' All-Russia University Olympiad;
- 8.14.4.6. based on the average score of points obtained for the state attestation assessment according to the information contained in the standard credential.
- 8.15. The applicants must submit their documents to have them considered for the purpose of ranking before the deadline stipulated in Clauses 8.9. 6.10. of these Regulations. The Admissions Committee shall have the authority to envisage other additional criteria for the ranking. The information on the additional criteria for ranking purposes shall be made available in the manner prescribed in clause 1.6. hereof.

- 8.16. Upon the decision of the Admissions Committee, the results of the respective international certification examinations provided by the applicant together with the admission application and other required documents shall be recognised upon the applicant's personal request as the results of the admissions test of foreign language skills and the results of the admissions test aimed at assessing applicants' foreign language competencies and professional competencies required for successful completion of the selected master programme. The applicants must submit the said documents not later than the deadline for submitting the documents specified in clause 8.10 of these Regulations. The said decision shall be made known to the applicants not later than the day the admissions tests begin in the manner prescribed by clause 1.6 hereof. At the same time, the results of international certification exams can be recognised once with admissions to degree programmes of the applicant's choice or as the results of admissions tests in foreign languages, admissions tests aimed at evaluating applicants' competencies in foreign languages and professional competencies needed for successful acquisition of a respective degree programme, or as individual achievement evidences (if recognition of international certification exams results has been provided by the List and Procedures for Recognition of Applicants' Individual Achievements)
- 8.17. Should the admissions been organised for a set of master's degree programmes within one subject area, the applicants shall participate in competitive selection for the admissions to pursue each of the master's degree programmes that comprise a single competitive group.
- 8.18. In case specified in Clause 8.17. of these Regulations, the procedures of both launching the certain degree programme and distributing (enrolling) the applicants to the certain degree programme shall be specified by the Procedures for Distribution (Enrolment) to Master's Degree Programmes within the Admissions to a Set of Degree Programmes within a Field of Study (a single competitive selection for a group of degree programmes) (Annex 6 hereto).

# 9. Admissions to Bachelor's or Specialist's Degree Programmes of Persons with Degrees

- 9 1. Persons with higher education, which is confirmed by a degree diploma of higher education and a qualification, and applying for the admission to bachelor's or specialist's degree programmes shall be enrolled to study at St Petersburg University on the places under contracts for chargeable educational services.
- 9 2. The procedures for the admission of applicants with higher education to bachelor's or specialist's degree programmes, acceptance of documents for the admission to study, organisation of admissions tests, development of competitive lists, and enrolment to study shall be governed by the provisions of Chapter 7 of these Regulations observing the same deadlines and the same procedures applied to the admissions of the those persons who apply for the admission to bachelor's or specialist's degree programmes to the places under contracts for chargeable educational services and pursue degree programmes of the appropriate level for the first time.
- 9 3. The applicants with higher education applying for the admission shall be subject to the admissions tests correspondent to their professional education as follows:
- 9.3.1. within admissions (individuals with higher education) to the full-time degree programme within the field of study 40.03.01 'Law, the admissions tests shall be held in the form of the competitive selection of the documents (portfolio);
- 9.3.2. within admissions to other bachelor's or specialist's degree programmes, admissions tests shall be held in those disciplines in which the general education-based admissions test are usually held, and admissions tests of creativity and/or profession-oriented nature in those fields of study (specialities) that require certain creative abilities, physical and/or psychological qualities in the applicants.

#### Section 3. Admission Particulars for Foreign Citizens

# 10. Categories of foreign citizens to enter the competitive selection, the particulars of submitting their admission applications and other required documents, admissions tests and enrolment

- 10.1. This section of the Regulations sets out the University policy for admitting foreign citizens, stateless citizens, including compatriots residing abroad (hereinafter referred to as foreign citizens; persons; applicants) having equal rights with Russian citizens when applying for degree programmes. They have the right to apply for the enrolment to the government-funded places, as well as to the places under contracts for chargeable educational services.
- 10.2. The enrolments of foreign citizens to the government-funded places shall be organised:
- 10.2.1. based on the quota set forth by Decree No 2150 'On establishing educational quota for foreign citizens and stateless persons in the Russian Federation' by the Government of the Russian Federation dated 18 December 2020 under references issued by the istry of Science and Higher Education of the Russian Federation:
  - 10.2.2. in compliance with international treaties of the Russian Federation;
- 10.2.3. pursuant to Federal Law No 99-FZ 'On the State Policy of the Russian Federation Concerning Compatriots Residing Abroad' dated 24 May 1999. .
- 10.3. The University enrols foreign citizens specified in sub-clauses 10.2.2, 10.2.3 herein to study on the government-funded places based on the competition and equal admission terms as applied to citizens of the Russian Federation and specified herein.
- 10.4. Deadlines and admissions procedures specified by Clause 10.2.1 of these Regulations shall be governed by the Procedures for the Open Competition for Admissions of Foreign Citizens, Stateless Persons, and Compatriots Residing Abroad to the Government-Funded Places as approved by the Admissions Committee and and made available in the manner prescribed by clause 1.6 of the Regulations.
- 10.5. The University shall enrol foreign citizens to the places under contracts for chargeable educational services on the competitive basis to the vacancies determined by the Admissions Committee and under the terms and conditions stipulated by the Regulations.
- 10.6. The University shall enrol foreign citizens to the degree programmes under agreements made by the University for international cooperation, including the agreements for implementing double-diploma degree programmes, in the manner prescribed by Clause 10.4 of these Regulations, or in the event when the agreements provide for enrolments to the places under contracts for chargeable educational services, in the manner prescribed for the individuals specified by clause 10.5 herein.
- 10.7. The list of degree programmes, the nomenclature of admissions tests and the scores of points sufficient for passing admissions tests, as well as admissions tests syllabi, shall be made available to the applicants in the manner prescribed by Clause 1.6. herein.
- 10.8. To be enrolled to St Petersburg University, international applicants shall also provide, if required, a certificate that their education and/or qualification credentials issued in a foreign state granted recognition in the Russian Federation;
- 10.9. The University shall enrol foreign citizens specified in Clause 10.2.1 of these Regulations upon the references issued by the Ministry of Science and Higher Education of the Russian Federation, which shall contain information about the field of study (speciality), within which the degree programme is implemented.
- 10.10. Foreign citizens, upon submitting their admission applications shall provide documents specified by Clause 2.17 of Section 2 of the Regulations, documents confirming

their association with the categories of foreign citizens as specified by Clauses 10.2, 10.5 herein, as well as other documents as follows:

- 10.10.1. a copy of the entry visa to the Russian Federation (in case a foreign citizen entered Russia on an entry visa) specifying the purpose of the visit to the Russian Federation: "to study";
  - 10.10.2. a copy of the migration card;
- 10.10.3. an acknowledgement confirming that the applicant has read the Residence Regulations for International Applicants in the Russian Federation;
- 10.11. All translations into Russian must show the same given name and surname as those used in the entry visa (in the absence of an entry visa, the given name and surname appearing in the identity document).
- 10.12. The applicants when applying for the admission to the competitive selection have the right to submit their documents specified in Clause 10.9 herein electronically through the St Petersburg University official website. To do so, the applicants shall undertake to register on the St Petersburg University official website and receive their personal login and password ensuring their access to the Personal Account. Logins and passwords must not be disclosed to third parties. The responsibility for their Personal Account login and password security after they have obtained them shall rest with the applicants.
- 10.13. Admission applications for admission to the respective degree programmes shall be filled in electronically. To submit their application form, the applicant must type information in all boxes required, print the application out, and confirm the completeness and reliability of the provided information by putting their handwritten signature. Applicants shall submit their signed admission application with the required documents to the Admissions Committee either in person, or through the general post service operator, or electronically in the manner prescribed by Clause 10.13 herein.
- 10.14. Should the applicant choose to submit the specified documents electronically, the respective documents must be scanned and attached to the registration form through the applicant's Personal Account:
- 10.14.1. the electronic image of an identity document with the indication of their citizenship (the following pages must be scanned: the applicant's photograph, family name, first name, middle name, gender, date and place of birth, citizenship, document series and number, information about the issuing authority and the applicant's place of registration);
- 10.14.2. the electronic image of the enrolment application to the respective degree programmes signed by the applicant;
  - 10.14.3. the electronic image of their standard credential with a supplement;
- 10.14.4. the applicant's photograph (taken in the current calendar year, with vertical resolution of at least 800 pixels and horizontal resolution of at least 600 pixels);
- 10.14.5. the electronic image of the documents for the competitive selection of documents (portfolio) in compliance with the admissions test syllabus;
- 10.14.6. the electronic images of the documents confirming the applicant's individual achievements (if applicable);
  - 10.14.7. other required documents.
- 10.15. By signing the enrolment application, the applicant confirms, apart from the circumstances specified in Clauses 2.11., 2.12 thereof, that they have read the residence regulations for foreign nationals in the Russian Federation provided by Federal Law No 115-FZ 'On the Legal Status of Foreign Nationals in the Russian Federation' dated 25 July 2002 and other by-laws and regulations of the Russian Federation.
- 10.16. The foreign citizens admitted in compliance with the provisions of Clauses 10.2.2 10.2.3 herein shall take the same admissions tests as for citizens of the Russian Federation applying for the degree programmes of the respected level implemented at St Petersburg University.

- 10.17. The foreign citizens admitted in compliance with the provisions of clauses 10.2.2 10.2.3 herein for enrolment to the government-funded places or to the places under contracts for chargeable educational services shall be enrolled in compliance with procedures and deadlines specified for the enrolment of citizens of the Russian Federation applying for the degree programmes of the respective level.
- 10.18. The University accepts documents from foreign citizens specified in clause 10.5 hereof **from 15 April 2024 to 21 June 2024.**
- 10.19. The University shall consider documents from foreign citizens applying for enrolment within three (3) working days from the day of the documents were submitted. The applicant, who has provided the complete set of the documents as specified in Clause 10.10 of these Regulations, shall be listed among foreign citizen admitted to participating in the competitive selection for the enrolment. The above mentioned list shall be made available in the manner prescribed in Clause 1.6. herein. If a foreign citizen is not been listed among foreign citizen admitted to participating in the competitive selection within three (3) working days from the day the documents were submitted, it means he/she has failed to submit a complete set of documents or meet requirements for the paperwork.
- 10.20. The foreign citizens specified in Clause 10.5 herein shall take admissions tests in compliance with admissions test syllabus made available on the official web site of St Petersburg University. The applicants must provide all documents and materials specified by the syllabi of the admissions tests held in the form of competitive selection of documents (portfolios) are to be submitted before the deadline set forth by Clause 10.18 of these Regulations.
- 10.21. Applicants applying to bachelor's degree, specialist's degree, or master's degree programme implemented in Russian within admissions tests in the form of the competitive selection of documents (portfolio) can provide their credentials confirming the fact of successful completion of a lower level degree programme (general secondary education and higher) implemented in Russian as the confirmation of their TORFL-I or TORFL-II Russian language reference level.
- 10.22. The lists of the applicants specified in clause 10.5 hereof of those recommended for the enrolment being highlighted, shall be made available on the official web site of St Petersburg University at least before **15 July 2024.**
- 10.23. The University shall enrol foreign citizens applying for degree programmes as specified in clause 10.5 hereof at least before **31 August 2024** upon the decision made by the Admissions Committee based on the results of the admissions tests for respective degree programmes.
- 10.24. The University shall enrol foreign citizens applying for enrolment under contracts for chargeable educational services only after they have signed the respective contracts for chargeable educational services.

Section 4. Admissions to the branches of St Petersburg University

# 11. Admissions to the Branch of St Petersburg University in Tashkent

- 11 1. St Petersburg University carries out admissions to study at its branch in Tashkent (hereinafter referred to as the Branch of St Petersburg University in Tashkent) in bachelor's degree programmes and master's degree programmes to the places under contracts for chargeable educational services.
- 11 2. Admissions of Russian citizens to study at the Branch of St Petersburg University in Tashkent are held according to the procedures and within deadlines specified in Chapters 7 and 8 of these Regulations.
- 11 3. Admissions of foreign citizens to study at the Branch of St Petersburg University in Tashkent are held according to the procedures and within deadlines specified in Chapters 10 of these Regulations.

Annex 1

to the Regulations for Admission to Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

#### **PROCEDURES**

## for Admissions Tests to Degree Programmes at St Petersburg State University in 2024

#### 1. General Provisions

- 1.1. These Procedures for Admissions Tests to Degree Programmes at St Petersburg State University in 2024 (hereinafter referred to as the Procedures for Admissions) set the procedures for practical creative admissions tests and/or profession-oriented additional admissions tests, admissions tests held at the discretion of St Petersburg University within admissions of applicants to bachelor's degree, specialist's degree, and master's degree programmes in 2024.
- 1.2. These Procedures have been developed in compliance with the following documents. They are: Federal Law No 273-FZ 'On Education in the Russian Federation' as of 29 December 2012; the Regulations for Admissions to Degree Programmes Bachelor's Degree, Specialist's Degree, Master's Degree Programmes approved by Order No 1076 of the Ministry of Science and Higher Education as of 21 August 2020 (hereinafter referred to as the Procedures for Admissions); the Charter of the Federal State Budgetary Institution of Higher Education 'St Petersburg State University'; the Regulations for Admissions to Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg State University in 2024 (hereinafter referred to as the Regulations for Admissions).
- 1.3. St Petersburg University shall ensure that the applicants' admissions test papers be accepted for the evaluation depersonalised so that their authorship cannot be determined. Should the competitive selection be held in the form of the selection of documents (portfolios), the evaluation of depersonalised admissions tests papers shall be carried out in the manner prescribed by Section 5 of the Procedures for Admissions.
- 1.4. The syllabi of the admissions tests held at the discretion of St Petersburg University for applicants specified in Clause 7.2.3 of the Procedures for Admissions are developed in compliance with the federal state educational standard of secondary general education.
- 1.5. There are several formats of admissions test held at the discretion of St Petersburg University. They are: written tests or written-oral tests; competitive selection of documents (portfolio); creative admissions tests and others according to the decision of the Admissions Committee.
- 1.6. The format for the admissions test of creative and/or profession-oriented nature shall be specified by the Admissions Committee of St Petersburg University.
- 1.7. The admissions tests held at the discretion of St Petersburg University for the enrolment to a particular bachelor's degree or specialist's degree programme shall be held in general education-based subjects specified by the Admissions Committee based on Order No 722 'On Approving the List of Admissions Tests for Degree Programmes Bachelor's Degree Programmes, Specialist's Degree Programmes, Master's Degree Programmes' of the Ministry of Science and Higher Education of the Russian Federation dated 6 August 2021.
- 1.8. The admissions tests nomenclature and their forms for admissions to master's degree programmes at St Petersburg University are to be specified by the Admissions Committee.
- 1.9. Examination materials and organisation of admissions tests for applicants to St Petersburg University degree programmes are developed by examination boards created in in the manner set forth at St Petersburg University for admissions tests.

1.10. Examination boards are formed out of St Petersburg University research and teaching staff, as well as research and teaching staff of educational, research and other organisations, which areas of activity correspond to the respective St Petersburg University degree programmes. The heads of examination boards are appointed out of highly qualified research and teaching staff and other staff of educational, research and other organisations which areas of activity correspond to the respective St Petersburg University degree programmes.

#### 2. Organisation of admissions test sessions

- 2.1. The 2024 schedule of admissions tests for admission to St Petersburg University degree programmes is established in accordance with the Regulations for Admissions.
- 2.2. Prior to all admissions tests, except for the competitive selection of documents (portfolio), the University provides counselling sessions for the applicants to brief them on the structure of the admissions tests, their procedures, and assessment criteria applied to answers or papers.
- 2.3. Unless allowed, the applicants must not carry any personal belongings to the classrooms during the admissions tests held in the forms other than competitive selection of documents (portfolio). Communication devices and other technical gadgets, except laptops, video cameras and microphones that enable video conferencing, shall be inactivated beforehand.
  - 2.4. During admissions tests, the applicants are not allowed to have:
- 2.4.1. personal computers (including portable computers, pocket computers etc.), other computing machinery (including calculators) and other technology devices except those specified in clause 2.3. of the Procedures for Admissions;
- 2.4.2. mobile telephones (unless used to picture the applicant's work station) and other communication devices;
- 2.4.3. stationary unless approved to be used as outlined in Clause 2.6.2 of these Procedures for Admissions;
- 2.4.4. any reference materials (including learner's guides, monographs, lecture notes, geographic maps and atlases, etc.) unless their usage is specified by the syllabus of a relevant admissions test.
- 2.5. Admission testing other than competitive selection of documents (portfolios) shall be held using the audio and video conferencing systems (applicants must use a video camera and a microphone during the admission testing and provide St Petersburg University with the video recording of the admissions test session), enabling the applicants to undergo the admissions testing through the St Petersburg University system of distance teaching and learning via the Internet connection, as well as ensuring the applicants' compliance with the provisions of the Procedures for Admissions (proctoring).
- 2.6. In the course of organising admissions tests as specified by Clause 2.5 of these Procedures for Admission:
- 2.6.1 members of respective examination boards, other authorised St Petersburg University officials and/or authorised officials from other organisations shall undertake measures to carry out applicant identification to ensure that the very applicant that have submitted their admission application undergoes the admissions testing. To do so, the applicants before the start of the admissions test session shall present the original of their identity document (pages with their full name and their photograph);
- 2.6.2 the information about requirements applied to the admissions test equipment, the order of obtaining access to the St Petersburg University system of distance teaching and learning, admissions tests tasks, the order of entering answers to the the admissions test tasks, the routine prescribed to confirm applicant's identity before the admissions test session starts and ensure applicants' abidance by the Procedures for Admissions during the test sessions, the

possibility of terminating the test session earlier, whether it is allowed to take short leaves from the admissions test session room or not, and the order of taking such leaves when allowed in the manner prescribed by clause 1.6. of the Procedures for Admissions at least before the entire period of holding the admissions test sessions.

#### 3. Procedures for Admissions Tests.

- 3.1. The oral answer at the tests (interview) or the oral element of the tests of each applicant shall be evaluated at least by two members of the respective examination board. The applicant has 45 minutes to prepare their oral answer (at the foreign language test to work with a text 20 minutes). In the course of applicant's performance at the tests, they may be asked questions relevant to the content of their examination card questions. The members of the respective examination board shall note whether the answers to the examination card questions are correct and complete.
- 3.2. When holding an interview, the respective examination board keeps a record of questions they ask the applicant, brief commentary of the examiners (synopsis) of the answers to them and grades every part of the respective admissions test.
- 3.3. The final score of points for the oral part of the examination is put in a figure and in cursive into the sheet of oral (oral and written) answers (interview record) signed by at least two board members.
- 3.4. The length of the written admissions test is specified by the respective admissions test syllabus.
- 3.5. The length of a practical creative admissions test and/or a profession-oriented admissions test may be defined in each particular case in accordance with the syllabus of the respective admissions test.
- 3.6. Should written admission test papers contain any indication enabling the authorship disclosure, they shall not be accepted for the evaluation and the applicant shall be awarded the lowest score of points (zero points). In this case, there should the Violation Statement be drawn out stating that he applicant has failed to pass the test without a good reason, an electronic copy of which shall be forwarded to the applicant.
  - 3.7. Each applicant must accomplish their tasks independently.
- 3.8. In the course of the admissions tests session, the applicants must not communicate with any persons except examination board members, authorised personnel of St Petersburg University and/or authorised personnel of other organisation on the issues directly relating to the procedures of the admissions test session.
- 3.9. Should any applicant during the admissions test session carry or use during the mentioned period any unauthorised written and/or printed materials, as well as any technology devices (including any communication devices); should any applicant fail to abide by the provisions of clauses 3.7 and 3.8 of the Procedures for Admissions, they shall be expelled from the competition and conferred the lowest score of points (zero points). Should such violations be determined after the admissions test sessions have finished, the fault applicants shall be conferred the lowest score of points (zero points) for their written admissions test paper regardless of their content. In this case, there should the Violation Statement be drawn out stating that he applicant has failed to pass the test without a good reason, an electronic copy of which shall be forwarded to the applicant.

### 4. Organisation of Admissions Tests for Disabled Persons

- 4.1. The University arranges admissions tests for physically challenged persons and the disabled taking into account the specificities of their psychophysical development, their individual capabilities and health status.
  - 4.2. Such applicants in their admission application can request extra time to add to the

admissions test session. Physically challenged persons, who have requested such a need in their admission application, shall be granted extra 60 minutes to accomplish their admissions test tasks.

## **5.** Competitive Selection of Documents (Portfolios)

- 5.1. To enter admissions testing held in the format of the competitive selection of documents (portfolio), the applicants along with their documents required for being admitted to the selection shall provide documents and materials, the list of which is being specified by the admissions test syllabus for each degree programme, specified by applicants in their admission application.
- 5.2. All documents required by the admissions tests syllabi for the competitive selection of documents (portfolios) shall be submitted within the admission acceptance period set forth by the Procedures for Admissions.
- 5.3. The documents for the competitive selection of documents (portfolios) shall be provided in the form of digital images in the cross-platform open format of electronic documents PDF (Portable Document Files). The digital images of the documents must ensure visual identity to the hard copy original on the 1:1 scale. The quality of the provided digital images must ensure legibility of the entire document and their details. It is advisable to make scans of documents with the resolution of 300 dpi.

Annex 2
To the Procedures for Admission
to Degree Programmes (Bachelor's Programmes, Specialist's Programmes,
Master's Programmes) at St Petersburg University in 2024

## Procedures for Reviewing Written Test Papers, Filing and Considering the Appeals within Admissions to Degree Programmes at St Petersburg University in 2024

- 1. Should the applicant disagree with the results of the respective admissions test after the test results have been announces they shall have the right to review their written test paper, written part of their oral and written admissions test, documents provided within the portfolio (hereinafter referred to as a 'written paper') in the manner prescribed by the Regulations for Reviewing Written Test Papers, Appeal Submission and Appeal Proceedings (hereinafter referred to as the 'Regulations for Reviewing') and file a reasoned written appeal application (hereinafter referred to as the 'appeal'). Also, the applicant may file the appeal in case they suspect the Regulations for Admissions have been violated.
- 2. The day the admissions test results are considered announced shall be the day on which the results of the admissions tests were published in the manner prescribed by clause 1.6 of the Procedures for Admissions..
- 3. Information on the date when the admissions tests results shall be announced, the period appointed review these written papers, the period appointed to file appeals within each admissions test, the period appointed for considering the appeal within each admissions testing shall be included into the schedule of the admissions tests and made available to the applicants in the manner prescribed by clause 1.6 herein.
- 4. Information about evaluation of documents, constituting the portfolio, shall be made available to applicants through posting these evaluation results. When the admissions tests are held in the form other than the competitive selection of documents (portfolios), the applicants shall be granted access to the results for the admissions tests through the St Petersburg University system of distance teaching and learning according to the approved schedule of the admissions tests sessions.
- 5. An appeal shall be filed in the name of the Chair of the Appeals Panel on the day the results are announced or on the working day following the day the admissions tests results have been announced. Applicants shall file appeals revealing the violation of the Regulations for Admissions on the day of the respective admissions tests session. Claimants shall submit their appeals via e-mail to the address of the respective committee for accepting admissions applications specified in the manner prescribed by Clause 1.6 of the Procedures for Admissions using the standard St Petersburg University template; because of that they do not meet the requirements specified in Clauses 2.4, 2.5, 8.2, 8.6 herein.
- 6. The appeal must contain specific reasons why the applicant disagrees with the grade obtained or certain facts proving that the Regulations for Admissions have been violated.
- 7. Periods for considering appeals for a particular admissions testing shall be specified in the admissions tests schedule.
- 8. The composition of the Appeals Panel shall be approved by an authorised St Petersburg University official.
  - 9. At least three members of the Appeal Panel shall consider the appeal.
- 10. The applicant shall have the right to witness the appeal consideration via the system of audio and video conferencing.
- 11. The appeal considering shall not be deemed a retaking the admissions test. The appeal considering aims at determining the compliance of both the admissions tests session with the Regulations for Admissions and the admissions tests evaluation with the assessment criteria.

- 12. Upon the decision of the Appeals Panel, the grade conferred may be made higher, lower or left unchanged.
- 13. Upon the appeal considering, the Appeals Panel shall decide on the score of points for the admissions testing. All decisions of the Appeals Panel shall be taken by majority vote.
- 14. The decision of the Appeals Panel shall be formalised in the form of minutes and approved by the Admissions Committee executive secretary and shall be final and without appeal. The Admissions Committee's decision formalised in the minutes shall be made available to the applicant via e-mail at least on the working day following the day of the appeal considering in accordance with the admissions tests schedule.
- 15. Should the Appeals Panel decide on the adjusting the admissions test grade (both making it higher or lower), such an adjustment shall effect the admissions tests results and be published in the manner prescribed by Clause 1.6 hereto.
- 16. The minutes of the Appeals Panel shall be stored as long as it is regulated by the nomenclature of cases.

Anex 3 to the Regulations for Admission to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Degree Programmes, Master's Programmes) at St Petersburg University in 2024

# Procedures for Applicants' Individual Achievement Evidences Recognition when applying for the admission to degree programmes

Main field of study title	Programme title	Individual achievement evidence	Evaluation criterion	Score of points
01.00.00 MATHEMAT	ICS AND MECHANICS			
01.04.01 Mathematics	Advanced Mathematics	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computational Technologies (Fundamental Informatics and Information Technologies)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Mathematics	Students-medal holders - 100 points Winners - 25 points Awardees - 10 points
		'Excellent' grade for the bachelor's degree or specialist's degree graduation project		15 points
		Scientific publications in 'Mathematics'  Notes: evaluated can be but one publication If	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases	50 points
		there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points. If the provided	in publications indexed by RSCI scientometric database / or from the Higher Attestation Committee lists.	25 points

		document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	In other scientific publications or in scientific archives of organisation s of higher education and research organisations	10 points
		Documents and materials confirming the applicant's participation in scientific events as a presenter on topics relevant to the respective enlarged group of fields of training (specialities) to which the admission to the degree programme is being carried out. The participation shall be confirmed by the conference programme.		10 points
		Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implemented projects;		10 points
01.04.02 Applied Mathematics and Informatics	Software Development and Data Science	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Mathematics Computational Technologies (Fundamental Informatics and Information Technologies)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Mathematics	Students-medal holders - 100 points Winners - 25 points Awardees - 10 points
		'Excellent' grade for the bachelor's degree or specialist's degree graduation project		15 points
		Scientific publications in 'computer sciences'  Notes: evaluated is but one publication If there are two or more articles submitted, the points shall not	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases	50 points

		be summed; evaluated is an article with the highest number of points. If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause	In periodicals indexed by RSCI scientometric database / or from the Higher Attestation Committee lists.	25 points
		shall not be scored	In other scientific publications or in scientific archives of organisation s of higher education and research organisations	
		Documents and materials confirming the applicant's participation in scientific events as a presenter on topics relevant to the respective enlarged group of fields of training (specialities) to which the admission to the degree programme is being carried out. The participation shall be confirmed by the conference programme.		10 points
		Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implemented projects;		10 points
01.04.02 Applied Mathematics and Informatics	All degree programmes in the main field of study except 'Software and Data Science' programme	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Security of Information Systems and Technologies of Critical Facilities Information and Cyber Security Artificial Intelligence Mathematics Mathematical Modelling Programming and Information Technology Robotics	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Applied Mathematics and Informatics	Winners - 10 points Awardees - 5 points

	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics Game theory	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
All degree programmes in the main field of study except 'Software and Data Science' and 'Mathematical Modelling, Programming	Bachelor's degree /specialist's degree in the field of study and specialities related to the 01.00.00, 02.00.00, 09.00.00 enlarged group of fields of study and specialities	Degree with distinction	7 points
and Artificial Intelligence' degree programmes		Degree without distinction with grade average point 4.5 and higher	5 points
		Degree without distinction with grade average point 4.0 and higher	2 points
	Information about publications on topics relevant to the scientific speciality to which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases,a s well as in periodicals included into the RSCI core and peerreviewed journals, recommended by HAC	14 points
Certificates/diplomas/letters of appreciation so not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the points shall not be summed; evaluated is an ar with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientification, points under this clause shall not be sco	Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication. If there are two or more articles submitted, the	Scientific publications in indexed by the RSCI scientometric database	9 points
	with the highest number of points. If the provided document proposes that the	Articles in other scientific periodicals	3 points

less than five (pages) of length if its imprint data supposes that it has been presented at a conference.  It is recommended to provide a link to the web page of the article in a scientometric database to		
speed up the verification of the work.  Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) to which the admission to the degree programme is being carried out.  Accepted can be but two documents. If there are three documents being provided, points shall be	Participation in a scientific event as a presenter.  The delivery of the report shall be confirmed by the programme of the conference with presenters named.	5 pints for one event
scored based on two of them.  Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.  Accepted can be one document. Should there be two or more documents, the points shall not be summed.	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.  Members of research teams, a grant, a project under the order from stateowned science foundatoins, reaserch organnisations or educational organisations of higher education.	8 points 9 points
Documents confirming respective working experience addressing the used in the professional activity competences correlated with the competences to be developed in the course of	12 months and more	3 points

mastering a respective degree programme (experience in professional practical activity). Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the fixed-term employment contract; - a confirmation from the place of work (sealed and signed by the official)	From 6 months to 12 months  From 1 month to 6 months	2 points  1 point
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Accepted can be one document. If there are two or more documents, the points shall not be summed.		3 points
Winners' or laureate's diploma for contests of research, design work, international moot courtcontests, Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged group of fields of study (speciality) within which the admission to the degree programme is being carried out. Considered shall be Olympiads and	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	4 points
	Diploma of an awardee of open international competitions of students'research projects, international game competitions, student Olympiads	3 points
contests held by educational organisations of higher education  Year of 2022/2023, 2023/2024	Diploma of the winner of Russian national competitions of students' research projects and student Olympiads	3 points
Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	2 points

		Diploma of the winner of open all- Russia competitions of students' research projects, student Olympiads	1 point
Mathematical Modelling, Programming and	Bachelor's degree /specialist's degree in the field of study and specialities related to the 01.00.00,	Degree with distinction	10 points
Artificial intelligence	al Intelligence 02.00.00 or 09.00.00 enlarged group of fields of study and specialities	Degree with 4.0 grade average point	7 points
	Information about publications on topics relevant to the scientific speciality to which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20 points
	collection, and indexing confirmation must be provided).	Scientific publications in periodicals recommended by the HAC	12 points
	Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.		
	If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored. Report abstracts shall be deemed a publication less than five (pages) of length if its imprint data supposes that it has been presented at a conference.	Scientific publications in indexed by the RSCI scientometric database	7 points
	It is recommended to provide a link to the web page of the article in a scientometric database to speed up the verification of the work.		

Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) to which the admission to the degree programme is being carried out.  Accepted can be two documents. If there are three documents being provided, points shall be scored based on two of them.	Participation in a scientific event as a presenter.  The delivery of the report shall be confirmed by the programme of the conference with presenters named.	5 points
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.	8 points
Only one document is considered. Should there be two or more documents, the points shall not be summed.	Members of research teams, a grant, a project under the order from state-owned science foundatoins, reaserch organnisations or educational organisations of higher education.	10 points
Documents confirming respective working experience addressing the used in the professional activity competences correlated with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity).	12 months and more  From 6 months to 12 months	5 points 2 points
Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the fixed-term employment contract; - a confirmation from the place of work (sealed and signed by the official)	1 Tom 6 months to 12 months	2 points

01.04.03 Mechanics and Mathematical Modelling	All degree programmes in the main field of study	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computational Technologies (Fundamental Informatics and Information Technologies)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Mathematics Mathematical Modelling	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Applied Mathematics and Informatics Mathematics and Mechanics	Winners - 10 points Awardees - 5 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		Bachelor's/Specialist's degree	Bachelor's/ specialist's degree with distinction in the fields of study correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes	9 points
			Bachelor's/ specialist's degree with more than 4.0 of grade average point without distinction in the fields of study correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes.	6 points
			Bachelor's/ specialist's degree with less than 4.0 of grade average point in the fields of study correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes.	3 points
			Bachelor's/specialist's degree with distinction in other fields of study	5 points
			Bachelor's/specialist's degree with more than 4.0 of grade average point without distinction in other fields of study	3 points

	Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	3 points
'Excellent' grade for the bachelor's degree/ specialist's degree graduation project	'Excellent' grade for the bachelor's degree/ specialist's degree graduation project in degree programmes correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes	2 points
Information about publications on topics relevant to the 01.00.00, 02.00.00, 03.00.00 and 09.00.0 enlarged groups of fields of study (specialities) (a	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	12 points
copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and confirmation must be provided).	Scientific publications in periodicals recommended by the HAC	10 points
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in indexed by the RSCI scientometric database	8 points
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the	Articles in other scientific periodicals	3 points
publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	Publications in electronic scientific archives of organisations of higher education and research organisations	2 points
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to 01.00.00, 02.00.00, 03.00.00, 09.00.00 respective enlarged groups of fields of study (specialities) to which the admission to the degree programme is being carried out.	Documents confirming the applicant's participation in scientific events.  The participation must be confirmed by the published abstract to the report or the event programme	5 points
Only one document is considered. Should there be two or more documents, the points shall not be summed.	Documents confirming the applicants participating in science schools/forums at least seven (7) days long held by	3 points

	organisations of higher education or research organisations	
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.	8 points
relevant to the fields of study correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	6 points
Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Documents confirming respective working experience addressing the used in the professional activity competences correlating with the competences to be developed in the course of mastering a respective degree programme	12 months and more	3 points
(experience in professional practical activity).  Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource	From 6 months to 12 months	2 points
department officer; - a copy of the employment contract; - a confirmation from the place of work (sealed and signed by the official)	From 1 month to 6 months	1 point
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):	English language reference credentials: TOEFL (score 65 and higher); IELTS (score 5.5 and higher); Cambridge CPE - C, B, A; Cambridge CAE - C, B, A; Cambridge FCE - B, A; Cambridge English: Business (BEC Vantage - grades B, A; BEC Higher - grades C, B, A); Level 2 Test in English from the Language Testing Centre at St	3 points

Only one document is considered. Should there be two or more documents, the points shall not be summed.

Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University(CEFR Level C1) (Grades C,B,A 51% and higher). In German Goethe Zertificate (B2/C1-70-100; C2); Test DaF (TDN 16-20); Deutsche Sprachprüfung für den Hochschulzugang (DSH 1 - DSH 3: 57% and higher); Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1: 60 – 100; ÖSD C2: 100

The European Language Certificate (TELC B2: 240 - 300, TELC C1: 151 -214; TELC C2: 112 – 160)

in French DELF B2 – 61-100; DALF C1 - 50-100

In Spanish DELE B2/C1 – 61-100

GMAT (Graduate Management Admission Test) (500 and higher)

In Italian

-160);

European Consortium for the Certificate of Attainment in Modern Languages

ECL: C1 CEFR (60% and higher)

In Chinese

Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher);

	Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	
	'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	2 points
Documents confirming internship and training in a foreign language.  Only one document is considered. Should there be two or more documents, the points shall not be	Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	3 points
summed.	Short-term training abroad in relevant fields of study (summer, autumn, winter and other schools, scientific seminars,	2 points  1 point
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be	etc.)	2 points
winners' or laureate's diploma for contests of research, design work, international moot courtcontests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	3 points
Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged	Diploma of an awardee of open international competitions of	2 points

group of fields of study (speciality) correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes Considered shall be Olympiads and contests held by educational organisations of higher education  Year of 2022/2023, 2023/2024  Only one document is considered. Should there be two or more certificates, the points shall not be summed	students'research projects, international game competitions, student Olympiads  Diploma of the winner of Russian national competitions of students' research projects and student  Olympiads  Diploma of the winner of open all-Russia competitions of students' research projects, student Olympiads  Diploma of the winner and an awardee of open regional competitions of	1 point
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.	students' research projects, student Olympiads	2 points
Only one document is considered. Should there be two or more certificates, the points shall not be summed  Certificates for online courses hosted by educational platforms 'Coursera' and 'Open Education' (except online courses of St Petersburg University specified by a particular clause of the List and Procedures for the Recognition of Individual	Online course correspondent to 01.00.00, 02.00.00, 03.00.00 and 09.00.0 state codes	1 point
Achievements of Applicants Applying to Study Department		

in Master's Degree Programmes) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.	
Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.	

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 01.00.00 'Mathematics and Mechanics' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

<b>02.00.00 COMPUTER</b>	02.00.00 COMPUTER AND INFORMATION SCIENCES				
02.04.02 Fundamental Informatics and Information Technology	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points	
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information Technology Robotics Security of Information Systems and Technologies of Critical Facilities	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points	
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points	
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points	

'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points
'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
CASE-IN International Engineering Championship Year of 2022/2023, 2023/2024	'Automation and IT' subject area	Winners - 10 points
WorldSkills Russia all-Russia championships in professional competence Year of 2022/2023, 2023/2024	DigitalSkills championship in high-tech profession	Winners - 10 points
Bachelor's/Specialist's degree	A bachelor's degree,/specialist's degree diploma with distinction in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admission within a 02.00.00 degree programme	25
	A bachelor's degree/ specialist's degree diploma with grade average point and higher than 4.5 n in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admission within a 02.00.00 degree programme	20
	A bachelor's degree, specialist's degree diploma with grade average point higher than 4.0 in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admission within a 02.00.00 degree programme	15

	A bachelor's degree/ specialist's degree diploma with grade average point lower than 4.0 in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admission within a 02.00.00 degree programme	10
	Bachelor's degree /specialist's degree with distinction in the field of study and specialities relevant to the 01.00.00, 09.00.00, 10.00.00 programmes of the enlarged group of fields of study and specialities	20
	Bachelor's/ specialist's degree with grade average point higher than 4.0 without distinction in the fields of study correspondent to 01.00.00, 09.00.00, and 10.00.0 programmes.	10
	Bachelor's/ specialist's degree with grade average point lower than 4.0 in the fields of study correspondent to 01.00.00, 09.00.00, and 10.00.0 programmes.	5
	Bachelor's/specialist's degree in other fields of study and specialities	5
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree	Articles in publications indexed by scientometric Web of Science and Scopus databases	11
programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents)	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	6
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication	Articles in other scientific periodicals	4

If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	The participation must be confirmed by the published abstract to the report or the event programme	4
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be	Scholarship from the President of the Russian Federation  Scholarship from the Government	4
two or more certificates, the points shall not be summed	Scholarships from foundations, public associations and other organisations	2
	Scholarship from an educational organisation of higher education	1
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	6
Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged group of fields of study (specialities) within which	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	5
the admission to the degree programme is being carried out. Considered shall be Olympiads and	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	4

		contests held by educational organisations of higher education  Year of 2022/2023, 2023/2024  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of an awardee of open all-Russia competitions of students' research projects, student Olympiads Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	2
02.04.03 Information Systems Administration and Mathematical Support	Software and Administration of Information Systems	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Information and Cyber Security Artificial Intelligence Mathematical Modelling Programming and Information Technology Robotics Security of Information Systems and Technologies of Critical Facilities	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points

'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs  Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
CASE-IN International Engineering Championship Year of 2022/2023, 2023/2024	'Automation and IT' subject area	Winners - 10 points
WorldSkills Russia all-Russia championships in professional competence Year of 2022/2023, 2023/2024	DigitalSkills championship in high-tech professions	Winners - 10 points
Bachelor's degree /specialist's degree in the field of study and specialities related to the enlarged group of fields of study and specialities of	Degree with distinction	10
02.00.00 or 09.00.00 programmes	Degree without distinction	7
Information about publications on topics relevant to the scientific speciality to which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases,a s well as in periodicals included into the RSCI core and peerreviewed journals, recommended by HAC.	15
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Scientific publications in indexed by the RSCI scientometric database.	10

If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored. Report abstracts shall be deemed a publication less than five (pages) of length if its imprint data supposes that it has been presented at a conference.  It is recommended to provide a link to the web page of the article in a scientometric database to speed up the verification of the work.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) to which the admission to the degree programme is being carried out.  Accepted can be two documents. If there are three documents being provided, points shall be scored based on two of them.	Participation in a scientific event as a presenter.  The delivery of the report shall be confirmed by the programme of the conference with presenters named.	5 (+5)
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.	8
relevant to the field of study to which programme the admissions is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed.	Members of research teams, a grant, a project under the order from state-owned science foundatoins, reaserch organnisations or educational organisations of higher education.	10
Documents confirming respective working experience addressing the used in the professional activity competences correlated with the	12 months and more	5

competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity).	From 6 months to 12 months	2
Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the fixed-term employment contract; - a confirmation from the place of work (sealed and signed by the official)		

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 02.00.00 'Computer and Information Science' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

03.00.00 PHYSICS AND ASTRONOMY				
03.04.01 Applied	All degree programmes	Petropolitan Science (Re)Search, St Petersburg	Physics	Winners - 100 points
Mathematics and	within this subject area	University Open International Olympiad for	Mathematics	Awardee- 50 points
Physics		university students and young professionals	Fundamental Mechanics	
			Computer Technology (Fundamental	
		Year of 2022/2023, 2023/2024	Computer Science and Information	
			Technology)	
		'I am a Professional' All-Russia Olympiad for	Physics and Technologies of the unique	Students-medal holders -
		students	MEGA-Science facilities	100 points
			Programming and Information	Winners - 50 points
		Year of 2022/2023, 2023/2024	Technology	Awardee- 25 points
			Quantum Technologies	
			Laser, plasma and radiation	
			technologies	
		'Vysshaia Liga' Olympiad for students and alumni	Physics	Students-medal holders - 15
				points
		Year of 2022/2023, 2023/2024		Winners - 10 points
				Awardees - 5 points

'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Basic and Applied Physics Computer sciences, theoretical and ;aser physics	Winners - 10 points Awardees - 5 points
National Research Nuclear University Moscow Engineering Physics Institute International Physics and Engineering Olympiad for students Year of 2022/2023, 2023/2024	Physics	Winners - 10 points Awardees - 5 points
Bachelor's/Specialist's degree	A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Degree with distinction - 20 points Degree without distinction - 15 points
	Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction - 10 points Degree without distinction - 5 points
Information about publications on topics relevant to the enlarged group of fields of training	Scientific publications in scientific periodicals	30
(specialities) within which the admission to degree programmes is carried out (a copy of the article/screen shot, a copy of the publication's imprint page, and a copy of the collection contents, indexing confirmation must be provided)	Articles in other scientific periodicals	15
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective	Participation in international or national events confirmed by the factually published (including e-publications)	15

enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	report abstracts or by the event programme	
Documents and materials confirming the applicant's participation in research projects supported by grants or work experience in the area of specialisation relevant to the respective enlarged group of fields of training and specialities in which the admission to the degree programme is being carried out.  Only one document is considered. Should there be	Member of a research team, a grant, a project or a employment record	10
two or more documents, the points shall not be summed.		
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.		10
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Certificates of the winners and awardees of competitions of research and development projects, international game competitions, as well as of student Olympiads in the subject relevant to	Diploma of the winner of open competitions of students' research projects, game competitions, student Olympiads	10
the field of study. Considered shall be Olympiads and contests held by educational organisations of higher education	Diploma of an awardee of open competitions of students' research projects, game competitions, student Olympiads	5
Year of 2022/2023, 2023/2024 Only one document is considered. Should there be two or more certificates, the points shall not be summed		

		Documents confirming the reference level of English (international certificates, 'excellent'/passed grade in the transcript):  Only one document is considered. Should there be two or more documents, the points shall not be summed.		5
03.04.02 Physics	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Physics	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Physics Physics and Technologies of the unique MEGA-Science facilities Nuclear Physics and Technology Quantum Technologies Laser, plasma and radiation technologies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Physics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Basic and Applied Physics Computer sciences, theoretical and ;aser physics	Winners - 10 points Awardees - 5 points
		National Research Nuclear University Moscow Engineering Physics Institute International Physics and Engineering Olympiad for students Year of 2022/2023, 2023/2024	Physics	Winners - 10 points Awardees - 5 points
		Bachelor/Specialist degree	A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Degree with distinction - 20 points Degree without distinction - 15 points

	Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction - 10 points Degree without distinction - 5 points
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree	Scientific publications in scientific periodicals	30
programmes is carried out (a copy of the article/screen shot, a copy of the publication's imprint page, and a copy of the collection contents, indexing confirmation must be provided)	Articles in other scientific periodicals	15
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	15
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents and materials confirming the applicant's participation in research projects supported by grants or work experience in the area of specialisation relevant to the respective enlarged group of fields of training and specialities in which	Member of a research team, a grant, a project or a employment record	10

the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be		10
two or more certificates, the points shall not be summed		
Certificates of the winners and awardees of competitions of research and development projects, international game competitions, as well as of student Olympiads in the subject relevant to	Diploma of the winner of open competitions of students' research projects, game competitions, student Olympiads	10
the field of study. Considered shall be Olympiads and contests held by educational organisations of higher education Year of 2022/2023, 2023/2024	Diploma of an awardee of open competitions of students' research projects, game competitions, student Olympiads	5
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents confirming the reference level of English (international certificates, 'excellent'/passed grade in the transcript): Only one document is considered. Should there be two or more documents, the points shall not be summed.		5

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 03.00.00 'Physics and Astronomy' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

04.00.00 CHEMISTR	Y			
04.04.01 Chemistry	Chemistry	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	International Tournament in Natural Sciences	Winners - 50 points Awardee- 25 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Chemistry Chemical Technology Petroleum Engineering	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		CASE-IN International Engineering Championship Year of 2022/2023, 2023/2024	Subject area 'Oil Chemistry'	Winners - 10 points
		Information about publications in journals with a non-zero impact-factor	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 25 points
		Documents and materials confirming the applicant's participation in scientific events	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 12 points
		Reference Letters	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 3 points

<sup>1.</sup> Documents and individual achievement evidences within admissions to the enlarged group of specialities and fields of study of 04.00.00 programme 'Chemistry' shall be accepted only if they meet the following requirements.

## Information about publications in journals with a non-zero impact-factor

If there are no articles published in journals with a non-zero impact-factor, assigned shall be 0 (zero) points thereto.

Requirements applied to the content and design of the list of publications in journals with a non-zero impact-factor

<sup>2.</sup> Points for the established types of individual achievements within the enlarged group of specialities and fields of study 04.00.00 'Chemistry' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

Failure to comply with the requirements herein shall result in rejecting to consider the respective application and the applicant's publication activity and, consequently, in assigning zero points therefor.

The first page must contain a general list of articles, whose teams of authors include the applicant, according to the specified form, and bear a handwritten signature of the applicant. All table columns, except the one for the journal's impact-factor, must be filled out.

Publication activity Applicant's Full name: articles

No	Publisher's imprint (authors, title of the publication, title of the journal, year volume, edition, pages)	Journal's impact-factor <sup>1</sup>	Link to the article on the Internet or DOI
1			
2			
3			

I confirm that the information provided is correct:

Signature Applicant's full name

Date

- 1. Impact-factor information is available for free on the journal's website. Its absence in the application shall not be the grounds for refusal in considering the documents submitted.
- 2. The second and subsequent pages must contain confirmations full texts of the articles organised in the order of their appearance in the table.
- 3. If the article has not been published but accepted for publication, the column 'Online link or DOI' shall contain 'in print', and either a confirmation from an editorial office certified by its seal (Russian translated journals) or an email confirmation of acceptance for publication (publication status: accepted).

#### **Evaluation criteria**

Points for publications shall be scored based on the statistic approach commonly used by the Federal State Statistic Service. For each applicant, the total impact-factor shall be calculated to the nearest hundredths (Average Impact-Factor) of the journals, which have published applicants' articles, based on the journal's impact-factor valid for the previous year according to Journal Citation Reports (Web of Science Core Collection, Thomson Reuters) (priority) and CWTS Journal Indicators (Scopus bibliographic database produced by Elsevier, SNIP) data.

The median value of the Average Impact-Factor (AIF) shall be determined, on the basis of which the sample shall be divided into four quartiles. The median shall be a certain mark or level of an indicator (a value in the totality of data), dividing the ranked data (sorted in our case in descending order of AIF values) into two equal parts. One part of the reference data is below the mark, the other part is above.

Further, for each quartile a maximum score value shall be established, which shall correspond to the maximum AIF (see the table).

Place of the total	Quartile 1	Quartile 2	Quartile 3	Quartile 4
impact-factor in sampling, %	100 -75	75 -5	50 -25	<25
Points maximum	25	20	15	10
Points minimum	20	15	10	5

<sup>&</sup>lt;sup>1</sup> Impact-factor of the recent valid year; after the figure in brackets, please, indicate Journal Citation Reports (Web of Science Core Collection, Thomson Reuters) (preferably) and CWTS Journal Indicators (Scopus bibliographic database produced by Elsevier, SNIP).

Within each quartile, a ranking is carried out from the highest value of the total impact factor to the lowest, and points are awarded to a particular applicant according to the following formula:

#### Documents and materials confirming the applicant's participation in scientific events

Accepted are the abstracts of full-time conferences in three recent years (2020 - 2024), with more than 50 attendants. The applicant must be a presenting author, which should be outlined in the abstracts.

If documents are submitted online, information about the abstracts must be provided in a single file in .pdf format.

A confirmation of the presence of abstracts shall be the title page of the collection of article for the respective conference, a conference programme or a copy of the abstracts from the collection of articles.

## Requirements applied to the contents and layout of information about conference attendance

- 1. The first page must contain a general list of reports, whose teams of authors include the applicant, according to the specified form, and bear a handwritten signature of the applicant. All fields of the table must be filled in.
- 2. If there is no links to the conference web site, abstracts of this conference shall be rejected.
- 3. To have the conference abstracts considered, the number of attendees must be at least 50 persons; the conference area must be chemistry or relevant sciences.
- 4. The first page of the set of documents must meet requirements for design according to the sample below and contain the general list of the applicant's report abstracts and bear their handwritten signature.

Publication activity Applicant's Full name: report abstracts

v 11	No	Publisher's imprint (authors, title of the publication, title of the conference, year, volume, pages)	Conference type <sup>1</sup>	Number of attendees <sup>2</sup>	Report typea <sup>3</sup>	Diploma (available, not available) <sup>4</sup>	Web address of the conference web site
------	----	---	------------------------------	----------------------------------	------------------------------	--	--

<sup>&</sup>lt;sup>1</sup> Please, indicate the conference type according to its title: All-Russia Youth, International Youth, All-Russia, International

<sup>&</sup>lt;sup>2</sup> Please, indicate the number of the conference attendees (usually it is sufficient to indicate a named list of reporters at the end of the abstract book or calculate the number of presented reports).

<sup>&</sup>lt;sup>3</sup> oral/stand-based

<sup>&</sup>lt;sup>4</sup> If the report has won a Diploma of I, II or III degree, enter 'yes'.

1				
2				
3				

I confirm that the information provided is correct:

Signature

Applicant's full name

Date

- 5. The second and subsequent pages must contain confirmations organised in the order of their appearance in the table:
  - Title page
  - ❖ Abstracts from the programme of the conference, which unambiguously discloses the report type
  - Abstracts
  - ❖ Diploma (if it is of I, II or III degree)
- 6. If reports were delivered orally, the initiate sore of points shall be doubled. Confirmation of the report type shall be the programme of the conference, which clearly shows that the report has been delivered at the oral session.
- 7. If the report has won a diploma (of I, II, or III degree provided there is a confirmation; attendee's diploma or certificate shall not be accepted), a multiplying factor 1.5 is used.

#### **Evaluation criteria**

Assignment of initial points shall be conducted in accordance with the reference table.

Conference level	Initial points
Youth all-Russia	1
Youth international	2
All-Russia	3
International	5

For the conferences of the similar level with several report abstracts, initial points shall be multiplied by their number.

Each applicant's initial points scored shall be summed; a maximum score of initial points shall be determined, which shall be equal to 12 points.

Calculations of points scored by each applicant shall be carried out according to the following proportion:

The totality of primary points scored by the applicant

Points, scored by the applicant = 12 x -----

The highest score of primary points among all applicants

If resulting points are fractions, they shall be rounded to whole values.

## **Reference Letters**

Letters of reference (as many as three (3)) from the faculty and research staff of the university that the applicant has graduated, as well as from employers correspondent to the field of study of the programme.

The signature must be verified by HR office of the place of work.

The letter shall indicate a degree programme for which the applicant is applying to study.

If the documents are submitted online, the scans of the letters shall be attached in a single .pdf file named 'Recommendations'.

Evaluated is the number of recommendation letters and the argumentation of the provided recommendation.

05.00.00 EARTH SC			T. 4.0.	100
05.01.04 Geology	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Geology Petroleum Engineering Sustainable Development and Environmental Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Geology	Winners - 10 points Awardees - 5 points
		CASE-IN International Engineering Championship Year of 2022/2023, 2023/2024	Subject areas 'Mining and Mineral Engineering', Geological Prospecting', 'Oil Chemistry'	Winners - 10 points
		A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to	Degree with distinction	10
		the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5
			Grade average point 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities)	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20

		within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database  Articles in publications indexed by	15
		provided)	RSCI scientometric database or from the Higher Attestation Committee list.	10
		Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
		Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10
05.04.02 Geography	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Geography Geourbanistics Sustainable Development and Environmental Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Geography and Tourism	Winners - 10 points Awardees - 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points

A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
Bachelor's/specialist's /master's degree in other fields of studies and specialities	Degree with distinction	5
	Grade average point more than 4.0 of the diploma without distinction	3
Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10

05.04.03 Cartography and Geoinformatics	Geographical Information Mapping	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Earth Sciences	Winners - 100 points Awardees- 50 points
		Year of 2022/2023, 2023/2024  'I am a Professional' All– Russia Olympiad for students Year of 2022/2023, 2023/2024	Geography	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
		specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other fields of studies and specialities	Degree with distinction	5
			Grade average point more than 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be provided)	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
			Articles in publications indexed by RSCI (RSCI core) scientometric database	15
		providedy	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
		Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10

		Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10
05.04.04 Hydrometeorology	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Geography Sustainable Development and Environmental Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
		specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5
			Grade average point more than 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
		is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be provided)	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
		provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10

		Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
		Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10
05.04.06 Ecology and Nature Management	Geoecology: Monitoring, Nature Management and Environmental Safety	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Biotechnologies Geography Sustainable Development and Environmental Management	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Soil Science and Ecology	Winners - 10 points Awardees - 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'Ecology of my native country' nomination Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points
		A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to	Degree with distinction	10
		the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5

			Grade average point more than 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
		is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
		provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
		Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
		Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10
05.04.06 Ecology and Nature Management	Ecology: Biodiversity and Nature Protection	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Current issues in biology Soil Science and Ecology	Winners - 10 points Awardees - 5 points
_		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Biology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points

Points for the established types of individual achievements in the enlarged group of specialities and fields of study 05.00.00 Earth Sciences (except degree programme 'Ecology. Biodiversity and Environment Protection') shall be summed; at the same time, only one documents shall be considered for one type of individual achievements unless otherwise is provided by this List of individual achievements of those applying for the admission to mater's degree programmes.

6.04.01 Biology	All degree programmes within this subject area	'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Bioengineering and Bioinformatics Biotechnologies Biology	Students-medal holders - 100 points Winners - 50 points
		1 ear of 2022/2023, 2023/2024	Biology	Awardee- 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Current issues in biology	Winners - 10 points Awardees - 5 points
		Degree credential	A bachelor's degree/specialist degree diploma in the field of study and	5
			speciality correspondent to the enlarged group of the fields of study and	
			specialities currently open for admissions within a respective degree	
			programme.  Bachelor's/specialist's degree in other	3
		Information about publications in journals indexed	fields of study and specialities  Articles in publications indexed by	20
		by scientometric databases Web of Science Core Collection, Scopus, Russian Science Citation Index (RSCI) (RSCI core) on relevant to the	Web of Science Core Collection and Scopus scientometric databases.	
		enlarged group of fields of training (specialities) within which the admission to the degree		
		programme is being carried out.  (A copy of the article, a copy of the publication's		
		imprint page, and indexing confirmation must be provided) Points shall be scored for each article: if there are several articles, the points shall be	Articles in publications indexed by RSCI (RSCI core) scientometric database	10
		summed.  Documents and materials confirming the	The lead of the research grant (grants	20
		applicant's participation in research projects supported by grants on topics relevant to the	from Federal Executive bodies, Governments of the entities constituting	20
		respective enlarged group of fields of training (specialities) in which the admission to the degree	the Russian Federation, state and science foundations)	
		programme is being carried out. Points shall be scored for each project: if there are several	The executor of the research grant (grants from the president, Federal	10
		projects, the points shall be summed.	Executive bodies, Governments of the	

			entities constituting the Russian Federation, state and science foundations) The lead of the research grant (grants from research and educational organisations of higher education) Member of the research team (grants	10
			from research and educational organisations of higher education)  Travel-grant for the trip to conference	5
06.04.02 Soil Science	Soil Science	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardees- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Biotechnologies Biology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Soil Science and Ecology	Winners - 10 points Awardees - 5 points
		A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
		specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5
			Grade average point more than 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities)	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20

within which the admission to degree programmes is carried out  (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database  Articles in publications indexed by	10
provided)	RSCI scientometric database or from the Higher Attestation Committee list.	
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 06.00.00 'Biology' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

09.00.00 INFORMA	TICS AND COMPUTER SC	IENCE		
09.00.00 Applied	All degree programmes	Petropolitan Science (Re)Search, St Petersburg	Mathematics	Winners - 100 points
Informatics	within this subject area	University Open International Olympiad for	Computer Technology (Fundamental	Awardee- 50 points
		university students and young professionals	Computer Science and Information	
			Technology)	
		Year of 2022/2023, 2023/2024	International Tournament in Natural	
			Sciences	
		'I am a Professional' All-Russia Olympiad for students	Information and Cyber Security Artificial Intelligence	Students-medal holders - 100 points
		Year of 2022/2023, 2023/2024	Mathematical Modelling	Winners - 50 points
		Teal of 2022/2023, 2023/2024	Wathematical Wodelling	Awardee- 25 points
				Awardee- 23 points

	'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024	Programming and Information Technology Robotics Security of Information Systems and Technologies of Critical Facilities Chemistry Chemical Technology	Winners - 10 points
	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
	'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points
	'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
Artificial Intelligence and Data Science	Bachelor's degree /specialist's degree in the field of study and specialities related to the enlarged group of fields of study and specialities of 02.00.00 or 09.00.00 programmes	Degree with distinction	10
		Degree without distinction	8
	Information about publications on topics relevant to the scientific speciality to which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases,a s well as in periodicals included into the RSCI core and peerreviewed journals, recommended by HAC.	15

Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the	Scientific publications in indexed by the RSCI scientometric database.	10
points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored. Report abstracts shall be deemed a publication less than five (pages) of length if its imprint data supposes that it has been presented at a conference.  It is recommended to provide a link to the web	Any other scientific publication	5
page of the article in a scientometric database to speed up the verification of the work.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) to which the	Participation in a scientific event as a presenter.  The delivery of the report shall be confirmed by the programme of the	5 (+5)
admission to the degree programme is being carried out. Accepted can be but two documents. If there are three documents being provided, points shall be scored based on two of them.	conference with presenters named.	
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.	8
the admissions is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed.	Members of research teams, a grant, a project under the order from state-owned science foundatoins, reaserch organnisations or educational organisations of higher education.	10

		Documents confirming respective working experience addressing the used in the professional activity competences correlated with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity).	12 months and more  From 6 months to 12 months	5
		Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the fixed-term employment contract; - a confirmation from the place of work (sealed and signed by the official)		
	Digital Technologies in Chemistry	Information about publications in journals with a non-zero impact-factor	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 25 points
		Documents and materials confirming the applicant's participation in scientific events	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 12 points
		Reference Letters	Requirement for the content and design and evaluation criteria are applicable for assigning	up to 3 points
09.04.04 Software Engineering	Software Engineering	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Fundamental Mechanics Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Information and Cyber Security Artificial Intelligence Mathematical Modelling	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points

'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024	Programming and Information Technologies Software Engineering Robotics Security of Information Systems and Technologies of Critical Facilities	Winners - 10 points
'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points
'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs  Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
Bachelor's degree /specialist's degree in the field of study and specialities related to the enlarged group of fields of study and specialities of 02.00.00 or 09.00.00 programmes	Degree with distinction  Degree without distinction	10 8
Information about publications on topics relevant to the scientific speciality to which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases, as well as in periodicals included into the RSCI core and peerreviewed journals, recommended by HAC.	15

Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific	Scientific publications in indexed by the RSCI scientometric database.	10
event, points under this clause shall not be scored. Report abstracts shall be deemed a publication less than five (pages) of length if its imprint data supposes that it has been presented at a conference.	Any other scientific publication	5
It is recommended to provide a link to the web page of the article in a scientometric database to speed up the verification of the work.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) to which the admission to the degree programme is being carried out.	Participation in a scientific event as a presenter.  The delivery of the report shall be confirmed by the programme of the conference with presenters named.	5 (+5)
Accepted can be but two documents. If there are three documents being provided, points shall be scored based on two of them.		
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial desilgns.	8

	Only one document is considered. Should there be two or more documents, the points shall not be summed.	Members of research teams, a grant, a project under the order from state-owned science foundatoins, reaserch organnisations or educational organisations of higher education.	10
	Documents confirming respective working experience addressing the used in the professional activity competences correlated with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity). Provided must be one of the following documents:  - a copy of the Employment Book (extracts) certified by the organisation's human resource	12 months and more	5
Degree programmes 'Artificial Intelligence and Da	department officer; - a copy of the fixed-term employment contract; - a confirmation from the place of work (sealed and signed by the official)	From 6 months to 12 months	2

### Degree programmes 'Artificial Intelligence and Data Science' and 'Software Engineering'

Points for the established types of individual achievements shall be summed; at the same time, only one documents shall be considered for one type of individual achievements unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

# Degree programmes 'Digital Technologies in Chemistry'

- 1. Documents and individual achievement evidences within admissions to the enlarged group of specialities and fields of study of 04.00.00 programme 'Chemistry' shall be accepted only if they meet the following requirements.
- 2. Points for the established types of individual achievements within the degree programme 'Digital Technologies in Chemistry' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

## Information about publications in journals with a non-zero impact-factor

If there are no articles published in journals with a non-zero impact-factor, assigned shall be 0 (zero) points thereto. Requirements applied to the content and design of the list of publications in journals with a non-zero impact-factor

Date

Failure to comply with the requirements herein shall result in rejecting to consider the respective application and the applicant's publication activity and, consequently, in assigning zero points therefor.

The first page must contain a general list of articles, whose teams of authors include the applicant, according to the specified form, and bear a handwritten signature of the applicant. All table columns, except the one for the journal's impact-factor, must be filled out.

Publication activity Applicant's Full name: articles

No	Publisher's imprint (authors, title of the publication, title of the journal, year, volume, edition, pages)	Journal's impact-factor <sup>1</sup>	Link to the article on the Internet or DOI
1			
2			
3			

I confirm that the information provided is correct:

Signature Applicant's full name

- 2. Impact-factor information is available for free on the journal's website. Its absence in the application shall not be the grounds for refusal in considering the documents submitted.
- 3. The second and subsequent pages must contain confirmations full texts of the articles organised in the order of their appearance in the table.
- 4. If the article has not been published but accepted for publication, the column 'Online link or DOI' shall contain 'in print', and either a confirmation from an editorial office certified by its seal (Russian translated journals) or an email confirmation of acceptance for publication (publication status: accepted).

### **Evaluation criteria**

Points for publications shall be scored based on the statistic approach commonly used by the Federal State Statistic Service. For each applicant, the total impact-factor shall be calculated to the nearest hundredths (Average Impact-Factor) of the journals, which have published applicants' articles, based on the journal's impact-factor valid for the previous year according to Journal Citation Reports (Web of Science Core Collection, Thomson Reuters) (priority) and CWTS Journal Indicators (Scopus bibliographic database produced by Elsevier, SNIP) data.

The median value of the Average Impact-Factor (AIF) shall be determined, on the basis of which the sample shall be divided into four quartiles. The median shall be a certain mark or level of an indicator (a value in the totality of data), dividing the ranked data (sorted in our case in descending order of AIF values) into two equal parts. One part of the reference data is below the mark, the other part is above.

Further, for each quartile a maximum score value shall be established, which shall correspond to the maximum AIF (see the table).

Place of the total	Quartile 1	Quartile 2	Quartile 3	Quartile 4
impact-factor in sampling, %	100 -75	75 -50	50 -25	<25
Points maximum	25	20	15	10

<sup>&</sup>lt;sup>1</sup> Impact-factor of the recent valid year, after the figure in brackets, please, indicate Journal Citation Reports (Web of Science Core Collection, Thomson Reuters) (preferably) and CWTS Journal Indicators (Scopus bibliographic database produced by Elsevier, SNIP).

Points minimum	20	15	10	5
-------------------	----	----	----	---

Within each quartile, a ranking is carried out from the highest value of the total impact factor to the lowest, and points are awarded to a particular applicant according to the following formula:

## Documents and materials confirming the applicant's participation in scientific events

Accepted are the abstracts of full-time conferences in three recent years (2020 - 2024), with more than 50 attendants. The applicant must be a presenting author, which should be outlined in the abstracts.

If documents are submitted online, information about the abstracts must be provided in a single file in .pdf format.

A confirmation of the presence of abstracts shall be the title page of the collection of article for the respective conference, a conference programme or a copy of the abstracts from the collection of articles.

# Requirements applied to the contents and layout of information about conference attendance

- 1. The first page must contain a general list of reports, whose teams of authors include the applicant, according to the specified form, and bear a handwritten signature of the applicant. All fields of the table must be filled in.
- 2. If there is no links to the conference web site, abstracts of this conference shall be rejected.
- 3. To have the conference abstracts considered, the number of attendees must be at least 50 persons; the conference area must be chemistry or relevant sciences. of Science
- 4. The first page of the set of documents must meet requirements for design according to the sample below and contain the general list of the applicant's report abstracts and bear their handwritten signature.

Publication activity Applicant's Full name: report abstracts

No	Publisher's imprint (authors, title of the publication, title of the conference, year, volume, pages)	Conference type <sup>1</sup>	Number of attendees <sup>2</sup>	Report typea <sup>3</sup>	Diploma (available, not available) <sup>4</sup>	Web address of the conference web site
1						
2						
3						

I confirm that the information provided is correct:

Signature *Applicant's full name* 

Date

- 5. The second and subsequent pages must contain confirmations organised in the order of their appearance in the table:
  - Title page
  - ❖ Abstracts from the programme of the conference, which unambiguously discloses the report type
  - Abstracts
  - ❖ Diploma (if it is of I, II or III degree)
- 6. If reports were delivered orally, the initiate sore of points shall be doubled. Confirmation of the report type shall be the programme of the conference, which clearly shows that the report has been delivered at the oral session.
- 7. If the report has won a diploma (of I, II, or III degree provided there is a confirmation; attendee's diploma or certificate shall not be accepted), a multiplying factor 1.5 is used.

### **Evaluation criteria**

Assignment of initial points shall be conducted in accordance with the reference table.

Conference level	Initial points
Youth all-Russia	1
Youth international	2
All-Russia	3
International	5

<sup>&</sup>lt;sup>1</sup> Please, indicate the conference type according to its title: All-Russia Youth, International Youth, All-Russia, International

<sup>&</sup>lt;sup>2</sup> Please, indicate the number of the conference attendees (usually it is sufficient to indicate a named list of reporters at the end of the abstract book or calculate the number of presented reports).

<sup>&</sup>lt;sup>3</sup> oral/stand-based

<sup>&</sup>lt;sup>4</sup> If the report has won a Diploma of I, II or III degree, enter 'yes'.

For the conferences of the similar level with several report abstracts, initial points shall be multiplied by their number.

Each applicant's initial points scored shall be summed; a maximum score of initial points shall be determined, which shall be equal to 12 points.

Calculations of points scored by each applicant shall be carried out according to the following proportion:

The totality of primary points scored by the applicant Points, scored by the applicant = 12 x ------

The highest score of primary points among all applicants

If resulting points are fractions, they shall be rounded to whole values.

### **Reference Letters**

Letters of reference (as many as three (3)) from the faculty and research staff of the university that the applicant has graduated, as well as from employers correspondent to the field of study of the programme.

The signature must be verified by HR office of the place of work.

The letter shall indicate a degree programme for which the applicant is applying to study.

If the documents are submitted online, the scans of the letters shall be attached in a single .pdf file named 'Recommendations'.

Evaluated is the number of recommendation letters and the argumentation of the provided recommendation.

## 21.00.00 APPLIED GEOLOGY, MINING, PETROLEUM ENGINEERING AND GEODESY

21.04.01 Petroleum Engineering	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Earth Sciences	Winners - 100 points Awardee- 50 points
		Year of 2022/2023, 2023/2024 'I am a Professional' All-Russia Olympiad for	Geology	Students-medal holders -
		students	Mining Engineering	100 points
			Petroleum Engineering	Winners - 50 points
		Year of 2022/2023, 2023/2024		Awardee- 25 points
		CASE-IN International Engineering Championship Year of 2022/2023, 2023/2024	Subject area 'Petroleum Engineering'	Winners - 10 points
		'Gazprom' olympiad for students Year of 2022/2023, 2023/2024	Profile 'Petroleum Engineering'	Winners– 10 points

		A bachelor's degree/specialist's degree diploma in	Degree with distinction	10
		the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
		Bachelor's/specialist's /master's degree in other	Degree with distinction	5
		fields of study and specialities	Grade average point more than 4.0 of the diploma without distinction	3
		Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
		is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
		provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
		Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
		Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10
21.04.02 Land Management and Cadastres	Management of Real Estate Objects and Development of Territories	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardee- 50 points

'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Geography	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5
	Grade average point more than 4.0 of the diploma without distinction	3
Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10
programme is being carried out.  Documents and materials confirming the applicant's participation in research projects supported by grants on topics relevant to the respective enlarged group of fields of training (specialities) in which the admission to the degree programme is being carried out (including students' start-ups supported by grants).		10

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 21.00.00 'Applied Geology, Mining, Oil and Gas Engineering and Geodesy' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

27.00.00 Systems	Engineering-Oriented	Petropolitan Science (Re)Search, St Petersburg	Computer Technology (Fundamental	Winners - 100 points
Analysis and Control	Computer Science and Artificial Intelligence	University Open International Olympiad for university students and young professionals	Computer Science and Information Technology)	Awardee- 50 points
		Year of 2022/2023, 2023/2024		
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Artificial Intelligence Robotics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics Innovation studies	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Innovation studies	Winners - 10 points Awardee- 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points
		'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
		Bachelor's degree /specialist's degree in the field of study and specialities the following the	Degree with distinction	30

enlarged group of fields of study and specialities: 01.00.00 Mathematics and Mechanics, 02.00.00 Computer and Information Sciences, 03.00.00 Physics and Astronomy, 09.00.00 Informatics and	Grade average point higher than 4.5 of the diploma without distinction	25
Computer Science, 16.00.00 'Physical and Technical Sciences and Technologies' 27.00.00 Control in Engineering Systems, 28.00.00 'Nanotechnologies and Nanomaterials	Degree with higher than 4.0 grade point average	15
Bachelor's/specialist's degree in other fields of study and specialities	Degree with distinction	15
	Grade average point more than 4 of the diploma without distinction	10
	Grade average point lower than 4 of the diploma without distinction	5
Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection must be provided).	Scientific publications in periodicals indexed by Scopus, WoS or included in the HAC list, as well as chapters in the preer-reviewed monographs and collections of articles published by leading international and national publishing houses.	10
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report	Scientific publications in non- preiodicals (collections of articles, proceedings of conferences, symposia, seminars) indexed by Scopus, WoS, or RSCI	7
delivered at the conference or another scientific event, points under this clause shall not be scored.	Articles in other scientific periodicals	4
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on	The participation must be confirmed by the published abstract to the report or the event programme	10

topics relevant to the following enlarged groups of fields of study (specialities): 01.00.00 Mathematics and Mechanics, 02.00.00 Computer and Information Sciences, 03.00.00 Physics and Astronomy, 09.00.00 Informatics and Computer Science, 16.00.00 'Physical and Technical Sciences and Technologies' 27.00.00 Control in Engineering Systems, 28.00.00 'Nanotechnologies and Nanomaterials.  Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.	Scholarship of the President of the Russian Federation	5
Only one document is considered. Should there be two or more documents, the points shall not be	Scholarship from the Government	5
summed.	Scholarship from an educational organisation of higher education	3
	Scholarships from foundations, public associations and other organisations	3
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except Open International Olympiad of St Petersburg University	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5
for university students and young specialists 'Petropolitan Science (Re)Search' and 'I am a Professional' Olympiad) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out. Considered shall	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	3
be Olympiads and contests held by educational organisations of higher education.  Only one document is considered. Should there be	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	5

		two or more documents, the points shall not be summed.  Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	3
			Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	3
27.04.07. Science- Intensive Technologies and the Economics of Innovation	Project Management in Digital Technology and Science-Intensive Economics	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Computer Technology (Fundamental Computer Science and Information Technology) Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Computer Sciences Artificial Intelligence Robotics Business Informatics Economic and Corporate Security Strategic project Management Economics	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024	240.00.000	Winners - 10 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Computer Systems and Networks Applied Mathematics Applied Mathematics and Informatics Innovation studies Economics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Innovation studies Mathematics and Mechanics Econometrics Innovation Management	Winners - 10 points Awardee- 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	Nomination 'Bolshaya Tekhnologicheskaya Razvedka Moyei Strany (Deep Technological Survey of My Country)'	Winners - 10 points

'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Database administration A developer of the virtual and augmented reality Industrial Robotics	Winners - 10 points
Bachelor's degree /specialist's degree in the field of study and specialities the following the enlarged group of fields of study and specialities:	Degree with distinction	30
01.03.01. Mathematics, 01.03.02 Applied Mathematics and Informatics; 09.03.03 Applied Informatics; 27.03.03 Systemic Analysis; 27.03.07 Innovation; 38.03.05 Business Informatics;	Grade average point higher than 4.5 of the diploma without distinction	25
38.03.01 (02)	Degree with higher than 4.0 grade point average	15
Bachelor's/specialist's degree in other fields of studies and specialities	Degree with distinction	15
	Grade average point higher than 4 of the diploma without distinction	10
	Grade average point lower than 4 of the diploma without distinction	5
Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection must be provided).	Scientific publications in periodicals indexed by Scopus, WoS or included in the HAC list, as well as chapters in the preer-reviewed monographs and collections of articles published by leading international and national publishing houses.	10
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report	Scientific publications in non- preiodicals (collections of articles, proceedings of conferences, symposia, seminars) indexed by Scopus, WoS, or RSCI	7

delivered at the conference or another scientific event, points under this clause shall not be scored.	Articles in other scientific periodicals	4
Documents and materials confirming the applicant's participating in scientific events (conferences, seminars, scientific schools) on topics relevant to the following enlarged groups of fields of study (specialities): 01/03/2001.  Mathematics, 01.03.02 Applied Mathematics and Informatics; 09.03.03 Applied Informatics; 27.03.03 Systemic Analysis; 27.03.07 Innovation; 38.03.05 Business Informatics; 38.03.01 (02)  Only one document is considered. Should there be two or more documents, the points shall not be summed.	The participation must be confirmed by the published abstract to the report or the event programme	10
Documents confirming scholarships granted by ministries, agencies, foundations, educational	Scholarship of the President of the Russian Federation	5
institutions of higher education.  Only one document is considered. Should there be two or more documents, the points shall not be	Scholarship from the Government	5
summed.	Scholarship from an educational organisation of higher education	3
	Scholarships from foundations, public associations and other organisations	3
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except Open International Olympiad of St Petersburg University	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5
for university students and young specialists 'Petropolitan Science (Re)Search' and 'I am a Professional' Olympiad) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	3

programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of higher education.  Only one document is considered. Should there be two or more documents, the points shall not be	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	5
summed. Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	3
	Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	3

- 1. Assigned shall be the points in the main field of study 27.04.03 Systemic Analysis and Control only for one established type of individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates of completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.
- 2. Points for the established types of individual achievements within the degree programme 27.04.07 'Science-Intensive Technologies and the Economics of Innovation shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

### 28.00.00 NANOTECHNOLOGY AND NANOMATERIALS

28.04.04 Nanosystems and Nanomaterials	High Technology Tech Materials	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	International Tournament in Natural Sciences	Winners - 50 points Awardee- 25 points
		Year of 2022/2023, 2023/2024		
		'I am a Professional' All-Russia Olympiad for	Physics and Technologies of the unique	Students-medal holders - 50
		students	MEGA-Science facilities	points
			Robotics	Winners - 25 points
		Year of 2022/2023, 2023/2024	Material Science and Technology	Awardee- 15 points
		'Vysshaia Liga' Olympiad for students and alumni	Physics	Students-medal holders - 15
		Year of 2022/2023, 2023/2024		points
				Winners - 10 points
				Awardee- 5 points

'Lomonosov' Universiad	Chemistry, Physics and Material Mechanics	Winners - 10 points Awardee- 5 points
Year of 2022/2023, 2023/2024	Chemistry	Tiwardee 5 points
Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to degree	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases.	10
programmes is carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing	Scientific publications in periodicals included into Russian Science Citetion Index.	8
confirmation must be provided).  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	7
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Articles in other scientific periodicals	3
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	Publications in electronic scientific archives of organisations of higher education and research organisations	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study and specialities of 28.00.00	Documents confirming the applicant's participation in scientific events.  The participation must be confirmed by the published abstract to the report or the event programme	4
Nanitechnologies and Nanomaterials programme within which the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be	Documents confirming the applicants participating in science schools/forums at least five (5) days long held by organisations of higher education or research organisations.	4
summed.	Documents confirming the applicant's awards in the form of diplomas/appreciation letters, etc., for their activities within scientific schools on topics relevant to the enlarged group of fields of study and specialities of the 28.00.00 Nanotechnologies and Nanomaterials programme.	20

Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the enlarged group of the fields of study and speciality 28.00.00 Nanotechnologies and Nanomaterials programme to which the admissions is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed.	Authors of the protected results of intellectual activity: software for computers, databases, inventions, utility models and industrial designs.  Members of research teams, a grant, a project under the order from state-owned science foundatoins, reaserch organnisations or educational organisations of higher education.	10
Letters of reference (as many as three (3)) from the faculty and research staff of the university that the applicant has graduated, as well as from employers correspondent to the field of study of the programme.  Evaluated is the number of recommendation letters and the argumentation of the provided recommendation.  The letter shall indicate a degree programme for which the applicant is applying to study.  The signature must be verified by HR office of the place of work.  If the documents are submitted online, the scans of the letters shall be attached in a single .pdf file named 'Recommendations'.		0 -21

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 28.00.00 'Nanosystems and Nanomaterials' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

37.00.00 PSYCHOLOGICAL SCIENCES					
37.04.01 Psychology	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Psychology	Winners - 100 points Awardee- 50 points	

	'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024  'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Psychology  Psychology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Psychology	Winners - 10 points Awardees - 5 points
General and Cognitive Psychology	Bachelor's/Specialist's degree  Accepted is only one document within the section. Should there be two or more documents, the documents points shall not be summed	Bachelor's/specialist's degree in the fields of study and specialities relevant to the main fields of study: 37.03.01 'Psychology', 37.05.01 'Clinical Psychology' or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study in degree programmes  Bachelor's/ specialist's or master's degree in the fields of study relevant to: Computer and Information Sciences', Biological Sciences', Arts and Humanities'. 'Philosophy', 'Linguistics and Literary Studies', or relevant to them fields of study and specialities included into the conglomerate of cognitive sciences approved by the	Degree with distinction - 20 points Degree without distinction - 15 points  Degree with distinction - 10 points Degree without distinction - 5 points
		Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study of degree programmes Bachelor's/ specialist's or master's degree in other fields of study	5

Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).  For the publications submitted to RSCI, a link to	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases	10
the article in e-library must be provided.  Certificates/diplomas/in-print confirmations shall	Scientific publications in periodicals recommended by the HAC	7
not be deemed as the confirmation of the publication	Articles in other scientific periodicals	5
If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective of fields of study to which the admission to the degree programme is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme  As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well	5
Only one document is considered. Should there be two or more documents, the points shall not be summed.	as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme	
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics	Members of research teams, a grant, a project under the order from state-owned science foundations, research	10

		relevant to the field of study to which programme the admissions is being carried out.	organisations and educational organisations of higher education	
		Only one document is considered. Should there be two or more certificates, the points shall not be summed		
		Documents confirming the foreign language competence (international certificates and other documents).	English language reference credentials: TOEFL (score 65 and higher ), IELTS (5.5 points and higher ), Cambridge CAE – C, B, A; Cambridge CPE – C,	5
		Only one document is considered. Should there be two or more documents, the points shall not be summed.	B, A, Cambridge FCE – B, A; Level 2 Test in English (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher) at the Centre of Language	
			Testing at St Petersburg University; Level 3 Test in English (CEFR Level C1) (Grades C,B,A 51% and higher) at the Centre of Language Testing at St	
		Documents confirming the assignment of personal scholarship.	Petersburg University  Documents confirming personal scholarships granted by ministries, agencies, state foundations and	5
		Only one document is considered. Should there be two or more documents, the points shall not be summed.	educational institutions of higher education	
		Standard credential of professional retraining in the non-degree programme 'Psychology' issued by an educational organisations ranked among top 300 by		5
27.04.01 P. 1-1-1	Children's M. a. I.H. 19	international ratings (the Shanghai Rating, Three Missions of University International Moscow International Rating)	A bashalawa daaway (1942)	Description of the control of the co
37.04.01 Psychology	Children's Mental Health and Early Interventions	Bachelor's/Specialist's degree	A bachelor's degree/specialist's degree diploma in the field of study and speciality relevant to the enlarged group of the fields of study and specialities	Degree with distinction - 35 points Degree without distinction - 30 points

	currently open for admissions within a respective degree programme: 37.03.01 'Psychology', 37.05.01 'Clinical Psychology'  Bachelor's/specialist's degree in the fields of study and specialities relevant to the main fields of study: 44.04.01 Pedagogical Education; 44.03.02 Psychologist-Pedagogical Education; 44.03.03 Special (Defectological) Education; 44.03.05 Pedagogical Education (with two training profiles); 06.03.01 Biology; 39.03.01 Sociology or relevant to them fields of study and specialities according to the lists of specialities and fields of study of degree programmes approved by the Ministry of Science and Higher Education of the Russian Federation.	Degree with distinction - 25 points Degree without distinction - 20 points
'Excellent' grade for the bachelor's degree/ specialist's degree graduation project  Only one document is considered. Should there be two or more certificates, the points shall not be summed	'Excellent' grade for the graduation project	4
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).  For the publications submitted to RSCI, a link to	Scientific articles in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases, RSCI and periodicals indexed by HAC.	2
the article in e-library must be provided.	Articles in other scientific periodicals	1

Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication  If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective of fields of study to which the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	The participation must be confirmed by the published abstracts to the report or by the event programme.  As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme	1
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out. The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Members of research teams, a grant, or a project under the order from state- owned science foundations, research organisations or educational organisations of higher education	1
Documents confirming the foreign language competence (international certificates and other documents).	English language reference credentials: TOEFL (score 65 and higher); IELTS ( 5.5 points and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A,	3

	Only one document is considered. Should there be two or more certificates, the points shall not be summed	Cambridge FCE – B, A; Level 2 Test in English (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher) at the Centre of Language Testing at St Petersburg University; Level 3 Test in English (CEFR Level C1) (Grades C,B,A 51% and higher) at the Centre of Language Testing at St Petersburg University	
		Certificates from state-owned educational organisations confirming the Upper Intermediate level of proficiency in English and higher.	2
	Winners' or laureate's diploma for contests of research, design work, international Olympiads for university students (except Olympiads and contests accepted as individual achievements	Diploma of the winner of open international competitions of students' research projects, international student Olympiads	2
	within admissions to the respective degree programmes) on the subject and/or topics relevant to the field of study to which the admission to the degree programme is being	Diploma of an awardee of open international competitions of students' research projects, international student Olympiads	1
	carried out.  Year of 2022/2023, 2023/2024	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	1
	Considered shall be Olympiads and contests held by educational organisations of higher education.		
	Only one document is considered. Should there be two or more documents, the points shall not be summed.		
	Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational		2

		programme to which the admission is being carried out (with the workload at least 72 hours).  Only one document is considered. Should there be two or more certificates, the points shall not be summed		
37.04.01 Psychology	Personality Psychology	Bachelor's/Specialist's /Master's degree  Accepted can be but one evidence in of the section. Should there be two or more documents, the documents points shall not be summed	Bachelor's/Specialist's /Master's degree in the fields of study 37.03.01 'Psychology', 37.05.01 'Clinical Psychology', 37.05.02 Employment Psychology, 37.04.01 'Psychology' or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher education lists of specialities and fields of study in degree programmes	Degree with distinction— 30 Degree without distinction— 25
			Bachelor's/Specialist's /Master's degree in the fields of study 44.03.02 Psychologist-Pedagogical Education, 44.04.02 Psychologist-Pedagogical Education or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study in degree programmes	Degree with distinction— 20
			Bachelor's/Specialist's /Master's degree in the fields of study 37.03.02 'Conflictology' 37.04.02 'Conflictology' or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study in degree programmes  Bachelor's/Specialist's /Master's degree in the fields of study 37.03.02 'Conflictology', 44.04.02	Grade average point 4.0 and more- of the diploma without distinction - 5

	Psychologist-Pedagogical Education or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study in degree programmes	3
	Bachelor's/Specialist's /Master's degree as the result of mastering an educational programme implemented online.	3
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases, as well as in periodicals included in the HAC list. (of at least 0.3 of author's sheet volume and	5
For the publications submitted to RSCI, a link to the article in e-library must be provided.	with as many as three authors) Scientific publications in periodicals	3
Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication	indexed by RSCI scientometric database; an author's article or an article with one/two co-authors (as long as 0.3 of author's sheets	
If there are two or more publications submitted, the points shall not be summed; evaluated is an	and longer), where the applicant is the first author	
article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	Scientific publications in periodicals indexed by RSCI scientometric database; Abstracts or an article with more than two co-authors (as long as 0.3 author's sheets)	2
Information about publications on topics not relevant to the Unified Decimal Code (UDC) 159.9 'Psychology' (a copy of the article, a copy of the publication's imprint page, and a copy of the	Scientific publications in periodicals indexed by Web of Science Core Collection, Scopus, RSCI scientometric databases (an author's article or an	1

collection contents must be provided). For the publications in RSCI, a link to the article in elibrary must be provided.  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication. If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.	article with one or two co-authors (as long as 0.3 of author's sheets or longer)	
Documents and materials confirming the applicant's participation in conferences (international or all-Russia (only in-person)), scientific schools (the St Petersburg University Winter Psychological School, Scientific schools at Graduate School of Economy, Moscow State University) on topics relevant to the respective master's degree programme or the field of study or specialities to which the admission is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme	4
Documents and materials confirming the applicant's participation in conferences (international or all-Russia (only in-person)), scientific schools on topics relevant to the respective master's degree programme or field of study or specialities to which the admission to the degree programme is being carried out. Accepted can be but one evidence in of the section. Should there be two or more documents, the points shall not be summed.	The participation must be confirmed by the published abstract to the report or the event programme As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme	2
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations and educational organisations of higher education	3

Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Winners' or laureate's diploma for contests of research, design work,Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests considered as	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	2
individual achievements upon admission to respective degree programmes) on topics relevant to the field of study or speciality (except 'Conflictology') to which the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of higher education.	Diploma of the winner of open all- Russia competitions of students' research projects, student Olympiads	1
Year of 2022/2023, 2023/2024		
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents confirming the presence of experience in practical activity in psychology.	12 months and more	2
Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the s human resource department officer;	From 6 months to 12 months	1

		Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out (with the workload at least 72 hours). Only one document is considered. Should there be two or more documents, the points shall not be summed.		2
37.04.01 Psychology	Developmental Psychology	Educational credential  Accepted can be but one evidence in the section. Should there be two or more documents, the documents points shall not be summed	A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme  Bachelor's/specialist's /master's degree in fields of study and specialities, such as: 06.03.01 'Biology', 37.03.02 'Conflictology', 39.03.01 'Sociology', 44.03.02 'Psychologist-Pedagogical Education', 44.03.03 'Special (Defectological) Education' or relevant to them fields of study and specialities as approved by the Ministry of Science and Higher Education of the Russian Federation lists of specialities and fields of study in degree programmes  Diplopma in professional retraining in the speciality 'Psychology', issued by an educational organisation Ranked by top 300 international	Degree with distinction— 30 Degree without distinction— 28  Degree with distinction— 18 Degree without distinction— 16  15

	Ratings (the Shanghai Rating, The Three Missions of Universities Moscow International Rating)	
	Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction– 8
Information about publications published in the RSCI on topics relevant to a respective degree programme to which the admission to a degree programme is being carried out (a copy of pages with publication, a copy of the imprint page of the collection and a copy of the contents of the collection must be provided).  It is required to provide a link to the publication in	Single scientific publications in publications indexed by the RSCI scientometric database, within no more than 3 areas of knowledge (without coauthors)	8
e-library.		
Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication  If there are two or more publications submitted,	Single scientific publications in publications indexed by the RSCI scientometric database, within no more than 3 areas of knowledge with coauthors (the first authors)	6
the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific	Single scientific publications in publications indexed by the RSCI scientometric database, within no more than 3 areas of knowledge with coauthors (not the foirst author)	4
event, points under this clause shall not be scored  Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective of fields of study to which the admission to the degree programme is	The participation must be confirmed by the published abstract to the report or the event programme  As a confirmation, provided shall be a	3
being carried out.	copy of the published report abstracts or materials of the conference (as well as a copy of the imprint page of the	

Only one document is considered. Should there be two or more documents, the points shall not be summed.	collection and a copy of the content page of the collection), or a copy of the event programme.	
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	3
The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.		
Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Documents confirming the foreign language competence (international certificates and other documents).	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B; Cambridge CPE – C, B, A;	2
Only one document is considered. Should there be two or more documents, the points shall not be summed.	Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage-	
summed.	grades B, A; BEC Higher- grades C, B, A); Level 2 Test in English from the Language	
	Testing Centre at St Petersburg University (CEFR Level B2)	
	(Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test	
	in English from the Language Testing Centre at St Petersburg	
	University(CEFR Level C1) (Grades C,B,A 51% and higher).	

Documents confirming the applicant's internship and inclusive study in a foreign language (in the speciality relevant to the degree programme).		1
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.		1
Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	2
Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of higher education.	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	1
Year of 2022/2023, 2023/2024		
Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective		2

37.04.01 Psychology	Psychology of Physical Culture and Sports	educational programme to which the admission is being carried out. At least 72 hours.  If considered is the diploma of professional retarining in the speciality 'Psychology' issued by an educational organisation ranked by top 300 international ratings (the Shanghai Rating, the Three Missions of Universities Moscow International Rating) as an education credential according to criterion 4 points in this category shall not be scored.  Only one document is considered. Should there be two or more certificates, the points shall not be summed  Bachelor's/Specialist's /Master's degree	Bachelor's/Specialist's /Master's degree in the fields of study and specialities relevant to the field of study within the degree programme (including the field of study 'Physical Training and Sports').	Degree with distinction—10 Degree without distinction—6
			Bachelor's/specialist's /master's degree in other fields of study and specialities  Only one document is considered.  Should there be two or more certificates, the points shall not be summed  Bachelor's/Specialist's /Master's degree as the result of mastering an educational programme implemented online.  Only one document is considered.  Should there be two or more	Degree with distinction—4 Diploma grade average point 4.0 and more - 1

	certificates, the points shall not be summed	
'Excellent' grade for the bachelor's degree/ specialist's degree graduation project  Only one document is considered. Should there be two or more certificates, the points shall not be summed	The 'excellent' grade for the graduation project on the topic relevant to the field of study within the degree programme (including the field of study 'Physical Training and Sports').	4
	'Excellent' grade for the graduation project in the field of training and	2
Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	10
degree programme is being carried out (a copy of the screen shot, a copy of the imprint page, a copy of pages with the Country contents of the	Scientific publications in periodicals recommended by the HAC	7
collection must be provided).	Publications in RSCI or other scientific periodicals	4
For the publications in RSCI, a link to the article in e-library must be provided.		
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.		
If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of	The participation must be confirmed by the published abstract to the report or by the event programme.	4

fields of study (specialities) to which the admission to the degree programme is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme	
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.	Members of research teams, a grant, or a project under the order from state- owned science foundations, research organisations or educational organisations of higher education	4
The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.	Within this section, Accepted can be one document. Should there be two or more certificates, the points shall not be summed	
The points for the clauses can be summed	Statements of Acceptance of research and methodology developments aimed to optimise training of athletes (including physically challenged athletes) and those who are engaged in physical culture and health activities. Within this section, Accepted can be one document. Should there be two or more certificates, the points shall not be summed	4
Documents confirming respective working experience addressing the used in the professional	12 months and more	4
activity competences correlating with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity in physical training and sports).  Provided must be one of the following documents:	From 6 months to 12 months	3

		1
- a copy of the Employment Book (extracts)		
certified by the organisation's human resource		
department officer;		
- a copy of the employment contract;		
- a confirmation from the place of work (sealed		
and signed by the official)		
Documents confirming the foreign language	English language reference credentials:	3
competence (international certificates and other	TOEFL (score 60 and higher); IELTS	
documents).	(score 6.0 and higher); Cambridge CAE	
documents).	- C, B, A; Cambridge CPE - C, B, A;	
Accepted can be one document (unless there is	Cambridge FCE – B, A; Cambridge	
evidence of the proficiency in two foreign	English: Business (BEC Vantage -	
languages) confirming the foreign language	grades B, A; BEC Higher – grades C,	
proficiency. If there are two documents, accepted	B, A)	
shall the one with the highest score of points.	German language reference credentials	
shall the one with the highest score of points.	Goethe Zertificate (B2/ C1– 70-100;	
If 41 : : 1 f 41 f : :- t	`	
If there is evidence of the proficiency in two	C2); Test DaF (TDN 16-20); Deutsche	
foreign languages, the points obtained by one	Sprachprüfung für den	
document that ensures the highest score of points	Hochschulzugang (DSH 1 - DSH 3:	
for one foreign language shall be added by the	57% and higher); Das Österreichische	
points according to the provisions of this clause:	Sprachdiplom Deutsch (ÖSD B2: 75 –	
	100; ÖSD C1: 60 – 100; ÖSD C2: 100	
	– 160); The European Language	
	Certificate (TELC B2: 240 – 300,	
	TELC C1: 151 – 214; TELC C2: 112 -	
	160); Level 2 Test in English from the	
	Language Testing Centre at St	
	Petersburg University (CEFR Level	
	B2) (Grade C 51-66%, Grades B, A	
	67% and higher); Level 3 Test in	
	English from the Language Testing	
	Centre at St Petersburg	
	University(CEFR Level C1) (Grades	
	C,B,A 51% and higher);	
	French language reference credentials	
	DELF B2 – 61-100; DALF C1 – 50-	
	100	
•		

	Spanish language reference credentials DELE B2/C1 – 61-100 GMAT (Graduate Management Admission Test) (500 and higher) Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR (60% and higher) Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	
	Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	2
	If there is evidence of the proficiency in two foreign languages confirmed by respective certificates/grade in foreign languages in bachelor's/mater's degree academic transcript	
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	3
Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged group of fields of study (specialities) within which	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	2
the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of	Diploma of the winner of open all- Russia competitions of students' research projects, student Olympiads	2
higher education.  Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	1

Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	1
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		2
Individual achievements in sports: sports category.  To confirm the sports category, the applicant must	Sports qualification: Honoured Master of Sports/Master of Sports of International Class	4
provide the athlete's classification book.	Sports qualification: Master of Sports	3
The points shall not be summarised; evaluated is a sports category with highest number of points.	Sports qualification: Candidate Master of Sports	2
	Sports qualification: first adult rank / ranks below first adult	1
Individual achievements in sports: sport competitions.  To confirm the status of the winner or an awardee,	Winners and awardees of the Olympic Games, Paralympic Games, World and European Cups, World or European Cups in adaptive kinds of sports.	4
the applicant must provide a certificate or a diploma.	Winners or awardees of all-Russia championships, as well as all-Russia and international tournaments of the	3
If there are two or more sports achievement evidences submitted, the points shall not be summed; evaluated is the achievement with the highest number of points.	similar level, including competitions for physically challenges athletes.	
Individual achievements in sports: membership in the national team.	A member of the Russian national sports team in a any kind of sports	2

		To confirm the status of a national team member, the applicant must provide a certificate from the respective kind of sports federation.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	(adult/ youth/ junior/ physically challenged)	
37.04.01 Psychology	Organisational Psychology and Psychology of Management	Bachelor's/Specialist's /Master's degree  Accepted can be but one evidence in the section. Should there be two or more documents, the documents points shall not be summed	A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme	Degree with distinction—27 Degree without distinction—22
			Bachelor's/specialist's /master's degree in the fields of study and specialities: 38.03.02 Management, 38.03.03 Personnel Management, 38.03.04 Public and Municipal Administration, 38.03.05 Business Informatics; 44.04.02 Psychologist-Pedagogical Education	Degree with distinction— 17 Degree without distinction— 12
			Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction—7
		'Excellent' grade for the bachelor's degree/ specialist's degree graduation project Only one document is considered. If there are two or more documents submitted, the points shall not be summed	'Excellent' grade for the graduation project	2
		Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out (a copy of	Scientific publications in the periodicals indexed by Web of Science Core Collection, Scopus and RSCI scientometric databases	3
		the screen shot, a copy of the imprint page, a copy of pages with the Country contents of the	RSCI	2
		collection must be provided).	Articles in other scientific periodicals	1

For the publications in RSCI, a link to the article in e-library must be provided.  Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication  If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts/materials of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of	The participation must be confirmed by the published abstract to the report or the programme of the event.	2
fields of study (specialities) to which the admission to the degree programme is being carried out.	As a confirmation, provided shall be a copy of the published report abstracts or materials of the conference (as well	
Only one document is considered. If there are two or more documents submitted, the points shall not be summed	as a copy of the imprint page of the collection and a copy of the content page of the collection), or a copy of the event programme.	
Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implemented projects; in research projects results in the topics relevant to the field of study to which the admission to the degree programme is being carried out.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	2

The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.  Only one document is considered. If there are two or more documents submitted, the points shall not be summed		
Documents confirming respective working experience addressing the used in the professional activity competences correlating with the competences to be developed in the course of mastering a respective degree programme	From 6 months to 12 months	1
(experience in professional practical activity).  Provided must be one of the following documents: - a copy of the Employment Book (extracts), certified by a Personnel Department officer of the organisation; - a copy of the employment contract; - a confirmation from the place of work (sealed and signed by the official)		
Documents confirming the foreign language competence (international certificates and other documents).  Only one document is considered. Should there be two or more certificates, the points shall not be summed	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage – grades B, A; BEC Higher – 112 - 160); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level C1) (Grades C,B,A 51% and higher).	2

	German language reference credentials Goethe Certificate (B2/C1–60-100; C2); Test DaF (TDN 4, TDN 5) French language reference credentials DELF B2 – 60- 100; DALF C1 – 50- 100 Spanish language reference credentials DELE B2/C1 – 60- 100 GMAT (Graduate Management Admission Test) (from 500 and higher) Other certificates confirming the Upper Intermediate reference level of the English language proficiency. The 'excellent' grade for 'Foreign Language' in the academic transcript of records; Training at universities ranked by top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International	1
Documents confirming internship and inclusive study in a foreign language.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	University Ranking)  Internship or inclusive study abroad at universities ranked among top 300 ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)  Internship or inclusive study at other	2
Documents confirming personal scholarships granted by ministries, agencies, state foundations and educational	foreign universities	2
institutions of higher education. Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Winners' or laureate's diploma for contests of research, design work, international gaming	The winner's/awardee's diploma of open	3

contests, Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests considered as individual achievements upon admission to respective degree	international competitions of students' research projects, international game competitions, student Olympiads  An awardee's diploma of open	2
programmes) and on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree	international competitions of students' research projects, international game competitions, student Olympiads	2
programme is being carried out.  Considered shall be Olympiads and contests held by educational organisations of higher education.	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	2
Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	1
Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	1
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out (with the workload at least 72 hours).  Only one document is considered. Should there be two or more documents, the points shall not be summed.		2
Certificates confirming mastering online courses on educational platforms 'Coursera' and 'open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to		1

		the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed.  The points shall only be assigned provided there is availability to verify the submitted documents through open resources.		
37.04.01 Psychology	Social Psychology and Political Psychology	Bachelor's/specialist's degree in the fields of study and specialities relevant to the main fields of	Degree with distinction	26
		study: 37.03.01 'Psychology', 37.05.01 'Clinical Psychology' 44.04.01 Pedagogical Education; 44.03.02 Psychologist-Pedagogical Education; 44.03.03 Special (Defectological) Education; 44.03.05 Pedagogical Education (with two training profiles); 06.03.01 Biology; 39.03.01 Sociology or relevant to them fields of study and specialities according to the lists of specialities and fields of study of degree programmes approved by the Ministry of Science and Higher Education of the Russian Federation	Degree without distinction	15
		'Excellent' grade for the bachelor's degree/ specialist's degree graduation project Only one document is considered. Should there be two or more certificates, the points shall not be summed	'Excellent' grade for the graduation project	3
		Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases	5

For the publications submitted to RSCI, a link to the article in e-library must be provided.  Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication		
If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points. If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations and educational organisations of higher education	4
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	3
Olympiads and contests which to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	1
within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) within which	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	2
the admission to the degree programme is being carried out. Considered shall be Olympiads and	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	1

contests held by educational organisations of higher education.  Year of 2022/2023, 2023/2024  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	1
Standard credential of professional retraining in the non-degree programme 'Psychology' issued by an educational organisations ranked among top 300 by international ratings (the Shanghai Rating, Three Missions of University Moscow International Rating)		4
Documents confirming the foreign language competence (international certificates and other documents).  Only one document is considered. Should there be two or more certificates, the points shall not be summed	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE  – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage- grades B, A; BEC Higher— grades C, B, A); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level C1) (Grades C,B,A 51% and higher);	5

			German language reference credentials Goethe Certificate (B2/ C1– 60-100; C2); Test DaF (TDN 4, TDN 5) French language reference credentials DELF B2 – 60-100; DALF C1 – 50-100 GMAT (Graduate Management Admission Test) Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	
37.04.01 Psychology	Educational Psychology	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Psychology	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Psychology	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
			Preschool Pedagogy Basic Pedagogy Special (defectological) education	Students-medal holders - 50 points Winners - 25 points Awardee- 10 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Psychology Education and Pedagogics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Psychology	Winners - 10 points Awardees - 5 points
		'Urok dlya Uchitelya (A Lesson for a Teacher)' interdisciplinary contest for teams. Year of 2022/2023, 2023/2024		Winners - 10 points

'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Pre-school education	Winners - 10 points
'Flagmany Obrazovaniia' all-Russia professional contest Year of 2022/2023, 2023/2024		Winners - 10 points
Bachelor's/specialist's degree in the fields of study and specialities relevant to the main fields of	Degree with distinction	35
study: 37.03.01 'Psychology', 37.05.01 'Clinical Psychology' 44.04.01 Pedagogical Education; 44.03.02 Psychologist-Pedagogical Education; 44.03.03 Special (Defectological) Education; 44.03.05 Pedagogical Education (with two training profiles); 06.03.01 Biology; 39.03.01 Sociology or relevant to them fields of study and specialities according to the lists of specialities and fields of study of degree programmes approved by the Ministry of Science and Higher Education of the Russian Federation	Degree without distinction	30
'Excellent' grade for the bachelor's degree/ specialist's degree graduation project  Only one document is considered. Should there be two or more certificates, the points shall not be summed	'Excellent' grade for the graduation project	5
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes is carried out (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).  For the publications submitted to RSCI, a link to	Articles in publications indexed by Web of Science Core Collection and Scopus scientometric databases	3
the article in e-library must be provided.		

	<u> </u>	
Certificates/diplomas/in-print confirmations shall not be deemed as the confirmation of the publication		
If there are two or more publications submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the field of study to which programme the admissions to the degree programme is being carried out.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations and educational organisations of higher education	2
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	2
Olympiads and contests considered as individual achievements upon admission to respective degree programmes) on topics relevant to the enlarged group of fields of study (specialities) within which	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	1
the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	2
higher education.  Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	1

		Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner or an awardee of open regional competitions of students' research projects, student Olympiads	3
		Standard credential of professional retraining in the non-degree programme 'Psychology' issued by an educational organisations ranked among top 300 by international ratings (the Shanghai Rating, Three Missions of University Moscow		3
37.04.02 Conflictology	Conflict Management	International Moscow International Rating)  Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Social Science (Sociology, Social Work) Psychology	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Sociology Psychology History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Sociology Psychology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Sociology Sociology and Public Processes Management Public Administration Psychology	Winners - 10 points Awardees - 5 points
		'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners - 10 points
		The Olympian of federal universities for those applying for the admissions to master's degree programmes Year of 2022/2023, 2023/2024	Conflictology	Winners - 10 points Awardees - 5 points

- 1. For the degree programmes in the field of study 37.04.01 Psychology, points for the established types of individual achievements shall be summed; at the same time, only one documents shall be considered for one type of individual achievements unless otherwise is provided by this List of individual achievements of those applying for admissions to master's degree programmes.
- 2. For the degree programmes in the field of study 37.04.02 'Conflictology', assigned shall be the points only for one established type of individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates of completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

29.04.01 E	A 11 . 4	Detropolitor Coiones (De)Connels Ct Determinant	E	Winner 100 maints
38.04.01 Economics	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Economics and Finance	Winners - 100 points Awardee- 50 points
		Year of 2022/2023, 2023/2024		0.1.1.11
		'I am a Professional' All-Russia Olympiad for	Economics	Students-medal holders -
		students	Finance and Credit	100 points
				Winners - 50 points
		Year of 2022/2023, 2023/2024		Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni	Economics	Students-medal holders - 15
		Year of 2022/2023, 2023/2024		points
				Winners - 10 points
				Awardee- 5 points
		'Lomonosov' Universiad	Economic and Financial Strategy	Winners - 10 points
			Current issued of the global economy	Awardee- 5 points
		Year of 2022/2023, 2023/2024		
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'My entrepreneur initiative. Creative industries for regional development' nomination	Winners - 10 points
		'Ambilimpix' National championship in	Entrepreneurship	Winners - 10 points
		professional competency among individuals with		
		a disability or additional support needs		
		Year of 2022/2023, 2023/2024		

A bachelor's degree/specialist degree with distinction in the field of study and speciality correspondent to the enlarged group of the fields of	A degree from St Petersburg University, Moscow State University, a research institute	30
study and specialities within which the admissions to the degree programme is being carried out.	Degrees from other educational organisations	26
Bachelor's/specialist's degree with distinction in other fields of study and specialities	A degree from St Petersburg University, Moscow State University, a research institute	28
	Degrees from other educational organisations	24
Information about publications on topics relevant to the Enlarged Group of specialities and fields of study 38.00.00 Economics and Management (a copy of the article, a copy of the publication's	Articles in scientific periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	15
imprint page, a copy of the collection contents, as well as a link or a screen shot of the web page, which	Scientific articles in periodicals indexed by the RSCI scientometric database	10
confirms the fact that the publication has been accepted by the scientometric database):	Publications in other scientific periodicals with ISBN or ISSN	10
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Abstracts to conference reports in periodicals indexed by the RSCI scientometric database	5
Documents confirming scholarships granted by ministries, agencies, foundations, educational organisations of higher education.		3
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Online course certificates (except online courses of St Petersburg University) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective		1

	educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall be scored provided there is an availability to verify the evidences through open resources		
All degree programmes in the field of study (except degree programmes 'International Trading System (in English)' and 'Russia and China in Global Politics and Trade (in English)')	Documents confirming The foreign language proficiency (international certificates and other evidences)  Only one certificate is considered If there are two or more evidences, the points shall not be summed	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage - grades B, A; BEC Higher - grades C, B, A)  Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grades B, A 67% and higher),  Level 3 Test in English (CEFR Level C1) from the Language Testing Centre at St Petersburg University (Grades B, A 51% and higher), In any foreign languages: European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	6
		German language reference credentials Goethe Certificate (B2/C1– 60-100; C2); Test DaF (TDN 4, TDN 5)	6
		Spanish language reference credentials DELE B2/C1 – 60-100	6

		French language reference credentials DELF B2 – 60-100; DALF C1 – 50-100	6
		GMAT (Graduate Management Admission Test) (500 and higher) Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University	4
		(Grade C 51-66%) Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	2
		The 'excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
Accounting, Analysis and Audit Economics and Business Development	'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024	Economics and Accounting Entrepreneurship	Winners - 10 points
Management Economics and Enterprise Management	'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Marketing	Winners - 10 points Awardee- 5 points
Economic Policy and Public Strategic Management Russian Business in the Global Economy and Eurasian Integration	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Introduction to Economic Policy	Winners - 10 points Awardee- 5 points
Mathematical Methods in Economics	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Mathematics	Winners - 100 points Awardees - 50 points

	Year of 2022/2023, 2023/2024		
	'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Mathematics Mathematical Modelling	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Mathematics Applied Mathematics Applied Mathematics and Informatics Game theory	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Mathematical Methods in Economics	Winners - 10 points Awardee- 5 points
International Trading System (in English) Russia and China in Global Politics and Trade	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	World Economy	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
(in English)	Documents confirming The foreign language proficiency (international certificates and other evidences)  Only one certificate is considered If there are two or more evidences, the points shall not be summed	Bachelor's/Specialist's/Master's degree with confirmation of the fact that the full-time training was implemented in English during the entire period of mastering the degree programme.  The state standard credential confirming additional training to the degree programme 'Translator / Interpreter in Professional Communication	30
		(English)  Foreign language proficiency certificate with the following scores of points:  TOEFL (hard copy) – from 600 and higher;  IBT (Internet-) – from 100 and higher;	30

IELTS – from 7,0 and higher;	
Cambridge CPE (Certificate of	
Proficiency in English);	
Cambridge CAE (Certificate in	
Advanced English) – A	
Level 3 Test in English	
(CEFR Level C1)	
from the Language Testing Centre	
at St Petersburg University	
at Streetsburg Oniversity	
(Grades C, B, A 51% and higher);	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	25
Foreign language proficiency certificate	25
with the following scores of points:	
TOEFL - 575-599 points;	
iBT 90-99 points;	
IELTS – 6,5;	
Cambridge CAE (Certificate in	
Advanced English) – B	
Level 2 Test in English (CEFR Level	
B2) from the Language Testing Centre	
at St Petersburg University (Grades B,	
A 67% and higher)	
Foreign language proficiency certificate	20
with the following scores of points:	
TOEFL 557-574 points;	
iBT 80-89 points;	
ISSN – 5.5-6.0	
Cambridge CAE (Certificate in	
Advanced English) – C	
Cambridge CAE (First Certificate in	
English) – A	
English) – A	

	•		
Occupational Safety	Petropolitan Science (Re)Search, St Petersburg	Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)  Foreign language proficiency certificate with the following scores of points: TOEFL - 540-556 points; iBT - 76-79 points; IELTS - 5,0 Cambridge CAE (First Certificate in English) - B, C  Other certificates confirming the Upper Intermediate reference level of the English language proficiency.  The 'excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	20 15 Winners - 100 points
Occupational Safety Management and Occupational Risks	University Open International Olympiad for university students and young professionals		Winners - 100 points Awardee- 50 points
	Year of 2022/2023, 2023/2024		
	'I am a Professional' All-Russia Olympiad for students	Mathematics Mathematical Modelling	Students-medal holders - 100 points
		Strategic Project Management	Winners - 50 points
	Year of 2022/2023, 2023/2024		Awardees- 25 points
	'Vysshaia Liga' Olympiad for students and alumni	Management	Students-medal holders - 15
	v yssiiaia Liga Olympiau ioi students and alumin	Management	Bradenis incadi noiders

Year of 2022/2023, 2023/2024	Applied Mathematics Applied Mathematics and Informatics Game Theory	Winners - 10 points Awardees- 5 points
'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners and awardees (10 points)
'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
'My Pilot Business Project' international contest Year of 2022/2023, 2023/2024		Winners - 10 points
Management of the Future (conference)		Finalists (top 100) - 10 points
A bachelor's degree/specialist degree with distinction in the field of study and speciality correspondent to the enlarged group of the fields of	A degree from St Petersburg University, Moscow State University, a research institute	30
study and specialities within which the admissions to the degree programme is being carried out	Degrees from other educational organisations	26
Bachelor's/specialist's degree with distinction in other fields of study and specialities	A degree from St Petersburg University, Moscow State University, a research institute	28
	Degrees from other educational organisations	24
Information on the applicant's papers on topics correspondent to the enlarged group of the fields of study (specialities) within which the admission to a	Articles in scientific periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	25
respective degree programme is being carried out ( a copy of the publication, a copy of the	Scientific articles in periodicals indexed by the RSCI scientometric database	20
imprint page of the collection, a copy of the content page, a link or a screen shot of the web page, which confirms the fact that the publication has been accepted by the scientometric database must be provided):	Publications in other scientific periodicals with ISBN or ISSN	10
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
Abstracts to conference reports in periodicals indexed by the RSCI scientometric database		5

		Documents confirming scholarships granted by ministries, agencies, foundations, educational organisations of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		3
		Online course certificates (except online courses of St Petersburg University) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.		1
		Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall be scored provided there is an availability to verify the evidences through open resources		
38.04.02 Management	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Mathematics Economics and Finance	Winners - 100 points Awardees- 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Mathematics Mathematical Modelling Strategic Project Management Economics	Students-medal holders - 100 points Winners - 50 points Awardees- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Management Mathematics Applied Mathematics Applied Mathematics and Informatics Game Theory Economics	Students-medal holders - 15 points Winners - 10 points Awardees- 5 points
		'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners and awardees (10 points)

'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
'My Pilot Business Project' international contest Year of 2022/2023, 2023/2024		Winners - 10 points
'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Entrepreneurship	Winners - 10 points
'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'My entrepreneur initiative. Creative industries for regional development' nomination	Winners - 10 points
Management of the Future (conference)		Finalists (top 100) - 10 points
Bachelor's/Specialist's /Master's degree		in the field of study 38.00.00 Economics and Management - 30 points In other fields of study - 15 points
Documents confirming analytical abilities (international certificates) obtained later than 2019:  - GMAT (Graduate Management Admission Test): from 480 points or  - GRE (Graduate Record Examinations) Revised General test (Quantitative Reasoning): from 143 points or  - GRE (Graduate Record Examinations): Subject Test in Mathematics: from 500 points		20 points
Publications  A copy of the publication, a copy of the imprint page of the collection, a copy of the content page,		for 38.00.00 Economics and Management In RSCI - 6 points In HAC - 10 points

	as well as the link confirming the fact that the publication has been included in the scientometric database		In Scopus or WoSCC - 20 points for other fields of study In RSCI - 3 points In HAC - 5 points In Scopus or WoSCC - 10 points
	Volunteering confirmed by the Volunteer Book (including an extract from the State Services)		5 points
	Winter school 'Digital Weekend' under the auspices of the the Market Performance and Applied Finance Research Centre of St Petersburg University		Participants - 10 points Winners - 25 points
Management (Master in Management (MiM))	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Social Sciences (Sociology, Social Work) International Relations	Winners - 100 points Awardees - 50 points
	'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Finance and Credit	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Finances and Investing	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
	Russian Electronic School Olympiad Year of 2022/2023, 2023/2024		Winners - 10 points Awardees - 5 points
Master in Corporate Finance (MCF)	'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Finance and Credit	Medal holders - 100 points Winners - 50 points Awardees- 25 points

	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Finances and Investing	Medal holders - 15 points Winners - 10 points Awardees- 5 points
	Russian Electronic School Olympiad Year of 2022/2023, 2023/2024		Winners - 10 points Awardees- 5 points
	Formal evidence of the passed CFA Institute certificate (any level)		25 points
	Formal evidence of the passed at least one ACCA out of 13 available (including formal evidence of the discipline retaking)		25 points
	CFA Russia Challenge Olympiad Year of 2022/2023, 2023/2024		Participants - 5 points Finalists - 10 points Members of the 1st place team-winner at the finals - 25 points
	NES Investment Research Challenge Olympiad Year of 2022/2023, 2023/2024		Participants - 5 points Finalists - 10 points Members of the 1st place team-winner at the finals - 25 points
Business Analysis and Big Data (Master in Business Analysis and Big Data– MiBA)	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Computer Technology (Fundamental Computer Science and Information Technology)	Winners - 100 points Awardees- 50 points
	'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Business Informatics Artificial Intelligence Programming and Information Technologies Software Engineering	Medal holders - 100 points Winners - 50 points Awardees- 25 points
	Russian Electronic School Olympiad Year of 2022/2023, 2023/2024		Winners - 10 points Awardees- 5 points

38.04.03 Personnel Management	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Economics and Finance	Winners - 100 points Awardees- 50 points
		Year of 2022/2023, 2023/2024  'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Economics	Students-medal holders - 100 points Winners - 50 points Awardees- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Economics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Entrepreneurship	Winners - 10 points
		'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners and awardees (10 points)
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		A bachelor's degree/specialist degree with distinction in the field of study and speciality correspondent to the enlarged group of the fields of	A degree from St Petersburg University, Moscow State University, a research institute	30
		study and specialities within which the admissions to the degree programme is being carried out.	Degrees from other educational organisations	26
		Bachelor's/specialist's degree with distinction in other fields of study and specialities	A degree from St Petersburg University, Moscow State University, a research institute	28
			Degrees from other educational organisations	24
		Information about publications on topics relevant to the Enlarged Group of specialities and fields of study 38.00.00 Economics and Management (a copy of the article, a copy of the publication's	Articles in scientific periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	15

imprint page, a copy of the collection contents, as well as	Scientific articles in periodicals indexed by the RSCI scientometric database	10
a link or a screen shot of the web page, which confirms the fact that the publication has been accepted by the scientometric database must be	Publications in other scientific periodicals with ISBN or ISSN	10
provided):  If there are two or more articles submitted, the points shall not be summed; evaluated is an article	Abstracts to conference reports in periodicals indexed by the RSCI scientometric database	5
with the highest number of points.		
Documents confirming scholarships granted by ministries, agencies, foundations, educational organisations of higher education. Only one document is considered. Should there be two or more certificates, the points shall not be summed		3
Online course certificates (except online courses of St Petersburg University) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.		1
Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall be scored provided there is an availability to verify the evidences through open resources		
Documents confirming The foreign language proficiency (international certificates and other evidences)  Only one certificate is considered If there are two or more evidences, the points shall not be summed	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage - grades B, A; BEC Higher - grades C, B, A)	6

Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grades B, A 67% and higher),	
Level 3 Test in English (CEFR Level C1) from the Language Testing Centre at St Petersburg University (Grades B, A 51% and higher), In any foreign languages: European Consortium for the Certificate of Attainment in Modern Languages ECL:	
Goethe Certificate (B2/C1– 60-100; C2); Test DaF (TDN 4, TDN 5)	6
DELE B2/C1 – 60-100	
French language reference credentials DELF B2 – 60-100; DALF C1 – 50- 100	6
GMAT (Graduate Management Admission Test) (500 and higher) Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)	4
Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	2
The 'excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three	1

			University Missions Moscow International University Ranking)	
38.04.04 Public and Municipal Administration	Master in Smart City Management (MSC)	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Urban Development Management (Management) Economics and Finance Mathematics Social Sciences (Sociology, Social Work)	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Public Administration Business Informatics Economics Mathematics Mathematical Modelling Sustainable Development and Rational Nature Management Urbanistics Strategic Project Management	Medal holders - 100 points Winners - 50 points Awardees- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Public Administration Urban Spatial Development Management Management Mathematics Economics	Medal holders - 15 points Winners - 10 points Awardees- 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Public Administration	Winners - 10 points Awardee- 5 points
		'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners and awardees (10 points)
		Russian Electronic School Olympiad Year of 2022/2023, 2023/2024		Winners - 10 points Awardee- 5 points

'History of my local governance' all-Russia contest  Year of 2022/2023, 2023/2024	Nominations: My experience in regional body of social administration, community committee activities; Research 'The History of My Local Governance', a video clip 'The History of My Local Governance'.	Winners - 5 points
'My Native Country - My Russia' all-Russia contest  Year of 2022/2023, 2023/2024	Nominations: 'My Country's Environment' 'My Far East' 'My Arctic' 'My Country's Intellectual Property' 'Digital Media for Improving Standards of Life in Regions'	Winners - 10 points
Management of the Future (conference)		Finalists (top 100) - 10 points
Bachelor's/Specialist's /Master's degree	for 38.00.00 Economics and Management	30 points
	For other fields of study	15 points
Documents confirming analytical abilities (international certificates) obtained later than 2019:  - GMAT (Graduate Management Admission Test): from 480 points or  - GRE (Graduate Record Examinations) Revised General test (Quantitative Reasoning): from 143 points or  - GRE (Graduate Record Examinations): Subject Test in Mathematics: from 500 points		20 points

		Publications  A copy of the publication, a copy of the imprint page of the collection, a copy of the content page, as well as the link confirming the fact that the publication has been included in the scientometric database		for 38.00.00 Economics and Management In RSCI - 6 points In HAC - 10 points In Scopus or WoSCC - 20 points for other fields of study In RSCI - 3 points In HAC - 5 points In Scopus or WoSCC - 10 points
		Volunteering confirmed by the Volunteer Book (including an extract from the State Services)		5 points
		Summer school for an urbanist of the Graduate School of Management at St Petersburg University		Participants - 10 points Winners - 25 points
38.04.05 Business Informatics	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Economics and Finance Mathematics	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Business Informatics Mathematics Mathematical Modelling	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Business Informatics Mathematics Applied Mathematics Applied Mathematics and Informatics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		A bachelor's degree/specialist degree with distinction in the field of study and speciality correspondent to the enlarged group of the fields of	A degree from St Petersburg University, Moscow State University, a research institute	30
		study and specialities within which the admissions to the degree programme is being carried out.	Degrees from other educational organisations	26

Bachelor's/specialist's degree with distinction in other fields of study and specialities	A degree from St Petersburg University, Moscow State University, a research institute	28
	Degrees from other educational organisations	24
Information about publications on topics relevant to the Enlarged Group of specialities and fields of study 38.00.00 Economics and Management (a copy of the article, a copy of the publication's	Articles in scientific periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	15
imprint page, a copy of the collection contents, as well as	Scientific articles in periodicals indexed by the RSCI scientometric database	10
a link or a screen shot of the web page, which confirms the fact that the publication has been	Publications in other scientific periodicals with ISBN or ISSN	10
accepted by the scientometric database must be provided):	Abstracts to conference reports in periodicals indexed by the RSCI scientometric database	5
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
Documents confirming scholarships granted by ministries, agencies, foundations, educational organisations of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		3
Online course certificates (except online courses of St Petersburg University) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.		1
Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall be scored provided there is an availability to verify the evidences through open resources		

Documents confirming English language reference credentials: The foreign language proficiency TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Only one certificate Vantage - grades B, A; BEC Higher grades C, B, A) Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grades B, A 67% and higher), Level 3 Test in English (CEFR Level C1) from the Language Testing Centre at St Petersburg University (Grades B, A 51% and higher), In any foreign languages: European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR German language reference credentials 6 Goethe Certificate (B2/C1- 60-100; C2); Test DaF (TDN 4, TDN 5) Spanish language reference credentials 6 DELE B2/C1 - 60-100 French language reference credentials 6 DELF B2 – 60-100; DALF C1 – 50-GMAT (Graduate Management 4 Admission Test) (500 and higher) Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)

(international certificates and other evidences)

is considered If there are two or more evidences, the points shall not be summed

			Other certificates confirming the Upper Intermediate reference level of the English language proficiency.  The 'excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
38.04.08 Finance and Credit	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Economics and Finance	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Finance and Credit Economics Taxes and Taxation	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Finances and Investing Economics	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Economic and Financial Strategy	Winners - 10 points Awardee- 5 points
		'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024	Economics and Accounting Entrepreneurship	Winners - 10 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Entrepreneurship	Winners - 10 points

A bachelor's degree/specialist degree with distinction in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities within which the admissions	A degree from St Petersburg University, Moscow State University, a research institute	30
to the degree programme is being carried out.	Degrees from other educational organisations	26
Bachelor's/specialist's degree with distinction in other fields of study and specialities	A degree from St Petersburg University, Moscow State University, a research institute	28
	Degrees from other educational organisations	24
Information about publications on topics relevant to the Enlarged Group of specialities and fields of study 38.00.00 Economics and Management (a copy of the article, a copy of the publication's	Articles in scientific periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	15
imprint page, a copy of the collection contents, as well as a link or a screen shot of the web page, which confirms the fact that the publication has been	Scientific articles in periodicals indexed by the RSCI scientometric database	10
accepted by the scientometric database must be provided):  If there are two or more articles submitted, the	Publications in other scientific periodicals with ISBN or ISSN	10
points shall not be summed; evaluated is an article with the highest number of points.	Abstracts to conference reports in periodicals indexed by the RSCI scientometric database	5
Documents confirming scholarships granted by ministries, agencies, foundations, educational organisations of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		3

Online course certificates (except online courses of St Petersburg University) provided a particular online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall be scored provided there is an availability		1
to verify the evidences through open resources  Documents confirming The foreign language proficiency (international certificates and other evidences)  Only one certificate is considered If there are two or more evidences, the points shall not be summed	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage - grades B, A; BEC Higher - grades C, B, A)  Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grades B, A 67% and higher),  Level 3 Test in English (CEFR Level C1) from the Language Testing Centre at St Petersburg University (Grades B, A 51% and higher), In any foreign languages: European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	6

German language reference credentials Goethe Certificate (B2/C1- 60-100; C2); Test DaF (TDN 4, TDN 5)
Spanish language reference credentials DELE B2/C1 – 60-100
French language reference credentials DELF B2 – 60-100; DALF C1 – 50- 100
GMAT (Graduate Management Admission Test) (500 and higher) Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)
Other certificates confirming the Upper Intermediate reference level of the English language proficiency.
The 'excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three
University Missions Moscow International University Ranking)

<sup>1.</sup> For the degree programmes in the fields of study 38.04.01 Economics, 38.04.03 Personnel Management, 38.04.05 Business Informatics, 38.04.08 Finance and Credit, points for the established types of individual achievements shall be summed; at the same time, only one documents shall be considered for one type of individual achievements unless otherwise is provided by this List of individual achievements of those applying for admissions to master's degree programmes.

## 39.00.00 SOCIOLOGY AND SOCIAL WORK

<sup>2.</sup> For the degree programmes in the fields of study 38.04.02 Management, 38.04.04 Public and Municipal Administration, assigned shall be the points only for one established type of individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates of completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

39.04.01 Sociology	All degree programmes within this subject area	Certificates of the winners and awardees of competitions of research, development projects, international game competitions and student Olympiads.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Main field of study: Social Science (Sociology, Social Work) Year of 2022/2023, 2023/2024 'I am a Professional' All-Russia Olympiad for students Main field of study: Sociology Year of 2022/2023, 2023/2024 'Vysshaia Liga' Olympiad for students and alumni Main field of study: Sociology Year of 2022/2023, 2023/2024 'Lomonosov' Universiad Main fields of study: Sociology Sociology and Public Processes	Winners - 100 points Awardees- 50 points  Students-medal holders - 100 points Winners - 50 points Awardee- 25 points  Students-medal holders - 15 points Winners - 10 points Awardee- 5 points  Winners - 10 points Awardees - 5 points
			Management Year of 2022/2023, 2023/2024 'History of My Local Governance' all-	Winners - 5 points
			Russia contest  Year of 2022/2023, 2023/2024	Williams S position
			Nominations: 'My Community (district, street) Self-Governing Committee working experience'; 'History of my local governance' Research'; 'History of my local governance' video-clip'	
		Information on the applicant's papers on topics correspondent to the enlarged group of the fields of study (specialities) within which the admission to a respective degree programme is being carried out	Articles in scientific journals indexed by the HAC of the Russian Federation.	25

(a copy of the publication, a copy of the imprint page of the collection or a scientific journal, a copy of the collection contents, as well as a link or a screen shot of the web page, which confirms the fact that the publication has been accepted by the scientometric database must be provided): If the publication is available in digital format, a link to the respective Internet resource must be provided.  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Publication in periodicals indexed by the RSCI scientometric database.	20
Documents and materials confirming the applicant's participation in grant-supported projects under the order from state-owned science foundations, research organisations or educational organisations of higher education  The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer	Member of a research team, a grant, a project.	10

		Bachelor's/specialist's degree with distinction in the fields of training and specialities correspondent to the following enlarged groups of fields of study and specialities:  37.00.00 Psychological Sciences  38.00.00 Economics and Management  39.00.00 Sociology and Social Work  40.00.00 Law  41.00.00 Political Sciences and Regional Studies  42.00.00 Mass Media and Information  Librarianship  43.00.00 Services and Tourism  44.00.00 Education and Pedagogical Sciences  46.00.00 History and Archaeology  47.00.00 Philosophy, Ethics, and Religious Studies  50.00.00 Art Studies  51.00.00 Cultural Studies and Sociocultural  Projects  58.00.00 Asian and African Studies	Bachelor's/Specialist's degree with distinction	10
39.04.02 Social Work	Social Work	Certificates of the winners and awardees of competitions of research, development projects, international game competitions, grant competitions and student Olympiads.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Field of study - Social Sciences (Sociology, Social Work) Year of 2022/2023, 2023/2024 'I am a Professional' All-Russia Olympiad for students in the main field of study Sociology Year of 2022/2023, 2023/2024 'Vysshaia Liga' Olympiad for students and alumni in the main field of study Sociology Year of 2022/2023, 2023/2024 'Lomonosov' Universiad Main fields of study: Sociology Sociology and Public Processes	Winners - 100 points Awardee- 50 points  Students-medal holders - 100 points Winners - 50 points Awardee- 25 points Students-medal holders - 15 points Winners - 10 points Awardees- 5 points Winners - 10 points Awardees - 5 points

1	Year of 2022/2023, 2023/2024	
	'#MYVMESTE' international award Year of 2022/2023, 2023/2024 Volunteers and NPOs Treck	Winners - 10 points
	'The Best of the Projects in Social Work' contest Year of 2022/2023, 2023/2024	Winners - 10 points
	'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Field of study - Social Work Year of 2022/2023, 2023/2024	Winners - 10 points
	The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Subject area 'Volunteering', 'Social upward mobility development' Year of 2022/2023, 2023/2024	Winners - 10 points
Information on the applicant's papers on topics correspondent to the enlarged group of the fields of study (specialities) within which the admission to a	Articles in scientific journals indexed by the HAC of the Russian Federation.	25
respective degree programme is being carried out (a copy of the publication, a copy of the imprint page of the collection or a scientific journal,, a copy of the contents of the collection or a scientific journal, as well as a link or a screen shot of the web page, which confirms the fact that the publication has been accepted by the scientometric database must be provided). If the publication is available in digital format, a link to the respective Internet resource must be provided.	Publication in periodicals indexed by the RSCI scientometric database.	20
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.		

If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
Documents and materials confirming the applicant's participation in grant-supported projects under the order from state-owned science foundations, research organisations or educational organisations of higher education  The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.	Member of a research team, a grant, a project.	10
Bachelor's /specialist's degree with distinction in the fields of training and specialities correspondent to the following enlarged groups of fields of study and specialities:  37.00.00 Psychological Sciences  38.00.00 Economics and Management  39.00.00 Sociology and Social Work  40.00.00 Law  41.00.00 Political Sciences and Regional Studies  42.00.00 Mass Media and Information  Librarianship  43.00.00 Services and Tourism  44.00.00 Education and Pedagogical Sciences  46.00.00 History and Archaeology  47.00.00 Philosophy, Ethics, and Religious Studies  50.00.00 Art Studies  51.00.00 Cultural Studies and Sociocultural  Projects  58.00.00 Asian and African Studies	Bachelor's/Specialist's degree with distinction	10

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 39.00.00 'Sociology and Social Work' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

40.00.00 LAW				
40.04.01 Law	All degree programmes within this subject area	'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Law studies (Civil law, Criminal law, and Constitutional law)	Medal-holders: Gold - 100 points Silver - 80 points Bronze - 60 points Winners - 50 points Awardees- 25 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'Intellectual property of my native country' nomination	Winners - 10 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Civil Law	Students-medal holders - 10 points
		All-Russian Student Olympiad in Law Year of 2022/2023, 2023/2024		Winners - 5 points
		'Moi Zakonoproekt (My Law Project' the St Petersburg Legislative Assembly contest Year of 2022/2023, 2023/2024		Winners - 10 points Awardee- 5 points
40.04.01 Law	International Public Law	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	International Law	Students-medal holders - 10 points
	Lawyer in the Field of Financial Markets  Legal Protection of Economic Competition (with The second qualification 'Procurement Specialist'/ 'Competition Law Specialist')	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Public law and public finance (Public law track), Economics (Economics track), Civil Law: digital technologies and intellectual property rights management	Students-medal holders - 10 points

40.04.01 Law	Civil Procedure, Arbitration Procedure  Lawyer in the Field of Criminal Justice  Legal Protection of	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024  'Tochka rosta' all-Russia essay contest for	Judicial Power and Procedural Law	Students-medal holders - 10 points  Winners - 20 points
	Economic Competition (with The second qualification 'Procurement Specialist'/ 'Competition Law Specialist')	undergraduate and graduate students Year of 2022/2023, 2023/2024		Awardees– 10 points
	Lawyer in the Field of Norm-Setting Activities	'History of my local governance' all-Russia contest Year of 2022/2023, 2023/2024	Nominations: 'My Community (district, street) Self-Governing Committee working experience'; 'History of my local governance' Research'; 'History of my local governance' video-clip'	Winners - 5 points
40.04.01 Law	Information Law and Data Protection	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Intellectual property rights, artificial intelligence and digital technologies	Winners - 10 points Awardees- 5 points
40.04.01 Law	Tax Law and Tax Consulting	Winner of the Tax Law game competition	Winner of the Tax Law International game competition	7
		Year of 2022/2023, 2023/2024	Winner of the open all-Russia Tax Law game competition	5
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the article/screen shot, a copy of the	Scientific publications in periodicals included in the RSCI core of teh RSCI scientometric database	10
		publication's imprint page, and a copy of the collection contents, indexing confirmation must be provided)	Scientific publications in periodicals included in the list of the HAC of the Russian Federation	5
		The article must be proved by a link to the onlibe resource.		

		Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report		
		delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
40.04.01 Law	Legal Protection of Economic Competition (with The second qualification 'Procurement Specialist'/ 'Competition Law	Information about publications relevant to the Unified decimal Code 346.545 Illegal Competition, 346.546 'Regulation (limitation) of Competition' on topics correspondent to the enlarged group of fields of study (specialities) within which the admission to a degree programme is being carried	Articles in scientific journals published under the auspices of the Federal Antimonopoly Service of Russia, included in the list of HAC - the total volume at length 8,000 characters with spaces	15 points
	Specialist')	out. (A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided)  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Articles in scientific journals published unbder the auspices of St Petersburg University, included in the list of the HAC, as well as in the journals indexed in Web of Science and Scopus scientometric databases - the total length at least 8,000 characters with spaces	10 points
		If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	Other scientific publications (essays, surveys, reviews, etc.): the total length at least 4,000 printed characters with spaces	5 points

If by the admission deadline a publication which published the article is not indexed in a respective scientometric database army more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database    Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, roundable discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.    Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.    Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12   Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12   Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12   Documents controlling and supervisory agencies for the period between 6 months and 12   Documents controlling and supervisory agencies for the period between 6 months and 12   Documents controlling and supervisory agencies for the period between 6 months and 12   Documents controlling a			
scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database  Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, roundtable discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition', Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database  Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, roundtable discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition', Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Document must be provided:  - a copy of the Employment Book (extracts) errified by the organisation's letterhead sealed and signed by the official)  Should the event of the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	*		
moment of the release of the article the publication was indexed in the respective scientometric database  Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, roundtable discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'IRegulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  - a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  The participation in scientific events organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
was indexed in the respective scientometric database  Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, roundable discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  The participation in scientific events organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, round-table discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Either document must be provided:  Either document field by the organisation's human resource department officer;  - a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  The participation in scientific events organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
applicant's participation in scientific events (conferences, workshops, research schools, round- table discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the Published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the Federal Antimonopoly service of Russia, confirmed by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or t			
(conferences, workshops, research schools, round- table discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			10 points
table discussions, etc.) with a report on the topic correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Document must be provided:  Either document must be provided:  Either document must be provided:  Either document must be provided:  a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  Service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	1		
correspondent to the Unified decimal Code 346.545 'Illegal Competition', 346.546 'Regulation (limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Document must be provided:  Either document must be provided:  - a copy of the Employment Book (extracts) etertified by the organisation's letterhead sealed and signed by the official)  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
The participation in scientific events considered. Should there be two or more documents, the points shall not be summed.   The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report			
(limitation) of Competition'. Only one document is considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Either document must be provided:  Either document officer;  - a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  The participation in scientific events except those organised by St Petersburg University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	1 *	1 *	
considered. Should there be two or more documents, the points shall not be summed.  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering to the provided in the course of mastering to the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			5 points
documents, the points shall not be summed.  University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  University and/or teh Federal Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			3 points
Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's letterhead sealed and signed by the official)  Antimonopoly service of Russia, confirmed by the published extracts/abstracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the course of mastering the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a copy of the place of work (on the organisation's letterhead sealed and signed by the official)  Confirmed by the published extracts/dastracts of the report  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering to the course of mastering the master's degree programme 'Legal Protection of Economic Competition'  To points  10 points  10 points  Practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'  To points  To points  To points  To points  To points	F		
Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  - b course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Competition'.  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  - a cotivities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition'.  Either document must be provided:  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	Documents confirming practical experience in	Documents confirming practical	25 points
degree programme 'Legal Protection of Economic Competition'.  the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be department officer; a copy of the contract; a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  the course of mastering the master's degree programme 'Legal Protection of Economic Competition' practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
Competition'.  degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
Either document must be provided:  Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
Either document must be provided:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  controlling and supervisory agencies for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	Competition'.		
Either document must be provided:  for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be department officer; a copy of the contract; a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  for the period from 12 months or more  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period from 12 months or more			
- a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	Fither decomment must be mustided.		
- a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	Enther document must be provided:	-	10 points
certified by the organisation's human resource department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12	- a copy of the Employment Book (extracts)		10 points
department officer; - a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12		1 * *	
- a copy of the contract; - a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  the master's degree programme 'Legal Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
- a confirmation from the place of work (on the organisation's letterhead sealed and signed by the official)  Protection of Economic Competition' in state controlling and supervisory agencies for the period between 6 months and 12			
organisation's letterhead sealed and signed by the official) in state controlling and supervisory agencies for the period between 6 months and 12			
for the period between 6 months and 12		•	
· · · · ·	official)		
		·	
months		months	

			Documents confirming practical experience in activities correspondent to the competences to be developed in the course of mastering the master's degree programme 'Legal Protection of Economic Competition' in other organisations	5 points
40.04.01 Law	Business Law	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'	Degree with distinction - 3 points
			Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	3 points
			To be confirmed by a formal document on the letter head of an educational organisation with the seal and a signature an authorised officer	
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10
		with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of teh RSCI scientometric database	10
		The article must be proved by a link to the online resource.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	2
		Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
		If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Articles in other scientific periodicals	1

If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
Documents confirming the reference level of a foreign language (international certificates,	English language reference credentials: TOEFL iBT (score 87 and	4
'excellent'/passed grade in the transcript):	higher), TOEFL CBT (score 227/567 and higher), ; IELTS (score 5.5 and	
	higher); Cambridge CPE – C, B, A;	
Only one document is considered.	Cambridge CAE – C, B, A; Cambridge	
Should there be two or more certificates, the points	FCE – B, A; Cambridge English:	
shall not be summed	Business (BEC Vantage – grades B, A;	
	BEC Higher – grades C, B, A);	
	European Consortium for the Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR; GMAT	
	(Graduate Management Admission	
	Test) (from 500 and higher);	
	Level 2 Test in English from the	
	Language Testing Centre at St	
	Petersburg University (CEFR Level	
	B2) (Grade C 51-66%, Grades B, A	
	67% and higher); Level 3 Test in	
	English from the Language Testing	
	Centre at St Petersburg	
	University(CEFR Level C1) (Grades	
	C,B,A 51% and higher);	
	German language reference	4
	credentials Goethe	
	Zertificate (B2/ C1– 70-100; C2);	

Test DaF (TDN 16-20); Deutsche Sprachprüfung für den Hochschulzugang (DSH 2 - DSH 3); Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)  Level 2 Test in English (CEFR Level		T
Hochschulzugang (DSH 2 - DSH 3); Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Test DaF (TDN 16-20);	
Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Deutsche Sprachprüfung für den	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Hochschulzugang (DSH 2 - DSH 3);	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Das Österreichische Sprachdiplom	
ÖSD C2); The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
(TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	/ ·	
European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	· ·	
French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		4
C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)  [Speaking 60% and higher) 2		
the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	The state of the s	
Languages ECL: C1 CEFR  Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		4
European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		•
Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
Italian language reference credentials  European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		4
Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	*	
ECL: C1 CEFR  Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Languages	
Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		2
5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)		
60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	Chinese Proficiency Test: HSK (Level	
Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	5/Level 6: Reading, Writing, Listening	
HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	60% and higher);	
and higher); HSKK: Advanced (Speaking 60% and higher)	Chinese Proficiency Speaking Test:	
and higher); HSKK: Advanced (Speaking 60% and higher)	HSKK: Intermediate (Speaking 60%	
2	and higher); HSKK: Advanced	
-	(Speaking 60% and higher)	
Level 2 Test in English (CEFR Level		2
	Level 2 Test in English (CEFR Level	
B2) from the Language Testing Centre	B2) from the Language Testing Centre	

			at St Petersburg University (Grade C 51-66%)	
			The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Real Estate Lawyer	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'	Degree with distinction	3
		Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	To be confirmed by a formal document on the letter head of the educational organisation with the seal and an official's signature	3
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10
		with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of teh RSCI scientometric database	10
		The article must be proved by a link to the online resource.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	2
		Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
		If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.	Articles in other scientific periodicals	1

If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
If by the admission deadline a publication which		
published the article is not indexed in a respective		
scientometric database any more, the applicant		
should provide an evidence confirming that at the		
moment of the release of the article the publication was indexed in the respective scientometric		
database		
Documents confirming the reference level of a	English language reference	4
foreign language (international certificates,	credentials: TOEFL iBT (score 87 and	7
'excellent'/passed grade in the transcript):	higher), TOEFL CBT (score 227/567	
1 8	and higher); IELTS (score 5.5 and	
Only one document is considered.	higher); Cambridge CPE - C, B, A;	
Should there be two or more certificates, the points	Cambridge CAE – C, B, A; Cambridge	
shall not be summed	FCE – B, A; Cambridge English:	
	Business (BEC Vantage – grades B, A;	
	BEC Higher – grades C, B, A);	
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR; GMAT	
	(Graduate Management Admission Test) (from 500 and higher);	
	Level 2 Test in English from the	
	Language Testing Centre at St	
	Petersburg University (CEFR Level	
	B2) (Grade C 51-66%, Grades B, A	
	67% and higher); Level 3 Test in	
	English from the Language Testing	
	Centre at St Petersburg	
	University(CEFR Level C1) (Grades	
	C,B,A 51% and higher);	
	German language reference	4
	credentials Goethe	

	Zertificate (B2/ C1– 70-100; C2);	
	Test DaF (TDN 16-20);	
	Deutsche Sprachprüfung für den	
	Hochschulzugang (DSH 2 - DSH 3);	
	Das Österreichische Sprachdiplom	
	Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
	ÖSD C2); The European Language	
	Certificate (TELC B2, TELC C1;	
	TELC C2); European Consortium for	
	the Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR	
	French language reference	4
	credentials DELF B2 – 61-100; DALF	
	C1 – 50-100; European Consortium for	
	the Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR	
	Spanish language reference	4
	<b>credentials</b> DELE B2/C1 – 61-100;	
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR	
	Italian language reference credentials	4
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR	
	Chinese language reference	2
	credentials	
	Chinese Proficiency Test: HSK (Level	
	5/Level 6: Reading, Writing, Listening	
	60% and higher);	
	Chinese Proficiency Speaking Test:	
	HSKK: Intermediate (Speaking 60%	
	and higher); HSKK: Advanced	
	(Speaking 60% and higher)	
L	· · · · · · · · · · · · · · · · · · ·	

			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)  The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Lawyer in the Field of Norm-Setting Activities	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'  Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.  To be confirmed by a formal document on the letter head of an educational organisation with the seal and  a signature an authorised officer	Degree with distinction - 3 points  3 points
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).  The article must be proved by a link to the online resource.	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases  Scientific publications in periodicals included in the RSCI core of teh RSCI scientometric database  Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	10

Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database	Articles in other scientific periodicals	1
Documents confirming the reference level of a	English language reference	4
foreign language (international certificates,	credentials: TOEFL iBT (score 87 and	
'excellent'/passed grade in the transcript):	higher), TOEFL CBT (score 227/567	
	and higher), ; IELTS (score 5.5 and	
	higher); Cambridge CPE - C, B, A;	
Only one document is considered.	Cambridge CAE – C, B, A; Cambridge	
Should there be two or more certificates, the points	FCE – B, A; Cambridge English:	
shall not be summed	Business (BEC Vantage – grades B, A;	
	BEC Higher – grades C, B, A);	
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR; GMAT	
	(Graduate Management Admission	
	Test) (from 500 and higher);	
	Level 2 Test in English from the	
	Language Testing Centre at St	
	Petersburg University (CEFR Level	
	B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in	
1	07/0 and higher), Level 3 Test III	

English from the Language Testing	
Centre at St Petersburg	
University(CEFR Level C1) (Grades	
C,B,A 51% and higher);	
German language reference	4
credentials Goethe	
Zertificate (B2/ C1– 70-100; C2);	
Test DaF (TDN 16-20);	
Deutsche Sprachprüfung für den	
Hochschulzugang (DSH 2 - DSH 3);	
Das Österreichische Sprachdiplom	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
ÖSD C2);	
The European Language Certificate	
(TELC B2, TELC C1; TELC C2);	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
French language reference	4
credentials DELF B2 – 61-100; DALF	
C1 – 50-100; European Consortium for	
the Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	4
Spanish language reference	4
credentials DELE B2/C1 – 61-100;	
European Consortium for the Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
	4
Italian language reference credentials	4
European Consortium for the Certificate of Attainment in Modern	
Languages	
ECL: C1 CEFR	2
Chinese language reference credentials	
Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening	
60% and higher);	

			Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	
			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)	2
			The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Civil Law, Family Law	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'	Degree with distinction - 3 points
			Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	3 points
			To be confirmed by a formal document on the letter head of an educational organisation with the seal and a signature an authorised officer	
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10
		with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of teh RSCI scientometric database	10

The article must be proved by a link to the online resource.  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list: Scientific publications in periodicals included into Russian Science Citation Index	2
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database	Articles in other scientific periodicals	1
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):  Only one document is considered. Should there be two or more certificates, the points shall not be summed	English language reference credentials: TOEFL iBT (score 87 and higher), TOEFL CBT (score 227/567 and higher), ; IELTS (score 5.5 and higher); Cambridge CPE – C, B, A; Cambridge CAE – C, B, A; Cambridge English: Business (BEC Vantage – grades B, A; BEC Higher – grades C, B, A); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR; GMAT (Graduate Management Admission Test) (from 500 and higher);	4

Level 2 Test in English from the	
Language Testing Centre at St	
Petersburg University (CEFR Level	
B2) (Grade C 51-66%, Grades B, A	
67% and higher); Level 3 Test in	
English from the Language Testing	
Centre at St Petersburg	
University(CEFR Level C1) (Grades	
C,B,A 51% and higher);	
German language reference	4
credentials Goethe	
Zertificate (B2/ C1– 70-100; C2);	
Test DaF (TDN 16-20);	
Deutsche Sprachprüfung für den	
Hochschulzugang (DSH 2 - DSH 3);	
Das Österreichische Sprachdiplom	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
ÖSD C2);	
The European Language Certificate	
(TELC B2, TELC C1; TELC C2);	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
French language reference	4
credentials DELF B2 – 61-100; DALF	
C1 – 50-100; European Consortium for	
the Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
Spanish language reference	4
credentials DELE B2/C1 – 61-100;	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	4
Italian language reference credentials	4
European Consortium for the	
Certificate of Attainment in Modern	
Languages	
ECL: C1 CEFR	

			Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	2
			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)	2
			The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Legal Support of Business	Education credential	Bachelor's/specialist's degree in the enlarged groups of fields of study and specialities 40.00.00 'Law', 38.00.00 'Economics', 39.00.00 Sociology and Social Work	Degree with distinction - 3 points
			Bachelor's/specialist's degree in other fields of study and specialities	Degree with distinction - 2 points Grade average point 4.0 of the diploma without distinction - 1 point
		Information about publications on topics relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law', 38.00.00 'Economics', 39.00.00 Sociology and Social Work	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10

(A copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided)  The article must be proved by a link to the online resource.	Scientific publications in periodicals included in the RSCI core of the RSCI scientometric database  Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	3
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database	Articles in other scientific periodicals	1
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) in the enlarged groups of fields of study and specialities 40.00.00 'Law', 38.00.00 'Economics', 39.00.00 'Sociology and Social Work'  Only one document is considered. Should there be two or more certificates, the points shall not be	The participation must be confirmed by the published abstract to the report or the event programme	4
summed	Lead of a research team,	8

Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implementing the projects on topics Relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law', 38.00.00 'Economics', 39.00.00 'Sociology and Social Work'  Only one document is considered. Should there be two or more certificates, the points shall not be	a grant, project under the order of a state-owned science foundations, research organisations or educational organisations of higher education  Members of a research team, a grant, project under the order of a state-owned science foundations,  research organisations or educational organisations of higher education	6
summed  Documents confirming respective working experience addressing the used in the professional	12 months and more under the employment contract	3
activity competences correlating with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity).  Provided must be the following documents: - a copy of the Employment Book (extracts)		1
certified by the organisation's human resource department officer; - a copy of the employment contract; - a copy of the civil contract		
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5
Olympiads and contests which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	4

within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of higher education.  Year of 2022/2023, 2023/2024  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner of Russian national competitions of students' research projects and student Olympiads  Diploma of the winner of open all-Russia competitions of students' research projects, student Olympiads  Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads  Olympiads	3
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies to be developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		2
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):  Only one document is considered. Should there be two or more certificates, the points shall not be summed	English language reference credentials: TOEFL iBT (score 87 and higher), TOEFL CBT (score 227/567 and higher), ; IELTS (score 5.5 and higher); Cambridge CPE – C, B, A; Cambridge CAE – C, B, A; Cambridge English: Business (BEC Vantage – grades B, A; BEC Higher – grades C, B, A); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR; GMAT	4

Italian language reference credentials	4
Languages ECL: C1 CEFR	
European Consortium for the Certificate of Attainment in Modern	
credentials DELE B2/C1 – 61-100;	
Spanish language reference	4
Languages ECL: C1 CEFR	4
the Certificate of Attainment in Modern	
C1 – 50-100; European Consortium for	
credentials DELF B2 – 61-100; DALF	
French language reference	4
Languages ECL: C1 CEFR	
Certificate of Attainment in Modern	
European Consortium for the	
(TELC B2, TELC C1; TELC C2);	
The European Language Certificate	
ÖSD C2);	
Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
Hochschulzugang (DSH 2 - DSH 3);	
Deutsche Sprachprüfung für den	
Test DaF (TDN 16-20);	
Zertificate (B2/C1-70-100; C2);	
credentials Goethe	
German language reference	4
C,B,A 51% and higher);	
University(CEFR Level C1) (Grades	
Centre at St Petersburg	
English from the Language Testing	
67% and higher); Level 3 Test in	
B2) (Grade C 51-66%, Grades B, A	
Language Testing Centre at St Petersburg University (CEFR Level	
Level 2 Test in English from the	
Test) (from 500 and higher);	
(Graduate Management Admission	

			European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	
			Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	4
			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)	2
			The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Criminal Law	Education credential	Bachelor's /specialist's degree in 'Law'	Degree with distinction - 5 points Grade average point 4.0 of the diploma without distinction - 3 point
			Bachelor's/specialist's degree in other fields of study and specialities	Degree with distinction - 2 points Grade average point at least 4.0 of the diploma without distinction - 1 point

Grade for the final state assessment (exam) in interdisciplinary test (exam) or in the major subject (exam)	'Excellent' grade	2
Information about publications on topics not relevant to the field of study 'Law' (a copy of the article, a copy of the publication's imprint page,	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	10
and a copy of the collection contents must be provided).  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the	Scientific publications in periodicals included in the list of the HAC of the Russian Federation	7
publication	Scientific publications in indexed by the RSCI scientometric database	5
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article  Is not indexed by a scientometric database any more, the applicant should provide an evidence confirming that by the moment of the release of the article this publication was indexed	Articles in other scientific periodicals	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the following enlarged groups of fields of study (specialities) 40.00.00 'Law'  Accepted can be one document Should there be two or more certificates, the points shall not be summed	The participation must be confirmed by the published abstract to the report or by the event programme.	4

Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implemented projects.	Participation must be confirmed by monitoring of law enforcement - 10 points  Participation in other projects	6
The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.		
Only one document is considered. Should there be two or more documents, the points shall not be summed.		
Documents confirming respective working experience addressing the used in the professional activity	The held positions in judicial or law enforcement agencies confirmed by a evidence of employment - 10 points	5
competences correlating with the competences to be developed in the course of mastering the master's degree programme 'Criminal Law' (experience in professional practical activity). Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the employment contract; - a confirming certificate from the place of work (on the letter-head of the organisation with a wet seal and an official's signature)	Work as a public assistant to a prosecutor or investigator, an assistant to a lawyer, confirmed by a certificate or certificate from the head of the relevant law enforcement agency, bar association	2
Certificates confirming the level of the foreign language proficiency:	International certificates: and other certificates confirming the CEFR B2 reference level of a foreign language proficiency.	2
Documents confirming internship and training in a foreign language.  Only one document is considered. Should there be	Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow	4
Only one document is considered. Should there be	International University Ranking)	

two or more documents, the points shall not be summed.	Training at other foreign universities	2
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		5
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests which are to be recognised	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5
according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree	Diploma of an awardee of open international competitions of students'research projects, international game competitions, student Olympiads	4
Programmes) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out. Considered shall be Olympiads and	Diploma of the winner of Russian national competitions of students' research projects and student Olympiads	4
contests held by educational organisations of higher education	Diploma of the winner of open all- Russia competitions of students' research projects, student Olympiads	3
Year of 2022/2023, 2023/2024  Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent to the competencies to be developed in the course of mastering the respective degree programme to which the admission is being carried out.		2

		Accepted can be one document. Should there be two or more certificates, the points shall not be summed  Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall only be assigned		1
40.04.01 Law	Civil Procedure, Arbitration Procedure	provided there is an availability to verify the submitted documents through open resources.  Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of	Degree with distinction - 3 points
			fields of study and specialities 'Law'  Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	3
			To be confirmed by a formal document on the letter head of an educational organisation with the seal and a signature an authorised officer	
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10

with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of the RSCI scientometric database	10
The article must be proved by a link to the online resource.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	2
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database	Articles in other scientific periodicals	1
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):	English language reference credentials: TOEFL iBT (score 87 and higher), TOEFL CBT (score 227/567	4
	and higher), ; IELTS (score 5.5 and higher); Cambridge CPE – C, B, A;	
Only one document is considered. Should there be two or more certificates, the points	Cambridge CAE – C, B, A; Cambridge FCE – B, A; Cambridge English:	
shall not be summed	Business (BEC Vantage – grades B, A;	
	BEC Higher – grades C, B, A); European Consortium for the	
	Certificate of Attainment in Modern Languages ECL: C1 CEFR; GMAT	

Italian language reference credentials	4
Languages ECL: C1 CEFR	
Certificate of Attainment in Modern	
European Consortium for the	
credentials DELE B2/C1 – 61-100;	
Spanish language reference	4
Languages ECL: C1 CEFR	
the Certificate of Attainment in Modern	
credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for	
French language reference	4
Languages ECL: C1 CEFR	4
Certificate of Attainment in Modern	
European Consortium for the	
(TELC B2, TELC C1; TELC C2);	
The European Language Certificate	
ÖSD C2);	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
Das Österreichische Sprachdiplom	
Hochschulzugang (DSH 2 - DSH 3);	
Deutsche Sprachprüfung für den	
Test DaF (TDN 16-20);	
Zertificate (B2/ C1– 70-100; C2);	
credentials Goethe	
German language reference	4
C,B,A 51% and higher);	
University(CEFR Level C1) (Grades	
English from the Language Testing Centre at St Petersburg	
67% and higher); Level 3 Test in	
B2) (Grade C 51-66%, Grades B, A	
Petersburg University (CEFR Level	
Language Testing Centre at St	
Level 2 Test in English from the	
Test) (from 500 and higher);	
(Graduate Management Admission	

			European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	
			Chinese language reference credentials Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher)	2
			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)	2
			The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Environmental and Legal Support of Business Activities	Bachelor's / specialist's degree in the fields of study 40.04.01 'Law', 05.04.06 'Ecology and Nature Management', 20.03.01 'Technosphere	Degree with 4.0 - 4.4 grade average point	2
	Activities	Safety'	Degree with 4.5 - 4.9 grade average point	3
			Degree with distinction	4
		The grade for the bachelor's degree/ specialist's degree/ master's degree graduation project correspondent to the Unified Decimal Code 342 'State Law. Constitutional Law. Administrative Law', UDC 346 'Economic Law' or UDC 347	The 'Excellent' grade for the graduation project	1

'Civil law. Legal Proceedings', 349 'Legal Problems of Environmental Protection', 349.4 'Land Law. Settlement Planning Law', UDC 502/504 'Nature. Environmental Protection'  Information about publications on topics relevant to the UDC 342 'State Law/ Constitutional Law. Administrative Law', UDC 346 'Economic Law' or UDC 347 'Civil law. Legal Proceedings', 349 'Legal Problems of Environmental Protection', 349.4 'Land Law. Settlement Planning Law', UDC 502/504 'Nature. Environmental Protection' (A copy of the article, a copy of the publication's	Two or more scientific publications in publications indexed by Web of Science Core Collection and Scopus scientometric databases and/or RSCI  One scientific publication in publications not indexed by Web of Science Core Collection and Scopus scientometric databases and/or RSCI	8
imprint page, and a copy of the collection contents must be provided)  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database	Publications in other scientific publications not indexed by Web of Science Core Collection and Scopus scientometric databases and/or RSCI	3
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on	Participation in two or more scientific events (conferences, seminars, scientific schools)	5
topics relevant to the UDC 342 'State Law. Constitutional Law. Administrative Law', UDC 346 'Economic Law' or UDC 347 'Civil law. Legal Proceedings', 349 'Legal Problems of Environmental Protection', 349.4 'Land Law.	Participation in one scientific event (conferences, seminars, scientific schools)	2

Settlement Planning Law', UDC 502/504 'Nature. Environmental Protection' The participation must be confirmed by the published abstract to the report or by a certificate, or by the event programme.  Documents confirming respective working experience addressing the used in the professional	12 months and more	3
activity competences correlating with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity). Provided must be one of the following documents: - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the employment contract; - a copy of the contract of the civil law nature; - a confirmation from the place of work (sealed and signed by an official)	From 6 months to 12 months	2
Documents confirming the foreign language proficiency (international certificates and other documents)  Only one document is considered. Should there be two or more certificates, the points shall not be summed	English language reference credentials: TOEFL, IELTS, Cambridge CAE Cambridge CPE, Cambridge FCE, Level 2 Test in English (CEFR Level B2) from the Language Testing Center at St Petersburg University (Grade C 51-66%); Grades B, A 67% and higher); Level 3 Test in English (CEFR Level C1) from the Language Testing Centre at St Petersburg University (Grade C 51-66%) (Grades C,B,A 51% and higher)  German language reference credentials Goethe Certificate	2

	French language reference credentials DALF (Diplome Approfondi de Langue Fraciaise)  GMAT (Graduate Management Admission Test) The 'excellent' grade for 'Foreign Language' in the bachelor's degree academic transcript of records;	1
Documents confirming internship and inclusive study in a foreign language  Only one document is considered. Should there be	Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow	2
two or more certificates, the points shall not be summed	International University Ranking) Training at other foreign universities	1
Documents confirming personal scholarships granted by ministries, departments, foundations, educational organisations of higher education  Only one document is considered. Should there be two or more certificates, the points shall not be summed		2
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	4
Olympiads and contests which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition	Diploma of an awardee of open international competitions of students'research projects, international game competitions, student Olympiads	3
within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being	Diploma of the winner of Russian national competitions of students' research projects and student Olympiads	3

		carried out. Considered shall be Olympiads and contests held by educational organisations of higher education	Diploma of the winner of open all- Russia competitions of students' research projects, student Olympiads	2
		Year of 2022/2023, 2023/2024 Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
		Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.		3
40.04.01 Law	Lawyer in the Field of Financial Markets	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'  Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	Degree with distinction - 3 points
			To be confirmed by a formal document	

	on the letter head of an educational	
	organisation with the seal and	
Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	a signature an authorised officer  Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10
with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of the RSCI scientometric database	10
The article must be proved by a link to the online resource.  Certificates/diplomas/letters of appreciation shall	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list:	2
not be deemed as the confirmation of the publication.	Scientific publications in periodicals included into Russian Science Citation Index	
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective	Articles in other scientific periodicals	1
scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):	English language reference credentials: TOEFL iBT (score 87 and higher), TOEFL CBT (score 227/567 and higher), ; IELTS (score 5.5 and higher); Cambridge CPE – C, B, A;	4

Only one document is considered. Should there be two or more certificates, the points shall not be summed	Cambridge CAE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage – grades B, A; BEC Higher – grades C, B, A); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR; GMAT (Graduate Management Admission Test) (from 500 and higher); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University(CEFR Level C1) (Grades C,B,A 51% and higher);	
	German language reference	4
	credentials Goethe Zertificate (B2/C1-70-100; C2); Test DaF (TDN 16-20); Deutsche Sprachprüfung für den Hochschulzugang (DSH 2 - DSH 3); Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2);	
	The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	
	French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	4

			Spanish language reference credentials DELE B2/C1 – 61-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Italian language reference credentials European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  Chinese language reference credentials	4 2
			Chinese Proficiency Test: HSK (Level 5/Level 6: Reading, Writing, Listening 60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher) Level 2 Test in English (CEFR Level	2
			B2) from the Language Testing Centre at St Petersburg University Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%) The 'Excellent' grade in a foreign	1
			language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	
40.04.01 Law	Lawyer in the Field of Criminal Justice	Bachelor's/specialist's /master's degree diploma with the supplement in the main fields of study and specialities	Grade point average from 4.6 to 5.0 of the diploma in main fields of study and specialities (inclusively)	10

1		
(History, Cultural Studies, International Relations,	Grade point average from 4.0 to 4.5 of	5
Political Science, Regional Studies,	the diploma in main fields of study and	
Sociology, Theory and Practice of Intercultural	specialities (inclusively)	
Communication, Philosophy, Economics, Law)	Grade point average to 3.9 of the	3
	diploma in main fields of study and	
Bachelor's/specialist's /master's degree diploma	specialities (inclusively)	
with the supplement in non- main fields of study	Grade point average from 4.6 to 5.0 of	8
and specialities	the diploma in non- main fields of study	
	and specialities (inclusively)	
The number of points shall not be increased if there	Grade point average from 4.0 to 4.5 of	4
are two or more documents in this subsection.	the diploma in non- main fields of study	
	and specialities (inclusively)	
	Grade point average to 3.9 of the	2
	diploma in non- main fields of study	
	and specialities (inclusively)	
Grade for the final state assessment (exam) in	A grade for the interdisciplinary test	5 ("Excellent") — 6 points
interdisciplinary test (exam) or in the major subject	(examination) confirmed by the	4 ("Good") — 5 points
(exam)	educational credential of the respective	3 ("Satisfactory") — 4
To be confirmed by the education credential.	level:	points
If there are two or more grades for final state	A grade for the profession-oriented test	5 ("Excellent") — 5 points
assessment (exam)	(examination) confirmed by the	4 ("Good") — 4 points
in the interdisciplinary test (exam) and/or major	educational credential of the respective	3 ("Satisfactory") — 3
subject (exam), the score of points in this	level:	points
subsection shall		
not be increased		
Information about publications on topics relevant	Articles in scientific journals included	4
to the enlarged group of fields of study and	into the following academic citation	
specialities 40.00.00	indexes: Russian Science Citation	
Law (a copy of the article, a copy of the	Index, Web of Science or Scopus: the	
publication's imprint page, and a copy of the	total length from 8000 printed	
collection contents must be provided).	characters with spaces	
	Articles in scientific editions not	2
The article must be proved by a link to the online	included into the Russian Science	
resource.	Citation Index (journals, digests, etc.):	
	the total length from 8000 printed	
Certificates/diplomas/letters of appreciation shall	characters with spaces	
not be deemed as the confirmation of the	Other scientific publications (essays,	2
publication.	surveys, reviews, etc.): the total length	
	l	

If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the most number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	from 4000 printed characters with spaces	
Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools, round-table discussions, etc.) on topics relevant to teh enlarged group of fields of study and specialities 40.00.00  Law  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Participation confirmed by a published event programme or a participant's certificate	4
Documents and materials confirming the applicant's participating in research projects on	Participation must be confirmed by monitoring of law enforcement	10
topics relevant to the enlarged groups of fields of study and specialities 40.00.00 Law  Participation in two or more scientific projects shall not increase the number of points for this subsection	Participation in other research projects	4
Documents confirming respective working experience provided the competencies used in the practical activity correspond to the ones to be	The held positions in judicial or law enforcement agencies confirmed by an evidence of employment	10
developed in the course of mastering the mater's degree programme 'Lawyer in the Field of Criminal Justice' (professional practical experience).	Work as a public assistant to a prosecutor or investigator, an assistant to a lawyer, confirmed by a certificate or certificate from the head of the	6

<ul> <li>- a copy of the Employment Book (extracts) certified by the organisation's human resource department officer;</li> <li>- a copy of the employment contract;</li> <li>- a confirmation from the place of work (sealed and signed by the official)</li> </ul>	relevant law enforcement agency, bar association	
Certificates confirming the level of the foreign language proficiency	International certificates and other certificates confirming at least B2 reference level of a foreign language proficiency.	2
Documents confirming the fact of training or internship in a foreign state or in a foreign language	Documents confirming short-term training in the main fields of study and specialities (summer, autumn, winter and other schools, scientific seminars, etc.)  The number of points will be increased by 2 points for two or more documents in this subsection.	2
Evidences confirming granting personal scholarships from ministries, departments, foundations, educational organisations of higher education. Two or more scholarship evidences shall not increase the number of points for this subsection.	Statements or other documents issued by the dean's office of the faculty / academic affiars department of the relevant educational organisation confirming the following scholarships from	the President of the Russian Federation - 5 points the Government of the Russian Federation - 5 points republics (established by educational organisations of public authority of the constituting federal entity) - 2 points Scholarships from foundations, public associations and other organisations — 2 points
Certificates of the winners and awardees of competitions of research and development projects, international	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	6

Game competitions, student Olympiads (except Olympiads and competitions recognised under the provisions of a separate clause of the List and Procedures for Applicants' Individual Achievement	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	3
Evidences to be recognised within admissions to master's degree programmes) o topics relevant to the enlarged groups of fields of	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	3
study and specialities 40.00.00 Law  Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	2
Teal of 2022/2023, 2023/2024	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
	Diploma of the winner or an awardee of other all-Russia competitions.	1
	Diploma of the winner or an awardee of competitions held by educational organisations	1
Standard credentials of professional training or professional retraining in accordance with the main	Documents confirming professional retraining	3
fields of study and specialities (History, Cultural Studies, International Relations, Political Science, Regional Studies, Sociology, Theory and Practice of Intercultural Communication, Philosophy, Economics, Law) provided the mastered degree	Internship placements (including the placement in the course of study - to be confirmed by the education credential of the respective level of education or another certificate)	2
Implied addressing competences relevant to the ones to be developed in the course of mastering the	Documents confirming professional internship	2
master's degree programme 'Lawyer in the Field of Criminal Justice'.	Documents confirming professional training within mastering	3
The score shall not be increased for two or more documents in this subsection	a non-degree programme with the workload of 72 academic hours and more	
	Documents confirming professional training within mastering	2

		a non-degree programme with the workload of least 72 academic hours	
	Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Accepted can be only one document. Should there be two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the		
Lawyer in the Field of Sports Law (Sports Lawyer)	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 40.00.00 'Law'  Bachelor's/specialist's degree in other fields of study and specialities  Articles in publications indexed by	Degree with distinction - 10 points Grade average point 4.0 of the diploma without distinction - 4 points Degree with distinction - 6 points Grade average point more than 4.0 of the diploma without distinction - 4 points  15
	relevant to the enlarged groups of fields of study and specialities 40.00.00 Law (a copy of the publication/a screen shot, a copy with the	scientometric Web of Science Core Collection and Scopus databases Scientific publications in indexed by	10
	Sports Law (Sports	educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Accepted can be only one document. Should there be two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.  Education credential  Education credential  Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 Law (a copy	Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  **Accepted can be only one document. Should there be two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.  **Education credential**  **Education credential**  **Education credential**  **Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 40.00.00 'Law'  **Bachelor's/specialist's degree in other fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields of study and specialities 40.00.00 'Law'  **Information about publications relevant to the enlarged groups of fields o

of the collection and a copy of the contents page must be provided)	Articles in other scientific periodicals	3
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication		
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
If by the admissions deadline the publication which published the article is not indexed by a respective scientometric database any more, the applicant must provide a confirmation of the fact that, by the moment of the release of the article, the publication was indexed by the respective scientometric database		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the enlarged group of fields of study and specialities 40.00.00 'Law'.		4
The participation must be confirmed by the published abstract to the report or by the event programme.		
Only one document is considered. Should there be two or more documents, the points shall not be summed.		

i		
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the enlarged group of the fields of study and specialities 40.00.00 'Law'.  The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.  Only one document is considered. Should there be two or more documents, the points shall not be	Lead of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education  Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	7
Summed.  Documents confirming the foreign language competence (international certificates and other documents).  Accepted can be only one document. Should there be two or more documents, the points shall not be summed.	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE - C, B, A; Cambridge CPE - C, B, A; Cambridge FCE - B, A; Cambridge English: Business (BEC Vantage— grades B, A; BEC Higher— grades C, B, A); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level C1) (Grades C,B,A 51% and higher);	5

	German language reference credentials Goethe Certificate (B2/C1– 60-100; C2); Test DaF (TDN 4, TDN 5)	5
	French language reference credentials DELF B2 – 60-100; DALF C1 – 50- 100	5
	Spanish language reference credentials DELE B2/C1 – 60-100	5
	GMAT (Graduate Management Admission Test) (500 and higher)	5
	Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	2
	The 'excellent' grade in a foreign language in the degree transcript /'passed' grade in the degree the transcript issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	2
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Chrysty Runking)	6
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants'	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5

		Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out. Considered shall be Olympiads and contests held by educational organisations of higher education  Year of 2022/2023, 2023/2024	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads  Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads  Diploma of an awardee of open all-Russia competitions of students' research projects, student Olympiads	4 4 3
		Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
		Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.  Only one document is considered. Should there be		1
		two or more documents, the points shall not be summed.  The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.		
40.04.01 Law	Bankruptcy Law	Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on		17

		topics relevant to the enlarged group of fields of study and specialities 40.00.00 'Law'. Only one document is considered. Should there be two or more certificates, the points shall not be summed		
		Only one document is considered. If there are two or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.		5
40.04.01 Law	Lawyer Activity	Bachelor's/Specialist's /Master's degree In the main fields of training and Specialities (History, Cultural Studies,	Grade point average from 4.6 to 5.0 of the diploma in main fields of study and specialities (inclusively)	10
		International Relations, Political Studies, Regional Studies, Sociology, Theory of Intercultural Communication	Grade point average from 4.0 to 4.5 of the diploma in main fields of study and specialities (inclusively)	7
		Philosophy, Economics, Law  Bachelor's/Specialist's /Master's degree	Grade point average to 3.9 of the diploma in main fields of study and specialities (inclusively)	3
		in non- main fields of training .  Only one document is considered.	Grade point average from 4.6 to 5.0 of the diploma in non-main fields of study and specialities (inclusively)	5
		Should there be two or more certificates, the points shall not be summed	Grade point average from 4.0 to 4.5 of the diploma in non-main fields of study and specialities (inclusively)	3
			Grade point average to 3.9 of the diploma in non-main fields of study and specialities (inclusively)	1
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	8

with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of the RSCI scientometric database	6
Upon submitting the publication indexed by RSCI, Web of Science, Scopus databases, a link to the eresource of the respective system that contains information about the publication must be provided If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list: Scientific publications in periodicals included into Russian Science Citation Index	5
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Articles in other scientific periodicals	4
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars,	Participation confirmed by a published event programme or a participant's certificate provided the organiser (coorganiser) of the event was any bar body, bar association	6

scientific schools on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law  Only one document is considered. If there are two or more documents, the points shall not be summed.  The participation must be proved by the evidence of the published abstracts to the report, a programme of the event or an participant's certificate	Participation confirmed by a published event programme or a participant's certificate	4
Documents and materials confirming the applicant's participating in research projects	Participation must be confirmed by monitoring of law enforcement	6
Participation in two or more scientific projects shall not increase the number of points for this subsection	Participation in other research projects	3
Documents confirming respective practical working experience provided this activity addressed competences relevant to the ones to be developed in the course of mastering the master's degree programme "Lawyer Activity (practical working experience).  Provided must be one of the following documents:	The current status of a lawyer confirmed by the standard certificate or a certificate from a bar association proving the successful completion of the qualification exam (before issuing a lawyer certificate), a certificate about the suspension of the lawyer status or a reference to the regional register of	8
- a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the employment contract;	lawyers  The work as an assistant or trainee lawyer confirmed by a certificate from the respective bar association	8
- a confirmation from the place of work (sealed and signed by the official) Accepted can be evidences confirming the continuous work in the period at least six (6) months by the moment of submitting the documents.	The work in a bar association or law firms included into the federal or regional rating PRAVO-300 in two or more categories confirmed by the follwing evidences: 1) a certified copy of the employment contract, and 2)	6

	order on the employment, and 3) a	
If there are evidences of the assigned lawyer status	certificate from the place of work	
or duly certified documents	The work in law sphere in other	4
confirming the work as an assistant or trainee	institutions or organisations confirmed	
lawyer, the required work period shall be irrelevant.	by 1) a certified copy of the	
mreievani.	employment contract, and 2) order on	
The work as an assistant or trainee lawyer shall be	the employment, and 3) a certificate	
recognised only upon providing the relevant	from the place of work	4
certificate from the bar association	The work as a an assistant detective,	4
of the respective federal constituting entity.	attorney, judge, etc. confirmed by a certificate signed by the head of the	
	respective body of HR department, or	
	employment contract, employment	
The work with civil contracts shall not be	order and/or a certificate from the place	
considered.	of work	
Only one document is considered.		
Should there be two or more certificates, the points		
shall not be summed		
Standard credentials of professional training or	Documents confirming	4
professional retraining	professional retraining	
provided the mastered non-degree programme	Completion of the professional lawyer	4
implied addressing competences relevant to the	training courses organised in	·
ones	compliance with the provisions of the	
to be developed in the course of mastering the	Methodology Regulations for	
degree programme to which the admission is being carried out.	Professional Lawyer, Trainee Lawyer	
Carried out.	Training and Retraining (approved by	
Only one document is considered.	protocol No 3 of the Federal Chamber	
Should there be two or more certificates, the points	of Advocates of the Russian Federation	
shall not be summed	30 November 2007) at least with the workload 20 hours per one year and	
	confirmed by one or several certificates	
	Credentials of	2
	Professional training within mastering a	-
	non-degree	
	programme with the workload	
	72 academic hours and more	

			Documents confirming professional training Within mastering a non-degree programme with the workload Less than 72 academic hours Documents confirming	2
			professional internship,	
40.04.01 Law	International Private Law	Bachelor's degree/ specialist's degree in the fields of study and specialities relevant to the enlarged	Degree with distinction	5
		group of fields of study and specialities 40.00.00 'Law'	Grade average point 4 and higher of the diploma without distinction	3
		Bachelor's/specialist's degree in other fields of	Degree with distinction	2
		study and specialities	Grade average point 4 and higher of the diploma without distinction	1
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Articles in publications indexed by scientometric Web of Science and Scopus databases	10
		The article must be proved by a link to the online resource.	Scientific publications in indexed by the RSCI scientometric database	8
		Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the		
		publication.	Articles in other scientific periodicals	3
		If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		

If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the following enlarged groups of fields of study (specialities) 40.00.00 'Law'  Accepted can be one document If there are two or more documents, the points shall not be summed. The participation must be confirmed by the published abstract to the report or the event programme		4
Documents and materials confirming the applicant's participation in research projects, supported by grants, and results obtained by them in the course of implementing projects on topics relevant to the enlarged group of the fields of study	Lead of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	8
and specialities 40.00.00 'Law'.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	6
Documents confirming internship and inclusive study in a foreign language.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	3

	Training at other foreign universities	2
Documents confirming the foreign language proficiency (international certificates and other documents)	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE – C, B, A; Cambridge CPE – C, B, A;	3
Only one document is considered. Should there be	Cambridge FCE – B, A; Cambridge	
two or more certificates, the points shall not be	English: Business (BEC Vantage–	
summed	grades B, A; BEC Higher-	
	grades C, B, A); Level 2 Test in	
	English from the Language	
	Testing Centre at St Petersburg	
	University (CEFR Level B2)	
	(Grade C 51-66%, Grades B, A	
	67% and higher); Level 3 Test	
	in English from the Language	
	Testing Centre at St Petersburg	
	University(CEFR Level C1)	
	(Grades C,B,A 51% and higher);	
	German language reference credentials Goethe Certificate (B2/C1– 60-100; C2); Test DaF (TDN 4, TDN 5)	3
	French language reference credentials DELF B2 – 60-100; DALF C1 – 50- 100	3
	Spanish language reference credentials DELE B2/C1 – 60-100	3
	GMAT (Graduate Management Admission Test) (500 and higher)	2
	Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	1
	The 'excellent' grade in a foreign language in the degree transcript	3

Documents confirming respective working experience addressing the used in the professional activity competences correlating with the competences to be developed in the course of mastering the master's degree programme' International Private Law (experience in professional practical activity).  Provided must be one of the following documents:  - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer;  - a copy of the employment contract;  - a confirmation from the place of work (sealed and signed by an official)	/'passed' grade in the degree the transcript issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)  12 months and more	3
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.  Only one document is considered. Should there be two or more certificates, the points shall not be summed		3
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	5
Olympiads and contests which are to be recognised according to the provisions of the separate clause	Diploma of an awardee of open international competitions of students'	4

of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition	research projects, international game competitions, student Olympiads	
within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) 40.00.00	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	4
'Law'. Considered shall be Olympiads and contests held by educational organisations of higher education	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	3
Year of 2022/2023, 2023/2024	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
Standard credentials of professional training or professional retraining provided the mastered educational programme implied the acquisition of competencies correspondent the competencies to be developed in the course of mastering the master's degree programme 'International Private Law'.  Only one document is considered. Should there be two or more certificates, the points shall not be		2
summed		
Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the master's degree programme 'International Private Law'.		1
Only one document is considered. If there are two		

		or more documents, the points shall not be summed. The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.		
40.04.01 Law	Information Law and Data Protection	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 'Law'	Degree with distinction - 3 points
			Bachelor's/specialist's degree as the result of mastering an educational programme implemented online.	3
			To be confirmed by a formal document on the letter head of an educational organisation with the seal and a signature an authorised officer	
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the publication / screen shot, a copy of the imprint	Scientific publications in periodicals indexed by Web of Science Core Collection and Scopus scientometric databases	10
		page, a copy of pages with the contents of the collection or another publishing, and indexing confirmation must be provided).	Scientific publications in periodicals included in the RSCI core of the RSCI scientometric database	10
		The article must be proved by a link to the online resource.	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list: Scientific publications in periodicals	2
		Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	included into Russian Science Citation Index	
		If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report	Articles in other scientific periodicals	1

delivered at the conference or another scientific event, points under this clause shall not be scored.		
If by the admission deadline a publication which published the article is not indexed in a respective		
scientometric database any more, the applicant should provide an evidence confirming that at the		
moment of the release of the article the publication		
was indexed in the respective scientometric database		
Documents confirming the reference level of a	English language reference	4
foreign language (international certificates,	credentials: TOEFL iBT (score 87 and	
'excellent'/passed grade in the transcript):	higher), TOEFL CBT (score 227/567	
	and higher), ; IELTS (score 5.5 and	
	higher); Cambridge CPE – C, B, A;	
Only one document is considered.	Cambridge CAE – C, B, A; Cambridge	
Should there be two or more certificates, the points	FCE – B, A; Cambridge English:	
shall not be summed	Business (BEC Vantage – grades B, A;	
	BEC Higher – grades C, B, A);	
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR; GMAT	
	(Graduate Management Admission	
	Test) (from 500 and higher);	
	Level 2 Test in English from the	
	Language Testing Centre at St Petersburg University (CEFR Level	
	B2) (Grade C 51-66%, Grades B, A	
	67% and higher); Level 3 Test in	
	English from the Language Testing	
	Centre at St Petersburg	
	University(CEFR Level C1) (Grades	
	C,B,A 51% and higher);	
	German language reference	4
	credentials Goethe	
	Zertificate (B2/C1-70-100; C2);	
	Test DaF (TDN 16-20);	

Deutsche Sprachprüfung für den	
Hochschulzugang (DSH 2 - DSH 3);	
Das Österreichische Sprachdiplom	
Deutsch (ÖSD B2: 75 – 100; ÖSD C1;	
ÖSD C2);	
The European Language Certificate	
(TELC B2, TELC C1; TELC C2);	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
French language reference	4
credentials DELF B2 – 61-100; DALF	
C1 - 50-100; European Consortium for	
the Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
Spanish language reference	4
credentials DELE B2/C1 – 61-100;	
European Consortium for the	
Certificate of Attainment in Modern	
Languages ECL: C1 CEFR	
Italian language reference credentials	4
European Consortium for the	
Certificate of Attainment in Modern	
Languages	
ECL: C1 CEFR	
Chinese language reference	2
credentials	
Chinese Proficiency Test: HSK (Level	
5/Level 6: Reading, Writing, Listening	
60% and higher);	
Chinese Proficiency Speaking Test:	
HSKK: Intermediate (Speaking 60%	
and higher); HSKK: Advanced	
Level 2 Test in English (CEFR Level	2
B2) from the Language Testing Centre	
60% and higher); Chinese Proficiency Speaking Test: HSKK: Intermediate (Speaking 60% and higher); HSKK: Advanced (Speaking 60% and higher) Level 2 Test in English (CEFR Level	2

			Level 2 Test in English (CEFR Level B2) from the Language Testing Centre at St Petersburg University (Grade C 51-66%)  The 'Excellent' grade in a foreign language in degree transcript /'passed' grade in degree transcripts issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	1
40.04.01 Law	Transnational Legal Practice (in English)	Education credential	A bachelor's degree/specialist degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and specialities currently open for admissions within a respective degree programme.  Bachelor's/specialist's degree in other fields of study and specialities	Degree with distinction - 10 points Grade average point more than 4.0 of the diploma without distinction - 4 points  Degree with distinction - 6 points Grade average point more than 4.0 of the diploma without distinction - 4 points
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the article, a copy of the publication's imprint page, and a copy of the collection contents must be provided).  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication  If there are two or more articles submitted, the	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases Scientific publications in indexed by the RSCI scientometric database	5

points shall not be summed; evaluated is an article with the highest number of points.		
If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.		
If by the admissions deadline the publication which published the article Is not indexed by the respective database any more, the applicant should provide an evidence confirming that by the moment of the release of the article this publication was indexed		
Documents confirming the foreign language competence (international certificates and other documents).  Accepted can be only one document. If there are two or more documents, the points shall not be summed.	English language reference credentials: TOEFL (score 60 and higher); IELTS (score 6.0 and higher); Cambridge CAE - C, B, A; Cambridge CPE - C, B, A; Cambridge FCE - B, A; Cambridge English: Business (BEC Vantage - grades B, A; BEC Higher - grades C, B, A); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University(CEFR Level C1) (Grades C,B,A 51% and higher).	5
	German language reference credentials Goethe Certificate (B2/C1– 60-100; C2); Test DaF (TDN 4, TDN 5)	5
	French language reference credentials DELF B2 – 60-100; DALF C1 – 50- 100	5
	Spanish language reference credentials DELE B2/C1 – 60-100	5

	GMAT (Graduate Management Admission Test) (500 and higher)	5
	Other certificates confirming the Upper Intermediate reference level of the English language proficiency.	2
	The 'excellent' grade in a foreign language in the degree transcript /'passed' grade in the degree the transcript issued by the educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	2
Documents confirming scholarships granted by ministries, agencies, foundations, educational institutions of higher education.		3
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Winners' or laureate's diploma for contests of research, design work, international gaming contests, Olympiads for university students in subjects relevant to the field of study (except Olympiads and contests which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants'	Diploma of the winner of open international competitions of students' research projects, international game competitions, student Olympiads	4
Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) on topics relevant to the enlarged group of fields of study (specialities) 40.00.00	Diploma of an awardee of open international competitions of students' research projects, international game competitions, student Olympiads	4
'Law'. Considered shall be Olympiads and contests held by educational organisations of higher education	Diploma of the winner of All-Russia competitions of students' research projects and student Olympiads	4
Year of 2022/2023, 2023/2024	Diploma of an awardee of open all- Russia competitions of students' research projects, student Olympiads	3

		Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Diploma of the winner and an awardee of open regional competitions of students' research projects, student Olympiads	1
40.04.01 Law	Medical and Pharmaceutical Law	Education credential	Bachelor's/ specialist's / master's degree in the fields of study correspondent to the enlarged group of fields of study and specialities 40.00.00 'Law'  Bachelor's/ specialist's or master's degree in another fields of study and specialities	Degree with distinction- 3 points Grade average point 4.0 of the diploma without distinction- 2 point  Degree with distinction- 2 points Grade average point more than 4.0 of the diploma without distinction- 1 point
			Bachelor's/Specialist's /Master's degree obtained for mastering a degree programme implemented online  To be confirmed by a formal document on the letter head of the educational organisation with the seal and an official's signature	3
		The 'excellent' grade for the bachelor's / specialist's / master's graduation project or the 'excellent' grade for final state assessment in the form of the state exam in a single discipline, or the grade average point at least 4.5 for the final state assessment in the form of the state exam in several disciplines		2
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	10
		shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals included into Russian Science Citation Index	9

The article must be proved by a link to the online resource.  Certificates/diplomas/letters of appreciation shall	Scientific publications in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	8
not be deemed as the confirmation of the publication.	Articles in other scientific periodicals	3
If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific event, points under this clause shall not be scored.  If by the admission deadline a publication which published the article is not indexed in a respective scientometric database any more, the applicant should provide an evidence confirming that at the moment of the release of the article the publication was indexed in the respective scientometric database		
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the enlarged group of fields of study and specialities 40.00.00 'Law'.	Documents confirming the applicant's participation in scientific events.  The participation must be confirmed by the published abstract to the report or the event programme	4
Only one document is considered. Should there be two or more certificates, the points shall not be summed	Documents confirming the applicants participating in science schools/forums at least seven (7) days long held by organisations of higher education or research organisations	3
Documents confirming the reference level of a foreign language (international certificates, 'excellent'/passed grade in the transcript):	English language reference credentials: TOEFL iBT (score 87 and higher), TOEFL CBT (score 227/567 and higher), ; IELTS (score 5.5 and	3

Only one document is considered. Should there be two or more certificates, the points shall not be summed	higher); Cambridge CPE – C, B, A; Cambridge CAE – C, B, A; Cambridge CAE – C, B, A; Cambridge FCE – B, A; Cambridge English: Business (BEC Vantage – grades B, A; BEC Higher – grades C, B, A); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR; GMAT (Graduate Management Admission Test) (from 500 and higher); Level 2 Test in English from the Language Testing Centre at St Petersburg University (CEFR Level B2) (Grade C 51-66%, Grades B, A 67% and higher); Level 3 Test in English from the Language Testing Centre at St Petersburg University(CEFR Level C1) (Grades C,B,A 51% and higher);	
	German language reference credentials Goethe Zertificate (B2/C1-70-100; C2); Test DaF (TDN 16-20); Deutsche Sprachprüfung für den Hochschulzugang (DSH 2 - DSH 3); Das Österreichische Sprachdiplom Deutsch (ÖSD B2: 75 – 100; ÖSD C1; ÖSD C2);	3
	The European Language Certificate (TELC B2, TELC C1; TELC C2); European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR  French language reference credentials DELF B2 – 61-100; DALF C1 – 50-100; European Consortium for the Certificate of Attainment in Modern Languages ECL: C1 CEFR	3

	Spanish language reference credentials DELE B2/C1 – 61-100;	3
	European Consortium for the	
	Certificate of Attainment in Modern	
	Languages ECL: C1 CEFR	
	Italian language reference credentials	3
	European Consortium for the	3
	Certificate of Attainment in Modern	
	Languages	
	ECL: C1 CEFR	
	Chinese language reference	3
	credentials	3
	Chinese Proficiency Test: HSK (Level	
	5/Level 6: Reading, Writing, Listening	
	60% and higher);	
	Chinese Proficiency Speaking Test:	
	HSKK: Intermediate (Speaking 60%	
	and higher); HSKK: Advanced	
	(Speaking 60% and higher)	
	Level 2 Test in English (CEFR Level	2
	B2) from the Language Testing Centre	
	at St Petersburg University	
	Level 2 Test in English (CEFR Level	
	B2) from the Language Testing Centre	
	at St Petersburg University (Grade C	
	51-66%)	
	The 'Excellent' grade in a foreign	1
	language in degree transcript /'passed'	
	grade in degree transcripts issued by the	
	educational organisation ranked among	
	top 300 by international ratings	
	(Shanghai Ranking, The Three	
	University Missions Moscow	
	International University Ranking)	
Standard credentials of professional training or		2
professional retraining provided the mastered		
educational programme implied the acquisition of		
competencies correspondent the competencies to		

		be developed in the course of mastering the master's degree programme 'Medical and Pharmaceutical Law'  Only one document is considered.  If there are two or more documents, the points shall not be summed.		
40.04.01 Law	International Public Law	Diploma of the winner of the International Public Law essay contest at St Petersburg University		25
		Diploma of an awardee (second place) of the International Public Law essay contest at St Petersburg University		15
		Winner's or laureate diploma for the competition in the form of the moot court trial on topics relevant to the specialities within which the admission to the degree programme is being carried out (except Olympiads and competitions to be recognised under the provisions of a separate clause of the List and Procedures for the Recognition of Applicants' Individual Achievement Evidences within admissions to master's degree programmes)  (only one document is considered)	Criteria are specified in the Procedures for the Recognition of Applicant's Individual Achievements within admissions to the enlarged group of fields of study and specialities 40.00.00 'Law'	15
		Information about publications on topics relevant to the enlarged group of fields of study and specialities 40.00.00 Law (a copy of the screen shot, a copy of the imprint page, a copy of pages with the contents of the collection, and indexing confirmation must be provided).	Scientific publications in periodicals 1) indexed by Web of Science Core Collection and Scopus scientometric databases, 2) included into the RSCI core	10
		The article must be proved by a link to the online resource. Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.	Scientific publications in periodicals indexed by RSCI, included into the HAC list or RSCI database	5

If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		
Documents confirming the English language proficiency at least at B2 level (international certificates and other documents)  Only one document is considered. If there are two or more documents, the points shall not be summed.	Criteria are specified in the Procedures for the Recognition of Applicant's Individual Achievements within admissions to the enlarged group of fields of study and specialities 40.00.00 'Law'	10
Documents confirming the English language proficiency at least at CEFR B2 level (international certificates and other documents)		5
Only one document is considered. If there are two or more documents, the points shall not be summed.		
Education credential	A bachelor's degree/ specialist's degree in the fields of study and specialities correspondent to the enlarged group of the fields of study and specialities within which the admission to the degree programme or in other fields of study or specialities provided	Degree with distinction— 10
	A bachelor's degree/ specialist's degree in the fields of study and specialities correspondent to the enlarged group of the fields of study and specialities within which the admission to the degree programme or in other fields of study or specialities provided	Grade average point is more than 4 of the diploma without distinction - 5
	Bachelor's/specialist's degree in other fields of study and specialities	Degree with distinction—7
	Bachelor's/specialist's degree in other fields of study and specialities	Grade average point is more than 4 of the diploma without distinction - 5

	A letter of recommendation from the scientific supervisor, coach of the student team or another mentor. The letter should contain the information, such as: how long and in what capacity the recommending person knows the applicant, why they recommend the applicant for admission to the degree programme 'International Public Law' Only one document is considered. Should there be two or more certificates, the points shall not be summed		5
--	--	--	---

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 40.00.00 'Law' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

Assessment criteria of individual achievements within admissions to the degree programme 'International Public Law'

Winner's and awardees diplomas of international competitions in the form of moot court trials

Competitions specified in herein include

- 1. Main international and all-Russia competitions in Law in the form of moot courttrials:
  - a. Philip Jessup International Law Moot Court Competition, including team and individual competition at Russia's rounds (top places: first 5 places for team competition and written stage; first 10 places for individual competition; any top place according to the procedures of the competition starting from entering the category of advanced teams),
  - b. Willem C. Vis International Commercial Arbitration Moot (top places: entering the 1/64, special awards for written memorandums or individual honourable mention in individual competition, as well as the win or honourable mention in individual competition at preliminary rounds),
  - c. Jean-Pictet Competition on International Humanitarian Law,
  - d. Frankfurt Investment Moot Court
  - e. International Humanitarian Law Competition named after F. F. Martens (A winner-team and the second place team),
  - f. The international moot court competition "Dispute Resolution in the EAEU',
  - g. The committee can also consider the participation in other competitions in international law in the form of moot court trials. To decide on recognising the competition, the committee shall consider, in particular, the history of the competition, the composition of the organising committee and judges/arbitrators, the number and geographical distribution of the participating in the competition teams.

## The foreign language proficiency

- 1. Since the degree programme is implemented in Russian and English, the applicants should prove their foreign language proficiency
  - TOEFL, IELTS, Cambridge CAE Cambridge CPE, Cambridge FCE certificates, cetificates from the Language Testing Centre at St Petersburg University shall score 10 points;
  - The 'excellent' grade in the degree academic transcript of records 8 points;
  - other certificates 5 points
- 2. Evidences of another language proficiency, including, but not limited to, German (Goethe-Zertificate, etc.) and French: DALF (Diplome Approfondi de Langue Française и др.) shall score extra 5 points.

. 1 /	score extra 3 points.			
41.00.00 POLITICA	L SCIENCE AND REGION	AL STUDIES		
41.04.02 Russian Region Studies	Russian Region Studies	'I am a Professional' All-Russia Olympiad for students Year of 2023/ 2024	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
41.04.04 Political Science	All degree programmes within this subject area	Certificates of the winners and awardees of competitions of research, development projects, international game competitions and student Olympiads  If there are two or more articles	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals The field of study 'Political Science' Year of 2022/2023, 2023/2024  'Vysskaia Liga' Olympiad for students	Winners - 100 points Awardees - 50 points  Students-medal holders
		submitted, the points shall not be summed; evaluated is an article with the highest number of points.	'Vysshaia Liga' Olympiad for students and alumni The field of study 'Political Science' Year of 2022/2023, 2023/2024 'Lomonosov' Universiad The field of study 'Political Science' Year of 2022/2023, 2023/2024	- 15 points Winners - 10 points Awardees - 5 points Winners - 10 points Awardees - 5 points Awardees - 5 points
			'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024  'My entrepreneur initiative. Creative industries for regional development' nomination	Winners - 10 points

	'Leaders of Russia. Politics' all-Russia contest Year of 2022/2023, 2023/2024	Winners - 10 points
	'Contest for Polyglots' all-Russia project Year of 2022/2023, 2023/2024	Winners - 10 points
	'History of my local governance' all- Russia contest Year of 2022/2023, 2023/2024	Winners - 5 points
	Nominations: 'My Community (district, street) Self-Governing Committee working experience'; 'History of my local governance' Research'; 'History of my local governance' videoclip'	
Information on the applicant's papers on topics correspondent to the enlarged group of the fields of study (specialities) within which the admission to a	Scientific publications in periodicals included in the list of the HAC of the Russian Federation	25
respective degree programme is being carried out (a copy of the publication, a copy of the imprint page of the collection or a scientific journal, a copy of the collection contents, as well as a link or a screen shot of the web page, which confirms the fact that the publication has been accepted by the scientometric database must be provided): If the publication is available in digital format, a link to the respective Internet resource must be provided.	Scientific articles in periodicals indexed by the RSCI scientometric database	20
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication. If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.		

		Documents and materials confirming the applicant's participation in grant-supported projects under the order from state-owned science foundations, research organisations or educational organisations of higher education  The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer	Member of a research team, a grant, a project.	10
		Bachelor's /specialist's degree with distinction in the fields of training and specialities correspondent to the following enlarged groups of fields of study and specialities:  37.00.00 Psychological Sciences  38.00.00 Economics and Management  39.00.00 Sociology and Social Work  40.00.00 Law  41.00.00 Political Sciences and Regional Studies  42.00.00 Mass Media and Information  Librarianship  43.00.00 Services and Tourism  44.00.00 Education and Pedagogical Sciences  46.00.00 History and Archaeology  47.00.00 Philosophy, Ethics, and Religious Studies  50.00.00 Art Studies  51.00.00 Cultural Studies and Sociocultural  Projects  58.00.00 Asian and African Studies		10
41.04.05 International Relations	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	International Relations	Winners - 100 points Awardees - 50 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	International Relations	Students-medal holders - 15 points

		Winners - 10 points Awardee- 5 points
'Lomonosov' Universiad Year of 2022/2023, 2023/2024	International Relations	Winners - 10 points Awardee- 5 points
The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Sub-areas 'Promoting the strengthening of the youth inter-confessional and international agreement', 'Prevention and counteraction to extremism'	Winners - 10 points
Bachelor's/Specialist's degree with distinction	Degree with distinction	10
Information about publications on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree	Scientific publications in periodicals indexed in bibliographical foreign databases of publications and/or in RSCI	15
programmes is carried out (a copy/screen shot of the publication, a copy of the imprint page of the collection, and a copy/screen shot of the collection contents page, a copy/screen shot of the author's page or of the publication in the e-library, a copy/screen shot the author's page or of the publication in bibliographical foreign darabases of publications).	Scientific publications in indexed by the RSCI scientometric database	10
Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication  If there are two or more articles submitted, the points shall not be summed; evaluated is an article with the highest number of points.  If the provided document proposes that the publication represents abstracts of the report delivered at the conference or another scientific		
event, points under this clause shall not be scored.  Documents and materials confirming the applicant's participation in research projects supported by grants and confirming the results obtained by them in the course of implemented projects.	Members of research teams, a grant, a project under the order from state-owned science foundations, research organisations or educational organisations of higher education	10

The confirming documents must be issued on organisation's letter-head and certified by the seal and a signature of an authorised officer.  Only one document is considered. If there are two or more articles, the points shall not be summed;		
evaluated shall be the article that meets all criteria  Documents confirming respective working experience addressing the used in the professional activity competences correlating with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity). Provided must be one of the following documents: - a copy of the Employment Book (extracts), certified by a Personnel Department officer of the organisation; - a copy of the employment contract; - a confirmation from the place of work (sealed	12 months and more	10
and signed by an official)  Documents confirming the foreign language proficiency:  Only one document is considered.  Should there be two or more certificates, the points shall not be summed	The 'excellent' grade for 'Foreign Language' in the Russian standard academic transcript of records (if there is a grade other than 'excellent' the points shall not be scored),  or the 'passed' grade in the degree academic transcript of records From an educational organisation ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	10
	The Language Testing Centre at St Petersburg University certificate confirming at least the C1 language reference level	15

Artificial Intelligence and International Security	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	International Relations	Winners - 100 points Awardee- 50 points
	'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Artificial Intelligence	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	International Relations	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
	'Lomonosov' Universiad Year of 2022/2023, 2023/2024	International Relations	Winners - 10 points Awardees - 5 points
	The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Sub-areas 'Promoting the strengthening of the youth inter-confessional and international agreement', 'Prevention and counteraction to extremism'	Winners - 10 points

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 41.00.00 'Political Science and Regional Studies' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

42.00.00 MASS MEDI	42.00.00 MASS MEDIA AND INFORMATION LIBRARIANSHIP				
42.04.01 Advertising	All degree programmes	'I am a Professional' All-Russia Olympiad for	Advertising and Public Relations in	Students-medal holders	
and Public Relations	within this subject area	students	Politics	- 100 points	
		Year of 2022/2023, 2023/2024		Winners - 50 points	
				Awardee- 25 points	
		'Vysshaia Liga' Olympiad for students and alumni	Media Communications	Students-medal holders	
		Year of 2022/2023, 2023/2024		- 15 points	
				Winners - 10 points	
				Awardee- 5 points	
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Subject area 'Youth Media'	Winners - 10 points	

'Leader in the Internet Communication' all-Russia project Year of 2022/2023, 2023/2024		Winners - 10 points
'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Text processing Photographer-reporter	Winners - 10 points
'TopBLOG' all-Russia game project-contest Year of 2022/2023, 2023/2024		Winners - 10 points
'Upravlay' all-Russia youth cup in management Year of 2022/2023, 2023/2024		Winners - 10 points
Information about publications (a copy of the publication, a copy of the imprint page of the collection, and a copy of the collection contents page, a screen shot form the Scopus/Web of Science Core Collection/RSCI databases confirming the indexing of the publication must be provided)  Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools)		on topics relevant to the enlarged group of fields of study and specialities within which the admission to the degree programme is being carried out - 15 points On other topics - 10 points on topics relevant to the enlarged group of fields of study and specialities within which the admission to the degree programme is being
		carried out - 15 points On other topics - 10 points
Documents confirming respective working experience addressing the used in the professional activity competences correlating with the		12 months and longer - 10 points
competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity). To prove the labour record, it is required to provide one of the following (the working experience must		from 6 months to 12 months – 5 points

42.04.02 Journalism	All degree programmes	be evident unambiguously based on the submitted documents): - a copy of the Employment Book (extracts) certified by the organisation's human resource department officer; - a copy of the employment contract; - a confirmation from the place of work (sealed and signed by an official)  Winner's or laureate diplomas for other profession-oriented competitions and Olympiads in the field relevant to the field of study (except Olympiads and competitions to be recognised as individual achievements within admissions to th respective degree programmes) on topics relevant to the enlarged group of fields of study and specialities within which the admissions to degree programmes is being carried out.  Year of 2022/2023, 2023/2024  Petropolitan Science (Re)Search, St Petersburg	International Relations	Winners of international competitions - 10 points Awardees of international competitions - 9 points Winners of All-Russia competitions - 8 points Awardees of All-Russia competitions - 7 points Winners or awardees of regional competitions - 5 points Winners - 100 points
	within this subject area	University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Interdisciplinary research (Arts and Humanitarian Sciences)	Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students	Journalism Linguistics Philology	Students-medal holders - 100 points Winners - 50 points
		Year of 2022/2023, 2023/2024 'Lomonosov' Universiad	Linguistics and Literary Studies  Journalism ('Mediaproject')	Awardees- 25 points Winners - 10 points
		Year of 2022/2023, 2023/2024	Journalisii (Wediaproject)	Awardees - 5 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Field of study 'Youth Media'	Winners - 10 points

'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Text processing Photographer-reporter	Winners - 10 points
'TopBLOG' all-Russia game project-contest Year of 2022/2023, 2023/2024		Winners - 10 points
Information about publications (a copy of the publication, a copy of the imprint page of the collection, and a copy of the collection contents page, a screen shot form the Scopus/Web of Science Core Collection/RSCI databases confirming the indexing of the publication must be provided)		on topics relevant to the enlarged group of fields of study and specialities within which the admission to the degree programme is being carried out - 15 points On other topics - 10 points
Documents and materials confirming the applicant's participation in scientific events (conferences, workshops, research schools)		on topics relevant to the enlarged group of fields of study and specialities within which the admission to the degree programme is being carried out - 15 points On other topics - 10 points
Documents confirming respective working experience addressing the used in the professional activity competences correlating with the competences to be developed in the course of mastering a respective degree programme (experience in professional practical activity). To prove the labour record, it is required to provide one of the following (the working experience must be evident unambiguously based on the submitted documents): - a copy of the Labour Book (extracts) certified by the organisation's human resource department officer; - a copy of the employment contract; - a certificate-confirmation from the place		12 months and longer - 10 points  from 6 months to 12 months – 5 points

	of work (on the letter-head of the organisation with the seal and official's signature)	
	Winner's or laureate diplomas for other profession-oriented competitions and Olympiads in the field relevant to the field of study (except Olympiads and competitions to be recognised as individual achievements within admissions to th respective degree programmes) on topics relevant to the enlarged group of fields of study and specialities within which the admissions to degree programmes is being carried out.  Year of 2022/2023, 2023/2024	Winner's of international competitions - 10 points Awardees of international competitions - 9 points Winner's of All-Russia competitions - 8 points Awardees of All-Russia competitions - 7 points Winners or awardees of regional competitions - 5 points
Professional Speech Activity in Mass Media	Documents confirming the Chinese language proficiency (international certificates and other documents)	HSK 5– 10 points HSK 3.4 – 5 points

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 42.00.00 'Mass Media and Information Librarianship' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

43.00.00 SERVICES AND TOURISM					
43.04.02 Tourism	Tourist Destination Management	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals  Year of 2022/2023, 2023/2024	Earth Sciences	Winners - 100 points Awardees- 50 points	
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Tourism	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points	
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Geography and Tourism	Winners - 10 points Awardees - 5 points	

The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Sub-area 'Development of Youth Tourism and Regional Studies'	Winners - 10 points
'Masters of Hospitality' all-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'My Hospitable Russia' nomination	Winners - 10 points
A bachelor's degree/specialist's degree diploma in the field of study and speciality correspondent to the enlarged group of the fields of study and	Degree with distinction	10
specialities currently open for admissions within a respective degree programme	Grade average point more than 4.0 of the diploma without distinction	7
Bachelor's/specialist's /master's degree in other fields of study and specialities	Degree with distinction	5
	Grade average point more than 4.0 of the diploma without distinction	3
Information about articles on topics relevant to the enlarged group of fields of training (specialities) within which the admission to degree programmes	Articles in publications indexed by scientometric Web of Science Core Collection and Scopus databases	20
is carried out (A copy of the article, a copy of the publication's imprint page, and indexing confirmation must be	Articles in publications indexed by RSCI (RSCI core) scientometric database	15
provided)	Articles in publications indexed by RSCI scientometric database or from the Higher Attestation Committee list.	10
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) with reports on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	Participation in international or national events confirmed by the factually published (including e-publications) report abstracts or by the event programme	10

Documents and materials confirming the applicant's participating in research projects supported by grants on topics relevant to the	
respective enlarged group of fields of training (specialities) in which the admission to the degree	10
programme is being carried out (including students' start-ups supported by grants).	

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 43.00.00 'consult and Tourism' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

## 44.00.00 EDUCATION AND PEDAGOGICAL SCIENCES Petropolitan Science (Re)Search, St Petersburg Winners - 100 points 44.04.01 Pedagogical All degree programmes All subject areas within this subject area University Open International Olympiad for Education Awardee- 50 points university students and young professionals Year of 2022/2023, 2023/2024 'I am a Professional' All-Russia Olympiad for Class supervisor Students-medal holders Pedagogical education (preschool) - 100 points students Pedagogical education (basic) Winners - 50 points Special (defectological) education Year of 2022/2023, 2023/2024 Awardee- 25 points 'Vysshaia Liga' Olympiad for students and alumni **Education and Pedagogics** Students-medal holders Year of 2022/2023, 2023/2024 - 15 points Winners - 10 points Awardees - 5 points 'Flagmany Obrazovaniia' all-Russia professional Winners - 30 points contest 'Uchitel Buduschego' all-Russia contest Year of 2022/2023, 2023/2024 'Urok dlya Uchitelya (A Lesson for a Teacher)' Winners - 10 points interdisciplinary contest for teams. Year of 2022/2023, 2023/2024

'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Pre-school education	Winners - 10 points
'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	'Moya Pedagogicheskaia Initsiativa (National project 'Obrazovaniie' nomination)	Winners - 10 points
Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the journal's/collection of scientific articles imprint page, a copy of pages with the contents of the journal/ collection, screen shots of the page in e-library with evidence of the publication indexing must be provided).  The points shall be scored only after the evidences have been validated through e-library service.  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  Abstracts to the reports delivered at scientific conferences or other scientific events shall not be	Scientific publications indexed in the RSCI scientometric database	4 points for one publication, 8 points maximum
considered.  Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which are to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Master's Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies		1 points for a successfully completed course, 4 points maximum

developed in the course of mastering the respective	
educational programme to which the admission is	
being carried out.	
The points shall only be assigned provided there is	
an availability to verify the submitted documents	
through open resources.	

Points for the established types of individual achievements within the enlarged group of specialities and fields of study44.00.00 'Education and Educational Science' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

## 45.00.00 LINGUISTICS AND LITERARY STUDIES

45.04.01 Phil	ology All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Philology Linguistics	Winners - 100 points Awardee- 50 points
		'I am a Professional' All-Russia Olympiad for students Year of 2022/2023, 2023/2024	Linguistics Philology Linguistics and Literary Studies	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Philology	Winners - 10 points Awardees - 5 points
		'My native country - My Russia' all-Russia contest Year of 2022/2023, 2023/2024	''My native tongue' nomination	Winners - 10 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Sub-area 'Literature and History'	Winners - 10 points
45.04.01 Phil	ology Creative Writing	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philology	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points

45.04.02 Linguistics	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	Philology Linguistics	Winners - 100 points Awardee- 50 points
		Year of 2022/2023, 2023/2024		
		'I am a Professional' All-Russia Olympiad for	Linguistics	Students-medal holders
		students	Philology	- 100 points
			Linguistics and Literary Studies	Winners - 50 points
		Year of 2022/2023, 2023/2024		Awardees - 25 points
		'Ambilimpix' National championship in	Translator / interpreter	Winners - 10 points
		professional competency among individuals with a		
		disability or additional support needs		
		Year of 2022/2023, 2023/2024		
	Artificial Intelligence and	'I am a Professional' All-Russia Olympiad for	Linguistics	Students-medal holders
	Speech Technologies	students	Philology	- 100 points
			Linguistics and Literary Studies	Winners - 50 points
		Year of 2022/2023, 2023/2024	Artificial Intelligence	Awardees - 25 points

In case there are several types of individual achievements specified for one subject area/degree programme, the points shall be assigned only for a single type of the specified individual achievements (upon the applicant's choice). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates for the completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

46.04.01 History	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	History of Arts	Winners - 100 points Awardees - 50 points
		Year of 2022/2023, 2023/2024		
		'I am a Professional' All-Russia Olympiad for students	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points
		Year of 2023/ 2024		Awardees - 25 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Subject area 'Patriotic Education' Sub-area 'Literature and History'	Winners - 10 points

		'Vysshaia Liga' Olympiad for students and alumni	History	Students-medal holders
		Year of 2022/2023, 2023/2024	History of Arts	- 15 points Winners - 10 points
				Awardees - 5 points
		'Lomonosov' Universiad	Theory and History of Art	Winners - 10 points
		Year of 2022/2023, 2023/2024		Awardees- 5 points
46.04.03	All degree programmes	Petropolitan Science (Re)Search, St Petersburg	History of Arts	Winners - 100 points
Anthropology and	within this subject area	University Open International Olympiad for		Awardee- 50 points
Ethnology		university students and young professionals		
		Year of 2022/2023, 2023/2024		
		'I am a Professional' All-Russia Olympiad for	History and Culture of Russia	Students-medal holders
		students		- 100 points
		V		Winners - 50 points
		Year of 2023/2024	III at a ma	Awardees - 25 points
		'Vysshaia Liga' Olympiad for students and alumni	History	Students-medal holders
		Year of 2022/2023, 2023/2024	History of Arts	- 15 points
				Winners - 10 points Awardees - 5 points
		'Lomonosov' Universiad	Theory and History of Art	
		Year of 2022/2023, 2023/2024	Theory and history of Art	Winners - 10 points
		1 car or 2022/2023, 2023/2024		Awardees - 5 points

In case there are several types of individual achievements specified for one field of study/degree programme within admissions to the enlarged group of fields of study and specialities 46.00.00 History and Archaeology, the points shall be assigned only for a single type of the specified individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates for the completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

47.00.00 PHILOSOPH	Y, ETHICS, AND RELIGI	OUS STUDIES		
		'I am a Professional' All-Russia Olympiad for students Year of 2023/ 2024	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
47.04.01 Philosophy	Current Philosophical Discourse	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philosophy	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points

		International Philosophy contest for university students (ACM/ICPC or ICPC) Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
		The Olympiad of federal universities for those applying for the admissions to master's degree programmes Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
47.04.02 Applied Ethics	Applied Ethics	'I am a Professional' All-Russia Olympiad for students Year of 2023/ 2024	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philosophy	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		International Philosophy Olympiad for university students (ACM/ICPC or ICPC) Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
		The Olympiad of federal universities for those applying for the admissions to master's degree programmes Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2023/ 2024	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
47.04.03 Religious Studies	Religious Studies	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philosophy	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		International Philosophy Olympiad for university students (ACM/ICPC or ICPC) Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points

		The Olympiad of federal universities for those applying for the admissions to master's degree programmes Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
and specialities 47.00.0 discretion). In case their courses or online progr	00 Philosophy, Ethics, and Re re are several individual achie ammes), the points shall be co	achievements specified for one field of study/degree p ligious Studies, the points shall be assigned only for a sevements of a similar type (several Olympiad diplomas onferred for only one individual achievement evidence	single type of the specified individual achieve of a winner/awardee, several certificates for t	ements (at the applicant's
<b>48.00.00 THEOLOGY</b> 48.06.01 Theology	All degree programmes within this subject area	'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Philosophy	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
49.03.01 Physical Education	Physical Culture and Sports	Research activities	RSCI scientific publications, including the ones in peer-reviewed journals recommended by the HAC, and in periodicals, included in Scopus, Web of Science databases, shall be outnumbered in a separate list with topics of the reports or publications and their input data.  * The participation in regional, all-republic or all-Russia and international conferences; student conferences; scientific schools and associations, round tables; presentations at festivals, seminars, scientific congresses; grants for research activities; awards for the participation in student especially all-Russia competitions and Olympiads.	0- 15
		Project (labour) activities	Experience in practical participation in scientific projects, labour activities, internship, and evidences proving such activities.  Letters of recommendation from mentors, heads of organisations or institutions.	0-8

	Sports qualification	Sports titles	Merited Master of Sports
			of the Russian Federation
			- 7 points
			Master of Sports of
			International Level of the
			Russian Federation - 6
			points
			Master of Sports of the
			Russian Federation - 5
			points
		Sports categories	Candidate Master of
			Sports - 4 points
			Category 1 - 3 points
			Category 2 - 2 points
			Category 3 - 1 point

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 49.00.00 'Physical Culture and Sports' shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

50.00.00 ART STUDIES				
50.04.01 Arts and	Art Criticism	Petropolitan Science (Re)Search, St Petersburg	Interdisciplinary research (Arts and	Winners - 100 points
Humanities	Music Criticism	University Open International Olympiad for university students and young professionals	Humanities)	Awardee- 50 points
	Curatorial Research	Year of 2022/2023, 2023/2024		
	Historical Performance on Keyboard Musical Instruments	The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives  Year of 2022/2023, 2023/2024	Subject areas 'Theatre and Cinematography' 'Music and choreography', 'Artistic work'	Winners - 10 points
50.04.01 Arts and Humanities	Historical Performance on Keyboard Musical	Laureate of international contest		10
	Instruments	Year of 2022/2023, 2023/2024		

		Diploma winner of international contest		8
		Year of 2022/2023, 2023/2024		
		Laureate of all-Russia contest		6
		Year of 2022/2023, 2023/2024		
		Diploma winner of all-Russia contest		4
		Year of 2022/2023, 2023/2024		
50.04.01 Arts and Humanities	Intercultural Education	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Interdisciplinary research (Arts and Humanities) Linguistics Pedagogy Psychology	Winners - 100 points Awardees - 50 points
		'I am a Professional' All-Russia Olympiad for students  Year of 2022/2023, 2023/2024	Class supervisor Pedagogical education (preschool) Pedagogical education (basic) Special (defectological) education	Students-medal holders - 100 points Winners - 50 points Awardee- 25 points
		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Education and Pedagogics	Students-medal holders - 15 points Winners - 10 points Awardees - 5 points
		'Urok dlya Uchitelya (A Lesson for a Teacher)' interdisciplinary contest for teams. Year of 2022/2023, 2023/2024		Winners - 10 points
		'Flagmany Obrazovaniia' all-Russia professional contest 'Uchitel Buduschego' all-Russia contest Year of 2022/2023, 2023/2024		Winners - 30 points
		Information about publications on topics relevant to the enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out (a copy of the screen shot, a copy of the journal's/collection of scientific articles imprint page, a copy of pages	Scientific publications indexed in the RSCI scientometric database	4 points for one publication, 8 points maximum

		with the contents of the journal/collection, screen shots of the page in e-library with evidence of the publication indexing must be provided).  The points shall be scored only after the evidences have been validated through e-library service.  Certificates/diplomas/letters of appreciation shall not be deemed as the confirmation of the publication.  Abstracts to the reports delivered at scientific conferences or other scientific events shall not be considered.		
		Online course certificates for the courses on educational platforms 'Coursera' and 'Open Education' (except online course of St Petersburg University, which to be recognised according to the provisions of the separate clause of the List and the Procedures for Applicants' Individual Achievement Evidences Recognition within Admissions to Degree Programmes) provided the online course implied the acquisition of competencies correspondent to the competencies developed in the course of mastering the respective educational programme to which the admission is being carried out.		1 points for a successfully completed course, 4 points maximum
		The points shall only be assigned provided there is an availability to verify the submitted documents through open resources.		
50.04.01 Arts and Humanities	Cognitive Studies	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	Linguistics Interdisciplinary research (Arts and Humanities) Psychology	Winners - 100 points Awardee- 50 points

		Bachelor/Specialist degree	Degree with distinction (grade point average more than 4.75)	10
			Degree with 4.5 - 4. 75 grade point average	8
			Degree with 4 - 4. 5 grade point average	4
50.04.03 History of Arts	All degree programmes within this subject area	Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals Year of 2022/2023, 2023/2024	History of Arts	Winners - 100 points Awardees - 50 points
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Fields of study 'Theatre and Cinematography' 'Music and choreography', 'Artistic work'	Winners - 10 points

<sup>1.</sup> Scores for an individual achievement within the enlarged group of fields of study and specialities 50.00.00 'Art History' (except the degree programme 'Intercultural Education') shall only be assigned for one of the established types of individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates of completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

## 51.04.01 Cultural Studies All degree programmes within this subject area Year of 2023/2024 Students Students History and Culture of Russia History and Culture of Russia Students History and Culture of Russia Students Year of 2023/2024 Awardees - 25 points

<sup>2.</sup> Points for the established types of individual achievements within the degree programme 'Intercultural Education' shall be summed; at the same time, only one documents shall be considered for one type of individual achievements unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

		'Vysshaia Liga' Olympiad for students and alumni Year of 2022/2023, 2023/2024	Cultural Studies	Students-medal holders - 15 points Winners - 10 points Awardee- 5 points
		'Lomonosov' Universiad Year of 2022/2023, 2023/2024	Universiad in Linguistics, Regional Studies and Cultural Studies	Winners - 10 points Awardees - 5 points
		International Philosophy Olympiad for university students (ACM/ICPC or ICPC) Year of 2022/2023, 2023/2024	Philosophy	Winners - 10 points Awardees - 5 points
		The Olympiad of federal universities for those applying for the admissions to master's degree programmes Year of 2022/2023, 2023/2024	Cultural Studies Philosophy	Winners - 10 points Awardees - 5 points
51.03.04 Museology and Conservation of Cultural and Natural Heritage	All degree programmes within this field of study	'I am a Professional' All-Russia Olympiad for students  Year of 2023/ 2024	History and Culture of Russia	Students-medal holders - 100 points Winners - 50 points Awardees - 25 points
		Petropolitan Science (Re)Search, St Petersburg University Open International Olympiad for university students and young professionals	History of Arts	Winners - 100 points Awardees - 50 points
		Year of 2022/2023, 2023/2024		

In case there are several types of individual achievements specified for one field of study/degree programme within admissions to the enlarged group of fields of study and specialities 51.00.00 Cultural Studies and Sociocultural Projects, the points shall be assigned only for a single type of the specified individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates for the completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

54.00.00 FINE AND APPLIED ARTS				
54.04.01 Design	All degree programmes	'I am a Professional' All-Russia Olympiad for	Design	Students-medal holders
	within this subject area	students	Creative industries	- 100 points
				Winners - 50 points
		Year of 2022/2023, 2023/2024		Awardees - 25 points
		'Otlichno! (Excellent!)' the National award for the	Environmental Design	Winners - 10 points
		best graduate designers.	Communication Design	

		Year of 2022/2023, 2023/2024		
		The Federal Agency for Youth Affairs (among individuals) grant contest of the youth initiatives Year of 2022/2023, 2023/2024	Sub-areas 'Architecture, design, urbanism', 'Artistic work'	Winners - 10 points
		'Digital Breakthrough' All-Russia contest Year of 2022/2023, 2023/2024		Winners - 10 points
		'Ambilimpix' National championship in professional competency among individuals with a disability or additional support needs Year of 2022/2023, 2023/2024	Artistic design	Winners - 10 points
54.04.04 Restoration	Restoration of Fine and Applied Art Objects	'Restavrator goda' contest of the professional mastership Year of 2022/2023, 2023/2024	Restoration	Winners - 25 points Awardees - 10 points

In case there are several types of individual achievements specified for one field of study/degree programme within admissions to the enlarged group of fields of study and specialities 54.00.00 Fine and Applied Arts, the points shall be assigned only for a single type of the specified individual achievements (at the applicant's discretion). In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates for the completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

50 AA AA	ACTANIAND	A EDICAN CTUDIES

58.04.01 Asian and	All dagraa programmas	'I am a Professional' All-Russia Olympiad for	Asian Studies Chinese Studies, Korean	Students-medal holders
	All degree programmes	_ ·		
African Studies	within this subject area	students	Studies, Japanese Studies	- 100 points
				Winners - 50 points
		Year of 2022/2023, 2023/2024		Awardees - 25 points
		'Vysshaia Liga' Olympiad for students and alumni	Asian studies: contemporary Asia	Students-medal holders
				- 15 points
		Year of 2022/2023, 2023/2024		Winners - 10 points
				Awardees - 5 points
	Asian and African	Bachelor's/specialist's degree in other fields of	Bachelor's degree in 58.03.01 Asian and	
	Studies	study and specialities	African Studies, 41.03.03 Asian and	100
		relevant to the fields of study:	African Studies with distinction	
		58.03.01 Asian and African Studies. 41.03.03	Dashalan's dasmas in 50 02 01 Asian and	
		Asian and African Studies	Bachelor's degree in 58.03.01 Asian and	70
		Standard credentials confirming completion of	African Studies, 41.03.03 Asian and	

professional training courses correspondent to the main field of study of the master's degree programme*	African Studies without distinction (grade average point 4 and higher)	
Bachelor's/specialist's degree in other fields of study and specialities Credentials shall only be evaluated within one above-	Bachelor's/specialist's degree with distinction in other fields of study that implied mastering an Asian language	60
mentioned categories Points for several evidences shall not be summed.	Bachelor's/specialist's degree without distinction (grade average point 4 and higher) in other fields of study that implied mastering an Asian language	50
	Evidence of professional retraining in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	40
	Evidence of professional training	0
The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project	The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	10
	The 'Good' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	7
	The 'Excellent' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	7
	The 'Good' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	5
	The 'Excellent' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies,	5

1		T
	41.03.03 Asian and African Studies	
	The 'Good' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	3
A grade for an Asian language' in the academic transcript of records <i>If there are grades for two</i>	The 'excellent' grade for the final state assessment (exam)	10
Asian language in the transcript of records, considered shall be the one for the first Asian language	The 'good' grade for the final state assessment (exam)  The 'excellent' grade for the assessment	5
	(exam)	3
	The 'good' grade for the assessment (exam)	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme.	4
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents and materials confirming the applicant's participation in archaeological, ethnographic and linguistic expeditions (a certificate from the head of the expedition or an invitation/certificate form a foreign institution).		6
Only one document is considered. Should there be two or more certificates, the points shall not be summed		

	International certificates or formal course certificates confirming the Asian language proficiency or documents confirming the fact of studying abroad in an Asian language.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	International certificates with scored points correspondent to B2 level and higher (JLPT, HSK, TOPIC, etc.) Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)  The score shall not be increased for two or more documents in this subsection	10
	Winner's and laureate diplomas for competitions and student Olympiads Year of 2022/2023, 2023/2024  Only one document is considered. Should there be two or more certificates, the points shall not be summed	Diplomas: - of the winner or an awardee of all-Russia Olympiads and competitions in Chinese, Turkish, Persian, Korean and other Asian languages.	10
Contemporary China: Economy, Politics, Society (with the Chinese	Bachelor's/specialist's degree in other fields of study and specialities relevant to the fields of study, such as:	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies with distinction	100
language training)	58.03.01 Asian and African Studies. 41.03.03 Asian and African Studies Standard credentials confirming completion of professional training courses correspondent to the main field of study of the master's degree programme*	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies without distinction (grade average point 4 and higher)	70
	Bachelor's/specialist's degree in other fields of study and specialities Credentials shall only be evaluated within one above-	Bachelor's/specialist's degree with distinction in other fields of study that implied mastering an Asian language	60
	mentioned categories Points for several evidences shall not be summed.	Bachelor's/specialist's degree without distinction (grade average point 4 and higher) in other fields of study that implied mastering an Asian language	50
		Evidence of professional retraining in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	40

	Evidence of professional training	0
The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project	The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	10
	The 'Good' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	7
	The 'Excellent' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	7
	The 'Good' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	5
	The 'Excellent' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	5
	The 'Good' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	3
A grade for an Asian language' in the academic transcript of records <i>If there are grades for two Asian language in the transcript of records</i> ,	The 'excellent' grade for the final state assessment (exam)	10
considered shall be the one for the first Asian language	The 'good' grade for the final state assessment (exam)	5

	The 'excellent' grade for the assessment (exam)  The 'good' grade for the assessment (exam)	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme.	4
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents and materials confirming the applicant's participation in archaeological, ethnographic and linguistic expeditions (a certificate from the head of the expedition or an invitation/certificate form a foreign institution).		6
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
International certificates or formal course certificates confirming the Asian language proficiency or documents confirming the fact of studying abroad in a foreign language.	At least HSK level 5 international certificate	100
Winner's and laureate diplomas for competitions and student Olympiads Year of 2022/2023, 2023/2024	Diplomas: - of the winner or an awardee of all-Russia Olympiads and competitions in Chinese	10
Accepted can be one document. Should there be two or more certificates, the points shall not be summed		

Islamic Studies	Bachelor's/specialist's degree in other fields of study and specialities relevant to the fields of study, such as: 58.03.01 Asian and African Studies. 41.03.03	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology' with distinction	100
	Asian and African Studies, 48.03.01 'Theology' Standard credentials confirming completion of professional training courses correspondent to the main field of study of the master's degree programme*	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology' without distinction (grade average point 4 and higher)	70
	Bachelor's/specialist's degree in other fields of study and specialities Credentials shall only be evaluated within one above-	Bachelor's/specialist's degree with distinction in other fields of study that implied mastering an Asian language	60
	mentioned categories Points for several evidences shall not be summed.	Bachelor's/specialist's degree without distinction (grade average point 4 and higher) in other fields of study that implied mastering an Asian language	50
		Evidence of professional retraining in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology'	40
		Evidence of professional training	0
	The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project	The 'Excellent' grade for the graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology'	10
		The 'Good' grade for the graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology'	7
		The 'Excellent' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	7
		The 'Good' grade for the degree graduation project in fields of study and	5

	specialities implied mastering of an Asian language	
	The 'Excellent' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology'	5
	The 'Good' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies, 48.03.01 'Theology'	3
A grade for an Asian language' in the academic transcript of records <i>If there are grades for two</i>	The 'excellent' grade for the final state assessment (exam)	10
Asian language in the transcript of records,	The 'good' grade for the final state	10
considered shall be the one for the first Asian	assessment (exam)	5
language	The 'excellent' grade for the assessment	_
	(exam)	3
	The 'good' grade for the assessment (exam)	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme.	4
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents and materials confirming the applicant's participation in archaeological, ethnographic and linguistic expeditions (a certificate from the head of the expedition or an		6

	invitation/certificate form a foreign institution).		
	Only one document is considered. Should there be two or more certificates, the points shall not be summed		
	International certificates or formal course certificates confirming the Asian language proficiency or documents confirming the fact of studying abroad in an Asian language.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	International certificates with scored points correspondent to B2 level and higher (JLPT, HSK, TOPIC, etc.) Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)  The score shall not be increased for two or more documents in this subsection	10
	Winner's and laureate diplomas for competitions and student Olympiads Year of 2022/2023, 2023/2024  Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Diplomas: - of the winner or an awardee of all-Russia Olympiads and competitions in Arabic, Turkish, Persian.	10
Theory and Practice of Translation (Asian and African Languages)	ctice of Bachelor's/specialist's degree in other fields of study and specialities	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies with distinction	100
58.03.01 Asian and African Studies. 41.03.03 Asian and African Studies Standard credentials confirming completion of professional training courses correspondent to the main field of study of the master's degree programme*	Bachelor's degree in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies without distinction (grade average point 4 and higher)	70	
	Bachelor's/specialist's degree in other fields of study and specialities <i>Credentials</i> shall only be evaluated within one above-	Bachelor's/specialist's degree with distinction in other fields of study that implied mastering an Asian language	60
	mentioned categories Points for several evidences shall not be summed.	Bachelor's/specialist's degree without distinction (grade average point 4 and	50

	higher) in other fields of study that implied mastering an Asian language  Evidence of professional retraining in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	40
	Evidence of professional training	0
The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project	The 'Excellent' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	10
	The 'Good' grade for the bachelor's degree/ specialist's degree graduation project in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	7
	The 'Excellent' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	7
	The 'Good' grade for the degree graduation project in fields of study and specialities implied mastering of an Asian language	5
	The 'Excellent' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	5
	The 'Good' grade for the certification project (in professional retraining transcript of records in 58.03.01 Asian and African Studies, 41.03.03 Asian and African Studies	3

A grade for an Asian language' in the academic transcript of records If there are grades for two Asian language in the transcript of records,	The 'excellent' grade for the final state assessment (exam)  The 'good' grade for the final state	10
considered shall be the one for the first Asian language	assessment (exam)	5
	The 'excellent' grade for the assessment (exam)	3
	The 'good' grade for the assessment (exam)	2
Documents and materials confirming the applicant's participation in scientific events (conferences, seminars, scientific schools) on topics relevant to the respective enlarged group of fields of study (specialities) within which the admission to the degree programme is being carried out.	The participation must be confirmed by the published abstract to the report or the event programme.	4
Only one document is considered. Should there be two or more certificates, the points shall not be summed		
Documents and materials confirming the applicant's participation in archaeological, ethnographic and linguistic expeditions (a certificate from the head of the expedition or an invitation/certificate form a foreign institution).  Only one document is considered. Should there be		6
two or more certificates, the points shall not be summed		
International certificates or formal course certificates confirming the Asian language proficiency or documents confirming the fact of studying abroad in an Asian language.  Only one document is considered. Should there be two or more certificates, the points shall not be summed	International certificates with scored points correspondent to B2 level and higher (JLPT, HSK, TOPIC, etc.) Training at universities ranked among top 300 by international ratings (Shanghai Ranking, The Three University Missions Moscow International University Ranking)	10

	The score shall not be increased for two or more documents in this subsection	
Winner's and laureate diplomas for competitions and student Olympiads Year of 2022/2023, 2023/2024	Diplomas: - of the winner or an awardee of all-Russia Olympiads and competitions in Chinese,	10
Accepted can be one document. Should there be two or more certificates, the points shall not be summed	Turkish, Persian, Korean and other Asian languages.	10

Points for the established types of individual achievements within the enlarged group of specialities and fields of study 58.00.00 Asian and African Studies shall be summed; at the same time, only one documents shall be considered for one type of individual achievement unless otherwise is provided by this List of individual achievements of those applying for the admission to master's degree programmes.

0.0000000000000000000000000000000000000	T	1.45 . 5 . 6	 
01.00.00 Mathematics	All degree programmes	All-Russia Engineering Contest	Winners– 50 points
and Mechanics	within the mentioned		Awardees– 25 points
02.00.00 Computer	large groups of fields of	Year of 2022/2023, 2023/2024	
and Information	study		
Sciences			
03.00.00 Physics and			
Astronomy			
04.00.00 Chemistry			
05.00.00 Earth			
Sciences			
09.00.00 Informatics			
and Computer			
Engineering			

21.00.00 Applied				
Geology, Mining,				
Petroleum Engineering				
and Geodesy				
27.00.00 Control in				
Engineering Systems				
28.00.00				
Nanotachnology and				
nanomaterials				
38.00.00 Economics				
and Management				
All subject areas				
All subject areas	All degree programmes	'Leaders of Russia' all-Russia contest for		Winners - 25 points
		managers		
		Year of 2022/2023, 2023/2024		
		'Student of the year' Russian national award		Winners– 10 points
		Year of 2022/2023, 2023/2024		
		'Tvoi Khod' all-Russia contest		Winners– 10 points
		Year of 2022/2023, 2023/2024		
		'Zolotaia stazhirovka' all-Russia contest		Winners– 10 points
		Year of 2022/2023, 2023/2024		
		All-Russia Population Census		Censor – 5 points
		'My native country - My Russia' all-Russia contest	'Commonwealth of our countries'	Winners– 5 points
		Year of 2022/2023, 2023/2024	nomination. (The nomination for	
			foreigners and compatriots residing	
			abroad)	

In case there are several types of individual achievements specified for one field of study/degree programme, the points shall be assigned only for a single type of the specified individual achievements (upon the applicant's choice), unless otherwise provided by this List of individual achievements within admissions to master's degree programmes. In case there are several individual achievements of a similar type (several Olympiad diplomas of a winner/awardee, several certificates for the completed online courses or online programmes), the points shall be conferred for only one individual achievement evidence.

- 1. The University shall recognise the individual achievements of applicants to master's degree programmes by assigning points that are to be added to the points awarded for the admissions test (admissions tests).
- 2. The St Petersburg University Admissions Committee will confer points for individual achievements only if the applicant has obtained at least 65 points for their first-taken admission test based on the admissions tests priority set forth by the Admissions Committee.
- 3. If the applicant has several evidences of individual achievements, the points shall be assigned only for one individual achievement, unless otherwise set forth by the Procedures for the individual achievements recognition within the enlarged group of fields of study and specialities.
- 4. In their admission application, applicants must specify a particular individual achievement, for which they may be assigned additional points (if the applicant has several individual achievements when admitted to one subject area/degree programme, they must specify only a single of them) and specify the details of the document confirming this particular individual achievement.
  - 5. The applicant must attach documents confirming their individual achievements to their personal admission application.
- 6. If the admissions test is held in the form of a competitive selection of documents (portfolios) and points are conferred to the applicant based on the already provided particular documents in the framework of the portfolio, points for individual achievements based on the same documents considered earlier as the portfolio shall not be conferred.
- 7. The results of mastering the open online courses at St Petersburg University shall be considered as personal achievements during admission to master's degree programmes taking into account the congruity between open online courses at St Petersburg University and degree programmes (enlarged groups of subject areas), to which the admission has been announced at St Petersburg University. The applicant shall be assigned five (5) points if they provide a certificate of the successful completion of the open online courses or online programmes at St Petersburg University and meet requirements outlined in Clause 2 of the List and Procedures for Recognising Applicant's Individual Achievements when applying to master's degree programmes.

The congruency between the open online courses at St Petersburg University and the degree programmes (enlarged groups of subject areas) shall be established as follows:

COURSE OR PROGRAMME TITLE	FIELD OF STUDY	
PHYSICAL AND MATHEMATICAL SCIENCES		
Computational Geometry	02.00.00 Computer and Information Sciences	
Fundamentals of electron and ion microscopy	03.00.00 Physics and Astronomy	

Interface Design: interactive systems interface design (including any course of the online programme)	02.00.00 Computer and Information Sciences
Introduction to Data Analysis	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
LegalTech (including any course of the online programme)	06.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Machine Learning Based Econometrics	09.00.00 Informatics and Computer Engineering
Multi-agent Technologies	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
One-way Quantum Computation	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Quantum Computer Programming on Python with Qiskit	02.00.00 Computer and Information Sciences
Quantum Computing. Less Formulas— More Understanding	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 03.00.00 Physics and Astronomy 01.04.02 Applied Mathematics and Informatics
Quantum Computing: from Basics to the Cutting Edge (including any course of the online programme)	03.00.00 Physics and Astronomy
The Introduction to Quantum Computing	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 03.00.00 Physics and Astronomy 01.04.02 Applied Mathematics and Informatics
UX/UI-Design: Project Development, Management and Prototyping (previous title: Project Development, Management and Prototyping (UX/UI-Design))	02.00.00 Computer and Information Sciences

UX/UI-Design: Usability Testing of Interactive Systems (previous title: Usability Testing of Interactive Systems in UX/UI-Design)	02.00.00 Computer and Information Sciences
UX-Strategy	02.00.00 Computer and Information Sciences
Automation in Legal Activities: Analysis of Practices	06.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Computer Architecture	01.00.00 Mathematics and Mechanics
Databases	01.00.00 Mathematics and Mechanics
Database and Work with Data	01.00.00 Mathematics and Mechanics 09.00.00 Informatics and Computer Engineering
Block chain and finances: concepts, technologies, and control	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Block chain: Architecture and Operating Principles	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Blockchain: Objectives, Business-models and Controll	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Introduction to LegalTech	06.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Introduction to UX/UI- Design	02.00.00 Computer and Information Sciences
Introduction to Web Data Analysis	02.00.00 Computer and Information Sciences
Introduction to Bioinformatics	09.00.00 Informatics and Computer Engineering
Introduction to Bioinformatics: Metagenomics	09.00.00 Informatics and Computer Engineering
Introduction to Deformable Body Mechanics	01.00.00 Mathematics and Mechanics

	02.00.00 Computer and Information Sciences
Web Analytics	02.00.00 Computer and Information Sciences
Interaction of Science with Arts and Technologies (including any course of the online programme)	03.00.00 Physics and Astronomy
Interaction of Science with Arts and Technologies. Introductory Course	03.00.00 Physics and Astronomy
Higher Mathematics. Algebra: Introduction to Group Theory	09.00.00 Informatics and Computer Engineering
Introduction to R and basic statistics	01.00.00 Mathematics and Mechanics 27.00.00 Control in Engineering Systems
Tools for Creating LegalTech-Solutions	06.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Artificial Intelligence: Fundamentals	09.00.00 Informatics and Computer Engineering
Quantum Computing	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.02 Applied Mathematics and Informatics
Space Weather: environmental impact assessment (in any language)	03.00.00 Physics and Astronomy
Linear Regression	01.00.00 Mathematics and Mechanics
Linear Models with Discrete Predictors	01.00.00 Mathematics and Mechanics
Mathematical Game Theory (in any language)	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences 01.04.02 Applied Mathematics and Informatics
Mathematical Methods in Psychology: Basic Applications	01.00.00 Mathematics and Mechanics
Machine Learning: Fundamentals	09.00.00 Informatics and Computer Engineering
Mobile Analytics	02.00.00 Computer and Information Sciences
Data Science (previous title: Introduction to Data Science)	02.00.00 Computer and Information Sciences
Generalised Linear Models	01.00.00 Mathematics and Mechanics
Optical Science and Visual Arts New Era and Modernity	03.00.00 Physics and Astronomy

Optical Science and Visual Arts From Ancient Times to the 17th century Part 1.	03.00.00 Physics and Astronomy
Optical Science and Visual Arts From Ancient Times to the 17th century Part 2.	03.00.00 Physics and Astronomy
Java Basics Elementary level	09.00.00 Informatics and Computer Engineering
Introduction to digital environment	02.00.00 Computer and Information Sciences
Legal regulation of Legal Relations in Application of Distributed Ledger Technology	01.00.00 Mathematics and Mechanics 02.00.00 Computer and Information Sciences
Digital Transformation Practices (in any language)	27.00.00 Control in Engineering Systems
Programming Quantum Computers in Python with Qiskit	03.00.00 Physics and Astronomy
Web Site Promotion	02.00.00 Computer and Information Sciences
Website design (including any course of the online programme)	02.00.00 Computer and Information Sciences
Statistics Made Simple (Using R) (including any course of the online programme)	01.00.00 Mathematics and Mechanics
WordPress web-design	02.00.00 Computer and Information Sciences
Extended Full-Stack. Graphs in practice	02.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Extended Full-Stack. From an idea to a Minimal Viable Product	02.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Extended Full-Stack. From an Idea to a Minimal Viable Product Graphs in practice (including any course of the online programme)	02.00.00 Computer and Information Sciences 09.00.00 Informatics and Computer Engineering
Mixed Linear Models	01.00.00 Mathematics and Mechanics
3D-Modelling of objects based on digital images (previous title: Creation of three-dimensional models of objects based on digital images)	02.00.00 Computer and Information Sciences 01.04.02 Applied Mathematics and Informatics
Sports Programming	09.00.00 Informatics and Computer Engineering
Sport Algorithmic Programming	09.00.00 Informatics and Computer Engineering

Theory of Cybernetic Systems: Introduction (previous title:	01.00.00 Mathematics and Mechanics	
Introduction to Cybernetic Systems Theory)	02.00.00 Computer and Information Sciences	
introduction to Cybernetic Systems Theory)	01.04.02 Applied Mathematics and Informatics	
UAV Aerial Photography Technology	02.00.00 Computer and Information Sciences	
Phase Transitions in Bulk Materials and Low-Dimensional	02 00 00 Physics and Astronomy	
Systems	03.00.00 Physics and Astronomy	
Physical Basis of Quantum Computing (in any language)	01.00.00 Mathematics and Mechanics	
Friystear Basis of Quantum Computing (in any ranguage)	03.00.00 Physics and Astronomy	
Artistic Web Design	02.00.00 Computer and Information Sciences	
Digital Currencies and Plack Chain Technology	01.00.00 Mathematics and Mechanics	
Digital Currencies and Block Chain Technology	02.00.00 Computer and Information Sciences	
Nuclear Physics	03.00.00 Physics and Astronomy	
NATUR	RAL SCIENCES	
Computational Chemistry from Molecular Properties to	04.00.00 Chamistry	
Reactions: Learning-by-Doing	04.00.00 Chemistry	
General Pathophysiology	32.00.00 Health Sciences and Preventive Medicine	
Heart Physiology (in any language)	06.00.00 Biological Sciences	
Protists: Evolution and Ecology of Microbial Eukaryotes	06.00.00 Biological Sciences	
Structure of Matter: Atoms, Molecules, Nanomaterials. Part 1	04.00.00 Chemistry	
Structure of Matter: Atoms, Molecules, Nanomaterials. Part 2	04.00.00 Chemistry	
Visual Perception and Visual Illusions (in any language)	06.00.00 Biological Sciences	
Agrolandscapes: Fundamentals of rational management		
(previous title: Fundamentals of Rational Agrolandscape	06.00.00 Biological Sciences	
Management		
Introduction to Bioinformatics	06.00.00 Biological Sciences	
Introduction to Bioinformatics: Metagenomics	06.00.00 Biological Sciences	
Introduction to Geology of Mineral Resources	05.00.00 Earth Sciences	

Introduction to Parasitology	06.00.00 Biological Sciences
Introduction to Physiology	06.00.00 Biological Sciences
Genetics of Development of Plants	06.00.00 Biological Sciences
	04.00.00 Chemistry
Introduction to R and basic statistics	05.00.00 Earth Sciences
introduction to K and basic statistics	06.00.00 Biological Sciences
	32.00.00 Health Sciences and Preventive Medicine
Computer-based modelling of chemical reactions: a practical	04.00.00 Chemistry
way (in any language)	04.00.00 Chemistry
Coronaviruses SARS-Co-2 and Pathogens of Acute Respiratory	06.00.00 Biological Sciences
Viral Infections (in any language)	00.00.00 Biological Sciences
Space Weather: environmental impact assessment (in any	05.00.00 Earth Sciences
language)	
Linear Regression	06.00.00 Biological Sciences
Medical English Part 1.	32.00.00 Health Sciences and Preventive Medicine
Medical English Part 2.	32.00.00 Health Sciences and Preventive Medicine
Medical English Part 3.	32.00.00 Health Sciences and Preventive Medicine
Medical English Part 4.	32.00.00 Health Sciences and Preventive Medicine
Medical English Part 5.	32.00.00 Health Sciences and Preventive Medicine
Neurolinguistics (in any language)	06.00.00 Biological Sciences
Inorganic Chemistry: Introduction to the Chemistry of	04.00.00 Chemistry
Elements	
Pathology and Pathological Anatomy (previous title: General	32.00.00 Health Sciences and Preventive Medicine
Issues of Pathology and Pathological Anatomy)	32.00.00 Health Sciences and Preventive Medicine
Rehabilitation after COVID-19 (in any language)	32.00.00 Health Sciences and Preventive Medicine
Symbiosis Genetics: Principles of Formation of Plant-Microbe	06 00 00 Riological Sciences
Genetic Systems	06.00.00 Biological Sciences

The Composition of Matter: from Nuclei and Molecules to Materials and Nanoparticles	04.00.00 Chemistry	
UAV Aerial Photography Technology	05.00.00 Earth Sciences 21.00.00 Applied Geology, Mining, Petroleum Engineering and Geodesy	
Transcription and the RNA World	06.00.00 Biological Sciences	
Physiology of the heart and its Dysfunctions (in any language)	06.00.00 Biological Sciences	
Photometric Processing of UAV Aerial Photographs (in any language)	05.00.00 Earth Sciences 21.00.00 Applied Geology, Mining, Petroleum Engineering and Geodesy	
The Evolutionary History of Vertebrates: from Fish to Dinosaurs and Humans	06.00.00 Biological Sciences	
Ecology and Environment	05.00.00 Earth Sciences	
Crude Minerals Economy	05.00.00 Earth Sciences	
EDUCATION AND PEDAGOGICAL SCIENCES		
Interaction of Science with Arts and Technology (including any course of the online programme)	44.00.00 Education and Pedagogical Sciences	
Interaction of Science with Arts and Technologies. Introductory Course	44.00.00 Education and Pedagogical Sciences	
Graphic Visualisation of Research and Projects	44.00.00 Education and Pedagogical Sciences	
Continuous Education: Modern Issues (previous title: Modern Issues of Continuous Education)	44.00.00 Education and Pedagogical Sciences	
Optical Science and Visual Arts New Era and Modernity	44.00.00 Education and Pedagogical Sciences	
Optical Science and Visual Arts From Ancient Times to the 17th century Part 1.	44.00.00 Education and Pedagogical Sciences	
Optical Science and Visual Arts From Ancient Times to the 17th century Part 2.	44.00.00 Education and Pedagogical Sciences	
Pedagogy and Psychology in Higher Education	44.00.00 Education and Pedagogical Sciences	

Personalised Education in the Context of Digital			
Transformation of Society	44.00.00 Education and Pedagogical Sciences		
Preparing an Expert Opinion	44.00.00 Education and Pedagogical Sciences		
Psycholinguistics	44.00.00 Education and Pedagogical Sciences		
Psychology	44.00.00 Education and Pedagogical Sciences		
Psychology of Professional Health: from Theory to Practice (including any course of the online programme)	44.00.00 Education and Pedagogical Sciences		
Psychology of Professional Health: from Theory to Practice. Part 1.	44.00.00 Education and Pedagogical Sciences		
Psychology of Professional Health: from Theory to Practice. Part 2.	44.00.00 Education and Pedagogical Sciences		
Autistic Spectrum Disorders	44.00.00 Education and Pedagogical Sciences		
Cutting-Edge Pedagogics: How It Works (previous title: Cutting-Edge Pedagogics. How it Works)	44.00.00 Education and Pedagogical Sciences		
Modern educational technologies	44.00.00 Education and Pedagogical Sciences		
Creating Science, Technology, and Creativity Studio at School	44.00.00 Education and Pedagogical Sciences		
Success for Everyone. Psychology of Inclusive Education	44.00.00 Education and Pedagogical Sciences		
Study the Evident! Visualising in Teaching and Learning	44.00.00 Education and Pedagogical Sciences		
I want to become a proctor!	44.00.00 Education and Pedagogical Sciences		
Efficient Teaching and Learning Technologies in High School	44.00.00 Education and Pedagogical Sciences		
SOCIA	SOCIAL SCIENCES		
Brand management in Digital Economy	38.00.00 Economics and Management		
Conflict and Stress Management	38.00.00 Economics and Management		
Constitutional Reforms in Russia	40.00.00 Law		
Contemporary Finance (including any course of the online	20 00 00 F		
programme)	38.00.00 Economics and Management		
Creating Talent Leadership in Different Contexts	38.00.00 Economics and Management		
Doing business in Russia	38.00.00 Economics and Management		

Emerging Multinationals and Global Strategy	38.00.00 Economics and Management
ESG Finance	38.00.00 Economics and Management
ESG-Transformation of Companies under Today's Challenges (including any course of the online programme)	38.00.00 Economics and Management
ESG Transformation of Companies: Strategy and Tactics	38.00.00 Economics and Management
Financial Accounting	38.00.00 Economics and Management
Global Talent Management (including any course of the online programme)	38.00.00 Economics and Management
Introduction to the Topic of Finance Essentials	38.00.00 Economics and Management
Investments and Valuation	38.00.00 Economics and Management
LegalTech (including any course of the online programme)	38.00.00 Economics and Management 40.00.00 Law
Machine Learning Based Econometrics	38.00.00 Economics and Management
Managing Diverse Talent	38.00.00 Economics and Management
Performance Management and Decision-Making	38.00.00 Economics and Management
Political Governance and Public Policy in Russia	41.00.00 Political Sciences and Regional Studies
Russia and Nuclear Arms Control	41.00.00 Political Sciences and Regional Studies
Talent Management in Global Organisation	38.00.00 Economics and Management
Automation in Legal Activities: Analysis of Practices	38.00.00 Economics and Management 40.00.00 Law
Current Issues in Civil Law	40.00.00 Law
Analysis of Financial Statement	38.00.00 Economics and Management
Business Planning and Marketing: Fundamentals (previous title: Fundamentals of Business Planning and Marketing)	38.00.00 Economics and Management
Block-Chain in Public and Municipal Administration	38.00.00 Economics and Management
Block-chain and Finances: Concepts, Technologies, and Control	38.00.00 Economics and Management
Big Data in Law: Introduction to Subject Matter	40.00.00 Law

Accounting and Audit Advanced course	38.00.00 Economics and Management
Accounting and Audit: Fundamentals (previous title: Fundamentals of Accounting and Auditing)	38.00.00 Economics and Management
Introduction to LegalTech	38.00.00 Economics and Management 40.00.00 Law
Introduction to Gender Sociology	39.00.00 Sociology and Social Work
Data Visualisation	38.00.00 Economics and Management
Introduction to R and basic statistics	38.00.00 Economics and Management
Visual Perception and Visual Deception (in any language)	37.00.00 Psychological Sciences (degree programmes 'Psychology of Physical Culture and Sports', 'Psychology of Education', 'Social Psychology and Political Psychology')
Computer Games Industry: Key Legal Issues (in any language)	40.00.00 Law
Tools for Creating LegalTech-Solutions	38.00.00 Economics and Management 40.00.00 Law
Tools of Positive Psychology in Management Course for Managers	37.00.00 Psychological Sciences (degree programmes 'Psychology of Physical Culture and Sports', 'Psychology of Education', 'Social Psychology and Political Psychology') 38.00.00 Economics and Management 39.00.00 Sociology and Social Work
Information Security for Everyone	40.00.00 Law
Information Resources in Economic and Managerial Research	38.00.00 Economics and Management
The Art of Doing Business (including any course of the online	-
programme)	38.00.00 Economics and Management
The Art of Tax Planning	38.00.00 Economics and Management
The Art of Sales	38.00.00 Economics and Management

How to stand out in a replete market	38.00.00 Economics and Management
Remote Teams: Culture and Management	38.00.00 Economics and Management
Commercialisation of the Results of Innovative Activities	38.00.00 Economics and Management
	37.00.00 Psychological Sciences
Conflictology	38.00.00 Economics and Management
Connectology	39.00.00 Sociology and Social Work
	41.00.00 Political Sciences and Regional Studies
Creative Thinking and Business	38.00.00 Economics and Management
Criminology (in any language)	40.00.00 Law
Logistics and Procurement Chain Management: Today's	20 00 00 Feenemies and Management
Challenges	38.00.00 Economics and Management
Macroeconomics	38.00.00 Economics and Management
Marketing in VUCA-world	38.00.00 Economics and Management
Mathematical Methods in Psychology: Basic Applications	37.00.00 Psychological Sciences
International Labour Law (in any language)	40.00.00 Law
Methodological Issues in Psychology: Demarcation Criteria	37.00.00 Psychological Sciences (except degree programme
Distinguishing Scientific Knowledge from Non-Scientific	'Developmental Psychology')
Methodological Issues in Psychology: Paradoxical Path of	37.00.00 Psychological Sciences (except degree programme
Science	'Developmental Psychology')
Microeconomics	29 00 00 Feed amine and Management
Microeconomics	38.00.00 Economics and Management
Multilateral International Structures in Social and Political	38.00.00 Economics and Management 40.00.00 Law
Processes of Globalising World	40.00.00 Law 41.00.00 Political Sciences and Regional Studies
Municipal Law	40.00.00 Law
Tax Law	40.00.00 Law
Fundamentals of Business	38.00.00 Economics and Management
Fundamentals of Information Security	40.00.00 Law

Foundations of Constitution and Constitutional Law	40.00.00 Law
Fundamentals of Taxation Businesses in Russia	38.00.00 Economics and Management
Fundamentals of Law	40.00.00 Law
Fundamentals of Project Activities	38.00.00 Economics and Management
Foundations of Innovation and Intellectual Property	40.00.00 7
Management	40.00.00 Law
Fundamentals of Financial Literacy	38.00.00 Economics and Management
Fundamentals of Financial Culture	38.00.00 Economics and Management
	38.00.00 Economics and Management (38.04.01 Economics, 38.04.03
Fundamentals of Economics	Personnel Management, 38.04.05 Business Informatics, 38.04.08
	Finance and Credit
Fundamentals of Effective Business Communication	38.00.00 Economics and Management
Features of Doing Business in Arctic Zone of the Russian	38.00.00 Economics and Management
Federation	38.00.00 Economics and Management
Valuation of Businesses and Investment Projects	38.00.00 Economics and Management
Valuation and Managment of Company Value (including any	20 00 00 F
course of the online programme)	38.00.00 Economics and Management
Consumer Behaviour	38.00.00 Economics and Management
Build Your Career!	38.00.00 Economics and Management
Intellectual Property Rights in Digital Era (in any language)	40.00.00 Law
Legal Regulation of Natural Resource Management: Modern	
Concepts (previous title: Modern Legal Regulation of Natural	40.00.00 Law
Resource Management)	
I 1 C	38.00.00 Economics and Management
Legal Support of Business in Russia (in any language)	40.00.00 Law
Legal Regulation of Corporate Income Taxation of the Russian Federation	40.00.00 Law

Legal Regulation of Internet Relations: the Russian Perspective (previous title: Legal Regulation of Internet Relations. The Russia's Perspective	40.00.00 Law
Legal regulation of Legal Relations in Application of Distributed Ledger Technology	40.00.00 Law
Legal Regulation of Financial Investments and Services	38.00.00 Economics and Management 40.00.00 Law
Economic Data Analysis Practices: from Simple to Complicated (any course of the online programme)	38.00.00 Economics and Management
Operational Analysis Practice in MS Excel	38.00.00 Economics and Management
Digital Transformation Practices (in any language)	38.00.00 Economics and Management
Customs Checks	40.00.00 Law
Project Development (including any course of the online programme)	38.00.00 Economics and Management
Breakthrough Digital Technologies: Cases of Companies	38.00.00 Economics and Management
Countering Financial Fraud and Managing Individual Risks	38.00.00 Economics and Management
Psychological and Psychiatric Aspects of Traumatic Stress and Its Consequences	37.00.00 Psychological Sciences
Psychology	37.00.00 Psychological Sciences (except degree programmes 'Children's Mental Health and Early Intervention', 'Organisational Psychology and Psychology of Management')
Career Psychology	37.00.00 Psychological Sciences (degree programmes 'Psychology of Physical Culture and Sports', 'Psychology of Education', 'Social Psychology and Political Psychology') 38.00.00 Economics and Management 39.00.00 Sociology and Social Work

	37.00.00 Psychological Sciences (except degree programmes
Migration Psychology	'Children's Mental Health and Early Intervention', 'Organisational
	'Organisational Psychology and Psychology of
	Management'
Psychology of Professional Health: from Theory to Practice	37.00.00 Psychological Sciences
(including any course of the online programme)	39.00.00 Sociology and Social Work
Psychology of Professional Health: from Theory to Practice.	37.00.00 Psychological Sciences
Part 1.	39.00.00 Sociology and Social Work
Psychology of Professional Health: from Theory to Practice.	37.00.00 Psychological Sciences
Part 2.	39.00.00 Sociology and Social Work
Developmental Psychology (including any course of the online programme)	37.00.00 Psychological Sciences
Developmental Psychology: Adulthood and Old Age	37.00.00 Psychological Sciences
Developmental Psychology: Childhood and Adolescence	37.00.00 Psychological Sciences
Psychology of Consciousness	37.00.00 Psychological Sciences (degree programme 'Psychology of Education', Psychology of Physical Culture and Sports', 'Organisational Psychology and Psychology of Management', 'Personality Psychology')
Stress Psychology Introduction	37.00.00 Psychological Sciences
	37.00.00 Psychological Sciences (degree programmes 'Psychology of
Psychotechnologies for Working with Human Resources	Physical Culture and Sports', 'Psychology of Education', 'Social
	Psychology and Political Psychology')
	38.00.00 Economics and Management
	39.00.00 Sociology and Social Work
Psychotraumatic Events in a Person's Life: Signs of Mental Trauma, Psychological Support	37.00.00 Psychological Sciences
US Public Diplomacy	41.00.00 Political Sciences and Regional Studies

Working with Data in Power Query and Power Pivot	38.00.00 Economics and Management
Developing Organisation's Developmental Strategy	38.00.00 Economics and Management
Virtual Teams: Theories and Practices	38.00.00 Economics and Management
Autistic Spectrum Disorders	37.00.00 Psychological Sciences (degree programme 'Educational Psychology') 'Social Psychology and Political Psychology', General and Cognitive Psychology', 'Psychology of Physical Culture and Sports')
Segment Analysis	38.00.00 Economics and Management
Easement as an Encumbrance on the Right to Real Estate	40.00.00 Law
Modern International Relations Theories	41.00.00 Political Sciences and Regional Studies
Modern Technologies in Insurance	38.00.00 Economics and Management
Modern Financial Technologies	38.00.00 Economics and Management
Modern Management	38.00.00 Economics and Management
Creating Analytical Panels in Microsoft Power BI	38.00.00 Economics and Management
Project Development and Implementation	38.00.00 Economics and Management
Sociology	39.00.00 Sociology and Social Work 38.00.00 Economics and Management (38.04.02 Management, 38.04.04 Public and Municipal Administration)
Sociology of Labour	39.00.00 Sociology and Social Work
Start-up and Open Innovations	38.00.00 Economics and Management
Start-up: Growing into Successful Business (in any language)	38.00.00 Economics and Management
Product Sales Strategy in the Market of Government Contracts	38.00.00 Economics and Management
Insurance Business	38.00.00 Economics and Management
Stress and Family Life	37.00.00 Psychological Sciences
Stress and Workplace	37.00.00 Psychological Sciences
Stress, Crisis, Extreme Situation	37.00.00 Psychological Sciences
Theory of Law and State	40.00.00 Law

Distributed Ledger Technology: Legal Regulation	38.00.00 Economics and Management
Criminal Procedure	40.00.00 Law
Brand Management in the Digital Economy (in any language)	38.00.00 Economics and Management
Conflict and Stress Management in an Organisation (in any language)	38.00.00 Economics and Management
	37.00.00 Psychological Sciences (degree programmes 'Psychology of
	Physical Culture and Sports', 'Psychology of Education', 'Social
Managing People and Teams	Psychology and Political Psychology')
	38.00.00 Economics and Management
	39.00.00 Sociology and Social Work
Retail Sales Management	38.00.00 Economics and Management
Company Value Management	38.00.00 Economics and Management
Digital Business Transformation Management	38.00.00 Economics and Management
Managerial Analysis in MS Excel (previous title: Managerial	
Analysis Practices in MS Excel)	38.00.00 Economics and Management
	37.00.00 Psychological Sciences
Success for Everyone. Psychology of Inclusive Education	39.00.00 Sociology and Social Work
Financial Literacy	38.00.00 Economics and Management
Financial Literacy (including any course of the online	38.00.00 Economics and Management
programme)	38.00.00 Economics and Wanagement
Developing ESG-Principles of Companies as a Response to	38.00.00 Economics and Management
Global Challenges	38.00.00 Economics and Management
Formation of Customs Value	40.00.00 Law
On-Demand Supply Chain	38.00.00 Economics and Management
Circular Economics and Climate Agenda	38.00.00 Economics and Management
Digital Transformation of Business	38.00.00 Economics and Management

Digitalisation of International Relations (in any language)	41.00.00 Political Sciences and Regional Studies
Digitalisation and Human Resource Management	38.00.00 Economics and Management
Digital Marketing for Electronic Commerce	38.00.00 Economics and Management
Digital Currencies and Block Chain Technology	38.00.00 Economics and Management
Econometrics: Time Series Analysis	38.00.00 Economics and Management
Corporate Economics	38.00.00 Economics and Management
	37.00.00 Psychological Sciences (degree programmes 'Psychology of
	Physical Culture and Sports', 'Psychology of Education', 'Social
Efficient Leader (including any course of the online programme)	Psychology and Political Psychology')
	38.00.00 Economics and Management
	39.00.00 Sociology and Social Work
	37.00.00 Psychological Sciences
Legal Psychology	40.00.00 Law
Legal English	40.00.00 Law
Language of Efficient Communication	38.00.00 Economics and Management
Language of Efficient Communication in Professional Setting	38.00.00 Economics and Management
Language of Efficient Communication in Digital Media	38.00.00 Economics and Management
HUMANITIES	
Arabic for Beginners (including any course of the online programme)	45.00.00 Linguistics and Literary Studies
Arabic for Beginners. Part 1	45.00.00 Linguistics and Literary Studies
Arabic for Beginners. Part 2	45.00.00 Linguistics and Literary Studies
Arabic for Beginners. Part 3	45.00.00 Linguistics and Literary Studies
Arabic for Beginners. Part 4	45.00.00 Linguistics and Literary Studies
Arabic for Beginners. Part 5	45.00.00 Linguistics and Literary Studies

Business Russian Communication (including any course of the online programme)	45.00.00 Linguistics and Literary Studies
Business Russian Communication Part 1	45.00.00 Linguistics and Literary Studies
Business Russian Communication Part 2	45.00.00 Linguistics and Literary Studies
Business Russian Communication Part 3	45.00.00 Linguistics and Literary Studies
Business Russian Communication Part 4	45.00.00 Linguistics and Literary Studies
Chinese in 5 Steps (including any course of the onlline programme)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies
Fundamentals of the Chinese Character Writing (including any course of the online programme)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies
Fundamentals of the Chinese Character Writing. Part 1	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies
Fundamentals of the Chinese Character Writing. Part 2	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies
History of Russia (including any course of the online programme)	46.00.00 History and Archaeology
History of Russia: from Ancient Rus to the Time of Troubles	46.00.00 History and Archaeology
History of Russia: from NEP to Modern Times	46.00.00 History and Archaeology
History of Russia: From Peter the Great to the Revolution	46.00.00 History and Archaeology
Korean for Beginners (including any course of the online programme)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies
Korean for Beginners. Part 1	45.00.00 Linguistics and Literary Studies

	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
Korean for Beginners. Part 2	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
Korean for Beginners. Part 3	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
Korean for Beginners. Part 4	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
The Gems of Middle Eastern Literature	45.00.00 Linguistics and Literary Studies
The Magnificent Century: History and Culture of the Ottoman Empire	46.00.00 History and Archaeology
Pussion Academia Writing	45.00.00 Linguistics and Literary Studies
Russian Academic Writing	51.00.00 Cultural Studies and Sociocultural Projects
English fro Beginners (including any course of the online programme)	45.00.00 Linguistics and Literary Studies
English for Beginners Part 1.	45.00.00 Linguistics and Literary Studies
English for Beginners Part 2.	45.00.00 Linguistics and Literary Studies
English for Beginners Part 3.	45.00.00 Linguistics and Literary Studies
English for Beginners Part 4.	45.00.00 Linguistics and Literary Studies
English in Hospitality Industry	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Arabic Introductory Course	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies

	45 00 00 I in oviation and I itemany Chydica
Arabic Part 2.	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
A	
Argumentation and Critical Thinking	47.00.00 Philosophy, Ethics, and Religious Studies
Introduction to Literary Studies	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Introduction to Theology: History and Methodology	47.00.00 Philosophy, Ethics, and Religious Studies
Introduction to Translation Theory	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Hungarian for Beginners (including any course of the online	47.00.0074
programme)	45.00.00 Linguistics and Literary Studies
Hungarian for Beginners Part 1.	45.00.00 Lineariesia and Lisane Condina
	45.00.00 Linguistics and Literary Studies
Hungarian for Beginners Part 2.	45.00.00 Linguistics and Literary Studies
Interaction of science with arts and technology (including any	45.00.00.00.00.00
course of the online programme)	47.00.00 Philosophy, Ethics, and Religious Studies
Interaction of Science with Arts and Technologies. Introductory	
Course	47.00.00 Philosophy, Ethics, and Religious Studies
General History Part 1.	46.00.00 History and Archaeology
General History Part 2.	46.00.00 History and Archaeology
Official Language in Russia and Abroad	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Business English (including any course of the online	
programme)	45.00.00 Linguistics and Literary Studies
Business English Part 1.	45.00.00 Linguistics and Literary Studies
Business English Part 2.	45.00.00 Linguistics and Literary Studies
Business English Part 3.	45.00.00 Linguistics and Literary Studies
Business English Part 4.	45.00.00 Linguistics and Literary Studies

Current Processes in Spoken Russian	42.00.00 Mass Media and Information Librarianship
	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Hagiography and Other Sources on the History of the Russian	46.00.00 III 4 1 1
Orthodox Church	46.00.00 History and Archaeology
Western Literature: the 20th Century	45.00.00 Linguistics and Literary Studies
Visual Perception and Visual Deception (in any language)	50.00.00 Art History (degree programme 'Cognitive Studies')
Islam: History, Culture and Practice	46.00.00 History and Archaeology;4
	47.00.00 Philosophy, Ethics, and Religious Studies
	58.00.00 Asian and African Studies
Spanish for Beginners (including any course of the online	45.00.00 Linguistics and Literary Studies
programme)	51.00.00 Cultural Studies and Sociocultural Projects
Spanish for Beginners Part 1.	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
Spanish for Beginners Part 2.	45.00.00 Linguistics and Literary Studies
Spanish for Beginners Fare 2.	51.00.00 Cultural Studies and Sociocultural Projects
Spanish for Beginners Part 3.	45.00.00 Linguistics and Literary Studies
Spanion for Deginion furt 3.	51.00.00 Cultural Studies and Sociocultural Projects
Spanish for Beginners Part 4.	45.00.00 Linguistics and Literary Studies
	51.00.00 Cultural Studies and Sociocultural Projects
History of English Literature	45.04.01 Philology
History and Culture of St Petersburg and Leningrad	46.00.00 History and Archaeology
Region	40.00.00 Thistory and Archaeology
The History of Russia	46.00.00 History and Archaeology
History of Russian Literature: the Second Part of the 19th Century	45.00.00 Linguistics and Literary Studies

History of Russian Literature: Golden Age	45.00.00 Linguistics and Literary Studies	
	42.00.00 Mass Media and Information Librarianship	
History of Russian Written Texts	45.00.00 Linguistics and Literary Studies	
	51.00.00 Cultural Studies and Sociocultural Projects	
History of Soviet Literature	45.00.00 Linguistics and Literary Studies	
•	51.00.00 Cultural Studies and Sociocultural Projects	
History of Japan Antiquity and Middle Ages	46.00.00 History and Archaeology	
History of Japan New and Contemporary Times	46.00.00 History and Archaeology	
	45.00.00 Linguistics and Literary Studies	
Chinese Painting: Introduction	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
	45.00.00 Linguistics and Literary Studies	
Chinese Painting: Master Class	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
Chinese Hieroglyphics: Basics (previous title: Introduction to	45.00.00 Linguistics and Literary Studies	
Chinese Hieroglyphics)	51.00.00 Cultural Studies and Sociocultural Projects	
Cimiese Therogrypmes)	58.00.00 Asian and African Studies	
	45.00.00 Linguistics and Literary Studies	
Chinese for Beginners: Phonetic Aspects (in any language)	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
Chinese: 5 Steps to Success (including any course of the online	45.00.00 Linguistics and Literary Studies	
programme)	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
	45.00.00 Linguistics and Literary Studies	
Chinese: 5 Steps to Success Part 1 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
Chinese: 5 Steps to Success Part 2 (in any language)	45.00.00 Linguistics and Literary Studies	
\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	51.00.00 Cultural Studies and Sociocultural Projects	

	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Chinese: 5 Steps to Success Part 3 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Chinese: 5 Steps to Success Part 4 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Chinese: 5 Steps to Success Part 5 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
Korean for Beginners (including any course of the online	45.00.00 Linguistics and Literary Studies
programme)	51.00.00 Cultural Studies and Sociocultural Projects
programme)	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Korean for Beginners Part 1 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Korean for Beginners Part 2 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Korean for Beginners Part 3 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
	45.00.00 Linguistics and Literary Studies
Korean for Beginners Part 4 (in any language)	51.00.00 Cultural Studies and Sociocultural Projects
	58.00.00 Asian and African Studies
Cultural Studies	51.00.00 Cultural Studies and Sociocultural Projects
Latin for beginners	45.00.00 Linguistics and Literary Studies
-	51.00.00 Cultural Studies and Sociocultural Projects
English Lexicology	45.00.00 Linguistics and Literary Studies

Linguistic Expert Evaluation: Objectives and Techniques (previous title Objectives and Techniques of Linguistic Expert Evaluation)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects	
Russian Émigré Environment	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects	
The Germs of Middle East Literature	45.00.00 Linguistics and Literary Studies	
Logical Aspects of Cognitive Processes	47.00.00 Philosophy, Ethics, and Religious Studies	
Medical English (including any course of the online programme)	45.00.00 Linguistics and Literary Studies	
Medical English Part 1.	45.00.00 Linguistics and Literary Studies	
Medical English Part 2.	45.00.00 Linguistics and Literary Studies	
Medical English Part 3.	45.00.00 Linguistics and Literary Studies	
Medical English Part 4.	45.00.00 Linguistics and Literary Studies	
Medical English Part 5.	45.00.00 Linguistics and Literary Studies	
Intercultural Communication and Translations	45.00.00 Linguistics and Literary Studies	
Methods of Teaching Russian as a Foreign language (TORFL)	45.04.01 Philology	
Muslim Art (in any language)	45.04.01 Philology 58.00.00 Asian and African Studies	
Neurolinguistics (in any language)	50.00.00 Art History (degree programme 'Cognitive Studies')	
Optical Science and Visual Arts New Era and Modernity	47.00.00 Philosophy, Ethics, and Religious Studies	
Optical Science and Visual Arts From Ancient Times to the 17th century Part 1.	47.00.00 Philosophy, Ethics, and Religious Studies	
Optical Science and Visual Arts From Ancient Times to the 17th century Part 2.	47.00.00 Philosophy, Ethics, and Religious Studies	
Introduction to Religious Studies	47.00.00 Philosophy, Ethics, and Religious Studies	

	45.00.00 Linguistics and Literary Studies	
Persian: Conversational Course Part 1.	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
	45.00.00 Linguistics and Literary Studies	
Persian: Conversational Course Part 2.	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
Peter the Great: Myths and Reality	46.00.00 History and Archaeology	
Web Site Promotion	42.00.00 Mass Media and Information Librarianship	
Statistics Made Simple (Using R) (including any course of the online programme)	50.00.00 Art History (degree programme 'Cognitive Studies')	
Psycholinguistics	50.00.00 Art History (degree programme 'Cognitive Studies')	
Psychology of Consciousness	50.00.00 Art History (degree programme 'Cognitive Studies')	
Russian Literature: 18th Century: Routes of Development and	45.00.00 Linguistics and Literary Studies	
Cultural and Historical Context	51.00.00 Cultural Studies and Sociocultural Projects	
Contemporary Russian Literature (in any language)	45.00.00 Linguistics and Literary Studies	
Russian Stylistics	45.00.00 Linguistics and Literary Studies	
•	51.00.00 Cultural Studies and Sociocultural Projects	
Cultural Linguistics of Great Britain and the USA	45.04.01 Philology	
Telecommunication Computer Technologies in Advertising and Public Relations	42.00.00 Mass Media and Information Librarianship	
Theoretical Issues in English Language Grammar	51.00.00 Cultural Studies and Sociocultural Projects	
Traditional Spiritual and Moral Values of Russia: Axiology of Culture	47.00.00 Philosophy, Ethics, and Religious Studies	
Traditional Religions at Turning Points in Russian History	47.00.00 Philosophy, Ethics, and Religious Studies	
	45.00.00 Linguistics and Literary Studies	
Turkish for Beginners: Conversational Course	51.00.00 Cultural Studies and Sociocultural Projects	
	58.00.00 Asian and African Studies	
Philosophy (in any language)	47.00.00 Philosophy, Ethics, and Religious Studies	

Legal English	45.00.00 Linguistics and Literary Studies	
I am a Translator (including any course of the online programme)	45.00.00 Linguistics and Literary Studies	
Language of Modern Opinion Journalism	42.00.00 Mass Media and Information Librarianship 45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects	
Language of Modern Advertising and Mass Media	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 42.00.00 Mass Media and Information Librarianship	
Japanese for Beginners (including any course of the online programme)	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies	
Japanese for Beginners Part 1.	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies	
Japanese for Beginners Part 2.	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies	
Japanese for Beginners Part 3.	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies	
Japanese for Beginners Part 4.	45.00.00 Linguistics and Literary Studies 51.00.00 Cultural Studies and Sociocultural Projects 58.00.00 Asian and African Studies	
ART AND CULTURE		
Adobe Illustrator for Beginners (in any language)	54.00.00 Fine and Applied Arts	
Interface Design: interactive systems interface design (including any course of the online programme)	54.00.00 Fine and Applied Arts	

UX/UI-Design: Project Development, Management and Prototyping (previous title: Project Development, Management and Prototyping (UX/UI-Design))	54.00.00 Fine and Applied Arts
UX/UI-Design: Usability Testing of Interactive Systems (previous title: Usability Testing of Interactive Systems in UX/UI-Design)	54.00.00 Fine and Applied Arts
UX-Strategy	54.00.00 Fine and Applied Arts
Introduction to UX/UI-Design	54.00.00 Fine and Applied Arts
Graphic Design Fundamentals of Project Development	54.00.00 Fine and Applied Arts
Design Methodology: Inspiration Management	54.00.00 Fine and Applied Arts
Pastel Painting. Stage-Based Practising 'Still Life' and "Landscape' Assignments	54.00.00 Fine and Applied Arts
Creative Communication: Methods and Tools	54.00.00 Fine and Applied Arts
Creative Painting Techniques	54.00.00 Fine and Applied Arts
Polychrome Painting	54.00.00 Fine and Applied Arts
Portfolio Design in Figma and Tilda	54.00.00 Fine and Applied Arts
Portfolio Design in InDesign	54.00.00 Fine and Applied Arts
Developing Design Portfolio (including any course of the online programme)	54.00.00 Fine and Applied Arts
Artistic Web Design	54.00.00 Fine and Applied Arts
Six Steps to Portfolio	54.00.00 Fine and Applied Arts

Annex 4

to the Regulations for Admission to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

Correspondence of the profiles of the All-Russia School Student Olympiad, republican Olympiads of the Donetsk People's Republic, the Lugansk People's Republic, international Olympiads in general education subjects to general-education-based admissions tests

Profile of the All-Russia School Student Olympiad, republican Olympiads of the Donetsk People's Republic, the Lugansk People's Republic, international Olympiads in general education subjects	Subject of admissions tests
English	English
Astronomy	Physics
Biology	Biology
Geography	Geography
Informatics, Informatics and ICT	Informatics and ICT
Art (World Art)	History
Spanish	Spanish
History	History
Italian	_
Chinese	Chinese
Literature	Literature
Mathematics	Mathematics
German	German
Social Science	Social Science
Life Safety	_
Law	Social Science
Russian	Russian
Technology	_
Physics	Physics
Physical Education	Physical Education
French	French
Chemistry	Chemistry
Ecology	Biology, Geography
Economics	Social Science

Annex 5 to the Regulations for Admission to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

## Discipline correspondence (examination) of the Centralised Test to the general education-based admissions tests

Disciplines of the Centralised Test (examination)	Disciplines of the general education- based admissions test
English	English
Belorusian	_
General History	_
Geography	Geography
Spanish	Spanish
History of Belarus	_
Chinese	Chinese
Mathematics	Mathematics
German	German
Social Studies	Social Science
Russian	Russian
Physics	Physics
French	French
Chemistry	Chemistry

Annex 6
to the Regulations for Admission
to Higher Education Degree Programmes
(Bachelor's Programmes, Specialist's Programmes,
Master's Programmes)
at St Petersburg University in 2024

## Correspondence of the All-Russian School Olympiad profiles, international Olympiads in general education subjects to the fields of study and specialities to which St Petersburg University conducts the admission within degree programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
English	37.03.02 Conflictology	Bachelor's
		degree
		programmes
English	39.03.01 Sociology	Bachelor's
		degree
		programmes
English	39.03.02 Social Work	Bachelor's
		degree
		programmes
English	40.03.01 Law	Bachelor's
		degree
		programmes
English	41.03.05 International Relations	Bachelor's
		degree
		programmes
English	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
English	42.03.02 Journalism	Bachelor's
		degree
		programmes
English	43.03.02 Tourism	Bachelor's
		degree
		programmes
English	45.03.01 Philology	Bachelor's
		degree
		programmes
English	45.03.02 Linguistics (degree programmes 'English as a Foreign	Bachelor's
	Language in Teaching and Communication (with second	degree
	qualification 'Teacher of English'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programmes Balkan Studies and	Bachelor's
<del>-</del>	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'Foreign Languages in	Bachelor's
-	Global Communication (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'General and Applied	Bachelor's
-	Phonetics')	degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
English	45.03.02 Linguistics (degree programme 'Applied, Computer and Computational Linguistics (English)')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Russian as a Foreign Language ( (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Theoretical and Experimental Linguistics (English) (with second qualification 'Teacher of a Foreign Language')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Theory and Practice of Intercultural Communication (English) (with second qualification 'Teacher of a Foreign Language'/'Specialist in Translation')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Languages of the Slavic World and Russian as a Foreign Language (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
English	45.05.01 Translation and Translation Studies	Specialist's degree programme
English	47.03.01 Philosophy	Bachelor's degree programmes
English	47.03.03 Religious Studies	Bachelor's degree programmes
English	50.03.01 Arts and Humanities	Bachelor's degree programmes
English	50.03.03 History of Arts	Bachelor's degree programmes
English	51.03.01 Cultural Studies	Bachelor's degree programmes
English	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
English	58.03.01 Asian and African Studies	Bachelor's degree programmes
Astronomy	01.03.01 Mathematics	Bachelor's degree programmes
Astronomy	01.03.03 Applied Mathematics and Informatics (degree programme 'Modern Programming')	Bachelor's degree programmes
Astronomy	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	Bachelor's degree programmes
Astronomy	03.03.01 Applied Mathematics and Physics	Bachelor's degree programmes
Astronomy	03.03.02 Physics	Bachelor's degree programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Astronomy	03.05.01 Astronomy	Specialist's
		degree
		programme
Biology	05.03.02 Geography	Bachelor's
		degree
		programmes
Biology	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Biology	05.03.04 Hydrometeorology	Bachelor's
		degree
		programmes
Biology	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Biology	06.03.01 Biology	Bachelor's
		degree
		programmes
Biology	06.03.02 Soil Science	Bachelor's
		degree
		programmes
Biology	31.05.01 General Medicine	Specialist's
		degree
		programme
Biology	31.05.03 Dental Medicine	Specialist's
		degree
		programme
Biology	33 05 01 Pharmacy	Specialist's
		degree
		programme
Biology	34.02.01 Nursing	Bachelor's
		degree
		programmes
Biology	37.03.01 Psychology	Bachelor's
		degree
D: 1	27.07.04.011.1.1.1.1	programmes
Biology	37.05.01 Clinical Psychology	Specialist's
		degree
Diala and	27 05 02 Eurolannant Parahala	programme
Biology	37.05.02 Employment Psychology	Specialist's
		degree
Cacamanhy	05 02 01 Coology	programme
Geography	05.03.01 Geology	Bachelor's
		degree
Goography	05 03 02 Goography	programmes Bachelor's
Geography	05.03.02 Geography	degree
		•
Goography	05.03.03 Cartography and Geoinformatics	programmes Bachelor's
Geography	05.05.05 Cartography and Geomformatics	
		degree
Geography	05.03.04 Hydrometeorology	programmes Bachelor's
Geography	05.05.04 Hydrollieteorology	degree
		•
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Geography	05.03.06 Ecology and Nature Management	Bachelor's degree
		_
C1	21 02 01 Petrolous Engineering	programmes Bachelor's
Geography	21.03.01 Petroleum Engineering	
		degree
<u> </u>	21.02.021	programmes
Geography	21.03.02 Land Management and Cadastres	Bachelor's
		degree
<u> </u>	07.00.00 G . G! !	programmes
Geography	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Geography	38.03.01 Economics	Bachelor's
		degree
		programmes
Geography	41.03.02 Russian Region Studies	Bachelor's
		degree
		programmes
Geography	41.03.05 International Relations	Bachelor's
		degree
		programmes
Geography	43.03.02 Tourism	Bachelor's
		degree
		programmes
Geography	46.03.01 History	Bachelor's
0 1 0		degree
		programmes
Geography	47.03.01 Philosophy	Bachelor's
C 1 7		degree
		programmes
Geography	47.03.03 Religious Studies	Bachelor's
0 1 0		degree
		programmes
Geography	51.03.01 Cultural Studies	Bachelor's
C 1 7		degree
		programmes
Geography	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
& 1 J	Heritage Sites	degree
		programmes
Informatics	01.03.01 Mathematics	Bachelor's
	V - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	degree
		programmes
Informatics	01.03.02 Applied Mathematics and Informatics	Bachelor's
		degree
		programmes
Informatics	02.03.01 Mathematics and Computer Sciences (programme	Bachelor's
	'Data Sciences')	degree
	Jum Golohood )	programmes
Informatics	02.03.02 Fundamental Informatics and Information Technology	Bachelor's
mormanes	02.03.02 Fundamental informatics and information reciliology	degree
		_
Informatics	02 02 02 Information Systems Administration and Mathematical	programmes  Pachalor's
mormatics	02.03.03 Information Systems Administration and Mathematical	Bachelor's
	Support	degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Informatics	05.03.01 Geology	Bachelor's
		degree
		programmes
Informatics	05.03.02 Geography	Bachelor's
		degree
		programmes
Informatics	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Informatics	05.03.04 Hydrometeorology	Bachelor's
		degree
		programmes
Informatics	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Informatics	06.03.02 Soil Science	Bachelor's
		degree
		programmes
Informatics	09.03.03 Applied Informatics	Bachelor's
		degree
		programmes
Informatics	09.03.04 Software Engineering	Bachelor's
		degree
		programmes
Informatics	21.03.01 Petroleum Engineering	Bachelor's
		degree
		programmes
Informatics	21.03.02 Land Management and Cadastres	Bachelor's
		degree
		programmes
Informatics	27.03.03 Systems Analysis and Control	Bachelor's
		degree
		programmes
Informatics	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Informatics	38.03.01 Economics	Bachelor's
		degree
		programmes
Informatics	38.03.03 Personnel Management	Bachelor's
		degree
T. C	20.02.07 D	programmes
Informatics	38.03.05 Business Informatics	Bachelor's
		degree
T. C	20.02.01.0	programmes
Informatics	39.03.01 Sociology	Bachelor's
		degree
T. C:	47.02.01 PL'1	programmes
Informatics	47.03.01 Philosophy	Bachelor's
		degree
		programmes
T. C.	47.02.02 B. I'.'. G. I'.	
Informatics	47.03.03 Religious Studies	Bachelor's degree

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Informatics	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Informatics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Art (World Art)	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Art (World Art)	39.03.02 Social Work	Bachelor's
		degree
		programmes
Art (World Art)	46.03.01 History	Bachelor's
		degree
		programmes
Art (World Art)	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Art (World Art)	47.03.03 Religious Studies	Bachelor's
,		degree
		programmes
Art (World Art)	50.03.01 Arts and Humanities	Bachelor's
,		degree
		programmes
Art (World Art)	50.03.03 History of Arts	Bachelor's
(		degree
		programmes
Art (World Art)	51.03.01 Cultural Studies	Bachelor's
, , , ,		degree
		programmes
Art (World Art)	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
, , , ,	Heritage Sites	degree
	č	programmes
Spanish	37.03.02 Conflictology	Bachelor's
1		degree
		programmes
Spanish	39.03.01 Sociology	Bachelor's
•		degree
		programmes
Spanish	39.03.02 Social Work	Bachelor's
		degree
		programmes
Spanish	40.03.01 Law	Bachelor's
•		degree
		programmes
Spanish	41.03.05 International Relations	Bachelor's
=		degree
		programmes
Spanish	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Spanish	42.03.02 Journalism	Bachelor's
1		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Spanish	43.03.02 Tourism	Bachelor's
		degree
		programmes
Spanish	45.03.01 Philology	Bachelor's
		degree
		programmes
Spanish	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
Spanish	45.03.02 Linguistics (degree programme 'Spanish' (with second	Bachelor's
	qualifications 'Teacher of the Spanish'/'Specialist in	degree
	Translation')	programmes
Spanish	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
Spanish	45.03.02 Linguistics (degree programme 'Languages of the	Bachelor's
	Slavic World and Russian as a Foreign Language (with second	degree
	qualification 'Teacher of the Russian Language and Literature')	programmes
Spanish	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Spanish	47.03.03 Religious Studies	Bachelor's
•		degree
		programmes
Spanish	50.03.01 Arts and Humanities	Bachelor's
1		degree
		programmes
Spanish	50.03.03 History of Arts	Bachelor's
•		degree
		programmes
Spanish	51.03.01 Cultural Studies	Bachelor's
1		degree
		programmes
Spanish	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
1	Heritage Sites	degree
		programmes
Spanish	58.03.01 Asian and African Studies	Bachelor's
1		degree
		programmes
History	37.03.02 Conflictology	Bachelor's
•		degree
		programmes
History	39.03.01 Sociology	Bachelor's
•		degree
		programmes
History	39.03.02 Social Work	Bachelor's
•		degree
		programmes
History	40.03.01 Law	Bachelor's
•		degree
		programmes
History	41.03.02 Russian Region Studies	Bachelor's
,		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
History	41.03.04 Political Science	Bachelor's
•		degree
		programmes
History	41.03.05 International Relations	Bachelor's
•		degree
		programmes
History	42.03.01 Advertising and Public Relations	Bachelor's
•		degree
		programmes
History	43.03.02 Tourism	Bachelor's
•		degree
		programmes
History	46.03.01 History	Bachelor's
•		degree
		programmes
History	47.03.01 Philosophy	Bachelor's
1115001	····obio11 miosopmy	degree
		programmes
History	47.03.03 Religious Studies	Bachelor's
Thistory	47.03.03 Religious Studies	degree
		programmes
History	48.06.01 Theology	Bachelor's
Thistory	40.00.01 Theology	degree
		-
History	50.03.01 Arts and Humanities	programmes Bachelor's
History	50.03.01 Arts and Humanities	
		degree
III: at a sec	50.02.02 History of Auto	programmes Bachelor's
History	50.03.03 History of Arts	
		degree
TT'	51 02 01 G Iv 1 G IV	programmes
History	51.03.01 Cultural Studies	Bachelor's
		degree
***		programmes
History	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
History	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Italian	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Italian	39.03.01 Sociology	Bachelor's
		degree
		programmes
Italian	39.03.02 Social Work	Bachelor's
		degree
		programmes
Italian	40.03.01 Law	Bachelor's
		degree
		programmes
Italian	41.03.05 International Relations	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Italian	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Italian	42.03.02 Journalism	Bachelor's
		degree
		programmes
Italian	43.03.02 Tourism	Bachelor's
		degree
		programmes
Italian	45.03.01 Philology	Bachelor's
		degree
		programmes
Italian	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
Italian	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
Italian	45.03.02 Linguistics (degree programme 'Languages of the	Bachelor's
	Slavic World and Russian as a Foreign Language (with second	degree
	qualification 'Teacher of the Russian Language and Literature')	programmes
Italian	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Italian	47.03.03 Religious Studies	Bachelor's
	g	degree
		programmes
Italian	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Italian	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Italian	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Italian	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Italian	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Chinese	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Chinese	39.03.01 Sociology	Bachelor's
		degree
		programmes
Chinese	39.03.02 Social Work	Bachelor's
		degree
		programmes
Chinese	40.03.01 Law	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Chinese	41.03.05 International Relations	Bachelor's
		degree
		programmes
Chinese	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Chinese	42.03.02 Journalism	Bachelor's
		degree
		programmes
Chinese	43.03.02 Tourism	Bachelor's
		degree
		programmes
Chinese	45.03.01 Philology	Bachelor's
		degree
		programmes
Chinese	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Chinese	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Chinese	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Chinese	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Chinese	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Chinese	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Chinese	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Literature	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Literature	42.03.02 Journalism	Bachelor's
		degree
		programmes
Literature	45.03.01 Philology	Bachelor's
		degree
		programmes
Literature	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Literature	47.03.03 Religious Studies	Bachelor's
<del></del>	0-1-1-0-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1	degree
		programmes
Literature	50.03.01 Arts and Humanities	Bachelor's
	COOCOTT THE MICHIGANIA	
		degree

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Literature	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Literature	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Literature	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Mathematics	01.03.01 Mathematics	Bachelor's
		degree
		programmes
Mathematics	01.03.02Applied Mathematics and Informatics	Bachelor's
		degree
		programmes
Mathematics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
		degree
		programmes
Mathematics	01.05.01 Fundamental Mathematics and Mechanics	Specialist's
		degree
		programme
Mathematics	02.03.01 Mathematics and Computer Sciences	Bachelor's
		degree
		programmes
Mathematics	02.03.02 Fundamental Informatics and Information Technology	Bachelor's
		degree
		programmes
Mathematics	03.03.01 Applied Mathematics and Physics	Bachelor's
		degree
		programmes
Mathematics	03.03.02 Physics	Bachelor's
		degree
		programmes
Mathematics	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
		programmes
Mathematics	05.03.01 Geology	Bachelor's
		degree
		programmes
Mathematics	05.03.02 Geography	Bachelor's
		degree
		programmes
Mathematics	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Mathematics	05.03.04 Hydrometeorology	Bachelor's
		degree
		programmes
Mathematics	05.03.06 Ecology and Nature Management	Bachelor's
	-	degree
		programmes
Mathematics	06.03.02 Soil Science	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Mathematics	09.03.03 Applied Informatics (degree programme: Applied Informatics in Arts and Humanities)	Bachelor's degree
		programmes
Mathematics	21.03.01 Petroleum Engineering	Bachelor's
		degree programmes
Mathematics	21.03.02 Land Management and Cadastres	Bachelor's
wattematics	21.05.02 Land Management and Cadastres	degree
		programmes
Mathematics	27.03.03 Systems Analysis and Control	Bachelor's
		degree
		programmes
Mathematics	37.03.01 Psychology	Bachelor's
		degree
		programmes
Mathematics	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Mathematics	37.05.01 Clinical Psychology	Specialist's
	3	degree
		programme
Mathematics	37.05.02 Employment Psychology	Specialist's
1,140110111401105	e merez zimproyment i sythology	degree
		programme
Mathematics	38.03.01 Economics	Bachelor's
Mathematics	50.05.01 Leonomies	degree
		programmes
Mathematics	38.03.02 Management	Bachelor's
Mathematics	50.03.02 ividinagement	degree
		programmes
Mathematics	38.03.03 Personnel Management	Bachelor's
Mathematics	38.03.03 Fersonner Management	
		degree
Mathamatica	29 02 04 Dublic and Municipal Administration	programmes
Mathematics	38.03.04 Public and Municipal Administration	Bachelor's
		degree
Madamada	38.03.05 Business Informatics	programmes
Mathematics	38.03.05 Business Informatics	Bachelor's
		degree
37.1	20.00.04.0	programmes
Mathematics	39.03.01 Sociology	Bachelor's
		degree
		programmes
Mathematics	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Mathematics	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Mathematics	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Mathematics	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Mathematics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
German	37.03.02 Conflictology	Bachelor's
		degree
		programmes
German	39.03.01 Sociology	Bachelor's
		degree
		programmes
German	39.03.02 Social Work	Bachelor's
		degree
		programmes
German	40.03.01 Law	Bachelor's
		degree
		programmes
German	41.03.05 International Relations	Bachelor's
		degree
		programmes
German	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
German	42.03.02 Journalism	Bachelor's
Commi	12.03.02 #0dHdilbii	degree
		programmes
German	43.03.02 Tourism	Bachelor's
German	+3.03.02 Tourism	degree
		programmes
German	45.03.01 Philology	Bachelor's
German	43.03.01 Timology	degree
		programmes
German	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
German	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
German	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
Octinan	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
German	45.03.02 Linguistics (degree programme 'Languages of the	Bachelor's
Comman	Slavic World and Russian as a Foreign Language (with second	degree
	qualification 'Teacher of the Russian Language and Literature')	programmes
German	47.03.01 Philosophy	Bachelor's
German	47.03.01 1 miosophy	degree
		programmes
German	47.03.03 Religious Studies	Bachelor's
German	77.03.03 Religious Studies	degree
		programmes
German	50.03.01 Arts and Humanities	Bachelor's
OCHHAII	50.05.01 Arts and Humanities	degree
		_
Common	50.02.02 History of Auto	programmes  Deshalor's
German	50.03.03 History of Arts	Bachelor's
		degree
	51.02.01.C. k. 1.0. k.	programmes
German	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
German	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
German	58.03.01 Asian and African Studies	Bachelor's
	COTODIO I I ISIMI MIM I INITAMI SOMOLO	degree
		programmes
Social Science	37.03.02 Conflictology	Bachelor's
Boeiai Beienee	37.03.02 Commetology	degree
		programmes
Social Science	38.03.01 Economics	Bachelor's
Social Science	38.03.01 Economics	
		degree
0 '10'	20.02.02.14	programmes
Social Science	38.03.02 Management	Bachelor's
		degree
		programmes
Social Science	38.03.03 Personnel Management	Bachelor's
		degree
		programmes
Social Science	38.03.04 Public and Municipal Administration	Bachelor's
		degree
		programmes
Social Science	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Social Science	39.03.01 Sociology	Bachelor's
Bocial Belefice	37.03.01 Bociology	degree
		programmes
Social Science	39.03.02 Social Work	Bachelor's
Social Science	37.03.02 Social Work	degree
		programmes
Social Science	40.03.01 Law	Bachelor's
Social Science	40.03.01 Law	degree
		_
0 110 1	41.02.02 B B G . 1'	programmes
Social Science	41.03.02 Russian Region Studies	Bachelor's
		degree
		programmes
Social Science	41.03.04 Political Science	Bachelor's
		degree
		programmes
Social Science	41.03.05 International Relations	Bachelor's
		degree
		programmes
Social Science	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Social Science	43.03.02 Tourism	Bachelor's
		degree
		programmes
Social Science	46.03.01 History	Bachelor's
Social Science	10.03.01 1115101 y	degree
		_
Conial Cairne	47.02.01 Dhilosophy	programmes
Social Science	47.03.01 Philosophy	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Social Science	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Social Science	48.06.01 Theology	Bachelor's
		degree
		programmes
Social Science	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Social Science	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Social Science	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Life Safety	34.02.01 Nursing	Bachelor's
·		degree
		programmes
Life Safety	39.03.02 Social Work	Bachelor's
·		degree
		programmes
Law	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Law	39.03.02 Social Work	Bachelor's
		degree
		programmes
Law	40.03.01 Law	Bachelor's
		degree
		programmes
Law	41.03.05 International Relations	Bachelor's
		degree
		programmes
Law	46.03.01 History	Bachelor's
		degree
		programmes
Law	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Law	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Law	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Law	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Russian	37.03.01 Psychology	Bachelor's
	, 3,	degree
		programmes
Russian	37.03.02 Conflictology	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Russian	37.05.01 Clinical Psychology	Specialist's
		degree
		programme
Russian	37.05.02 Employment Psychology	Specialist's
		degree
		programme
Russian	40.03.01 Law	Bachelor's
		degree
		programmes
Russian	41.03.04 Political Science	Bachelor's
		degree
		programmes
Russian	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
Russian	42.03.02 Journalism	Bachelor's
		degree
		programmes
Russian	45.03.01 Philology	Bachelor's
		degree
		programmes
Russian	45.03.02 Linguistics	Bachelor's
		degree
		programmes
Russian	45.05.01 Translation and Translation Studies	Specialist's
		degree
		programme
Russian	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Russian	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Russian	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Russian	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Russian	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Russian	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Physics	01.03.01 Mathematics	Bachelor's
		degree
		programmes
Physics	01.03.03 Applied Mathematics and Informatics (degree	Bachelor's
	programme 'Modern Programming')	degree
		programmes
Physics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Physics	01.05.01 Fundamental Mathematics and Mechanics (programme 'Fundamental Mechanics')	Specialist's degree
Physics	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	programme Bachelor's degree programmes
Physics	03.03.01 Applied Mathematics and Physics	Bachelor's degree
Physics	03.03.02 Physics	Bachelor's degree programmes
Physics	03.05.01 Astronomy	Specialist's degree programme
Physics	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's degree programmes
Physics	05.03.01 Geology	Bachelor's degree programmes
Physics	21.03.01 Petroleum Engineering	Bachelor's degree programmes
Physics	21.03.02 Land Management and Cadastres	Bachelor's degree programmes
Physics	27.03.03 Systems Analysis and Control	Bachelor's degree programmes
Physical Education	49.03.01 Physical Education	Bachelor's degree programmes
French	37.03.02 Conflictology	Bachelor's degree programmes
French	39.03.01 Sociology	Bachelor's degree programmes
French	39.03.02 Social Work	Bachelor's degree programmes
French	40.03.01 Law	Bachelor's degree programmes
French	41.03.05 International Relations	Bachelor's degree programmes
French	42.03.01 Advertising and Public Relations	Bachelor's degree programmes
French	42.03.02 Journalism	Bachelor's degree programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
French	43.03.02 Tourism	Bachelor's
		degree
		programmes
French	45.03.01 Philology	Bachelor's
		degree
		programmes
French	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
French	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
Trenen	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
French	45.03.02 Linguistics (degree programme 'Languages of the	Bachelor's
Trenen	Slavic World and Russian as a Foreign Language (with second	degree
	qualification 'Teacher of the Russian Language and Literature')	programmes
French	47.03.01 Philosophy	Bachelor's
FIGURE	47.03.01 Filliosophy	
		degree
F1.	47.02.02 D.1'.'	programmes
French	47.03.03 Religious Studies	Bachelor's
		degree
T 1	#0.00.01 h	programmes
French	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
French	50.03.03 History of Arts	Bachelor's
		degree
		programmes
French	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
French	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
French	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Chemistry	04.03.01 Chemistry	Bachelor's
		degree
		programmes
Chemistry	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
	a endingly, a hybrid and material modification	degree
		programmes
Chemistry	05.03.01 Geology	Bachelor's
Chambu	05.05.01 3001053	degree
		programmes
Chemistry	05.03.06 Ecology and Nature Management	Bachelor's
Chemisu y	03.03.00 Leology and Practice Management	degree
		_
Chamiat	06 02 02 Sail Saignes	programmes
Chemistry	06.03.02 Soil Science	Bachelor's
		degree
C1	24.02.04 P	programmes
Chemistry	21.03.01 Petroleum Engineering	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Chemistry	21.03.02 Land Management and Cadastres	Bachelor's
		degree
		programmes
Chemistry	27.03.03 Systems Analysis and Control	Bachelor's
		degree
		programmes
Chemistry	31.05.01 General Medicine	Specialist's
		degree
		programme
Chemistry	31.05.03 Dental Medicine	Specialist's
		degree
		programme
Chemistry	33 05 01 Pharmacy	Specialist's
		degree
		programme
Chemistry	34.02.01 Nursing	Bachelor's
		degree
		programmes
Ecology	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Ecology	06.03.02 Soil Science	Bachelor's
		degree
		programmes
Ecology	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Ecology	39.03.02 Social Work	Bachelor's
		degree
		programmes
Ecology	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Ecology	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Ecology	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Ecology	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Economics	37.03.02 Conflictology	Bachelor's
		degree
г :	20.02.01 F	programmes
Economics	38.03.01 Economics	Bachelor's
		degree
	20.00.00.16	programmes
Economics	38.03.02 Management	Bachelor's
		degree
		programmes
Economics	38.03.03 Personnel Management	Bachelor's
		degree
		programmes

Disciplines of the All-Russian Olympiad for School Students	Titles of the main fields of study and specialities (winners and awardees shall not undergo admissions test to be enrolled to these fields of study and specialities)	Level of education
Economics	38.03.04 Public and Municipal Administration	Bachelor's
		degree
		programmes
Economics	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Economics	39.03.01 Sociology	Bachelor's
		degree
		programmes
Economics	39.03.02 Social Work	Bachelor's
		degree
		programmes
Economics	41.03.05 International Relations	Bachelor's
		degree
		programmes
Economics	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Economics	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Economics	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Economics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes

Annex 7 to the Regulations for Admission to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

List of Olympiads for school students held in accordance with the Procedures for Olympiads for school students, for the results of which St Petersburg University grants particular rights at admissions in 2024; correspondence of fields of study and specialities to degree programmes open for admissions at St Petersburg University in 2024, Olympiad profiles, and the procedures for granting particular rights to winners and awardees of Olympiads for school students held in accordance with the Procedures for Olympiads for school students

Particular rights shall be granted to the winners and awardees of Olympiads for school students held in 2023/2024, 2022/2023, 2021/2022, 2020/2021, 2019/2020 school years in accordance with the Procedures for Olympiads for school students and included in the List of Olympiads for school students approved by the Ministry of Science and Higher Education of the Russian Federation for the respective school year (Orders No 823 of 28 August 2023, No 828 of 30 August 2022, No 804 of 31 August 2021, No 1125 of 27 August 2020, No 658 of 30 August 2019 of the Ministry of Science and Higher Education of the Russian Federation provided they were **10th or 11th** year students at the time of the Olympiad.

Main field of study (speciality) title	Programme title	Type of study	AT No.	Subjects of admissions tests (AT)	Maximu m points to exercise the particul ar right	Olympia d level	FULL TITLE OF THE OLYMPIAD ACCORDING TO ORDER NO 823 OF 28 AUGUST 2023 OF THE MINISTRY OF SCIENCE AND HIGHER EDUCATION OF THE RUSSIAN FEDERATION	Olympiad profile	General education- based subject or the enlarged groups of fields of study and specialities	Category of the contestants	Particular rights to be granted to the winners and runners of the Olympiad
01.00.00 MATHEMATICS AND DEGREE	MECHANICS — BACHELOR'S										
01.03.01 Mathematics	Mathematics	05.04.0 1	1	Mathematics	80	I	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects elected by	Informatics and ICT	75	I	Vysshaia Proba All-Russia School Olympiad	Informatics	Informatics	Winner	admissions tests not required
			the applicant)			I	All-Siberian Open School Olympiad	Informatics	Informatics	Winner	admissions tests not required
						I	'Innopolis Open' international Olympiad	Informatics	Informatics	Winner	admissions tests not required
						I	Moscow School Olympiad	Informatics	Informatics	Winner	admissions tests not required
						I	School Olympiad in Informatics and Programming	Informatics	Informatics	Winner	admissions tests not required

						I	School olympiad in programming	Informatics	Informatics	Winner	admissions tests not required																		
						I	"TechnoCup"  St Petersburg University School Olympiad	Informatics	Informatics	Winner	admissions tests not required																		
						I	Open Olympiad for	Informatics	Informatics	Winner	admissions tests not required																		
							Schoolchildren in Programming																						
				Physics	_				lo particular rights granted																				
			3	Russian					lo particular rights granted	T	T																		
01.03.02Applied Mathematics and Informatics	Mathematical Modelling, Programming, and Artificial Intelligence	05.04.0 1 Geolog	1	Mathematics	75 75	I-III	_	Mathematics	Mathematics	\Winners, Awardees \Winners,	admissions tests not required																		
	intenigence	у	2 (subjects	Informatics and ICT	/5	I-III	_	Informatics	Informatics	Awardees	admissions tests not required																		
			elected by the			I-II	_	Artificial Intelligence	_	\Winners, Awardees	admissions tests not required																		
		applicant)	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required																			
				Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required																		
			3	Russian	_				lo particular rights granted																				
01.03.02Applied Mathematics and Informatics		05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required																		
	Geolog y				III.		Mathematical Modelling and Artificial Intelligence	_	\Winners, Awardees	admissions tests not required																			
		2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required																			
			elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required																		
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required																		
			3	Russian	_			N	lo particular rights granted	•	•																		
01.03.02Applied Mathematics and Informatics	Modern Programming	05.04.0 1 Geolog	1	Mathematics	80	I	Vysshaia Proba All-Russia School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required																		
		У				I	Moscow School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required																		
						I	School olympiad "Lomonosov"	Mathematics	Mathematics	Winner	admissions tests not required																		
						I	School olympiad "Phystech"	Mathematics	Mathematics	Winner	admissions tests not required																		
						I	St Petersburg University School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required																		
						I	St Petersburg School Olympiad	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required																		
						I	Tournament of Cities	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required																		
			2	Informatics and ICT	80	I	Vysshaia Proba All-Russia School Olympiad	Informatics	Informatics	Winner	admissions tests not required																		
																-	-								I	All-Siberian Open School Olympiad	Informatics	Informatics	Winner
						I	University Academic Olympiad in Informatics	Informatics	Informatics	Winner	admissions tests not required																		

						I	'Innopolis Open' international	Informatics	Informatics	Winner	admissions tests not required
						I	Olympiad Moscow School Olympiad	Informatics	Informatics	Winner	admissions tests not required
						I	School olympiad "Lomonosov"	Informatics	Informatics	Winner	admissions tests not required
						I	School Olympiad in Informatics and Programming	Informatics	Informatics	Winner	admissions tests not required
						I	School olympiad in programming "TechnoCup"	Informatics	Informatics	Winner	admissions tests not required
						I	St Petersburg University School Olympiad	Informatics	Informatics	Winner	admissions tests not required
						I	Open Olympiad for Schoolchildren in	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian			Programming	N.	particular rights granted		
01.03.03 Mechanics and Mathematical Modelling	Mechanics and Mathematical Modelling	05.04.0	1	Mathematics	75	I-III	_	Mathematics	Mathematics Mathematics	\Winners, Awardees	admissions tests not required
<i>g</i>		Geolog y				I-III	_	Mechanics and Mathematical Modelling	_	\Winners, Awardees	admissions tests not required
			2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
01 00 00 14 17 17 17 17 17 17 17 17 17 17 17 17 17	D MECHANICS — SPECIALIST'S		3	Russian			_	Ne	particular rights granted	1	_
DEGREE	D MECHANICS — SPECIALIST'S										
01.05.01 Fundamental Mathematics and Mechanics	Fundamental Mathematics	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			elected by the applicant)	Informatics and ICT	75	I-III	<b>—</b> .	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			аррисані)	Chemistry	75	I-III			Chemistry	\Winners, Awardees	admissions tests not required
			3	Russian	_			Ne	particular rights granted		
01.05.01 Fundamental Mathematics and Mechanics	Fundamental Mechanics	05.04.0	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y				I-III	_	Mechanics and Mathematical Modelling	_	\Winners, Awardees	admissions tests not required
			2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			elected by the	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry Russian	75	I-III	_		Chemistry  particular rights granted	\Winners, Awardees	admissions tests not required
02.00.00 COMPUTER AND IN	FORMATION SCIENCES —	1	ر	Aussian	<del>                                     </del>			IN	, particular rights granted		
BACHELOR'S DEGREE 02.03.01 Mathematics and	Mathematics and Computer Sciences	05.04.0	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners,	admissions tests not required
Computer Sciences	r	1 Geolog	2	Informatics and	75	I-III	_	Informatics	Informatics	Awardees \Winners,	admissions tests not required
		y	(subjects	ICT					mormatics	Awardees	admissions tests not required
			elected by	Physics	75	I-III	_	Physics	Physics	\Winners,	admissions tests not required

			the applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required		
			3	Russian				No	particular rights granted				
02.03.01 Mathematics and Computer Sciences	Data Sciences	05.04.0 1 Geolog	1	Mathematics	80	I	Vysshaia Proba All-Russia School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required		
		у				I	Moscow School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required		
						I	St Petersburg University School Olympiad	Mathematics	Mathematics	Winner	admissions tests not required		
								I	St Petersburg School Olympiad	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
						I	Tournament of Cities	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required		
			2	Informatics and ICT	80	I	Vysshaia Proba All-Russia School Olympiad	Informatics	Informatics	Winner	admissions tests not required		
						I	All-Siberian Open School Olympiad	Informatics	Informatics	Winner	admissions tests not required		
							I	'Innopolis Open' international Olympiad	Informatics	Informatics	Winner	admissions tests not required	
						I	Moscow School Olympiad	Informatics	Informatics	Winner	admissions tests not required		
					I	School Olympiad in Informatics and Programming	Informatics	Informatics	Winner	admissions tests not required			
						I	School olympiad in programming "TechnoCup"	Informatics	Informatics	Winner	admissions tests not required		
						I	St Petersburg University School Olympiad	Informatics	Informatics	Winner	admissions tests not required		
						I	Open Olympiad for Schoolchildren in Programming	Informatics	Informatics	Winner	admissions tests not required		
			3	Russian			1 8	No	particular rights granted	L.			
02.03.02 Fundamental Informatics and Information	Dig Data and Blockchain Digital Platform	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required		
Technology		Geolog y	2 (subjects	Informatics and ICT	75	I-III	=	Informatics	Informatics	\Winners, Awardees	admissions tests not required		
			elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required		
			applicant)	Chemistry	75	I-III	_		Chemistry	\Winners, Awardees	admissions tests not required		
02.03.02 Fundamental Informatics and Information	Programming and Information Technology	05.04.0	1	Russian Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required		
Technology	reciniology	Geolog y	2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required		
			elected by	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required		
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required		
			3	Russian				No	particular rights granted				
02.03.03 Information Systems Administration and Mathematical	Technologies of Programming	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required		
Support		Geolog y	2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required		
			elected by	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required		

			the	Chemistry	75	I-III	_	_	Chemistry	\Winners,	admissions tests not required
			applicant) 3	Russian	_			No	particular rights granted	Awardees	
03.00.00 PHYSICS AND ASTRO	NOMY — BACHELOR'S DEGREE			Tuosan				110	particular rights granted		
03.03.01 Applied Mathematics and Physics	Engineering-Oriented Physics	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			elected by the	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_		Chemistry	\Winners, Awardees	admissions tests not required
02.02.01 Applied Made and in	Analind Dhaning of Mathematics	05.04.0	3	Russian	75	I-III	_	- Mathematica	Russian	\Winners, Awardees	100 points for AT in Russian
03.03.01 Applied Mathematics and Physics	Applied Physics and Mathematics	05.04.0 1 Geolog	2	Mathematics	75 75	I-III I-III	_	Mathematics	Mathematics	\Winners, Awardees \Winners,	admissions tests not required
		y	(subjects elected by	Physics	/5	I-III		Physics	Physics	Awardees \Winners,	admissions tests not required
		the applicant)	Informatics and	75	I-III		Astronomy  Informatics	Astronomy Informatics	Awardees \Winners,	admissions tests not required admissions tests not required	
			аррисанс	ICT Chemistry	75	I-III			Chemistry	Awardees \Winners,	admissions tests not required
			3	Russian	75	I-III	_		Russian	Awardees \Winners,	100 points for AT in Russian
03.03.02 Physics	Physics	05.04.0	1	Physics	75	I-III		Physics	Physics	Awardees \Winners,	admissions tests not required
03.03.02 Physics Physics	rhysics	1 Geolog	1	rilysics	/3	I-III		Astronomy	Astronomy	Awardees \Winners,	admissions tests not required
		у	2	Mathematics	75	I-III		Mathematics	Mathematics	Awardees \Winners,	admissions tests not required
			(subjects elected by	Informatics and	75	I-III		Informatics	Informatics	Awardees \Winners,	admissions tests not required
			the applicant)	ICT Chemistry	75	I-III	_		Chemistry	Awardees \Winners,	admissions tests not required
			3	Russian	75	I-III			Russian	Awardees \Winners,	100 points for AT in Russian
03 00 00 PHYSICS AND ASTROI	NOMY — SPECIALIST'S DEGREE		,	Kussian	/3	1-111			Russian	Awardees	100 points for AT in Russian
03.05.01 Astronomy	Astronomy	05.04.0 1	1	Physics	75	I-III	_	Astronomy	Astronomy	\Winners, Awardees	admissions tests not required
		Geolog y				I-III	ı	Physics	Physics	\Winners, Awardees	admissions tests not required
			2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by the	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
AAAAAAAAA			3	Russian	_		1	No	particular rights granted	T	
04.00.00 CHEMISTRY — BACHELOR'S DEGREE											
04.03.01 Chemistry	Chemistry	05.04.0	1	Chemistry	80	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Mathematics	80	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by the applicant)	Physics	80	I-III	_	_	Physics	\Winners, Awardees	admissions tests not required
			3	Russian	80	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
04.03.02 Chemistry, Physics and Material Mechanics	Chemical Material Science	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required

	1	1		a .			1	T			T
		Geolog y	2 (subjects	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
			elected by the			I-III	_	Engineering Sciences	_	\Winners, Awardees	admissions tests not required
			applicant)			III.	_	Engineering systems	_	\Winners, Awardees	admissions tests not required
						II.	_	NanoSystems and Nanoengineering	_	\Winners, Awardees	admissions tests not required
				Physics	75	I-III	_	_	Physics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
05.00.00 EARTH SCIENCES —	BACHELOR'S DEGREE										
05.03.01 Geology	Geology	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y				I-III	_	Geology	_	\Winners, Awardees	admissions tests not required
			2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			elected by the	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
			applicant)	Biology	75	I-III	_		Biology	\Winners, Awardees	admissions tests not required
				Chemistry	75	I-III	_		Chemistry	\Winners, Awardees	admissions tests not required
				Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
05.03.02 Geography	Geography (with the qualification of 'Teacher of Geography)	05.04.0 1	1	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Ţ	elected by the	Biology	75	I-III	_		Biology	\Winners, Awardees	admissions tests not required
			applicant)	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
05.03.03 Cartography and Geoinformatics	Cartography and Geoinformatics	05.04.0 1	1	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by the	Biology	75	I-III	_		Biology	\Winners, Awardees	admissions tests not required
			applicant)	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
05.03.04 Hydrometeorology	Hydrometeorology	05.04.0 1	1	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by the	Biology	75	I-III	_	_	Biology	\Winners, Awardees	admissions tests not required
			applicant)	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
05.03.06 Ecology and Nature Management	Ecology and Nature Management	05.04.0 1	1	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
Management		Geolog v				II.	_	Ecology	_	\Winners, Awardees	admissions tests not required

			2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by the	Biology	75	I-III	=	_	Biology	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
				Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
06.00.00 BIOLOGICAL SCIENCE	S _ BACHELOD'S DECDEE									Tiwardees	
06.03.01Biology	Biology	05.04.0	1	Biology	75	ī			Biology	Winner	admissions tests not required
00.03.01Biology	Blology	1	1	Diology	13	Ī			Biology	Awardees	100 points for AT in Biology
		Geolog				II.	_	_	Biology	Winner	100 points for AT in Biology
		у	2	Chemistry	75	I	_	_	Chemistry	\Winners, Awardees	100 points for AT in Chemistry
						II.	_	_	Chemistry	Winner	100 points for AT in Chemistry
			3	Russian	75	I	_	_	Russian	\Winners,	100 points for AT in Russian
						TT			Russian	Awardees Winner	100 points for AT in Russian
06.03.01Biology	Biology:bioengineering technologies	05.04.0	1	Biology	75	II. I			Biology	Winner	admissions tests not required
00.03.01Biology	Biology.bioengineering technologies	1	1	Diology	13	I			Biology	Awardees	100 points for AT in Biology
		Geolog				II.			Biology	Winner	100 points for AT in Biology
		у	2	Mathematics	75	I	_	Mathematics	Mathematics	\Winners,	100 points for AT in
					II.	_	Mathematics	Mathematics	Awardees Winner	Mathematics 100 points for AT in Mathematics	
			3	Russian	75	I	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						II.	_		Russian	Winner	100 points for AT in Russian
06.03.02 Soil Science	Soil Science	05.04.0	1	Biology	75	I-III	_	_	Biology	\Winners, Awardees	admissions tests not required
		Geolog				II-III	_	Ecology	_	\Winners, Awardees	admissions tests not required
			2 (subjects	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by	Chemistry	75	I-III	_	_	Chemistry	\Winners,	admissions tests not required
			the applicant)	j						Awardees	
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
09.00.00 INFORMATICS AND CO BACHELOR'S DEGREE	OMPUTER SCIENCE —										
09.03.03 Applied Informatics	Applied Informatics in Arts and Humanities	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
09.03.03 Applied Informatics	Artificial Intelligence and Data Science	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog	2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			elected by the			I-II	_	Artificial Intelligence	_	\Winners, Awardees	admissions tests not required
			applicant)	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
				Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
		1	3	Russian				N	o particular rights granted	d	

09.03.04 Software Engineering	Software Engineering	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
		·	elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			applicant)	Chemistry	75	I-III	_		Chemistry	\Winners, Awardees	admissions tests not required
			3	Russian	_			No	particular rights granted		-1
21.00.00 APPLIED GEOLOGY, M	INING, PETROLEUM ENGINEERIN	G AND									
GEODESY — BACHELOR'S DEG	GREE									i	
21.03.01 Petroleum Engineering	Petroleum Engineering	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y	2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
		elected by the	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required	
		applicant)	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required	
				Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
		3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
	Real Estate Cadastre: Appraisal and Information Support	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
	Geolog y	2 (subjects	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required	
			elected by the	Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
			applicant)	Geography	75	I-III	_	_	Geography	\Winners, Awardees	admissions tests not required
				Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
DEGREE	RING SYSTEMS — BACHELOR'S										
27.03.03 Systems Analysis and Control	Applied Information Technology	05.04.0 1	1	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
		Geolog y				III.	_	Engineering systems	_	\Winners, Awardees	admissions tests not required
			2 (subjects	Informatics and ICT	75	I-III	_	Informatics	Informatics	\Winners, Awardees	admissions tests not required
			elected by the	Physics	75	I-III	_	Physics	Physics	\Winners, Awardees	admissions tests not required
			applicant)				_	Engineering systems	_	\Winners, Awardees	admissions tests not required
				Chemistry	75	I-III	_	_	Chemistry	\Winners, Awardees	admissions tests not required
			3	Russian	_		•	No	particular rights granted		_
31.00.00 CLINICAL MEDICINE —	- SPECIALIST'S DEGREE	1					1			1	1
PROGRAMME 31.05.01 General Medicine	General Medicine	05.04.0	1	Biology	75	I	_	_	Biology	\Winners,	admissions tests not required
		Geolog				II-III	_	_	Biology	Awardees \Winners, Awardees	100 points for AT in Biology
		у				I	_	Medicine	_	\Winners, Awardees	admissions tests not required
						II-III	_	Medicine	_	\Winners, Awardees	100 points for AT in Biology
		2	Chemistry	75	I	_	_	Chemistry	\Winners,	admissions tests not required	
										Awardees	· I

1												
10   10   10   10   10   10   10   10				3	Russian	75	I-III	_	_	Russian		100 points for AT in Russian
1   -	31.05.03 Dental Medicine	Dental Medicine	05.04.0	1	Biology	75	I	_		Biology		admissions tests not required
Marie			1				I	_	_			
March   Marc								_	_		\Winners,	100 points for AT in Biology
1			,				III		_	Riology		100 points for AT in Biology
								+				
Harmonia												
Medicine												
Additional Parameter   Param										_	Awardees	-
Augustion   Augu												
13.06.00 CLINICAL MEDICINE				2	Chemistry	75	I	_	_	Chemistry	Awardees	
1							II.	_	_	Chemistry		100 points for AT in Chemistry
1.   -   -				3	Russian	75	I	_	_	Russian		100 points for AT in Russian
							II.	_	_	Russian		100 points for AT in Russian
Pharmacy		IE — SPECIALIST'S DEGREE										
Second   S		Pharmacy	05.04.0 1	1	Chemistry	75	I	_	_	Chemistry		admissions tests not required
II-III							II-III	_	_	Chemistry		100 points for AT in Chemistry
Awardees   Awardees   Awardees				2	Biology	75	I	_	_	Biology	Awardees	admissions tests not required
Avantage								_		Biology	Awardees	100 points for AT in Biology
Augusta   Augu								_	Medicine	_	Awardees	admissions tests not required
Addition								_	Medicine	_	Awardees	100 points for AT in Biology
Nursing				3	Russian	75	I-III	_	_	Russian		100 points for AT in Russian
1												
A wardees   A wardees   A wardees	34.02.01 Nursing	Nursing	1	1	Biology	75		_			Awardees	
34.02.01 Nursing   Nursing (on the basis of secondary vocational education)   O5.04.0   1   Nursing   Nursing   O5.04.0   1   Russian   O5.04.0   O5								_	Medicine	_	Awardees	admissions tests not required
Adv. 2.01 Nursing Nursing (on the basis of secondary vocational education)  Nursing (on the basis of secondary vocational education)  Nursing (on the basis of secondary vocational education)  Russian					Chemistry				_	Chemistry	Awardees	admissions tests not required
vocational education)  1 Geolog y  37.00.00 PSYCHOLOGICAL SCIENCES — BACHELOR'S DEGREE  37.03.01 Psychology  Winners, admissions tests not require  II. — — Biology  Winners, admissions tests not require  III. — Psychology  III. — Psychology  III. — Psychology  Winner admissions tests not require  III. — Psychology  Winner 100 points for AT in Biolog  III. — Psychology  Winner 100 points for AT in Biolog  III. — Medicine  III. — Medicine  Winners, admissions tests not require  Winner admissions tests not require  III. — Medicine  Winner admissions tests not require				3	Russian	75	I-III	_	_	Russian		100 points for AT in Russian
Geolog y  37.00.00 PSYCHOLOGICAL SCIENCES — BACHELOR'S DEGREE  37.03.01 Psychology  Awardees  II. — — Biology  III. — — Psychology  III. — Medicine	34.02.01 Nursing		05.04.0		Nursing	_			N	o particular rights granted	ı	
Psychology   Psy		vocational education)	Geolog v	2	Russian	_			N	o particular rights granted		
Psychology   Psy	37.00.00 PSYCHOLOGICAL S	CIENCES — BACHELOR'S DEGREE										
Geolog y  II. — — Biology Winner admissions tests not require III. — — Biology Awardees 100 points for AT in Biolog III. — — Biology Winners, 100 points for AT in Biolog III. — Psychology — Winners, Awardees  II. — Psychology — Winner admissions tests not require II. — Psychology — Winner admissions tests not require III. — Psychology — Winner 100 points for AT in Biolog III. — Psychology — Winner 100 points for AT in Biolog III. — Psychology — Winner 4 admissions tests not require III. — Medicine — Winner 4 admissions tests not require III. — Medicine — Winner 4 admissions tests not require III. — Medicine — Winner 4 admissions tests not require III. — Medicine — Winner 4 admissions tests not require			05.04.0	1	Biology	75	I	_	_	Biology		admissions tests not required
II.   —   Biology   Awardees   100 points for AT in Biolog   Winners, Awardees   III.   —   Psychology   —   Winners, Awardees   III.   —   Psychology   —   Winners admissions tests not require   III.   —   Psychology   —   Winner admissions tests not require   III.   —   Psychology   —   Winner   100 points for AT in Biolog   III.   —   Psychology   —   Winner   100 points for AT in Biolog   III.   —   Psychology   —   Winner   4dmissions tests not require   Awardees   III.   —   Medicine   —   Winner   Awardees   III.   Winners, Admissions tests not require   Winner   Awardees   III.   —   Medicine   —   Winner   Admissions tests not require   Winner   Admissions tests not require   III.   Medicine   —   Winner   III.			Geolog				II.	_	1 -	Biology		admissions tests not required
III. — Psychology — Winners, admissions tests not require  II. — Psychology — Winner admissions tests not require  II. — Psychology — Winner admissions tests not require  II. — Psychology — Awardees  III. — Psychology — Winner admissions tests not require  III. — Psychology — Winner 100 points for AT in Biolog  III. — Psychology — Winner 100 points for AT in Biolog  III. — Medicine — Winner admissions tests not require  Awardees  III. — Medicine — Winner admissions tests not require												
I — Psychology — Winners, Awardees  II. — Psychology — Winner admissions tests not require  II. — Psychology — Winner admissions tests not require  II. — Psychology — Awardees 100 points for AT in Biolog  III. — Psychology — Winner 100 points for AT in Biolog  II. — Medicine — Winners, admissions tests not require  Awardees  II. — Medicine — Winner admissions tests not require								_			\Winners,	100 points for AT in Biology
II. — Psychology — Winner admissions tests not require  II. — Psychology — Awardees 100 points for AT in Biolog  III. — Psychology — Winner 100 points for AT in Biolog  I — Medicine — Winner Admissions tests not require  II. — Medicine — Winner admissions tests not require  III. — Medicine — Winner admissions tests not require							I	_	Psychology	_	\Winners,	admissions tests not required
II. — Psychology — Awardees 100 points for AT in Biolog III. — Psychology — Winner 100 points for AT in Biolog I — Medicine — Winner, admissions tests not require Awardees II. — Medicine — Winner admissions tests not require							II		Psychology	+		admissions tests not required
III. — Psychology — Winner 100 points for AT in Biolog  I — Medicine — Winners, admissions tests not require Awardees  II. — Medicine — Winner admissions tests not require												
I — Medicine — \\Winners, admissions tests not require \\Awardees\\ II. — Medicine — Winner admissions tests not require					1	1						
II. — Medicine — Winner admissions tests not require											\Winners,	admissions tests not required
							***	<b>+</b>	Madiair :	+		adminsions to to act as and 1
II. — Medicine — Awardees 100 points for AT in Biolog		1			İ	1	II.	<u> </u>	Medicine Medicine	+	Awardees	100 points for AT in Biology

			2	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
37.03.02 Conflictology	Conflictology	05.04.0 1	1	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
		Geolog				II.	_	_	Social Science	Winner	admissions tests not required
		у				II.	_	_	Social Science	Awardees	100 points for AT in Social Science
						I	_	Philosophy	_	\Winners, Awardees	admissions tests not required
						II.	_	Philosophy	_	Winner	admissions tests not required
						II.	_	Philosophy	_	Awardees	100 points for AT in Social Science
						III.	_	Philosophy	_	\Winners, Awardees	100 points for AT in Social Science
						I	_	Law	_	\Winners, Awardees	admissions tests not required
						II.	_	Law	_	Winner	admissions tests not required
						II.	_	Law		Awardees	100 points for AT in Social Science
			2	History	75	I	_	_	History	Winner	admissions tests not required
			(subjects elected by			I	_	_	History	Awardees	100 points for AT in Social Science
			the applicant)			II-III	_	_	History	\Winners, Awardees	100 points for AT in Social Science
				Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
				(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
				Spanish, or Chinese, or German, or French)		II.		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
37.00.00 PSYCHOLOGICAL SO	CIENCES — SPECIALIST'S DEGREE										
37.05.01 Clinical Psychology	Clinical Psychology	05.04.0 1	1	Biology	75	I	_	_	Biology	\Winners, Awardees	admissions tests not required
		Geolog				II.		_	Biology	Winner	admissions tests not required
		У				II.		_	Biology	Awardees	100 points for AT in Biology
						III.	_	_	Biology	\Winners, Awardees	100 points for AT in Biology
						Ι	_	Psychology		\Winners, Awardees	admissions tests not required
				Ì		II.	_	Psychology	_	Winner	admissions tests not required
				Ì		II.	_	Psychology	_	Awardees	100 points for AT in Biology
						III.	_	Psychology	_	Winner	100 points for AT in Biology
						I	_	Medicine	_	\Winners, Awardees	admissions tests not required
				Ì		II.	_	Medicine	_	Winner	admissions tests not required
				ļ	ļ	II.	_	Medicine	—	Awardees	100 points for AT in Biology
			2	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
37.05.02 Employment Psychology	Employment Psychology	05.04.0 1	1	Biology	75	I	_	_	Biology	\Winners, Awardees	admissions tests not required
		Geolog		Ì		II.	_		Biology	Winner	admissions tests not required
		у		1		II.	_		Biology	Awardees	100 points for AT in Biology
						III.	_	_	Biology	\Winners, Awardees	100 points for AT in Biology
						I	_	Psychology	_	\Winners, Awardees	admissions tests not required
		1		I	1	II.	_	Psychology	_	Winner	admissions tests not required

	T	1	1	ı	1		1	I n			1400 1 4 4 7 1 7 1
						II.	_	Psychology	_	Awardees	100 points for AT in Biology
						III.	_	Psychology	_	Winner	100 points for AT in Biology
						I	_	Medicine	_	\Winners, Awardees	admissions tests not required
						П.	_	Medicine	_	Winner	admissions tests not required
						II.	_	Medicine	_	Awardees	100 points for AT in Biology
			2	Mathematics	75	I-III	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.00.00 ECONOMICS AND DEGREE	MANAGEMENT — BACHELOR'S									Awardees	
38.03.01 Economics	Economics	05.04.0	1	Mathematics	75	I-II	<del> </del>	Mathematics	Mathematics	Winner	admissions tests not required
		1 Geolog				I-II	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		у				III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			2	Social Science	75	I-II			Social Science	Winner	admissions tests not required
			(subjects	Social Science	13	I-II	_		Social Science	Awardees	100 points for AT in Social
			elected by the			III.	_	_	Social Science	\Winners,	Science 100 points for AT in Social
			applicant)							Awardees	Science
						I-II		Economics	Economics	Winner	admissions tests not required
						I-II	_	Economics	Economics	Awardees	100 points for AT in Social Science
						III.	_	Economics	Economics	\Winners, Awardees	100 points for AT in Social Science
				Geography	75	I-II	_	_	Geography	Winner	admissions tests not required
						I-II	_	_	Geography	Awardees	100 points for admissions test in Geography
						III.	_	_	Geography	\Winners, Awardees	100 points for admissions test in Geography
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.01 Economics	Economics (with Advanced Studies in	05.04.0	1	Mathematics	75	I-II	_	Mathematics	Mathematics	Winner	admissions tests not required
	Chinese Economy and Language)	1 Geolog				I-II	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		у				III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			2 (subjects elected by the	Foreign Languages (English or Chinese)	75	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			applicant)	Geography	75	I-II	_	_	Geography	\Winners, Awardees	100 points for admissions test in Geography
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.01 Economics	Economic and mathematical methods	05.04.0	1	Mathematics	75	I-II	_	Mathematics	Mathematics	Winner	admissions tests not required
		1 Geolog				I-II	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		у				III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			2	Social Science	75	I-II	_	_	Social Science	Winner	admissions tests not required
			(subjects elected by		,,,	I-II	-	_	Social Science	Awardees	100 points for AT in Social Science
	1	1			1	III.	_		Social Science	\Winners,	100 points for AT in Social
			the applicant)							Awardees	Science
			applicant)			1_11	+	Economics	Economics	Awardees	Science
						I-II I-II		Economics Economics	Economics Economics	Awardees Winner Awardees	admissions tests not required 100 points for AT in Social
										Winner	admissions tests not required

						I-II	_	_	Geography	Awardees	100 points for admissions test in Geography
						III.	_	_	Geography	\Winners, Awardees	100 points for admissions test in Geography
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.02 Management	Management	05.04.0	1	Mathematics	75	I	_	Mathematics	Mathematics	Winner	admissions tests not required
		1 Geolog				I	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		y	2 (subjects elected by the applicant)	Foreign language (English, or Spanish, or Chinese, or German, or French)	75	I-II		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.02 Management	International Management	05.04.0	1	Mathematics	75	I	_	Mathematics	Mathematics	Winner	admissions tests not required
		1 Geolog				I	-	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
	y 5.03.03 Personnel Management Personnel Management 05		2	English	75	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in English
			3	Russian	75	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.03 Personnel Management	Personnel Management	05.04.0	1	Mathematics	75	I-II	_	Mathematics	Mathematics	Winner	admissions tests not required
·		1 Geolog				I-II	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		у				III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics
			2	Social Science	75	I-II	_		Social Science	Winner	admissions tests not required
			_	Bookin Belence	,,,	I-II	-	_	Social Science	Awardees	100 points for AT in Social Science
						III.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
						I-II	_	Economics	Economics	Winner	admissions tests not required
						I-II	_	Economics	Economics	Awardees	100 points for AT in Social Science
						III.	_	Economics	Economics	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.04 Public and Municipal	Public Administration	05.04.0	1	Mathematics	75	I	_	Mathematics	Mathematics	Winner	admissions tests not required
Administration		1 Geolog				I	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
		у	2 (subjects	Social Science	75	I-II	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			elected by the applicant)	Geography	_			No	particular rights granted	•	·
			3	Russian	75	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
38.03.05 Business Informatics	Business Informatics	05.04.0	1	Mathematics	75	I-II	_	Mathematics	Mathematics	Winner	admissions tests not required
18.05.05 Business Informatics  Business Informatics		1 Geolog				I-II		Mathematics	Mathematics	Awardees	100 points for AT in Mathematics
	у				III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics	
			2	Informatics and	75	I-II	_	Informatics	Informatics	Winner	admissions tests not required
				ICT		I-II		Informatics	Informatics	Awardees	100 points for AT in Social Science
						III.	_	Informatics	Informatics	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	=	Russian	\Winners, Awardees	100 points for AT in Russian

DEGREE	SOCIAL WORK — BACHELOR'S					_					
9.03.01 Sociology	Sociology	05.04.0 1	1	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
		Geolog				II.	_	_	Social Science	Winner	admissions tests not required
		У				II.	_	_	Social Science	Awardees	100 points for AT in Social Science
						III.	_	-	Social Science	Winner	100 points for AT in Social Science
			2	Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
			(subjects	(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
			elected by the applicant)	Spanish, or Chinese, or German, or French)		II-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russia
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
9.03.01 Sociology	Sociological Research in Digital Society	05.04.0 1	1	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
		Geolog				II.	_		Social Science	Winner	admissions tests not required
		У				II.	_	_	Social Science	Awardees	100 points for AT in Social Science
						III.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			2 (subjects	Mathematics	75	I	_	Mathematics	Mathematics	\Winners, Awardees	admissions tests not required
			elected by			II.	I —	Mathematics	Mathematics	Winner	admissions tests not required
		the applicant)			II.	_	Mathematics	Mathematics	Awardees	100 points for AT in Mathematics	
					III.	_	Mathematics	Mathematics	\Winners, Awardees	100 points for AT in Mathematics	
				Informatics and	75	I	_	Informatics	Informatics	Winner	admissions tests not required
				ICT		I	_	Informatics	Informatics	Awardees	100 points for AT in Social Science
						II-III	_	Informatics	Informatics	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
9.03.02 Social Work	Social Work	05.04.0	1	History	75	I-II	_	_	History	\Winners, Awardees	admissions tests not required
		Geolog y				III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
			2 (subjects	Social Science	75	I-II	_	_	Social Science	\Winners, Awardees	admissions tests not required
			elected by the			III.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			applicant)	Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
				(English, or Spanish, or Chinese, or		II-III	<u> </u>	Foreign Language Foreign Language	Foreign Language Foreign Language	Awardees \Winners, Awardees	100 points for AT in Russian 100 points for AT in Russian
				German, or French)							
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russia
0.00.00 LAW — BACHELO		05		0.110:			-			****	<del> </del>
0.03.01 Law	Law	05.04.0 1	1	Social Science	75	I I			Social Science Social Science	Winner Awardees	admissions tests not required 100 points for AT in Social
		Geolog y				II.	<del> </del>		Social Science	\Winners,	Science 100 points for AT in Social
					1	-	+	T		Awardees	Science
						1	<del>  -</del>	Law	_	Winner	admissions tests not required
						I	_	Law	_	Awardees	100 points for AT in Social Science

						II.	_	Law	_	\Winners, Awardees	100 points for AT in Social Science
			2	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
			3	History	75	I	_	_	History	Winner	admissions tests not required
						I	_		History	Awardees	100 points for AT in Social Science
						II.	_	-	History	\Winners, Awardees	100 points for AT in Social Science
40.03.01 Law	Law (with advanced studies in	05.04.0	1	Social Science	75	I	_		Social Science	Winner	admissions tests not required
	Chinese Language and Law)	1 Geolog				I	_	_	Social Science	Awardees	100 points for AT in Social Science
		у				II.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
						I	_	Law	_	Winner	admissions tests not required
						I	_	Law	_	Awardees	100 points for AT in Social Science
						II.	_	Law	_	\Winners, Awardees	100 points for AT in Social Science
			2	Russian	75	I-III	_	-	Russian	\Winners, Awardees	100 points for AT in Russian
			3	Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
			(subjects	(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
			elected by the applicant)	Spanish, or Chinese, or German, or French)		II.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
40.03.01 Law	Law (for postgraduates)	part- time	1	Legal Studies	_		1	No	particular rights granted		
41.00.00 POLITICAL SCIENCE BACHELOR'S DEGREE	AND REGIONAL STUDIES —	time									
41.03.02 Russian Region Studies	Russian Studies	05.04.0 1	1	History	75	I-III	_	_	History	\Winners, Awardees	100 points for AT in Social Science
		Geolog y	2	Social Science	75	I-III	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
41.03.04 Political Science	Political Science	05.04.0 1	1	History	75	I	_	_	History	\Winners, Awardees	admissions tests not required
		Geolog				П.	_		History	Winner	admissions tests not required
		У				II.	_	_	History	Awardees	100 points for AT in Social Science
						III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
			2 (subjects	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
		1	elected by			II.	_		Social Science	Winner	admissions tests not required
			the applicant)			II.	_	_	Social Science	Awardees	100 points for AT in Social Science
						III.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
						I-II	_	Political Science	Political Science and Regional Studies	Winner	admissions tests not required
						I-II	_	Political Science	Political Science and Regional Studies	Awardees	100 points for AT in Social Science
						III.	_	Political Science	Political Science and Regional Studies	\Winners, Awardees	100 points for AT in Social Science
				Geography	75	I-III	_	_	Geography	\Winners, Awardees	100 points for admissions test Geography
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
41.03.05 International Relations	International Relations	05.04.0	1	History	75	I			History	Winner	admissions tests not required
		1		ĺ		I	_	_	History	Awardees	100 points for AT in Social Science

		Geolog y				II-III	_	_	History	\Winners, Awardees	100 points for AT in Social Science
		y				I	_	International Relations	Political Science and Regional Studies	Winner	admissions tests not required
						I	_	International Relations	Political Science and Regional Studies	Awardees	100 points for AT in Social Science
						II-III	_	International Relations	Political Science and Regional Studies	\Winners, Awardees	100 points for AT in Social Science
			2	Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
			(subjects	(English, or		I	i	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
			elected by the applicant)	Spanish, or Chinese, or German, or French)		II-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
				Geography	75	I-III		_	Geography	\Winners, Awardees	100 points for ВИ in Geography
			3	Russian	75	I-III		_	Russian	\Winners, Awardees	100 points for AT in Russian
	FORMATION LIBRARIANSHIP — BA		S DEGREE								
42.03.01 Advertising and Public Relations	Advertising and Public Relations	05.04.0 1	1	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
		Geolog y	2	Social Science	75	I-II	_	_	Social Science	\Winners, Awardees	admissions tests not required
						III.	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			3 (subjects	History	75	I-II	_	_	History	\Winners, Awardees	admissions tests not required
			elected by the			III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
			applicant)	Foreign language (English, or	75	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	admissions tests not required
				German, or French)		III.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
				Informatics and ICT	_			No	particular rights granted		
42.03.02 Journalism	Journalism	05.04.0 1	1	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
		Geolog y				I		Journalism	Mass Media and Information Librarianship	\Winners, Awardees	admissions tests not required
						II-III		Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
			2	Literature	75	I-II	_	_	Literature	\Winners, Awardees	admissions tests not required
						III.	_	_	Literature	\Winners, Awardees	100 points for AT in Social Science
			3	Practical creative admissions test	75	I-III	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
42.03.02 Journalism	Journalism	part- time	1	Russian	75	I-III		_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	admissions tests not required
						III.		Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
			2	Literature	75	I-II	_	_	Literature	\Winners, Awardees	admissions tests not required
						III.	_	_	Literature	\Winners, Awardees	100 points for AT in Social Science

			3	Practical creative admissions test	75	I-III	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
42.03.02 Journalism	International Journalism	05.04.0 1	1	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
		Geolog y				I	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	admissions tests not required
						II-III	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
			2	Literature	75	I-II	_	_	Literature	\Winners, Awardees	admissions tests not required
						III.	_		Literature	\Winners, Awardees	100 points for AT in Social Science
			3	Practical creative admissions test	75	I-III	_	Journalism	Mass Media and Information Librarianship	\Winners, Awardees	100 points for the creative admissions test
43.00.00 SERVICES AND TO	URISM — BACHELOR'S DEGREE										
43.03.02 Tourism	Organisation of Tourist Activities (with Advanced Chinese Language	05.04.0 1	1	History	75	I-II	_		History	\Winners, Awardees	admissions tests not required
	Training) (with second qualification 'Head of Hotel Complex and Other	Geolog y				III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
	Accommodation Facilities) Tourism (with second qualification		2 (subjects	Geography	75	I-II	_	_	Geography	\Winners, Awardees	admissions tests not required
	'Head of Hotel Complex and Other Accommodation Facilities')		elected by the			III.	_	_	Geography	\Winners, Awardees	100 points for admissions test in Geography
	Organisation of Tourist Activities (with Advanced Thai Language		applicant)	Foreign language (English, or	75	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	admissions tests not required
	(with Advanced Thai Language Training)			Spanish, or Chinese, or German, or French)		III.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
45.00.00 LINGUISTICS AND DEGREE	LITERARY STUDIES — BACHELOR'S										
45.03.01 Philology	English Language and Literature (with the qualifications 'Teacher of	05.04.0 1	1	Literature	80	I	_	_	Literature	\Winners, Awardees	admissions tests not required
	the English Language and	Geolog				II.	_	_	Literature	Winner	admissions tests not required
	Literature'/'Specialist in Translation')	у				II.	_	_	Literature	Awardees	100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_		Russian	Winner	admissions tests not required
						I	_		Russian	Awardees	100 points for AT in Russian
						II.	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3	English	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in English
		<u> </u>				I-II	_	Philology	_	\Winners, Awardees	100 points for AT in English
45.03.01 Philology	Foreign Languages of the Baltic Cultural Area (Latvian / Estonian,	05.04.0 1	1	Literature	80	I	_	_	Literature	\Winners, Awardees	admissions tests not required
	German) (with the qualifications	Geolog				II.	_		Literature	Winner	admissions tests not required
	'Teacher of Foreign Language (Latvian/Estonian/German)'/'Speciali	у				II.	_	_	Literature	Awardees	100 points for AT in Social Science
	st in Translation')					I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_	_	Russian	Winner	admissions tests not required
			ĺ			I	_	_	Russian	Awardees	100 points for AT in Russian

						II.	<b></b>		Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3	English	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in English
						I-II	-	Philology	_	\Winners, Awardees	100 points for AT in English
45.03.01 Philology	Italian Language and Literature (with the qualifications 'Teacher of the	05.04.0 1	1	Literature	80	I	_	_	Literature	\Winners, Awardees	admissions tests not required
	Italian Language and Literature'/'Specialist in Translation')	Geolog y				II.			Literature Literature	Winner Awardees	admissions tests not required 100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_	_	Russian	Winner	admissions tests not required
						I	_	_	Russian	Awardees	100 points for AT in Russian
						II.	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3 (subjects	Foreign language (English, or	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			elected by the applicant)	Spanish, or Chinese, or German, or French)		I-II		Philology		\Winners, Awardees	100 points for AT in Russian
45.03.01 Philology	Classical Philology (Ancient Greek and Latin; Ancient Greek and Roman	05.04.0 1	1	Literature	80	I	_	_	Literature	\Winners, Awardees	admissions tests not required
	Literature) (with the qualification	Geolog				II.	_	_	Literature	Winner	admissions tests not required
	'Teacher of Ancient Greek and Latin')	у				II.	_	_	Literature	Awardees	100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_	_	Russian	Winner	admissions tests not required
						I	_	_	Russian	Awardees	100 points for AT in Russian
						II.	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3 (subjects	Foreign language (English, or	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			elected by the applicant)	Spanish, or Chinese, or German, or French)		I-II	_	Philology	_	\Winners, Awardees	100 points for AT in Russian
45.03.01 Philology	German Language and Literature and Translation (with the qualifications	05.04.0 1	1	Literature	80	I		_	Literature	\Winners, Awardees	admissions tests not required
	'Teacher of the German Language and	Geolog				II.		_	Literature	Winner	admissions tests not required
	Literature'/'Specialist in Translation')	у				II.	_	_	Literature	Awardees	100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_		Russian	Winner	admissions tests not required
			1			I	_		Russian	Awardees	100 points for AT in Russian
						II.	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3	German	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in German
					I-II	_	Philology	_	\Winners, Awardees	100 points for AT in German	

45.03.01 Philology	Russian Philology (Russian Language	05.04.0	1	Literature	80	I	_		Literature	\Winners,	admissions tests not required
	and Literature) (with the qualification	1								Awardees	
	'Teacher of the Russian Language and	Geolog				II.	_	_	Literature	Winner	admissions tests not required
	Literature')	у				II.	_		Literature	Awardees	100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_	_	Russian	Winner	admissions tests not required
						I	_	_	Russian	Awardees	100 points for AT in Russian
						II.	_		Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3 (subjects	Foreign language (English, or	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			elected by	Spanish, or		I-II	_	Philology	_	\Winners,	100 points for AT in Russian
			the applicant)	Chinese, or German, or French)						Awardees	
45.03.01 Philology	French Language and Literature and Translation (with the qualifications	05.04.0 1	1	Literature	80	I	_	_	Literature	\Winners, Awardees	admissions tests not required
	'Teacher of the French Language and	Geolog				II.	_	_	Literature	Winner	admissions tests not required
	Literature'/'Specialist in Translation')	у				II.	_	_	Literature	Awardees	100 points for AT in Social Science
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			2	Russian	80	I	_	_	Russian	Winner	admissions tests not required
						I	_	_	Russian	Awardees	100 points for AT in Russian
						II.	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
						I-II	_	Philology	_	\Winners, Awardees	admissions tests not required
			3	French	85	I-II	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in French
						I-II	_	Philology	_	\Winners, Awardees	100 points for AT in French
45.03.02 Linguistics	English as a Foreign Language in	05.04.0	1	English	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
	Teaching and Communication (with	1				I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in English
	second qualifications 'Teacher of English'/'Specialist in Translation')	Geolog y				II.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in English
			2	Russian	80	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
			3	Literature	80	I-II	_	_	Literature	\Winners, Awardees	100 points for AT in Social Science
45.03.02 Linguistics	Balkan Studies and Comparative	05.04.0	1	Foreign language	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
-	Linguistics (with second qualification	1	(subjects	(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
	'Teacher of a Foreign Language'/'Specialist in Translation')	Geolog y	elected by the applicant)	Spanish, or German, or French)		II.		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			2	Russian	80	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
			3	Literature	80	I-II	_	_	Literature	\Winners, Awardees	100 points for AT in Social Science
45.03.02 Linguistics	Foreign Languages in Global	05.04.0	1	English	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
	Communication (with the	1				I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in English
	qualifications 'Teacher of a Foreign Language'/'Specialist in Translation')	Geolog y				II.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in English
			2	Russian	80	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
			3	Literature	80	I-II	_	_	Literature	\Winners, Awardees	100 points for AT in Social Science
45.03.02 Linguistics		05.04.0	1	Foreign language	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
-		1	(subjects	(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian

Engages   Septimized   Face   Septimized   Face   Septimized   Septi		Spanish Language (with the qualifications 'Teacher of the Spanish	Geolog v	elected by	Spanish, or German, or		II.	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
Section   Processing   Section   S			,								Tiwardees	
1.		Zangaage / Specialist in Translation /				80	I-II	_	_	Russian		100 points for AT in Russian
\$5.00   Linguistics   General and Applied Phonoxies   \$0.54 at   1   Figlish   \$5   1   Firegis Language   Foreign Language   Manufactor   \$1.00 points for AT   \$1.00 points				3	Literature	80	I-II	_	_	Literature	\Winners,	100 points for AT in Social
Cooking   Cook	45 03 02 Linguistics	General and Applied Phonetics	05 04 0	1	English	85	ī	_	Foreign Language	Foreign Language		admissions tests not required
Concept   Computer and Computational Engineers (English)   Computational Engineers (English) (English)   Computational Engineers (English) (English)   Computational Engineers (English) (English) (English)   Computational Engineers (English) (Englis	13.103.102 Emganetics	General and Applied Fliorieties	1	-	Zinginon.	05	ī		0 0 0			100 points for AT in English
Applied Computer and Computational Linguistics (English)   Computer and Computational Linguistics (English)   Computational Linguistics (English			_								\Winners,	100 points for AT in English
Applied Computer and Computer and Computer and Computer and Computer and Linguistics (English) of Centry 2   Centry 2   Centry 3   Applied Computer and Computer and Computational Linguistics (English) of Centry 2   Centry 3   Centry 4   Centry 4   Centry 4   Centry 5   Centry 5   Centry 5   Centry 5   Centry 6   Centry 7   Cen			·	2	Russian	80	I-II	_	_	Russian	\Winners,	100 points for AT in Russian
Computational Linguistics (Inglish)   Cooleg   Page   Cooleg   C				3	Literature	80	I-II	_	_	Literature		100 points for AT in Social Science
Condex	45.03.02 Linguistics	Applied, Computer and	05.04.0	1	English	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
Association   Section	-	Computational Linguistics (English)	1		-		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in English
Augustics   Russian as a Foreiga Language (1998)   1   1   1   1   1   1   1   1   1							II.	_	Foreign Language	Foreign Language		100 points for AT in English
Avanches				2	Russian	80	I-II	_	_	Russian		100 points for AT in Russian
the qualification Teacher of the Russian Language and Literature'   1				3	Literature	80	I-II	_	_	Literature		100 points for AT in Social Science
Russin Language and Literature   Variety of the composition of the c	45.03.02 Linguistics		05.04.0	1	Foreign language	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
Available			1				I		Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
Astronomical Communication (English) (with the qualifications Teacher of a Foreign Language)   Astronomication (English) (with the qualifications Teacher of a Foreign Language)   Astronomication (English) (with the qualifications Teacher of a Foreign Language)   Astronomication (English) (with the qualifications Teacher of a Foreign Language)   Astronomication (English) (with the qualifications Teacher of English)   Astronomication (E		Russian Language and Literature')	-	the	German, or		II.	_	Foreign Language	Foreign Language		100 points for AT in Russian
Section   Sect						80	I-II	_	_	Russian		100 points for AT in Russian
Theoretical and Experimental Linguistics (English) (with the qualification Teacher of a Foreign Language   Vinner   Linguistics   Color				3	Literature	80	I-II	_	_	Literature	\Winners,	100 points for AT in Social Science
Linguistics (English) (with the qualification "Teacher of a Foreign Language")	45 03 02 Linguistics	Theoretical and Experimental	05.04.0	1	English	85	ī	_	Foreign Language	Foreign Language		admissions tests not required
Qualification 'Teacher of a Foreign Language'   Poreign Language   Poreign Language   Awardees			1	-	8		ī	_				100 points for AT in English
Abardees		qualification 'Teacher of a Foreign	-					_			\Winners,	100 points for AT in English
As   As   As   As   As   As   As   As				2	Russian	80	I-II	_	_	Russian		100 points for AT in Russian
Communication (English) (with the qualifications Teacher of English/"Specialist in Translation")  English/"Specialist in Translation")  Auxiliary Specialist in Translation (English) (with the qualifications Teacher of the Russian Language and Literature')  English (Specialist in Translation)  English/"Specialist in Translation (Winners)  Auxiliary Specialist in Translation (Winners)  Auxiliary Special Language (Winners)  Auxiliary Special Language (Winners)  Auxiliary Special Language Special Language (Winner)  Auxiliary Special Language Spe				3	Literature		I-II	_	_	Literature	Awardees	100 points for AT in Social Science
qualifications "Teacher of English"/Specialist in Translation")    A	45.03.02 Linguistics	Theory and Practice of Intercultural	05.04.0	1	English	85	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
English'/Specialist in Translation')  Part of the qualification Teacher of the Russian Language and Literature')  Ausain Subjects of the Slavic World and Russian Language and Literature')  Ausain Subjects of the Slavic World and Russian Language and Literature')  Ausain Subjects of the Russian Subjects of the Russian Language and Literature')  Ausain Subjects of the Russian Subjects of the Russian Subjects of Geology of Spanish, or German, or French)  Ausain Subjects of Spanish, or German, or French Spanish, or French Spanish, or Spanish,			1				I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in English
As titerature								_	Foreign Language		Awardees	100 points for AT in English
45.03.02 Linguistics  Languages of the Slavic World and Russian as a Foreign Language ((with the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((with the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((in) the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((in) the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((in) the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((in) the qualification "Teacher of the Russian Language and Literature")  **Precious Language of the Slavic World and Russian as a Foreign Language ((in) the qualification "Teacher of the Russian Language of the Slavic World and Teacher of the Russian Language of the Slavic World and Teacher of the Russian Language of the Slavic World and Teacher of the Russian Language of the Slavic World and Teacher of the Russian Language of the Slavic World and Teacher of the Geology of the qualification studies (English and Nordic Languages)  **Translation and Translation Studies (English and Nordic Languages)**  **Translation and Translation Studies (English and Nordic Languages)**  **Translation and Teacher of the Russian Language of the State of the Qualification Studies (English and Nordic Languages)**  **Translation and Teacher of the Russian Language of the State of the Augusta Science of the Precip Language of the State of the Precip Language of the State of the Precip Language of the State of the State of the Precip Language of the State of the									_		Awardees	100 points for AT in Russian
Russian as a Foreign Language ( with the qualification 'Teacher of the Russian Language and Literature')  Percentage and Literature')  Russian Language and Literature')  Russian Russian Russian   Winners, Awardees    Russian   Winner    Russian											Awardees	
the qualification 'Teacher of the Russian Language and Literature')  ### Russian Language and Literature'  ### Russian Language Awardees  ### Russian Language Awardees  ### Russian Language Foreign Language  ### Russian Winners, Awardees  ### Awardees  ### Russian Winners, Awardees  ### Russian Winners, Awardees  ### Russian Winners, Awardees  ### Russian Language	45.03.02 Linguistics		05.04.0			85	I					admissions tests not required
Russian Language and Literature') $y$ the applicant) $y$ the applicant $y$ $y$ the applicant $y$			Contra				I					100 points for AT in Russian
Awardees  3 Literature 80 I-II — — Literature Winners, 100 points for AT in Awardees Science  45.00.00 LINGUISTICS AND LITERARY STUDIES — SPECIALIST'S DEGREE  45.05.01 Translation and Translation Studies (English and Nordic Languages) 1 English 90 I — Foreign Language Foreign Language Winner admissions tests no I — Foreign Language Foreign Language Awardees 100 points for AT in Admissions tests no I — Foreign Language Foreign Language Awardees 100 points for AT in Awardees I I — Foreign Language Foreign Language Foreign Language I I I — Foreign Language Foreign Language I I I — Foreign Language I I I I I I I I I I I I I I I I I I I				the	German, or		11.	_	Foreign Language	Foreign Language		100 points for AT in Russian
Awardees Science  45.00.00 LINGUISTICS AND LITERARY STUDIES— SPECIALIST'S DEGREE  45.05.01 Translation and Translation and Translation Studies (English and Nordic Languages)  Translation Studies  Tr				2	Russian					Russian	Awardees	100 points for AT in Russian
SPECIALIST'S DEGREE  45.05.01 Translation and Translation Studies  Trans				3	Literature	80	I-II		_	Literature		100 points for AT in Social Science
Translation Studies (English and Nordic Languages) 1 I I — Foreign Language Foreign Language Awardees 100 points for AT i		ITERARY STUDIES —										
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	45.05.01 Translation and	Translation and Translation Studies	05.04.0	1	English	90	I		Foreign Language	Foreign Language	Winner	admissions tests not required
Geolog II. — Foreign Language Foreign Language   Winners. 100 points for AT i	Translation Studies	(English and Nordic Languages)	1				I		Foreign Language	Foreign Language	Awardees	100 points for AT in English
y Awardees			Geolog y				II.		Foreign Language	Foreign Language		100 points for AT in English

			2	Russian	80	I-II	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
			3	Literature	80	I-II	-	_	Literature	\Winners, Awardees	100 points for AT in Social Science
46.00.00 HISTORY AND AR DEGREE	CHAEOLOGY — BACHELOR'S										
46.03.01 History	History	05.04.0	1	History	75	I	_	_	History	\Winners, Awardees	admissions tests not required
		Geolog				II.	_	_	History	Winner	admissions tests not required
		У				П.	_	_	History	Awardees	100 points for AT in Social Science
						III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
			2	Social Science	75	I-III	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
46.03.01 History	Archaeology	05.04.0 1	1	History	75	I	_	_	History	\Winners, Awardees	admissions tests not required
		Geolog				II.	_	_	History	Winner	admissions tests not required
		у				II.	_	_	History	Awardees	100 points for AT in Social Science
						III.	_	_	History	\Winners, Awardees	100 points for AT in Social Science
			2	Social Science	75	I-III	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
BACHELOR'S DEGREE	HICS, AND RELIGIOUS STUDIES —										
47.03.01 Philosophy	Philosophy	05.04.0 1 Geolog y	1	Social Science	75	I-II	_	_	Social Science	\Winners, Awardees	admissions tests not required
						I-II	_	Philosophy	—	\Winners, Awardees	admissions tests not required
						III.	_	Philosophy	_	\Winners, Awardees	100 points for AT in Social Science
						I-II	_	Law	_	\Winners, Awardees	admissions tests not required
			2	History	75	I	_	_	History	Winner	admissions tests not required
			(subjects elected by the			I	_	_	History	Awardees	100 points for AT in Social Science
			applicant)			II-III	_	_	History	\Winners, Awardees	100 points for AT in Social Science
				Foreign language	75	I	_	Foreign Language	Foreign Language	Winner	admissions tests not required
				(English, or		I	_	Foreign Language	Foreign Language	Awardees	100 points for AT in Russian
				Spanish, or Chinese, or German, or French)		II-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
7.03.03 Religious Studies	Religious Studies	05.04.0 1	1	History	75	I	_	_	History	\Winners, Awardees	admissions tests not required
		Geolog				II.			History	Winner	admissions tests not required
		У				II.	_	_	History	Awardees	100 points for AT in Social Science
		2 (subjects	Social Science	75	I	_		Social Science	\Winners, Awardees	admissions tests not required	
			elected by the			II-III	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			applicant)			I-III	_	Philosophy	_	\Winners, Awardees	100 points for AT in Social Science
					I-III	_	Law	_	\Winners, Awardees	100 points for AT in Social Science	

				Foreign language (English, or	75	I	_	Foreign Language	Foreign Language	\Winners, Awardees	admissions tests not required	
				Spanish, or Chinese, or German, or French)		II-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian	
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
04.00.00 Theology — BACHELO												
48.06.01 Theology	Christian Theology	05.04.0	1	History	75	I-III	_	_	History	\Winners, Awardees	admissions tests not required	
		Geolog y	2 (subjects	Social Science	75	I-III	_	—	Social Science	\Winners, Awardees	admissions tests not required	
			elected by the applicant)			I-III	_	Philosophy	_	\Winners, Awardees	admissions tests not required	
			аррисант)	Foreign language (English, or	75	I	_	Foreign Language	Foreign Language	\Winners, Awardees	admissions tests not required	
				Spanish, or Chinese, or German, or French)		II-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian	
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
48.06.01 Theology	Jewish Theology	05.04.0 1	1	History	75	I-III	_	_	History	\Winners, Awardees	admissions tests not required	
		Geolog y	2 (subjects	Social Science	75	I-III	_	_	Social Science	\Winners, Awardees	admissions tests not required	
			elected by the			I-III	_	Philosophy	_	\Winners, Awardees	admissions tests not required	
			applicant)	Foreign language (English, or	75	I	_	Foreign Language	Foreign Language	\Winners, Awardees	admissions tests not required	
					Spanish, or Chinese, or German, or French)		II-III		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
49.00.00 PHYSICAL EDUCATI DEGREE	ON AND SPORTS — BACHELOR'S											
49.03.01 Physical Education	Physical Culture and Sports (with the qualification 'Specialist in Fitness	05.04.0 1	1	Biology	75	I-III	_	_	Biology	\Winners, Awardees	admissions tests not required	
	(Fitness Coach)')	Geolog y	2	Physical Education	_			No	particular rights granted			
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
50.00.00 ART STUDIES — BAC		05.04.0	1	History	75	т т	+		History	Winner	adminsions to the second of	
50.03.01 Arts and Humanities	Arts and Humanities	05.04.0 1 Geolog	1	History	75	I	_		History History	Winner Awardees	admissions tests not required 100 points for AT in Social Science	
		y				II-III	_	_	History	\Winners, Awardees	100 points for AT in Social Science	
			2 (subjects elected by the applicant)	Foreign language (English, or Spanish, or Chinese, or German, or French)	75	I-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian	
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian	
50.03.03 History of Arts	History of Arts	05.04.0	1	History	75	I-II			History	Winner	admissions tests not required	
		1				I-II	_	_	History	Awardees	100 points for AT in Social Science	

		Geolog y				I-II	_	Academic Drawing, Painting,	_	Winner	admissions tests not required
								Composition, History of Arts and Culture			
						I-II	_	Academic Drawing, Painting, Composition, History of Arts and Culture	_	Awardees	100 points for AT in Social Science
						I-II	_	_	Art Studies	Winner	admissions tests not required
						I-II	_	_	Art Studies	Awardees	100 points for AT in Social Science
						I-II	_	_	Fine and Applied Arts	Winner	admissions tests not required
						I-II	_	_	Fine and Applied Arts	Awardees	100 points for AT in Social Science
						I-II	_	_	Arts	Winner	admissions tests not required
			_			I-II	_	_	Arts	Awardees	100 points for AT in Social Science
			2 (subjects elected by the applicant)	Foreign language (English, or Spanish, or Chinese, or German, or French)	75	I-III		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
51.00.00 CULTURAL STUDIES PROJECTS — BACHELOR'S D	EGREE										
51.03.01 Cultural Studies	Cultural Studies	05.04.0	1	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
		Geolog				II.	_		Social Science	Winner	admissions tests not required
		у				II.	_	Di li i	Social Science	Awardees	100 points for AT in Social Science
						I	_	Philosophy	_	\Winners, Awardees	admissions tests not required
						II.	_	Philosophy	_	Winner	admissions tests not required
						II.	_	Philosophy	_	Awardees	100 points for AT in Social Science
						III.	_	Philosophy	_	\Winners, Awardees	100 points for AT in Social Science
						I	_	Law	_	\Winners, Awardees	admissions tests not required
						II.	_	Law Law	_	Winner Awardees	admissions tests not required
				TT' -	75	II.	_				100 points for AT in Social Science
			2 (subjects	History	75	I-III	_	- Andreis Bresie	History	\Winners, Awardees	100 points for AT in Social Science
			elected by the applicant)			I-II	_	Academic Drawing, Painting, Composition, History of Arts and Culture	_	\Winners, Awardees	100 points for AT in Social Science
				Foreign language (English, or Spanish, or Chinese, or German, or French)	75	I-III	_	Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
	Museology and Conservation of Cultural and Natural Heritage Sites	05.04.0 1	1	History	75	I	_		History	\Winners, Awardees	admissions tests not required

51.03.04 Museology and		Geolog				II.	I —	_	History	Winner	admissions tests not required
Conservation of Cultural and Natural Heritage Sites		у				II.	_	_	History	Awardees	100 points for AT in Social Science
						I-II	_	Academic Drawing, Painting, Composition, History of Arts and Culture	_	Winner	admissions tests not required
						I-II	_	Academic Drawing, Painting, Composition, History of Arts and Culture	_	Awardees	100 points for AT in Social Science
			2 (subjects	Social Science	75	I	_	_	Social Science	\Winners, Awardees	admissions tests not required
			elected by the			II-III	_	_	Social Science	\Winners, Awardees	100 points for AT in Social Science
			applicant)			I-III	_	Philosophy	_	\Winners, Awardees	100 points for AT in Social Science
						I-III	_	Law	_	\Winners, Awardees	100 points for AT in Social Science
				Foreign language (English, or Spanish, or Chinese, or German, or French)	75	I-III		Foreign Language	Foreign Language	\Winners, Awardees	100 points for AT in Russian
			3	Russian	75	I-III	_	_	Russian	\Winners, Awardees	100 points for AT in Russian
DEGREE	AND LITERARY ACTIVITY — SPEC										
52.05.01 Acting	Stage and Film Actor	05.04.0	1	Literature	_				particular rights granted		
		1	2	Russian	_	1			particular rights granted		
		Geolog y	3 4	Stage Speech Plastique and Vocals					particular rights granted particular rights granted		
			5	Acting Skills	_	+		No	particular rights granted		
53.00.00 MUSICAL ART — BAG	CHELOR'S DEGREE					1			F		
53.03.02 Instrumental Music	Instrumental Performance: Organ, Harpsichord and Carillon	05.04.0 1	1 2	Literature Russian	_				particular rights granted particular rights granted		
		Geolog	3	Solo playing	_				particular rights granted		
		У	4	Solfège	_			No	particular rights granted		
		1	5	Harmony					particular rights granted		
53.03.02 Instrumental Music	Instrumental Performance: Violin	05.04.0	1	Literature	_				particular rights granted		
		Casla	2	Russian		1			particular rights granted		
		Geolog y	3 4	Solo playing		+			particular rights granted		
		,	5	Solfège Harmony		1			particular rights granted particular rights granted		
53.03.03 Vocal Performance	Academic Singing	05.04.0	1	Literature		+			particular rights granted		
		1	2	Russian	_	1			particular rights granted		
		Geolog	3	Solo playing	_				particular rights granted		
		у	4	Basics of the Artistry	_				particular rights granted		
			5	Theory of Music and Solfège	_			No	particular rights granted		
54.00.00 FINE AND APPLIED A	RTS — BACHELOR'S DEGREE										
54.03.01 Design	Graphic Design	05.04.0	1	Literature					particular rights granted		
		1	2	Russian					particular rights granted		
	Geolog	3	Composition	_				particular rights granted			
				I D .	1	1		No	particular rights granted		
		У	4	Drawing							
		Ť	5	Painting				No	particular rights granted		
54.03.01 Design	Environmental Design	05.04.0						No No			

		Geolog	2	Composition		1			No	particular rights granted				
		Geolog	4	Drawing										
		У			_					particular rights granted				
#10000 h W 11		0.504.0	5	Painting	_					particular rights granted				
54.03.02 Applied Arts and	Applied Art	05.04.0	1	Literature	_		No particular rights granted							
Vernacular Arts			2	Russian						particular rights granted				
		Geolog	3	Drawing						particular rights granted				
		У	4	Painting	_					particular rights granted				
			5	Composition	_					particular rights granted				
54.03.04 Restoration	Restoration of Fine and Applied Art	05.04.0	1	Literature	_					particular rights granted				
	Objects	1	2	Russian	_					particular rights granted				
		Geolog	3	Painting	_				No	particular rights granted				
		У	4	Drawing	_				No	particular rights granted				
54.00.00 FINE AND APPLIED	O ARTS — SPECIALIST'S DEGREE													
54.05.02 Painting	Easel Painting	05.04.0	1	Literature	_					particular rights granted				
		1	2	Russian	_				No	particular rights granted				
		Geolog	3	Painting	_				No	particular rights granted				
		У	4	Drawing	_				No	particular rights granted				
			5	Composition	_				No	particular rights granted				
54.05.03 Graphic Arts	Animated Cartoon Artist	05.04.0	1	Literature	_				No	particular rights granted				
		1	2	Russian	_					particular rights granted				
		Geolog	3	Painting	_				No	particular rights granted				
		У	4	Drawing	_				No	particular rights granted				
			5	Composition	_				No	particular rights granted				
58.00.00 ASIAN AND AFRICA DEGREE	AN STUDIES — BACHELOR'S													
58.03.01 Asian and African Studies	Arabic Philology (with second qualification 'Specialist in	05.04.0 1	1	History	75	I	_		_	History	\Winners, Awardees	admissions tests not required		
	Translation')	Geolog				II.	_		_	History	Winner	admissions tests not required		
	Hebraismtics (with second qualification 'Specialist in	у				II.	_		_	History	Awardees	100 points for AT in Social Science		
	Translation') Iranian Philology (with second					III.	_		_	History	\Winners, Awardees	100 points for AT in Social Science		
	qualification 'Specialist in Translation')		2	Russian	75	I-II	_		_	Russian	\Winners, Awardees	100 points for AT in Russian		

History of the Caucasus (Georgia)	3	Foreign language	75	I-III	_	Foreign Language	Foreign Language	\Winners,	100 points for AT in Russian
(with second qualification 'Specialist	(subjects	(English, or						Awardees	
in Translation')	elected by	Spanish, or						1	
Turkic Philology (with second	the	Chinese, or							
qualification 'Specialist in	applicant)	German, or							
Translation')		French)							
Languages, History and Culture of									
Ancient Egypt (Egyptology)(with									
second qualification 'Specialist in									
Translation')									
Indo-Aryan Philology (Hindi,									
Sanskrit) (with second qualification								1	
'Specialist in Translation')									
Indonesian-Malay Philology (with									
second qualification 'Specialist in								1	
Translation')									
History of Vietnam (with second									
qualification 'Specialist in								1	
Translation')									
History of China (with second									
qualification 'Specialist in									
Translation')								1	
History of Korea (with second									
qualification 'Specialist in								1	
Translation')									
History of Japan (with second								1	1
qualification 'Specialist in									
Translation')								1	1
Thai Philology (with the study of Thai									
and Burmese) (with second									
qualification 'Specialist in									
Translation')								1	
Tibetan-Mongolian Philology (with								1	
second qualification 'Specialist in								1	
Translation')								1	1
Languages and Culture of east Africa									
(Swahili) (with second qualification									
'Specialist in Translation')									

Annex 8 to the Regulation for Admissions to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

## Correspondence of fields of study and specialities, within which St Petersburg University conducts admissions to degree programmes, to the All-Ukrainian School Student Olympiad profiles international Olympiads in general education subjects.

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
English	37.03.02 Conflictology	Bachelor's
		degree
		programmes
English	39.03.01 Sociology	Bachelor's
		degree
		programmes
English	39.03.02 Social Work	Bachelor's
		degree
		programmes
English	40.03.01 Law	Bachelor's
		degree
		programmes
English	41.03.05 International Relations	Bachelor's
		degree
		programmes
English	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
English	42.03.02 Journalism	Bachelor's
_		degree
		programmes
English	43.03.02 Tourism	Bachelor's
		degree
		programmes
English	45.03.01 Philology	Bachelor's
_		degree
		programmes
English	45.03.02 Linguistics (degree programmes 'English as a Foreign	Bachelor's
	Language in Teaching and Communication (with second	degree
	qualification 'Teacher of English'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
_	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'Foreign Languages in	Bachelor's
	Global Communication (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
English	45.03.02 Linguistics (degree programme 'General and Applied Phonetics')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Applied, Computer and Computational Linguistics (English)')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Russian as a Foreign Language ( (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Theoretical and Experimental Linguistics (English) (with second qualification 'Teacher of a Foreign Language')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Theory and Practice of Intercultural Communication (English) (with second qualification 'Teacher of a Foreign Language')'Specialist in Translation')	Bachelor's degree programmes
English	45.03.02 Linguistics (degree programme 'Languages of the Slavic World and Russian as a Foreign Language (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
English	45.05.01 Translation and Translation Studies	Specialist's degree programme
English	47.03.01 Philosophy	Bachelor's degree programmes
English	47.03.03 Religious Studies	Bachelor's degree programmes
English	50.03.01 Arts and Humanities	Bachelor's degree programmes
English	50.03.03 History of Arts	Bachelor's degree programmes
English	51.03.01 Cultural Studies	Bachelor's degree programmes
English	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
English	58.03.01 Asian and African Studies	Bachelor's degree programmes
Astronomy	01.03.01 Mathematics	Bachelor's degree programmes
Astronomy	01.03.03 Applied Mathematics and Informatics (degree programme 'Modern Programming')	Bachelor's degree programmes
Astronomy	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Astronomy	03.03.01 Applied Mathematics and Physics	Bachelor's degree programmes
Astronomy	03.03.02 Physics	Bachelor's degree programmes
Astronomy	03.05.01 Astronomy	Specialist's degree programme
Biology	05.03.02 Geography	Bachelor's degree programmes
Biology	05.03.03 Cartography and Geoinformatics	Bachelor's degree programmes
Biology	05.03.04 Hydrometeorology	Bachelor's degree programmes
Biology	05.03.06 Ecology and Nature Management	Bachelor's degree programmes
Biology	06.03.01Biology	Bachelor's degree programmes
Biology	06.03.02 Soil Science	Bachelor's degree programmes
Biology	31.05.01 General Medicine	Specialist's degree programme
Biology	31.05.03 Dental Medicine	Specialist's degree programme
Biology	33 05 01 Pharmacy	Specialist's degree programme
Biology	34.02.01 Nursing	Bachelor's degree programmes
Biology	37.03.01 Psychology	Bachelor's degree programmes
Biology	37.05.01 Clinical Psychology	Specialist's degree programme
Biology	37.05.02 Employment Psychology	Specialist's degree programme
Geography	05.03.01 Geology	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Geography	05.03.02 Geography	Bachelor's degree programmes
Geography	05.03.03 Cartography and Geoinformatics	Bachelor's degree programmes
Geography	05.03.04 Hydrometeorology	Bachelor's degree programmes
Geography	05.03.06 Ecology and Nature Management	Bachelor's degree programmes
Geography	21.03.01 Petroleum Engineering	Bachelor's degree programmes
Geography	21/03/2002 Land Management and Cadastres	Bachelor's degree programmes
Geography	37.03.02 Conflictology	Bachelor's degree programmes
Geography	38.03.01 Economics	Bachelor's degree programmes
Geography	41.03.02 Russian Region Studies	Bachelor's degree programmes
Geography	41.03.05 International Relations	Bachelor's degree programmes
Geography	43.03.02 Tourism	Bachelor's degree programmes
Geography	46.03.01 History	Bachelor's degree programmes
Geography	47.03.01 Philosophy	Bachelor's degree programmes
Geography	47.03.03 Religious Studies	Bachelor's degree programmes
Geography	51.03.01 Cultural Studies	Bachelor's degree programmes
Geography	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
Informatics	01.03.01 Mathematics	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Informatics	01.03.02Applied Mathematics and Informatics	Bachelor's degree programmes
Informatics	02.03.01 Mathematics and Computer Sciences (programme 'Data Sciences')	Bachelor's degree programmes
Informatics	02.03.02 Fundamental Informatics and Information Technology	Bachelor's degree programmes
Informatics	02.03.03 Information Systems Administration and Mathematical Support	Bachelor's degree programmes
Informatics	05.03.01 Geology	Bachelor's degree programmes
Informatics	05.03.02 Geography	Bachelor's degree programmes
Informatics	05.03.03 Cartography and Geoinformatics	Bachelor's degree programmes
Informatics	05.03.04 Hydrometeorology	Bachelor's degree programmes
Informatics	05.03.06 Ecology and Nature Management	Bachelor's degree programmes
Informatics	06.03.02 Soil Science	Bachelor's degree programmes
Informatics	09.03.03 Applied Informatics	Bachelor's degree programmes
Informatics	09.03.04 Software Engineering	Bachelor's degree programmes
Informatics	21.03.01 Petroleum Engineering	Bachelor's degree programmes
Informatics	21/03/2002 Land Management and Cadastres	Bachelor's degree programmes
Informatics	27.03.03 Systems Analysis and Control	Bachelor's degree programmes
Informatics	37.03.02 Conflictology	Bachelor's degree programmes
Informatics	38.03.01 Economics	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Informatics	38.03.03 Personnel Management	Bachelor's
		degree
		programmes
Informatics	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Informatics	39.03.01 Sociology	Bachelor's
		degree
		programmes
Informatics	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Informatics	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Informatics	51.03.01 Cultural Studies	Bachelor's
momates	51.05.01 Cultural Studies	degree
		programmes
Informatics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
informatics	Heritage Sites	degree
	Tierrage Sites	programmes
Spanish	37.03.02 Conflictology	Bachelor's
Spanish	37.03.02 Connictology	degree
		programmes
Spanish	39.03.01 Sociology	Bachelor's
Spanish	39.03.01 Sociology	degree
		_
Spanish	39.03.02 Social Work	programmes Bachelor's
Spanish	39.03.02 Social Work	
		degree
C	40.02.01 L	programmes
Spanish	40.03.01 Law	Bachelor's
		degree
~ · ·	44.02.077	programmes
Spanish	41.03.05 International Relations	Bachelor's
		degree
a	12.02.04.4.1	programmes
Spanish	42.03.01 Advertising and Public Relations	Bachelor's
		degree
a	10 00 00 Y	programmes
Spanish	42.03.02 Journalism	Bachelor's
		degree
~	42.22.22	programmes
Spanish	43.03.02 Tourism	Bachelor's
		degree
		programmes
Spanish	45.03.01 Philology	Bachelor's
		degree
		programmes
Spanish	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Spanish	45.03.02 Linguistics (degree programme 'Spanish' (with second qualifications 'Teacher of the Spanish'/'Specialist in Translation')	Bachelor's degree programmes
Spanish	45.03.02 Linguistics (degree programme 'Russian as a Foreign Language ( (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
Spanish	45.03.02 Linguistics (degree programme 'Languages of the Slavic World and Russian as a Foreign Language (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
Spanish	47.03.01 Philosophy	Bachelor's degree programmes
Spanish	47.03.03 Religious Studies	Bachelor's degree programmes
Spanish	50.03.01 Arts and Humanities	Bachelor's degree programmes
Spanish	50.03.03 History of Arts	Bachelor's degree programmes
Spanish	51.03.01 Cultural Studies	Bachelor's degree programmes
Spanish	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
Spanish	58.03.01 Asian and African Studies	Bachelor's degree programmes
Mathematics	01.03.01 Mathematics	Bachelor's degree programmes
Mathematics	01.03.02Applied Mathematics and Informatics	Bachelor's degree programmes
Mathematics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's degree programmes
Mathematics	01.05.01 Fundamental Mathematics and Mechanics	Specialist's degree programme
Mathematics	02.03.01 Mathematics and Computer Sciences	Bachelor's degree programmes
Mathematics	02.03.02 Fundamental Informatics and Information Technology	Bachelor's degree programmes
Mathematics	03.03.01 Applied Mathematics and Physics	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Mathematics	03.03.02 Physics	Bachelor's
		degree
		programmes
Mathematics	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
		programmes
Mathematics	05.03.01 Geology	Bachelor's
		degree
		programmes
Mathematics	05.03.02 Geography	Bachelor's
		degree
		programmes
Mathematics	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Mathematics	05.03.04 Hydrometeorology	Bachelor's
	, ,	degree
		programmes
Mathematics	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Mathematics	06.03.02 Soil Science	Bachelor's
	0010010201100101	degree
		programmes
Mathematics	09.03.03 Applied Informatics (degree programme: Applied	Bachelor's
Tradition delices	Informatics in Arts and Humanities)	degree
	informatios in 711ts and 11amantios)	programmes
Mathematics	21.03.01 Petroleum Engineering	Bachelor's
Wathematics	21.03.01 Fedoleum Engineering	degree
		programmes
Mathematics	21.03.02 Land Management and Cadastres	Bachelor's
Wathematics	21.05.02 Land Wanagement and Cadastres	degree
		programmes
Mathematics	27.03.03 Systems Analysis and Control	Bachelor's
wiamemanes	27.05.05 Systems Analysis and Control	degree
		programmes
Mathematics	37.03.01 Psychology	Bachelor's
wianicilianes	37.03.01 F Sychology	degree
		_
Mathematics	37.03.02 Conflictology	programmes Bachelor's
Maniemanes	57.05.02 Connictorogy	degree
Mathematics	27.05.01 Clinical Dayabalassy	programmes Specialist's
wiamemanes	37.05.01 Clinical Psychology	Specialist's
		degree
N.C. dans and the	27.05.02 Family and D. 1.1	programme
Mathematics	37.05.02 Employment Psychology	Specialist's
		degree
		programme
Mathematics	38.03.01 Economics	Bachelor's
		degree
		programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Mathematics	38.03.02 Management	Bachelor's
		degree
		programmes
Mathematics	38.03.03 Personnel Management	Bachelor's
		degree
		programmes
Mathematics	38.03.04 Public and Municipal Administration	Bachelor's
		degree
		programmes
Mathematics	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Mathematics	39.03.01 Sociology	Bachelor's
		degree
		programmes
Mathematics	42.03.01 Advertising and Public Relations	Bachelor's
	Č	degree
		programmes
Mathematics	47.03.01 Philosophy	Bachelor's
	· · · · · · · · · · · · · · · · · · ·	degree
		programmes
Mathematics	47.03.03 Religious Studies	Bachelor's
Tradio made 5	17705105 Rengrous Studies	degree
		programmes
Mathematics	51.03.01 Cultural Studies	Bachelor's
Wathematics	31.03.01 Cultural Studies	degree
		programmes
Mathematics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Wathematics	Heritage Sites	degree
	Tierrage Sites	programmes
German	37.03.02 Conflictology	Bachelor's
German	37.03.02 Connictology	degree
C	20.02.01 \$:-1	programmes
German	39.03.01 Sociology	Bachelor's
		degree
Campan	20.02.02 Social Worls	programmes Bachelor's
German	39.03.02 Social Work	
		degree
C	40.02.01 I	programmes
German	40.03.01 Law	Bachelor's
		degree
	41.02.05 I	programmes
German	41.03.05 International Relations	Bachelor's
		degree
~		programmes
German	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
German	42.03.02 Journalism	Bachelor's
		degree
		programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
German	43.03.02 Tourism	Bachelor's
		degree
		programmes
German	45.03.01 Philology	Bachelor's
		degree
		programmes
German	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
German	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
German	45.03.02 Linguistics (degree programme 'Languages of the	Bachelor's
	Slavic World and Russian as a Foreign Language (with second	degree
	qualification 'Teacher of the Russian Language and Literature')	programmes
German	47.03.01 Philosophy	Bachelor's
	, which is a massive of the same of the sa	degree
		programmes
German	47.03.03 Religious Studies	Bachelor's
German	17.05.05 Religious studies	degree
		programmes
German	50.03.01 Arts and Humanities	Bachelor's
German	50.05.01 7tts and 11umanties	degree
		programmes
German	50.03.03 History of Arts	Bachelor's
German	50.05.05 History of Arts	degree
		programmes
German	51.03.01 Cultural Studies	Bachelor's
German	31.03.01 Cultural Studies	degree
		programmes
German	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
German	Heritage Sites	
	Heritage Sites	degree
C	58.03.01 Asian and African Studies	programmes
German	58.05.01 Asian and African Studies	Bachelor's
		degree
D	27 02 01 D 1 . 1	programmes
Russian Language and	37.03.01 Psychology	Bachelor's
Literature		degree
D	27.02.02.02.03.4.1	programmes
Russian Language and	37.03.02 Conflictology	Bachelor's
Literature		degree
D	27.05.01.01°. 1.1.	programmes
Russian Language and	37.05.01 Clinical Psychology	Specialist's
Literature		degree
D	27.05.02 F P 1.1.	programme
Russian Language and	37.05.02 Employment Psychology	Specialist's
Literature		degree
	40.00.04 X	programme
Russian Language and	40.03.01 Law	Bachelor's
Literature		degree
		programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Russian Language and	41.03.04 Political Science	Bachelor's
Literature		degree
		programmes
Russian Language and	42.03.01 Advertising and Public Relations	Bachelor's
Literature		degree
Dussian I anaugus and	42.03.02 Journalism	programmes Bachelor's
Russian Language and Literature	42.03.02 Journalism	
Literature		degree
Dussian Language and	45.03.01 Philology	programmes Bachelor's
Russian Language and Literature	45.05.01 Filliology	degree
Literature		programmes
Russian Language and	45.03.02 Linguistics	Bachelor's
Literature	45.05.02 Elliguistics	degree
Literature		programmes
Russian Language and	45.05.01 Translation and Translation Studies	Specialist's
Literature	45.05.01 Translation and Translation Studies	degree
Enterature		programme
Russian Language and	47.03.01 Philosophy	Bachelor's
Literature	47.03.01 1 miosophy	degree
Enterature		programmes
Russian Language and	47.03.03 Religious Studies	Bachelor's
Literature	+7.03.03 Religious Studies	degree
Enterature		programmes
Russian Language and	50.03.01 Arts and Humanities	Bachelor's
Literature		degree
		programmes
Russian Language and	50.03.03 History of Arts	Bachelor's
Literature		degree
		programmes
Russian Language and	51.03.01 Cultural Studies	Bachelor's
Literature		degree
		programmes
Russian Language and	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Literature	Heritage Sites	degree
		programmes
Physics	01.03.01 Mathematics	Bachelor's
		degree
		programmes
Physics	01.03.03 Applied Mathematics and Informatics (degree	Bachelor's
	programme 'Modern Programming')	degree
		programmes
Physics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
		degree
		programmes
Physics	01.05.01 Fundamental Mathematics and Mechanics (programme	Specialist's
	'Fundamental Mechanics')	degree
		programme
Physics	02.03.01 Mathematics and Computer Sciences (programme 'Data	Bachelor's
	Sciences')	degree
		programmes

Physics03.03.01 Applied Mathematics and PhysicsPhysics03.03.02 PhysicsPhysics04.03.02 Chemistry, Physics and Material MemoryPhysics05.03.01 GeologyPhysics21.03.01 Petroleum EngineeringPhysics21/03/2002 Land Management and CadastresPhysics27.03.03 Systems Analysis and ControlFrench37.03.02 ConflictologyFrench39.03.01 SociologyFrench40.03.01 LawFrench41.03.05 International RelationsFrench42.03.01 Advertising and Public Relations	
Physics 03.05.01 Astronomy  Physics 04.03.02 Chemistry, Physics and Material Memory Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes Bachelor's degree programmes Specialist's degree programme echanics Bachelor's
Physics 03.05.01 Astronomy  Physics 04.03.02 Chemistry, Physics and Material Me Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's degree programmes Specialist's degree programme echanics Bachelor's
Physics 03.05.01 Astronomy  Od.03.02 Chemistry, Physics and Material Memory  Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree programmes Specialist's degree programme echanics Bachelor's
Physics 04.03.02 Chemistry, Physics and Material Meterial	programmes Specialist's degree programme echanics Bachelor's
Physics 04.03.02 Chemistry, Physics and Material Me Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Specialist's degree programme echanics Bachelor's
Physics 04.03.02 Chemistry, Physics and Material Me Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree programme echanics Bachelor's
Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programme echanics Bachelor's
Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	echanics Bachelor's
Physics 05.03.01 Geology  Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	
Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	i
Physics 21.03.01 Petroleum Engineering  21/03/2002 Land Management and Cadastres  27.03.03 Systems Analysis and Control  27.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
Physics 21.03.01 Petroleum Engineering  Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
Physics 21.03.01 Petroleum Engineering  21/03/2002 Land Management and Cadastres  27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
Physics 21/03/2002 Land Management and Cadastres  27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
Physics 21/03/2002 Land Management and Cadastres  Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
Physics 21/03/2002 Land Management and Cadastres  27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
Physics 27.03.03 Systems Analysis and Control  French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	
French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
French 37.03.02 Conflictology  French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
French 39.03.01 Sociology  French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	degree
French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	programmes
French 39.03.02 Social Work  French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
French 40.03.01 Law  French 41.03.05 International Relations	degree
French 40.03.01 Law  French 41.03.05 International Relations	programmes
French 40.03.01 Law  French 41.03.05 International Relations	Bachelor's
French 41.03.05 International Relations	degree
French 41.03.05 International Relations	programmes
French 41.03.05 International Relations	Bachelor's
	degree
	programmes
	Bachelor's
French 42.03.01 Advertising and Public Relations	degree
French 42.03.01 Advertising and Public Relations	programmes
72.03.01 Advertising and 1 uone Relations	Bachelor's
l	i aearee
French 42.03.02 Journalism	degree
72.03.02 Journalish	programmes
	programmes Bachelor's
French 43.03.02 Tourism	programmes Bachelor's degree
45.05.02 TOURISH	programmes Bachelor's degree programmes
	programmes Bachelor's degree programmes Bachelor's
French 45.03.01 Philology	programmes Bachelor's degree programmes Bachelor's degree
French 45.03.01 Philology	programmes Bachelor's degree programmes Bachelor's degree programmes
	programmes Bachelor's degree programmes Bachelor's degree

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
French	45.03.02 Linguistics (degree programmes 'Balkan Studies and Comparative Linguistics (with second qualification 'Teacher of a Foreign Language'/'Specialist in Translation')	Bachelor's degree programmes
French	45.03.02 Linguistics (degree programme 'Russian as a Foreign Language ( (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
French	45.03.02 Linguistics (degree programme 'Languages of the Slavic World and Russian as a Foreign Language (with second qualification 'Teacher of the Russian Language and Literature')	Bachelor's degree programmes
French	47.03.01 Philosophy	Bachelor's degree programmes
French	47.03.03 Religious Studies	Bachelor's degree programmes
French	50.03.01 Arts and Humanities	Bachelor's degree programmes
French	50.03.03 History of Arts	Bachelor's degree programmes
French	51.03.01 Cultural Studies	Bachelor's degree programmes
French	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
French	58.03.01 Asian and African Studies	Bachelor's degree programmes
Chemistry	04.03.01 Chemistry	Bachelor's degree programmes
Chemistry	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's degree programmes
Chemistry	05.03.01 Geology	Bachelor's degree programmes
Chemistry	05.03.06 Ecology and Nature Management	Bachelor's degree programmes
Chemistry	06.03.02 Soil Science	Bachelor's degree programmes
Chemistry	21.03.01 Petroleum Engineering	Bachelor's degree programmes
Chemistry	21.03.02 Land Management and Cadastres	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Chemistry	27.03.03 Systems Analysis and Control	Bachelor's degree
Chemistry	31.05.01 General Medicine	Specialist's degree programme
Chemistry	31.05.03 Dental Medicine	Specialist's degree programme
Chemistry	33 05 01 Pharmacy	Specialist's degree programme
Chemistry	34.02.01 Nursing	Bachelor's degree programmes
Ecology	05.03.06 Ecology and Nature Management	Bachelor's degree programmes
Ecology	06.03.02 Soil Science	Bachelor's degree programmes
Ecology	37.03.02 Conflictology	Bachelor's degree programmes
Ecology	39.03.02 Social Work	Bachelor's degree programmes
Ecology	47.03.01 Philosophy	Bachelor's degree programmes
Ecology	47.03.03 Religious Studies	Bachelor's degree programmes
Ecology	51.03.01 Cultural Studies	Bachelor's degree programmes
Ecology	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes
Economics	37.03.02 Conflictology	Bachelor's degree programmes
Economics	38.03.01 Economics	Bachelor's degree programmes
Economics	38.03.02 Management	Bachelor's degree programmes
Economics	38.03.03 Personnel Management	Bachelor's degree programmes

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Main field of study and speciality titles	Level of education
Economics	38.03.04 Public and Municipal Administration	Bachelor's degree programmes
Economics	38.03.05 Business Informatics	Bachelor's degree programmes
Economics	39.03.01 Sociology	Bachelor's degree programmes
Economics	39.03.02 Social Work	Bachelor's degree programmes
Economics	41.03.05 International Relations	Bachelor's degree programmes
Economics	47.03.01 Philosophy	Bachelor's degree programmes
Economics	47.03.03 Religious Studies	Bachelor's degree programmes
Economics	51.03.01 Cultural Studies	Bachelor's degree programmes
Economics	51.03.04 Museology and Conservation of Cultural and Natural Heritage Sites	Bachelor's degree programmes

Annex 9 to the Regulation for Admissions to Higher Education Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's Programmes) at St Petersburg University in 2024

Correspondence of fields of study and specialities, within which St Petersburg University conducts admissions to degree programmes, to the profiles of republican school students Olympiads of the Donetsk People's Republic, the Lugansk People's Republic.

Olympiad profile	Main field of study and speciality titles	Level of education
English	37.03.02 Conflictology	Bachelor's
	<i>C.</i> 7	degree
		programmes
English	39.03.01 Sociology	Bachelor's
		degree
		programmes
English	39.03.02 Social Work	Bachelor's
		degree
		programmes
English	40.03.01 Law	Bachelor's
		degree
		programmes
English	41.03.05 International Relations	Bachelor's
		degree
		programmes
English	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
English	42.03.02 Journalism	Bachelor's
		degree
		programmes
English	43.03.02 Tourism	Bachelor's
		degree
		programmes
English	45.03.01 Philology	Bachelor's
		degree
		programmes
English	45.03.02 Linguistics (degree programmes 'English as a Foreign	Bachelor's
	Language in Teaching and Communication (with second qualification	degree
	'Teacher of English'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'Foreign Languages in	Bachelor's
	Global Communication (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'General and Applied	Bachelor's
	Phonetics')	degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
English	45.03.02 Linguistics (degree programme 'Applied, Computer and	Bachelor's
U	Computational Linguistics (English)')	degree
		programmes
English	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
Liigiisii	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
English	45.03.02 Linguistics (degree programme 'Theoretical and	Bachelor's
Eligiisii		
	Experimental Linguistics (English) (with second qualification	degree
D 1' 1	'Teacher of a Foreign Language')	programmes
English	45.03.02 Linguistics (degree programme 'Theory and Practice of	Bachelor's
	Intercultural Communication (English) (with second qualification	degree
	'Teacher of a Foreign Language'/'Specialist in Translation')	programmes
English	45.03.02 Linguistics (degree programme 'Languages of the Slavic	Bachelor's
	World and Russian as a Foreign Language (with second qualification	degree
	'Teacher of the Russian Language and Literature')	programmes
English	45.05.01 Translation and Translation Studies	Specialist's
<b>U</b>		degree
		programme
English	47.03.01 Philosophy	Bachelor's
Liigiisii	47.03.01 1 miosophy	degree
		-
T 1' 1	47.02.02 D. I'. '	programmes
English	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
English	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
English	50.03.03 History of Arts	Bachelor's
· ·		degree
		programmes
English	51.03.01 Cultural Studies	Bachelor's
26	C TIOCHOT CURINITIES STUDIES	degree
		programmes
English	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Eligiisii		
	Heritage Sites	degree
		programmes
English	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Astronomy	01.03.01 Mathematics	Bachelor's
		degree
		programmes
Astronomy	01.03.03 Applied Mathematics and Informatics (degree programme	Bachelor's
· · · · <del>· · · ·</del> /	'Modern Programming')	degree
	in the state of th	programmes
Astronomy	02.03.01 Mathematics and Computer Sciences (programme 'Data	Bachelor's
rasu onomy		
	Sciences')	degree
<b>.</b>	00.00.01 4 11 116 4 11 126	programmes
Astronomy	03.03.01 Applied Mathematics and Physics	Bachelor's
		degree
		programmes
Astronomy	03.03.02 Physics	Bachelor's
-		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Astronomy	03.05.01 Astronomy	Specialist's
,		degree
		programme
Biology	05.03.02 Geography	Bachelor's
210105)	octobro 2 Geography	degree
		programmes
Biology	05.03.03 Cartography and Geoinformatics	Bachelor's
Diology	03.03.03 Cartography and Geomformatics	degree
		programmes
Biology	05.03.04 Hydrometeorology	Bachelor's
Diology	03.03.04 Hydrometeorology	degree
		_
D'-1	05 02 06 F - 1 1 N M	programmes
Biology	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Biology	06.03.01Biology	Bachelor's
		degree
		programmes
Biology	06.03.02 Soil Science	Bachelor's
		degree
		programmes
Biology	31.05.01 General Medicine	Specialist's
27		degree
		programme
Biology	31.05.03 Dental Medicine	Specialist's
Бююбу	31.03.03 Bentul Medicine	degree
		programme
Biology	33 05 01 Pharmacy	Specialist's
Diology	33 03 01 Filatiliacy	degree
D' 1	24.02.01 N'	programme
Biology	34.02.01 Nursing	Bachelor's
		degree
		programmes
Biology	37.03.01 Psychology	Bachelor's
		degree
		programmes
Biology	37.05.01 Clinical Psychology	Specialist's
		degree
		programme
Biology	37.05.02 Employment Psychology	Specialist's
•		degree
		programme
Geography	05.03.01 Geology	Bachelor's
o cogrup)	obiobio 1 Georgy	degree
		programmes
Geography	05.03.02 Geography	Bachelor's
Geography	03.03.02 Ocography	degree
		<u> </u>
C1	05 02 02 Carta annulus and Carta Garage	programmes
Geography	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Geography	05.03.04 Hydrometeorology	Bachelor's
		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Geography	05.03.06 Ecology and Nature Management	Bachelor's
Geography	00.00.00 Ecology und Patatac Planagement	degree
		programmes
Geography	21.03.01 Petroleum Engineering	Bachelor's
Geography	21.03.01 Tetroleum Engineering	degree
		programmes
Cooperative	21.03.02 Land Management and Cadastres	Bachelor's
Geography	21.03.02 Land Management and Cadastres	
		degree
<u> </u>	25 02 02 C	programmes
Geography	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Geography	38.03.01 Economics	Bachelor's
		degree
		programmes
Geography	41.03.02 Russian Region Studies	Bachelor's
		degree
		programmes
Geography	41.03.05 International Relations	Bachelor's
3 1 7		degree
		programmes
Geography	43.03.02 Tourism	Bachelor's
Geography	15.05.02 Tourish	degree
		programmes
Geography	46.03.01 History	Bachelor's
Geography	40.03.01 History	degree
		_
C 1.	47.02.01.DL'11	programmes
Geography	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Geography	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Geography	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Geography	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
<b>0</b> 1 <b>0</b>	Heritage Sites	degree
		programmes
Informatics,	01.03.01 Mathematics	Bachelor's
Informatics and ICT	V-100-100	degree
information and for		programmes
Informatics,	01.03.02 Applied Mathematics and Informatics	Bachelor's
Informatics and ICT	01.03.02 Applied Wathernatics and Informatics	degree
illiorillatics and IC1		_
Informatics	02 02 01 Mathematics and Computer Sciences (mrs aromes - 1D-4-	programmes  Pachalor's
Informatics,	02.03.01 Mathematics and Computer Sciences (programme 'Data	Bachelor's
Informatics and ICT	Sciences')	degree
<b>.</b>		programmes
Informatics,	02.03.02 Fundamental Informatics and Information Technology	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	02.03.03 Information Systems Administration and Mathematical	Bachelor's
Informatics and ICT	Support	degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Informatics,	05.03.01 Geology	Bachelor's
Informatics and ICT	03.03.01 Geology	
illiorinatics and ici		degree
T., C.,	05 02 02 C 1	programmes
Informatics,	05.03.02 Geography	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	05.03.03 Cartography and Geoinformatics	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	05.03.04 Hydrometeorology	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	05.03.06 Ecology and Nature Management	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	06.03.02 Soil Science	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	09.03.03 Applied Informatics	Bachelor's
Informatics and ICT	oyiocioc i ppinou imorniumo	degree
information and for		programmes
Informatics,	09.03.04 Software Engineering	Bachelor's
Informatics and ICT	07.03.04 Software Engineering	degree
informatics and ici		_
T., C 12	21.02.01 D.41 Facility	programmes
Informatics,	21.03.01 Petroleum Engineering	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	21.03.02 Land Management and Cadastres	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	27.03.03 Systems Analysis and Control	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	37.03.02 Conflictology	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	38.03.01 Economics	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	38.03.03 Personnel Management	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	38.03.05 Business Informatics	Bachelor's
Informatics and ICT	2 2 women amountained	degree
information and ICI		programmes
Informatics,	39.03.01 Sociology	Bachelor's
Informatics and ICT	57.03.01 Sociology	degree
mormanes and tel		_
Information	47.02.01 Philosophy	programmes Bachelor's
Informatics,	47.03.01 Philosophy	
Informatics and ICT		degree
T 0 .	45.00.00 P. W. L. G. W.	programmes
Informatics,	47.03.03 Religious Studies	Bachelor's
Informatics and ICT		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Informatics,	51.03.01 Cultural Studies	Bachelor's
Informatics and ICT		degree
		programmes
Informatics,	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Informatics and ICT	Heritage Sites	degree
informatics and 1e i	Tierrage sites	programmes
History	37.03.02 Conflictology	Bachelor's
THStory	37.03.02 Connictology	degree
		_
TT' . 4	20.02.01.0	programmes Bachelor's
History	39.03.01 Sociology	
		degree
		programmes
History	39.03.02 Social Work	Bachelor's
		degree
		programmes
History	40.03.01 Law	Bachelor's
		degree
		programmes
History	41.03.02 Russian Region Studies	Bachelor's
,		degree
		programmes
History	41.03.04 Political Science	Bachelor's
Thistory	41.03.04 I Officer Science	degree
		programmes
History	41.03.05 International Relations	Bachelor's
nistory	41.03.03 International Relations	
		degree
***	12.02.04.4.1	programmes
History	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
History	43.03.02 Tourism	Bachelor's
		degree
		programmes
History	46.03.01 History	Bachelor's
•		degree
		programmes
History	47.03.01 Philosophy	Bachelor's
		degree
		programmes
History	47.03.03 Religious Studies	Bachelor's
THStory	47.03.03 Religious Studies	degree
		programmes
Listom	49.06.01 Theology	Bachelor's
History	48.06.01 Theology	
		degree
TT' .	70.00.01 A	programmes
History	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
History	50.03.03 History of Arts	Bachelor's
		degree
		programmes
History	51.03.01 Cultural Studies	Bachelor's
History	51.03.01 Cultural Studies	Bachelor's degree

Olympiad profile	Main field of study and speciality titles	Level of education
History	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
•	Heritage Sites	degree
		programmes
History	58.03.01 Asian and African Studies	Bachelor's
,		degree
		programmes
Literature	37.03.02 Conflictology	Bachelor's
	0.1100.00_ 0.111110.0008,	degree
		programmes
Literature	42.03.02 Journalism	Bachelor's
Literature	12.03.02 voulhansin	degree
		programmes
Literature	45.03.01 Philology	Bachelor's
Literature	+5.05.01 1 mlology	degree
		programmes
Literature	47.03.01 Philosophy	Bachelor's
Literature	47.03.01 Filliosophy	
		degree
T *	47.02.02 P. II. 1	programmes
Literature	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Literature	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Literature	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Literature	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Literature	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
Mathematics	01.03.01 Mathematics	Bachelor's
		degree
		programmes
Mathematics	01.03.02Applied Mathematics and Informatics	Bachelor's
Wathematies	01.05.027 ipplied ividicinaties and informaties	degree
		programmes
Mathematics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
Mathematics	01.03.03 Weenames and Wathematical Wodering	degree
		-
Mathematics	01.05.01 Fundamental Mathematics and Mechanics	programmes Specialist's
wramemancs	01.03.01 Fundamental Mathematics and Mechanics	Specialist's
		degree
Mathamatica	02 02 01 Mathamatics and Communication Co.	programme
Mathematics	02.03.01 Mathematics and Computer Sciences	Bachelor's
		degree
37.4		programmes
Mathematics	02.03.02 Fundamental Informatics and Information Technology	Bachelor's
		degree
		programmes
Mathematics	03.03.01 Applied Mathematics and Physics	Bachelor's
		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Mathematics	03.03.02 Physics	Bachelor's
1viumemunes	03.03.02 Thysics	degree
		programmes
Mathematics	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
Maniemanes	04.03.02 Chemistry, Fnysics and Material Mechanics	degree
		_
Mathematics	05.03.01 Geology	programmes Bachelor's
Mamemanes	03.03.01 Geology	****
		degree
37.1	07.00.00	programmes
Mathematics	05.03.02 Geography	Bachelor's
		degree
		programmes
Mathematics	05.03.03 Cartography and Geoinformatics	Bachelor's
		degree
		programmes
Mathematics	05.03.04 Hydrometeorology	Bachelor's
		degree
		programmes
Mathematics	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Mathematics	06.03.02 Soil Science	Bachelor's
Maniematics	00.03.02 Boll Belence	degree
		programmes
Mathematics	09.03.03 Applied Informatics (degree programme: Applied	Bachelor's
Maniemanes	Informatics in Arts and Humanities)	
	informatics in Arts and Humanities)	degree
3.6.4	21 02 01 D + 1	programmes
Mathematics	21.03.01 Petroleum Engineering	Bachelor's
		degree
		programmes
Mathematics	21/03/2002 Land Management and Cadastres	Bachelor's
		degree
		programmes
Mathematics	27.03.03 Systems Analysis and Control	Bachelor's
		degree
		programmes
Mathematics	37.03.01 Psychology	Bachelor's
		degree
		programmes
Mathematics	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Mathematics	37.05.01 Clinical Psychology	Specialist's
1vianicinanes	57.03.01 Chinear I sychology	degree
		_
Mathematics	27 05 02 Employment Payahology	programme Specialist's
iviamemanes	37.05.02 Employment Psychology	Specialist's
		degree
	20.00.01.7	programme
Mathematics	38.03.01 Economics	Bachelor's
		degree
		programmes
Mathematics	38.03.02 Management	Bachelor's
		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Mathematics	38.03.03 Personnel Management	Bachelor's
	č	degree
		programmes
Mathematics	38.03.04 Public and Municipal Administration	Bachelor's
	The state of the s	degree
		programmes
Mathematics	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Mathematics	39.03.01 Sociology	Bachelor's
Tradicinatios	37.03.01 300101055	degree
		programmes
Mathematics	42.03.01 Advertising and Public Relations	Bachelor's
Wiathematics	42.03.01 Advertising and I done Relations	degree
		programmes
Mathematics	47.03.01 Philosophy	Bachelor's
Maniemanes	47.03.01 Filliosophy	degree
		_
Mathematics	47.02.02 D.1'.'	programmes
Mathematics	47.03.03 Religious Studies	Bachelor's
		degree
36.1	#4.00.04.G.11.G!!	programmes
Mathematics	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
Mathematics	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
German	37.03.02 Conflictology	Bachelor's
		degree
		programmes
German	39.03.01 Sociology	Bachelor's
		degree
		programmes
German	39.03.02 Social Work	Bachelor's
		degree
		programmes
German	40.03.01 Law	Bachelor's
		degree
		programmes
German	41.03.05 International Relations	Bachelor's
	1500000	degree
		programmes
German	42.03.01 Advertising and Public Relations	Bachelor's
German	42.03.01 Advertising and Fabric Relations	degree
		programmes
German	42.03.02 Journalism	Bachelor's
German	T2.03.02 JOUINGIISIII	degree
		programmes
Garman	43.03.02 Tourism	Bachelor's
German	43.03.02 TOURISH	
		degree
	45 02 01 PH '1 1	programmes
German	45.03.01 Philology	Bachelor's
		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
German	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
German	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
Cerman	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes
German	45.03.02 Linguistics (degree programme 'Languages of the Slavic	Bachelor's
German	World and Russian as a Foreign Language (with second qualification	degree
	'Teacher of the Russian Language and Literature')	-
C		programmes Bachelor's
German	47.03.01 Philosophy	
		degree
		programmes
German	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
German	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
German	50.03.03 History of Arts	Bachelor's
Commi	50.05.05 Thistory of Thes	degree
		programmes
German	51.03.01 Cultural Studies	Bachelor's
German	51.05.01 Cultural Studies	degree
		_
<u> </u>	51.02.04.W 1 1.C CC 1, 1 1.W 1	programmes
German	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
	Heritage Sites	degree
		programmes
German	58.03.01 Asian and African Studies	Bachelor's
		degree
		programmes
Social Science	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Social Science	38.03.01 Economics	Bachelor's
		degree
		programmes
Social Science	38.03.02 Management	Bachelor's
Social Science	30.03.02 Wanagement	degree
		programmes
Social Science	38.03.03 Personnel Management	Bachelor's
Social Science	38.03.03 Fersonner Management	
		degree
0 '10'	20 02 04 P 11' 1M 1' 1A 1 1' 1'	programmes
Social Science	38.03.04 Public and Municipal Administration	Bachelor's
		degree
		programmes
Social Science	38.03.05 Business Informatics	Bachelor's
		degree
		programmes
Social Science	39.03.01 Sociology	Bachelor's
		degree
		programmes
Social Science	39.03.02 Social Work	Bachelor's
Social Science	57.03.02 DOCIGI WOLK	degree
		programmes
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Social Science	40.03.01 Law	Bachelor's
	1000000 = 2000	degree
		programmes
Social Science	41.03.02 Russian Region Studies	Bachelor's
Social Social	Titoero 2 Transmir Trogram Studies	degree
		programmes
Social Science	41.03.04 Political Science	Bachelor's
Social Science	11.03.011 Officer Science	degree
		programmes
Social Science	41.03.05 International Relations	Bachelor's
Social Science	41.03.03 international Relations	degree
		programmes
Social Science	42.03.01 Advertising and Public Relations	Bachelor's
Social Science	42.03.01 Advertising and Fublic Relations	
		degree
0 '10'	42.02.02 F	programmes
Social Science	43.03.02 Tourism	Bachelor's
		degree
		programmes
Social Science	46.03.01 History	Bachelor's
		degree
		programmes
Social Science	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Social Science	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Social Science	48.06.01 Theology	Bachelor's
		degree
		programmes
Social Science	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Social Science	51.03.01 Cultural Studies	Bachelor's
Bociai Beience	51.05.01 Cultural Studies	degree
		programmes
Social Science	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Social Science	Heritage Sites	degree
	Hemage Sites	_
D	27 02 01 Paral ala arr	programmes
Russian	37.03.01 Psychology	Bachelor's
		degree
- ·	25 02 02 G G G G G	programmes
Russian	37.03.02 Conflictology	Bachelor's
		degree
		programmes
Russian	37.05.01 Clinical Psychology	Specialist's
		degree
		programme
Russian	37.05.02 Employment Psychology	Specialist's
		degree
		programme
Russian	40.03.01 Law	Bachelor's
		degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Russian	41.03.04 Political Science	Bachelor's
	100000000000000000000000000000000000000	degree
		programmes
Russian	42.03.01 Advertising and Public Relations	Bachelor's
	12.00.00 = 2.00.00.000 = 2.00.000.000	degree
		programmes
Russian	42.03.02 Journalism	Bachelor's
rassian	12.03.02 (Odificialism	degree
		programmes
Russian	45.03.01 Philology	Bachelor's
Kussian	+5.05.01 1 miology	degree
		programmes
Russian	45 02 02 Linguistics	Bachelor's
Russian	45.03.02 Linguistics	
		degree
D :	45.05.01 TP	programmes
Russian	45.05.01 Translation and Translation Studies	Specialist's
		degree
		programme
Russian	47.03.01 Philosophy	Bachelor's
		degree
		programmes
Russian	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
Russian	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
Russian	50.03.03 History of Arts	Bachelor's
		degree
		programmes
Russian	51.03.01 Cultural Studies	Bachelor's
Russiun	51.05.01 Cultural Studies	degree
		programmes
Russian	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
Russian	Heritage Sites	degree
	Tierrage Sites	programmes
Physics	01.03.01 Mathematics	Bachelor's
Physics	01.03.01 Mathematics	
		degree
DI '		programmes
Physics	01.03.03 Applied Mathematics and Informatics (degree programme	Bachelor's
	'Modern Programming')	degree
		programmes
Physics	01.03.03 Mechanics and Mathematical Modelling	Bachelor's
		degree
		programmes
Physics	01.05.01 Fundamental Mathematics and Mechanics (programme	Specialist's
	'Fundamental Mechanics')	degree
		programme
Physics	02.03.01 Mathematics and Computer Sciences (programme 'Data	Bachelor's
	Sciences')	degree
		programmes
Physics	03.03.01 Applied Mathematics and Physics	Bachelor's
•	11	degree
		programmes

Olympiad profile	Main field of study and speciality titles	Level of education
Physics	03.03.02 Physics	Bachelor's
Ž		degree
		programmes
Physics	03.05.01 Astronomy	Specialist's
111,510.5		degree
		programme
Physics	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
Tilysies	0 1.05.02 Chemistry, 1 hysics and Material Mechanics	degree
		programmes
Physics	05.03.01 Geology	Bachelor's
rilysics	03.03.01 deology	degree
		-
Di	21 02 01 Parada as Farina da	programmes
Physics	21.03.01 Petroleum Engineering	Bachelor's
		degree
		programmes
Physics	21.03.02 Land Management and Cadastres	Bachelor's
		degree
		programmes
Physics	27.03.03 Systems Analysis and Control	Bachelor's
		degree
		programmes
French	37.03.02 Conflictology	Bachelor's
		degree
		programmes
French	39.03.01 Sociology	Bachelor's
Trenen	57.03.01 Boolology	degree
		programmes
French	39.03.02 Social Work	Bachelor's
Pichch	39.03.02 Social Work	degree
		-
French	40.02.01 L	programmes
French	40.03.01 Law	Bachelor's
		degree
		programmes
French	41.03.05 International Relations	Bachelor's
		degree
		programmes
French	42.03.01 Advertising and Public Relations	Bachelor's
		degree
		programmes
French	42.03.02 Journalism	Bachelor's
		degree
		programmes
French	43.03.02 Tourism	Bachelor's
	iciocio 2 Tourism	degree
		programmes
French	45.03.01 Philology	Bachelor's
Tichch	43.03.01 1 mlology	degree
		-
Enomoh	45.02.02 Linguistics (do (D. 11 Gr. 1' 1	programmes  Pachalogia
French	45.03.02 Linguistics (degree programmes 'Balkan Studies and	Bachelor's
	Comparative Linguistics (with second qualification 'Teacher of a	degree
	Foreign Language'/'Specialist in Translation')	programmes
French	45.03.02 Linguistics (degree programme 'Russian as a Foreign	Bachelor's
	Language ( (with second qualification 'Teacher of the Russian	degree
	Language and Literature')	programmes

Olympiad profile	Main field of study and speciality titles	Level of education
French	45.03.02 Linguistics (degree programme 'Languages of the Slavic	Bachelor's
	World and Russian as a Foreign Language (with second qualification	degree
	'Teacher of the Russian Language and Literature')	programmes
French	47.03.01 Philosophy	Bachelor's
		degree
		programmes
French	47.03.03 Religious Studies	Bachelor's
		degree
		programmes
French	50.03.01 Arts and Humanities	Bachelor's
		degree
		programmes
French	50.03.03 History of Arts	Bachelor's
	·	degree
		programmes
French	51.03.01 Cultural Studies	Bachelor's
		degree
		programmes
French	51.03.04 Museology and Conservation of Cultural and Natural	Bachelor's
I TORCH	Heritage Sites	degree
	Tierrage Sites	programmes
French	58.03.01 Asian and African Studies	Bachelor's
FIGUCII	36.03.01 Asian and African Studies	
		degree
CI.	04.02.01.01	programmes
Chemistry	04.03.01 Chemistry	Bachelor's
		degree
		programmes
Chemistry	04.03.02 Chemistry, Physics and Material Mechanics	Bachelor's
		degree
		programmes
Chemistry	05.03.01 Geology	Bachelor's
		degree
		programmes
Chemistry	05.03.06 Ecology and Nature Management	Bachelor's
		degree
		programmes
Chemistry	06.03.02 Soil Science	Bachelor's
		degree
		programmes
Chemistry	21.03.01 Petroleum Engineering	Bachelor's
•		degree
		programmes
Chemistry	21/03/2002 Land Management and Cadastres	Bachelor's
J		degree
		programmes
Chemistry	27.03.03 Systems Analysis and Control	Bachelor's
	2	degree
		programmes
Chemistry	31.05.01 General Medicine	Specialist's
спенный у	51.05.01 Ocheral Medicilic	degree
		-
Claration and the contract of	21.05.02 Dental Madicine	programme
Chemistry	31.05.03 Dental Medicine	Specialist's
		degree
		programme

Olympiad profile	Main field of study and speciality titles	Level of education
Chemistry	33 05 01 Pharmacy	Specialist's
		degree
		programme
Chemistry	34.02.01 Nursing	Bachelor's
	-	degree
		programmes

Annex 10
to the Regulation for Admissions
to Degree Programmes (Bachelor's Programmes, Specialist's Programmes, Master's
Programmes)
at St Petersburg University
in 2024

## Correspondence of the profiles of all-Ukrainian Student Olympiad, international Olympiads in general education subjects to general-education-based admissions tests

Profile of all-Ukrainian School Student Olympiad, international Olympiads in general education subjects	Subject of admissions tests
English	English
Astronomy	Physics
Biology	Biology
Bulgarian Language and Literature	_
Hungarian Language and Literature	_
Geography	Geography
Informatics	Informatics and ICT
Information Technology	_
Spanish	Spanish
History	_
Mathematics	Mathematics
Moldavian Language and Literature	_
German	German
Modern Greek Language and Literature	_
Polish Language and Literature	_
Legal Studies	_
Romanian Language and Literature	_
Russian Language and Literature	Russian / Literature
Labour Training (Tecgnology)	_
Ukrainian Language and Literature	_
Physics	Physics
French	French
Chemistry	Chemistry
Ecology	Biology, Geography
Economics	Social Science
Hebrew and Hebrew Literature	_